

IntechOpen

Knowledge Management Strategies and Applications

Edited by Muhammad Mohiuddin, Norrin Halilem, SM Ahasanul Kobir and Cao Yuliang





KNOWLEDGE MANAGEMENT STRATEGIES AND APPLICATIONS

Edited by **Muhammad Mohiuddin, Norrin** Halilem, SM Ahasanul Kobir and Cao Yuliang

Knowledge Management Strategies and Applications

http://dx.doi.org/10.5772/63142 Edited by Muhammad Mohiuddin, Norrin Halilem, SM Ahasanul Kobir and Cao Yuliang

Contributors

Michael Beyerlein, Rachele Collins, Shinhee Jeong, Christi Phillips, Suravee Sunalai, Lei Xie, Maria De Fatima Ebole Santana, Vasja Roblek, Maja Meško, Jana Suklan, Dr Angela Siew Hoong Lee, Lip Sam Thi, Mei Hua Lin, Diego Monferrer Tirado, Marta Estrada Guillén, Tommy Wong, Mark Wickham, Linda French, Matjaz Roblek, Maja Zajec, Benjamin Urh, Ivy Chan, Shofang Chang, Tain-Junn Cheng, Chung-Hsien Chan, Karla Olmos-Sánchez, Jorge Rodas-Osollo, Jose Roberto Ferretti, Maria Da Conceição Afonso, Wolfgang Ossadnik, Sarah Kölbel, Stefan Gergeleit, Antonio Martin-Montes, Mauricio Burbano, Carlos Leon, Gabriele Berg-Beckhoff, Peter Wiedemann, Ádám Balázs, Joachim Schüz, Kristian Breum Ølgaard, Pernille Tangaard Andersen, Steven Ndugwa Kabwama, Jesper Bo Nielsen

© The Editor(s) and the Author(s) 2017

The moral rights of the and the author(s) have been asserted.

All rights to the book as a whole are reserved by INTECH. The book as a whole (compilation) cannot be reproduced, distributed or used for commercial or non-commercial purposes without INTECH's written permission. Enquiries concerning the use of the book should be directed to INTECH rights and permissions department (permissions@intechopen.com).

Violations are liable to prosecution under the governing Copyright Law.

CC BY

Individual chapters of this publication are distributed under the terms of the Creative Commons Attribution 3.0 Unported License which permits commercial use, distribution and reproduction of the individual chapters, provided the original author(s) and source publication are appropriately acknowledged. If so indicated, certain images may not be included under the Creative Commons license. In such cases users will need to obtain permission from the license holder to reproduce the material. More details and guidelines concerning content reuse and adaptation can be foundat http://www.intechopen.com/copyright-policy.html.

Notice

Statements and opinions expressed in the chapters are these of the individual contributors and not necessarily those of the editors or publisher. No responsibility is accepted for the accuracy of information contained in the published chapters. The publisher assumes no responsibility for any damage or injury to persons or property arising out of the use of any materials, instructions, methods or ideas contained in the book.

First published in Croatia, 2017 by INTECH d.o.o. eBook (PDF) Published by IN TECH d.o.o. Place and year of publication of eBook (PDF): Rijeka, 2019. IntechOpen is the global imprint of IN TECH d.o.o. Printed in Croatia

Legal deposit, Croatia: National and University Library in Zagreb

Additional hard and PDF copies can be obtained from orders@intechopen.com

Knowledge Management Strategies and Applications Edited by Muhammad Mohiuddin, Norrin Halilem, SM Ahasanul Kobir and Cao Yuliang p. cm. Print ISBN 978-953-51-3617-0 Online ISBN 978-953-51-3618-7 eBook (PDF) ISBN 978-953-51-4605-6

We are IntechOpen, the first native scientific publisher of Open Access books

3.350+ Open access books available

International authors and editors

108,000+ 114M+ Downloads

15Countries delivered to Our authors are among the

lop 1% most cited scientists

12.2%

Contributors from top 500 universities



WEB OF SCIENCE

Selection of our books indexed in the Book Citation Index in Web of Science[™] Core Collection (BKCI)

Interested in publishing with us? Contact book.department@intechopen.com

Numbers displayed above are based on latest data collected. For more information visit www.intechopen.com



Meet the editors

Muhammad Mohiuddin, PhD, is an assistant professor of International Business at Thompson Rivers University, Kamloops, British Columbia, Canada. He has also taught at Laval University, Quebec, Canada. Dr. Mohiuddin has also taught at the University of Paris-Est Creteil, Paris, France; Osnabruck University of Applied Science, Germany; Tianjin University of Technology, China; and Shanghai Institute of Technology, China, as guest lecturer/visiting professor. His research interests focus on knowledge management strategy, offshore outsourcing and developing dynamic capabilities, innovation and industry 4.0, emerging markets, and social entrepreneurship. He has presented his research in many international conferences including Strategic Management Society (SMS) and Academy of International Business (AIB). Dr. Mohiuddin has published more than 30 articles in peer-reviewed journals and several book chapters.

Norrin Halilem, PhD, is an associate professor in Knowledge and Innovation Management (Laval University, Canada), where he teaches to PhD and MBA students. His research contributions concern the development of a systemic view of innovation and knowledge management, which encompasses the study of several actors and themes: academic knowledge producers and entrepreneurs, knowledge intermediation organizations, public and private organizations, or else knowledge and innovation management policy instruments. In management, he has published in Research Policy, Scientometrics, Technovation, Higher Education, and other leading journals, while in health sciences, he has published in Public Health Genomics, Familial Cancer, and other leading journals.

SM Ahasanul Kobir was born in Bangladesh in 1978. He received the BSc (Hons) degree in Biochemistry from Aligarh Muslim University, India, in 2001; MSc degree in Nutrition and Food Science from Dhaka University, Bangladesh, in 2004; MSc degree in Molecular Genetics and Physiology from Linkoping University, Sweden, in 2007; and MSc degree in Biotechnology from AgroParisTech and PhD degree in Molecular Genetics from the University of Paris XI in Paris, France, in 2008 and in 2012, respectively. In 2013, he joined as a postdoctoral researcher at the Department of Biotechnology and Microbiology at the Universidad Autonoma de Madrid, Spain; in 2014, he started his career as a research associate at the Department of Experimental Medicine at the Scripps Research Institute, San Diego, California, USA; and since 2015, he has been working as a postdoctoral fellow in the School of Medicine at the University of Pittsburgh, Pittsburgh, USA. His main areas of research interests are molecular mechanism of pulmonary hypertension and its drug designing, molecular detection and live cell imaging in cancer biology, molecular mechanism of HIV and its drug discovery, biotechnological interest of signaling and regulation in

bacterial cells, etc. He is a member of the American Thoracic Society and also received the best abstract award from ATS Conference in 2016. Dr. Kobir published in top journals in his field.

Cao Yuliang, PhD, is an associate professor of Management Science at the International College of Business and Technology, Tianjin University of Technology, China. He is also serving as an assistant dean and a master tutor of the International College of Business and Technology. His research interests focus on innovation and entrepreneurship management, supply chain management, and applied economics. He is also currently collaborating with the Chinese Ministry of Science and Technology.

Contents

Preface XI

Section 1	Knowledge Management Strategies 1
Chapter 1	Tacit Knowledge Sharing: A Literature Review Applied to the Context of the Brazilian Judiciary 3 José Roberto Ferretti and Maria da Conceição L. Afonso
Chapter 2	Knowledge Sharing and Human Resource Development in Innovative Organizations 25 Michael Beyerlein, Rachele Collins, Shinhee Jeong, Christi Phillips, Suravee Sunalai and Lei Xie
Chapter 3	Network Market Orientation, Knowledge Management and Born Globals' Competitiveness 47 Diego Monferrer Tirado and Marta Estrada Guillén
Chapter 4	Knowledge Management Hybrid Strategy with People, Technology and Process Pillars 69 Ivy Chan
Chapter 5	Knowledge Management for Informally Structured Domains: Challenges and Proposals 85 Karla Olmos-Sánchez and Jorge Rodas-Osollo
Chapter 6	Performance Management by Causal Mapping: An Application Field of Knowledge Management 103 Sarah Kölbel, Wolfgang Ossadnik and Stefan Gergeleit
Chapter 7	Intelligent Knowledge Retrieval from Industrial Repositories 121 Antonio Martin, Mauricio Burbano and Carlos León

Chapter 8 Affective Technology Acceptance Model: Extending Technology Acceptance Model with Positive and Negative Affect 147

Angela Lee Siew Hoong, Lip Sam Thi and Mei-Hua Lin

- Section 2 Knowledge Management Applications 167
- Chapter 9 The Impact of the Internet of Things to Value Added in Knowledge-Intensive Organizations 169 Maja Meško, Jana Suklan and Vasja Roblek
- Chapter 10 Knowledge Management Trends in Biotechnology in Brazil 187 Maria de Fátima Ebole Santana
- Chapter 11 Exploring the Impact of Online Clinical Guidelines on Individual Knowledge Management Behaviors and Individual Net Benefits 207 Shofang Chang, Tain-Junn Cheng and Chung-Hsien Chan
- Chapter 12 Knowledge-Based Assignment Model for Allocation of Employees in Engineering-to-Order Production 217 Matjaz Roblek, Maja Zajec and Benjamin Urh
- Chapter 13 An Examination of the Knowledge Management Process in the Emerging Chinese Hotel Industry 239 Tommy Wong, Linda French and Mark Wickham
- Chapter 14 Widening the Understanding of Risk Approaches by Comparing Definitions from Different Disciplines 253 Gabriele Berg-Beckhoff, Peter Wiedemann, Balázs Ádám, Joachim Schüz, Kristian Breum Ølgaard, Pernille Tanggaard Andersen, Steven Ndugwa Kabwama and Jesper Bo Nielsen

Preface

Since the use of Knowledge work by the management guru of all time Peter Drucker (1994), the world economy has gone through more structural changes and transformations. The globalization, digitalization and accelerated servitization of the economy have created a formidable context of accelerated competition and coopetition and the role of "knowledge management" becomes fundamental for organizations to survive and sustain in a dynamic competitive world. Rise of information and communication technologies (ICT) and social media have accelerated flow of information and data to such extent that it becomes challenging for managers to retain, read and keep organized of all those knowledge flows. A set of behaviors, processes and technologies are needed to efficiently manage the accelerated flow of information in order to improve learning, decision-making and innovation for business successes. The objective of knowledge management (KM) is to effectively use, develop and mobilize the expertise of the organization and apply them for achieving organization's strategic goals such as competitive advantage. Knowledge management involves with knowledge creation, knowledge retention and knowledge transfers. It can help to improve operational excellence by improving internal practices and processes of the organization to make it better, faster, cheaper, safer and cleaner. It can also help to improve innovation capability and growth of the organization. There are different kinds of knowledge such as tacit and explicit knowledge in organization. There are also different purposes of knowledge management in different kind of organizations from resource economy, industrial organizations as well as service economy including knowledge intensive business services (KIBS). The context of these different organizations for knowledge creation, retention and dissemination differs from one another and requires cross-disciplinary approaches. This book on "Knowledge Management Strategies and Applications" addresses the cross-disciplinary approaches of knowledge management of various kinds of organizations by proposing 14 chapters divided into two parts. The first part of the book proposes 8 chapters addressing knowledge management strategies and the second part of the book addresses the knowledge management application in different sector of activities.

In part 1, the first chapter addresses the tacit knowledge sharing strategies. Author Jose Roberto Ferretti and Maria da Conceicao L. Afonso conclude that organizations are moving from traditional approach of tangible assets toward intangible such as tacit knowledge, which is becoming an increasingly generator of growth and innovation. The second chapter explores the way organizations can become more innovative by developing capabilities of its human resources. Authors Michael Beyerlein, Rachele Collins, Shinhee Jeong, Christi Phillips, Suravee Sunalai and Lei Xie propose a model that ties human resource development practices to knowledge management strategies for developing innovation capabilities. The third chapter addresses the strategies on how the influence of network market orientation (NMO) on born global (BG) competitiveness is mediated by absorptive capacity. Authors Diego Monferrer Tirado and Marta Estrada Guillen conclude that absorptive capacity positively influences the firm's capacity to exploit the new relational knowledge and improve the competitiveness and international performance of born global firms. The fourth chapter examines three major KM pillars, including people, process and technology for effective KM deployment. Authors Ivy Chan shows that a hybrid KM strategy is more effective than a strategy focusing on any particular pillar. The fifth chapter presents the challenges and strategies of KM in informally structured domains that need "Requirements Elicitation (RE)" approach. Authors Karla Olmos-Sánchez and Jorge Rodas-Osollo propose an integrated approach to face challenges posed by "Requirement elicitation (RE)" issue. Chapter 6 deals with "causal mapping" approach of knowledge management for performance management (PM) of organizations. However, Causal mapping offers a low quality of the identified causal structures of success factors (SFs). Authors Sarah Kölbel, Wolfgang Ossadnik and Stefan Gergeleit propose to apply DEMATEL in the mapping context to minimize subjectivity by a systematic and transparent pairwise evaluation of the SFs. Chapter 7 explores the strategies of intelligent knowledge retrieval from industrial repositories. Authors Antonio Martin, Mauricio Burbano, and Carlos León present a model that supports efficient retrieval knowledge from digital repositories. Chapter 8 attempts to understand knowledge workers' behavioral intention to use the knowledge sharing tools (KS tools) in their day-to-day tasks. By using affective technology acceptance model, authors Angela Lee Siew Hoong, Tong-Ming Lim, Lip Sam, and Thi, Mee Hua, Lin conclude that negative affect (NA) does not have any impact on perceived usefulness (PU). However, NA has negative influence on behavioral intention (BI) but very significant negative influence on perceived ease of use (PEOU). The outcomes also highlight that PA has very significant positive influence on PU, PEOU and BI with impact on PEOU being the greatest.

In part 2, Chapter 9 explores the knowledge management application in industry 4.0 in an era of Internet of Things (IoT) and how IoT influence organizational changes in knowledge intensive organizations. Authors Maja Meško and Vasja Roblek show the knowledge sharing applications within the organization and with the external partners such as suppliers and customers. Chapter 10 explores knowledge management trends and applications in biotechnology field in Brazil. Author Maria de Fatima Ebole Santana finds that KM applications in biotechnology in Brazil are focused on its local context as well as follow the world trends. Chapter 11 explores the KM applications in online clinical guidelines and how the use of online clinical guidelines influences individual knowledge management behaviors and the net benefits. Authors Chung-Hsien Chan, Tain-Junn Cheng, and Shofang Chang demonstrate different influential models of online clinical guidelines between physicians and physician assistants. Chapter 12 addresses the application of knowledge-based assignment model for allocation of employees in Engineering-to-Order production system. Authors Matjaz Roblek, Maja Zajec, and Benjamin Urh studied an option for adjusting processes to available knowledge. Chapter 13 addresses the application of KM in hotel industry. Authors Tommy Wong, Linda French and Mark Wickham present an analysis of the KM issues and processes experienced in hotel industry in China. Chapter 14 explores the meaning of "risk" from across the disciplines. Authors Gabriele Berg-Beckhoff, Peter Wiedemann, Balázs Ádám, Joachim Schüz; Kristian Breum Ølgaard, Pernille Tanggaard Andersen, Steven Ndugwa Kabwama, Arounda Jesper and Bo Nielsen conclude that discipline differences are mainly connected to the terminology and interpretation of key concepts but the differences are based on different tasks and perspectives.

The proposed 14 articles constitute an important body of knowledge (BoK) on KM strategies and applications. These articles promote both emergent and managerial approach of KM in the organization. KM strategies' applications in both production and knowledge intensive business service (KIBS) firms demonstrated the context and applications of KM strategies and highlighted differences in applying KM in different sector of firms. We believe our reader will enjoy reading these timely chapters on KM strategies and applications.

We acknowledge the contributions of and indebted to anonymous reviewers of this book.

With warmest regards,

Dr. Muhammad Mohiuddin,

Thompson Rivers University, Kamloops, British Columbia, Canada

> **Dr. Norrin Halilem** Laval University, Canada

Dr. SM Ahasanul Kobir University of Pittsburgh, Pittsburgh, USA

Dr. Cao Yuliang, Tianjin University of Technology, China

References

[1] Drucker, P. F. (1994). The Age of Social Transformation. Atlantic Monthly, 274(5): 53-80.

Knowledge Management Strategies

Tacit Knowledge Sharing: A Literature Review Applied to the Context of the Brazilian Judiciary

José Roberto Ferretti and Maria da Conceição L. Afonso

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/intechopen.70066

Abstract

Knowledge has been occupying the center of studies and research for some time, and this chapter seeks to contribute to the depth of this collection and aims to offer articles that deal with tacit knowledge and the sharing of that knowledge. The observed understanding with this literature review was that organizations are moving from a traditional model that is attached to physical and tangible assets to another, where knowledge, especially tacit, is treated as an organizational asset that, despite being immaterial, becomes valuable and a generator of wealth.

Keywords: tacit knowledge sharing, knowledge management, intangible assets

1. Introduction

The aim of this chapter is to present a review of literature on the phenomenon of tacit knowledge sharing (TKS). Knowledge represents such a critical feature for the modern dynamics of organizations that Grant [1] interprets it as a central element in the search for a competitive advantage.

This work is the product of a study done between the years of 2012 and 2016 on the exchange of tacit knowledge among judges of the Small Claims Federal Courts of Brazil.

TKS will be addressed within the context of a new era of knowledge or learning society [2], where efficiency gaps are still present [3] and when sharing is assessed as an organizational process. During this research, TKS was seen and treated as a phenomenon, i.e.,: (a) "an established fact that is a science field" [4], characterized by its social interactions, in person or



remotely; (b) a display of human knowledge that appears under specific conditions [5]; and (c) "knowledge flow" [6], and "relational phenomenon" [7].

The literature review was structured following an epistemological process defined by the following questions: what historical background has been recorded in the literature that showed an investigative awakening about organizational knowledge? How are tacit knowledge and its sharing registered in the literature? And are there any theoretical propositions that deal with TKS as a significant process within the organizational context? To answer these questions, organizational knowledge was understood as the union of two types of knowledge, explicit and tacit [8], or codified knowledge and personal knowledge [2].

Explicit knowledge is understood as information, also referred to knowledge that has already been expressed, taken from the individual's mind, and materialized into any medium (rock, paper, plastic, electronic), established in any media (radio, television, press, publishing, adhesive, photo, film, internet and its content, Whatsapp, Twitter, homepage, blog). In this sense, according to Wolf [9], explicit knowledge is thus expressed and characterized in physical media (clay, papyrus, paper, plastics, CD, DVD); and digital, in bits. The second type of knowledge, tacit knowledge, the main substance of this study, is the content that is stored in human memory, where it exists and from where stories, moods, images, questions, and answers arise [10].

Besides the introduction and conclusion, this article is divided into five sections, which seek to answer the questions previously expressed. The first section discusses the knowledge society, seeking a better understanding of the handling of knowledge in the context of today's society. The second involves the theory of organizational growth, the resource-based view (RBV), and the emergence of the theory based on knowledge. The third introduces studies done on tacit knowledge. The fourth discusses the sharing of tacit knowledge. The last section points out some works where the TKS is the central topic of discussion.

Lastly, some aspects related to the practice within the judicial system are cited, in part because this work is derived from a doctoral thesis where the locus of the research was the Federal Small Claims Courts of Brazil. Thus, this review can extend its application to scholars and practitioners of the judicial system, albeit as a starting point, of some examples of TKS in the judicial context.

2. The knowledge society

It seems to be more reliable to name the current society, less as the information society and more as a society (or era) of knowledge [11, 12]. It is argued that a transition is underway for an economy and society based on knowledge [13].

The knowledge society that serves as a parameter for the development of this research in the Brazilian Judiciary was designed as a spiral of webs, where each web represents the meeting of networks or groups of people that share their tacit knowledge regardless of organizational boundaries or technological tools. This knowledge has been revived, especially since the 1990s, by knowledge management as a source of competitive advantage and intellectual capital [14].

The appearance and the intensive use of new digital technologies led to new and important mechanisms for the political and social development of nations and individuals [15]. Thus, it progresses society toward an increase in its activities based on knowledge, where the organization and sharing of ideas occur simultaneously [16].

The organizational environment, within the context of this new society, is based more on knowledge than on the societies that represented the Modern Era and the industrial society [17] does not differentiate between public and private entities. In both cases, the development of social relations, among others predetermines the willingness to transfer knowledge. With the increasing interconnectedness and the need for rapid change, people and organizations are forced to share what they know and learn from each other [18].

This transit ends up causing an exchange of intangible assets (knowledge), among people and organizations, it is what Van Caenegem treats as "intangible market" [19] that, although relevant, has not yet found the appropriate space within some organizations.

In the case of the Brazilian Judiciary, especially in the Federal Small Claims Courts, there is still a greater concern in dealing with the visible flow of knowledge (information) than with the invisible flow, or its tacit format. The first encompasses what is produced by the parties, what is stated in the sentences and jurisprudence, in the legal codes and norms that regulate the law, and is registered in some type of media. The second mainly covers all the content that is in the minds of the judges and their staff, in other words, fodder for the first. In general terms, in the judicial rite, the author (or advocate) uses tacit knowledge to explain his initial request and the judge in turn uses "[his] tacit knowledge to analyze the law and give the sentence" [20]. This flow is the most critical process, and, at the same time, the "judicial business" axis. **Table 1** below reflects some of these particularities.

In the interchangeable socialization of knowledge, new ideas are offered and foundation is created for the development of new products and services with the possibility of developing innovative mechanisms and workflows. For this reason, Casselman and Samson [21] point out that the use of tacit knowledge maintains a significant relationship with innovative products and financial performance. Similarly, Gomes [22] while studying the Brazilian judiciary claimed knowledge as being the main organizational asset.

New knowledge and sharing redefine problems and solutions and give new form to the organization and its surroundings. Exchanges alter the amount of prior knowledge and increase the permeability of the institutional fabric, allowing for the movement of knowledge from the inside out and vice versa, by causing individual, sectoral, organizational, or inter-organizational exchanges [8].

In the era of knowledge, there is a diminishing emphasis on the tangible flow of information and the emergence of a new worldview where uniqueness, flexibility, and performance, redirect organizations toward a market that is increasingly closer and more related to the intangible knowledge [23].

Present times require a management style that is aligned with a knowledge flow stemming from a new atmosphere created and developed by organizations to enable the recovery and

Industry paradigm	Knowledge paradigm	SCFC context
People under supervision	People mutually linked as a network	Intense TKS
Information control	Accessible information	Integrated electronic process systems
People process tangible resources into products	Knowledge workers convert knowledge into intangible structures	Judges and servants who, at certain stages of the procedural rite, gather knowledge of the parties and of the lawyers as aids to the (mental) decision-making process of the judge
Flow of information via organizational hierarchy	Network flow among co-workers	Network flow among servers, judges, lawyers, experts, doctors, social workers, judicial units, located in the same city or not
Process bottlenecks arising from lack of capital	Process bottlenecks arising from lack of knowledge	There are judges who practice and create an atmosphere of TKS
Production related to tangible products	Production recognizes the importance of intangible assets	In the SCFC, the relationship of judges with the parties, with lawyers and with the servers is stimulated for exchanging knowledge
Unilateral relationship with customers via the market	Relationship according to personal interaction	Judges and servants' relationship with the parties and with lawyers takes place face-to-face (during hearings and over-the-counter visits to provide information and resolve doubts). The relationship of the judges with the servers happens daily, in person and, almost always, without "scheduled time"

Source: adapted from Sveiby [68].

 Table 1. Comparative paradigm of principles of a knowledge organization within the context of Small Claims Federal Courts (SCFC).

use of this knowledge, especially the more effective personal knowledge [24], this tacit knowledge favors the emergence of new processes and innovation [12, 25].

Studies done in six organizations of the Asian public sector [16] showed concern for a better understanding of the handling of knowledge. Those organizations opted for increasing efficiency according to the knowledge shared by the most experienced to improve the operational and strategic excellence. The new approach adopted by these six organizations required a rediscussing on how individual and organizational knowledge was (or not) being created and shared, since they saw this knowledge as a fundamental stage of any process that aimed at dealing with change in mentality and behavior.

When analyzing knowledge management carried out at the Department of National Defense in Canada, Girard and McIntyre [26] argued that this process is not characterized as a novelty. However, the significant results highlight what knowledge sharing continues to generate in terms of socialization and an elevated degree of cooperation and learning, something similar to what Kerckhove [27] treated as connected intelligence.

This century, more than others, has compelled organizations to associate their internal structure and work processes to a new asset of intangible nature–knowledge [28]. However, it seems that some organizations, despite being surrounded by information, lack this resource [29, 30].

3. The valuable immateriality: tacit knowledge

In search of the background of what today is referred to as (with relative comfort and understanding) "organizational knowledge", this research took into consideration studies done by Penrose [31]¹ when she analyzed companies in the 1950s.

Organizational knowledge is knowledge that is registered in the media under the control of organizations and especially the one that lies within the people minds and flows according to social, formal, or informal interactions [76]. This type of knowledge is composed of personal knowledge, translated by their experience, know-how, and practices developed individually or from within an environment provided by the organization [32].

Penrose sought to find out if there was something that could differentiate organizations regarding its growth or setbacks. In her studies, she assessed and questioned the possibility of there being something else in the firm's growth that was beyond the organization's traditional view on handling human resources (labor) and products (supply chain). She found out that the administrators' experiences translated by their knowledge were revealed as the differentiating factor from one organization to the other. This experience had been transformed into tacit knowledge; it was not available in the form of a manual or coded; in other words, it was not explicit, to be physically incorporated by the organization; she pointed out that this knowledge could not be acquired in the market and concluded that those professionals could bolster or hinder the development of the company. Penrose [31], consequently, asserted knowledge held by more experienced professionals as a particular resource that could be associated with the growth in firms, especially one's individual knowledge, subsequently treated as tacit [8, 33] that if accumulated over time, is capable of providing more efficient opportunities for organizational growth [34].

Penrose [31] developed one of the fundamental assumptions of this theory that the growth of an organization "is essentially an evolutionary process and is based on the cumulative increase of the collective knowledge." Something that Inkpen [3] referred to as 'collaborative knowledge,' and Nonaka and Takeuchi [8] would later represent it as the 'spiral' of the epistemological and ontological knowledge.

In the early 1990s, within the field of strategic management and economics, the theory of 'resource-based view (RBV)' is established and the work of Penrose [31] as a notable

¹The original work dates from 1959.

contribution [35]. For Bierly et al. [36], such theory almost superseded the traditional view related to industrial organizations (I/O approach)² dealt by Porter 10 years earlier [38].

This RBV theory associates the performance of the organization to the combination of two elements, Resources and Capabilities (R & C),³ it highlights skills and expertise in dealing with knowledge, already showing an early interest in tacit knowledge. This theory seeks to explain and predict why some organizations are able to sustain a competitive advantage and achieve better financial returns.

Considering knowledge as a usable resource in order to improve efficiency and organizational effectiveness [39], RBV appears as a precursor to a new theory, the 'knowledge-based view (KBV)' where this asset, knowledge, is regarded as the most strategically important of the resources in a company [40] and the organization as a vehicle, for creation, sharing, storage, and implementation of this source [41].

Kogut and Zander [42], oriented by the KBV, associate organizational growth with knowledge sharing, defining firms as social communities specialized in the fast and efficient transfer of knowledge, especially knowledge that represent know-how, i.e., tacit knowledge [43].

If we live in a society or knowledge era, where tacit knowledge is seen as strategic in the business field and public organizations, what has the literature offered to conceptualize it?

4. Sharing the intangible

Knowledge is the core element of an emerging production model whose most important process is learning [2]. Through this, it is granted by the use of a sensory organ, the description, calculation or verifiable prediction of any entity, fact, thing or reality [5]. Knowledge consists of a "set of ideas and principles that a person acquires through study, observation, or experience and that can be an integral part of one's abilities" [44] and in this sense, it is related to the cognitive processes where assimilation, association, construction, and deconstruction of concepts occur [45].

Kofman [46] sees knowledge as a product of itself, that is, by acting in the search of results, each stage adds more knowledge, as the spiral spinning of "eternal" growth proposed by Nonaka and Takeuchi [8], taking into account all types of processes, including the most basic, such as, hunting, fishing, farming, and weaving [47].

In the framework of organizations, authors have emphasized the knowledge of the individual, tacit, since Taylor's [48] work in the early twentieth century. This interest is intensified

²The approach (I) Industrial/(O) Organizational refers to a business model valued by Porter [33] in his work: Competitive Strategy, where, to ensure a good performance, the organization should resemble the structure of an industry using one of the following strategies: leadership by cost (selling cheap); leadership by distinction (doing something different); or leadership by focus (focuses on meeting specific segments and markets). Later, Porter [37] through another work, Competitive Advantage, analyzes the introduction of concepts already flagged by RVB, such as the importance of internal skills and human expertise as critical to the good performance of the essential activities of the organization. ³The literature uses the binomial as "Resources & Capabilities".

when there is a perception of the transition of wealth from that of natural resources to another, provided by brainpower [49], as well as the diversity of concepts and applications within and outside academia⁴ [50].

For Davenport and Prusak [51], tacit knowledge "is a fluid mixture of condensed experience, values, and insight,⁵ experimented [...] originates and is applied in the minds of connoisseurs."

Tacit knowledge, or that which accumulates in the mind of the individual in the form of ideas and experiences [53], has been on the agenda, in some way or the other, in the last 40 years [54]. It presents itself as a knowledge that escapes the formality and materiality; therefore, not captured in licenses and patents [55]. It is an attribute of those who know what they are doing and better understand the needs to be met, as a skilled craftsman whose knowledge is highly personal and deeply rooted in his experiences, values, and emotions [56]. It is of a highly personal nature [57–59], better shared through conversation [60], and with a more pronounced fluidity in informal environments [61].

Alavi and Leidner [62] understand it as something rooted in the actions and experiences of individuals and categorized it as cognitive (mental maps, beliefs, paradigms and view-points) and technical (know-how, skills) that allows for thinking and examining a problem and its solutions by a variety of perspectives [63]. Others treat tacit knowledge as "things that originate in someone's brain: electronic programming, trademark, and marketing force" [64]. Lemos and Jóia [65] translate it as a direct result of experiences, reflections, and dialogue.

Most of the organizational knowledge is tacit and includes intuition, perspectives, beliefs, and values [66]. It is estimated at 80% of the useful knowledge of an organization [67] and when not shared brings more problems or difficulties than useful results [33].

Although it seems like a minor issue for some managers, it must be observed that all the practical knowledge is, in almost all its extension, tacit [68] whose most common application is to solve problems [56]. According to Martins et al. [69], these sets of "tacit knowledge" remain scattered throughout the organization and arise from the professional practical activity.

Tacit knowledge is dynamic, nonquantitative, inherent to the practice, and know-how of people [70]. As a brand of organizational knowledge, it is something created in a complex interaction between people [71].

This intangible asset whose origin is directly related to the creative agents of organizations generates economic value [72] and can lead them to competitive advantages [73].

Leonard and Sensiper [74] linked tacit knowledge to innovation, when the capacity of the human mind (worker's expertise) creates new products, services, and processes. Other authors associate it to organizational performance [56, 75–78].

⁴Study conducted taking into account the period between 1958 and 2002, mapped out 149 different meanings for tacit knowledge, among them, skills, know-how, communities of practice (CoP), best practice and mental models.

³Insight is the result of a mental process of analyzing complex situations or problems, featuring accurate and deep understanding of them [52].

Tacit knowledge is like an intangible mental product, susceptible to sharing and in constant reformulation, comprised of abstract elements (ideas, reasonings, know-how, skills, allegories, models, constructs, insights, experiences, memories, sounds, images, smell, taste, touch, etc.) capable of generating new knowledge and innovative practices.

Considering the existence of this "available" asset, one must also note the importance of its sharing, given that this dynamic has its own peculiarities, such as that tacit knowledge is best shared through conversation [60], has a more noticeable fluidity in informal settings [61], and, in some organizations, its flow is neglected [79, 80].

Key points relating to TK	References	
Ability	[44]	
Cognitive process	[45, 49, 62, 64]	
Competitive advantage	[73]	
Experience, ideas, emotions, beliefs	[51, 53, 56, 62, 65, 66]	
Informality, imateriality, intangibility	[55, 61, 72]	
Innovation	[74]	
Interaction	[71]	
Organizational performance	[56, 75–78]	
Personal nature	[57–59]	
Practical activity, know-how	[50, 62, 69, 70]	
Process of learning	[2]	
Process relating	[47]	
Product of itself	[46]	
Sharing	[33, 60, 79, 80]	
Solution of problems	[56, 63]	
Spiral spinning	[8]	
Topic in agenda of firms	[54]	
Transferring	[48]	
Utility	[67]	

Table 2 below summarizes key concepts from this section and cited authors.

Table 2. Summary table of the literature and key points relating to tacit knowledge (TK).

5. Tacit knowledge: the invisible asset

Krogh [81] argues that TKS is the main vector when firms want to raise their own value. He highlights the (invisible) transit of experiences and practices of members in the organization

that can leverage knowledge creation, increase TKS, and, as result, provide a higher organizational value.

Bierly et al. [36] argue that the wisdom of each employee can be transformed into an organizational wisdom through knowledge sharing, and what Eraut [82] defines as a type of learning.

The essence of tacit knowledge is its sharing and not its control [83]. Any unilateral displacement from tacit knowledge, when only one party is benefited, is not characterized by an interactive action among people [84], and it does not constitute tacit knowledge transfer, which requires intense interpersonal contact [51]. When interviewing a contributor to the Sematech Company about sharing ideas, Davenport and Prusak [51] registered the comment: "We have documents, databases, intranet, web, groupware, whatever you can think of. But representatives and face to face meetings that we have are by far the most important transfer channels."

Taylor [48] in his book *Principles of Scientific Management*, although seeking a rational, precise, formal, and mathematical model, the best way, also stated the importance of ferreting out the most efficient knowledge, tacit [85], which was found lodged in craftsmen in order to transmit it to engineers, the current scientists and managers of knowledge. This flow has increased over the past decades [86] and positively affected the development of business innovation [87].

Understanding the nature of knowledge as being more tacit than explicit, as well as representing multidirectional and intangible inter-relations, can serve as a source of advantage for the organization, in the transfer, for example, of best practices and lessons learned [37, 83, 84, 88].

Transmission occurs most effectively when set amidst informal and unscheduled conversations, or through structured meetings and actions that enable the free movement of knowledge [67]. In "an economy governed by knowledge, to chat is to work" [51].

Vieira [89] contextualizes one of those moments when he quotes an experience seen as successful by the civil servants of Justice in the city of Casca.⁶ "Regular meetings: breakfast held in the dependencies of the Forum, with all civil servants and the Judge, followed by a meeting with a predefined agenda and open to suggestions from staff throughout the group, in order to have functional issues addressed (...) First they all enjoy a delicious breakfast with casual conversations and then the meeting is held. In these meetings solutions to improve judicial services are addressed" [89].

Regarding the public, Carvalho et al. [67] highlight that sharing may lead to a delay of organizational deterioration and an improvement in performance. Caution, however, must be taken in order that this phenomenon is not held down by bureaucratic assumptions of management or personal presumption to understand knowledge as individual property [27].

Van Caenegem [19] when dealing with a possible 'regulation' for sharing through a migration of people between companies, he draws attention to the fact that the socialization of knowledge is less of a normative protocol of relationship and more of a free act between people who have valuable knowledge and desire to share it. Leading organizations often encourage the use of tacit knowledge of their workforce [54], considering crucial its transfer within the organization [65].

⁶A town in the state of Rio Grande do Sul in Brazil.

Although little studied and understood in the Brazilian judiciary, TKS can increase the chances of diffusion of knowledge [90], especially where, "in general, there is no dialogue or exchange of experiences" [91]. Taal et al. [92], however, when examining the phenomenon in the European judiciary, notes that many judges declared that sharing is not a new activity and meetings, seminars, symposiums, and learning events provide the environment conducive to sharing. This phenomenon thus encourages the formulation of policies contrary to the permanence of outdated or dysfunctional routines.

The observation of the existence of a prior knowledge (good practices and lessons learned), for example, can prevent the occurrence of errors of repetition [93], as state judge reported: "In fact, we learn from situations as they arise, we have no strategy (...) errors may occur and occur frequently, hence the difficulty of getting it right from errors" [94]. Even from the perspective of administrative modernization, the transferability of knowledge and management practices gain in sharing a relevant factor and an increasingly intensified practice [95].

The importance of knowledge sharing by tacit means ("face to face" interactions) was reaffirmed by Stefanovitz [96] when dealing with innovation in product development processes. Von Hippel [97] reported that rapid transfers of knowledge actually increase a company's ability to benefit from its own know-how.

From the literature, aspects related to not sharing are also present. Amabile [98] argues that the immobility or inertia of tacit knowledge may be related to lack of motivation of the holder to share. Szulanski [99] analyzed, from the organization's level, its inability to create an atmosphere for TKS; and from the personal level, the lack of interest in the receiver and his/her low absorption capacity, as well as relationship problems with the transmitter. The author perceived an association between these aspects and a possible loss of a firm's ability to build competitive advantage. In communities of practice some causes were identified: fear of misuse; the possibility for personal attacks, and little objectivity concerning the shared knowledge [100].

When dealing with successful companies, Choo [101] pointed out that organizations that have demonstrated ability to adapt to constant change, innovate continuously, and to make decisions that lead toward their goals demonstrate ability to handle knowledge. He noted that an organization with a properly harnessed knowledge has a differentiated ability from others to perceive, discern, and act smart. Thus, the performance of an organization in the era of knowledge is associated with a continuous learning process, generating a model where creativity, innovation, and sharing of tacit knowledge are integrated and reveal themselves in the form of experience.

The United Nations (UN) in recognizing the importance of sharing, points out in its study on innovation in governance the importance of governments working together and better, and so, "knowledge sharing is the first step" [102]. Knowledge sharing can modify the installed mental baggage [103], allows for the creation of new knowledge [104], boosting the organization in its ability to innovate [105], and even grants a greater market value for the organization [106].

With impetus, sharing of tacit knowledge enables the formation of new ideas and knowledge, food for innovation [13]; this sharing being essential to the organization's intelligence [107].

As the whirlwind holds dust immobilizing it, the sharing phenomenon brings stimuli to the knowledge flow. Examples of this dynamic are: the conversation that transfers knowledge [51], the spiral structure [8], and "interpreting communities" [108], which provide new channels, spaces, and organizational arrangements to knowledge.

The sharing terminology encompasses other words such as transfer, distribution, communication, collaboration, diffusion, dissemination, allocation, network, cooperation [109], passing [106], interaction, exchange, and socializing. Because it is a growing challenge for most organizations [18] and because it provides for an opportunity of interpretations [35], there is a growing interest in studying the sharing process, even under the warning of being difficult to transfer [88].

Itami [110] addressing knowledge and the ability to pass it as an invisible organizational asset, points out that, although they are very important ingredients for the development of a successful strategy, they are difficult to measure. Christensen [111] denotes that there are few references in the literature and few companies that include sharing as a key component to increase organizational performance, due to the difficulty of measuring this phenomenon.

Since, it involves intangibles internalized in beliefs, experiences, and personal values, Inkpen [3] stated that the formalization of tacit knowledge becomes complex in certain contexts and difficult to share. Ruschel [90], while researching the Labor Court, found that, although precious, tacit assets were restricted only to the minds of the judges.

The results from the research on knowledge management in public administration [112] follow in the same direction when evaluating the intangible assets (intellectual property assets), and although available in the institutional environment, they are difficult to be qualified and measured. Alvarenga Neto [12] endorses such views, stressing that for being innate to human beings, one cannot share tacit knowledge with ease and spontaneity, however, it is suggested in his conclusion the "urgency to create a set of indicators for measuring the benefits of knowledge management." Likewise, Sveiby [68] supports the considerations of Alvarenga Neto [12]. Noting that a focused strategy on knowledge can be quantified if an organization seeks something beyond measuring monetary flows, but also flows related to intangible assets such as knowledge.

Longanezi et al. [113] presented some nonnumeric indicators based on idea generation: brainstorming sessions, specific training, number of submitted ideas, and the identification of best practices. Another tool that according to the authors encompasses a continuous measuring process is benchmarking. Regarding this topic, however, they pointed out a limitation that restricts its use, "the unwillingness of leaders to share real relevant indicators" [113].

The TKS concept proposed in this chapter is an interpersonal exchange phenomenon describable and measurable. Represents the interaction of tacit knowledge flows established between people. Can occur within organizations or among them, requires a social environment of trust, and promotes the incorporation of new content with multilateral gains of aggregating nature [114].

The following Table 3 shows some key points concerning TKS and their respective references.

Key points relating to TKS	References
Communities of practices (CoP)	[100, 108]
Competitive advantage	[99]
Crucial	[54, 65]
Difficulty relating to TKS	[12, 88, 110–112]
Firms value	[81, 106]
Indicator to TKS	[12, 68, 113]
Individual property	[27, 90]
Informality	[3, 51, 67, 89]
Innovation	[13, 86, 87, 101–105]
Interactive action, socialization	[19, 48, 51, 84, 96, 114]
Learning	[37, 82–84, 88]
Management practices	[95]
Motivation	[98]
Organization intelligence	[107]
Organizational deterioration, performance	[67, 97]
Organizational wisdom	[36, 90]
Other words to TKS	[106, 109]
Prevention of errors	[93, 94]
Sharing as essence of TK	[83, 90–92]
Spiral structure	[8]

Table 3. Summary table of the literature and key points relating to tacit knowledge sharing (TKS).

6. Some models that have TKS as its core

Ikujiro Nonaka, Japanese organizational theorist, in contextualizing the creation of knowledge in an organizational environment, understood that the interaction between tacit and explicit knowledge was the main dynamic generator of knowledge. He conceptualized tacit knowledge as something difficult to visualize and express, particularly in regards to the actions and experiences of the individual, as well as emotions, values, or ideals. His model gives emphasis to personal interaction, where the key is in the sharing of knowledge [8].

Sveiby [68] introduces his model dealing with visible and invisible knowledge structures. He details among the intangible goods (assets), the competence of employees, customer relationships, brands, management mode, knowledge, and experience. He defines some organizations as 'organizations of knowledge,' featuring its employees as 'knowledge workers.'

In these organizations, intangible assets have more value than its tangible ones, knowledge gains value when it is shared and those that take action in the TKS process end up having an increase in their stock of knowledge through interaction.

Leonard and Sensiper [74] highlighted that collaborators' individual and collective expertise is the source of innovation and a competitive base for most organizations They suggested three applications to the tacit knowledge to leverage the innovation (problem solving, problem finding, prediction, and anticipation future occurrences). They stressed that creative ideas "do not arise spontaneously from the air," but they are originated by a mental process by social interactions where the tacit knowledge are sorted, grouped, matched, and melted. Still, they emphasized that new solutions are rarely constructed in an isolated fashion and that cooperative productions are essential to organizational development.

Choo [101] states that the construction of knowledge is achieved when one recognizes the synergistic relationship between tacit and explicit knowledge, and organizations need to learn how to convert tacit knowledge into explicit knowledge, becoming capable of promoting innovation and the development of new products. For Choo [101], there are four ways of converting knowledge: through a process of socialization (tacit knowledge into tacit), in other words, allowing the acquisition of tacit knowledge through the sharing of experiences; through externalization (tacit knowledge into explicit) a process in which tacit knowledge is translated into explicit concepts through the use of metaphors, analogies, and models; by combination (what is the word that the author used? Combination, blending, mixing, mixture, matching, hybrid... choose whichever is original) (explicit knowledge into explicit), a process that builds explicit knowledge through the gathering of explicit knowledge from various sources; through internalization (explicit knowledge into tacit) a process in which the experiences acquired in other ways of building knowledge are internalized by the individual in the form of mental models or common work routines.

Davenport and Prusak [51] stressed the importance for knowledge-based organizations to adopt sharing as one of the essential elements. They alerted to the fact that the existence of large numbers of individual tacit knowledge does not necessarily lead the organization to use this knowledge. The importance of organizations turning their attention to the spontaneous and unstructured sharing of knowledge strengthens and encourages the emergence of social movements in favor of TKS. Regarding the inherent informality of sharing, they stressed that conversations around water coolers or in restaurants are sharing opportunities and to designate these moments as a 'waste of time,' is part of an outdated theory on the nature of work. They stress, however, that it would not be a sensible attitude to wait around the breakroom for an answer to a specific knowledge, but, they reaffirm that 'spontaneous meetings of minds' (unstructured transfers) have demonstrated a capacity to generate new ideas or solve old problems unexpectedly. In the culture of knowledge transfer, the authors pointed out, some friction can slow or even block the communication flow:

- Lack of mutual trust;
- Lack of time and of places to meet;

- 'Not invented here' syndrome'⁷;
- Intolerance toward mistakes or for needing assistance.

Miranda and Simeão's model [108] proposes the existence of elements, not so discernible (such as, technological communication systems), but of high value for the transfer of knowledge, such as, "a set of knowledge and practices"; the practice of "Granting specific know-how," or the sharing of "how-to-do". The authors point out to the irrevocability of human interaction in the transfer of tacit knowledge, setting up an 'extensive model of communication.' They argue that in TKS, the process occurs in an interpersonal way, developing increasingly complex structures, such as communities of practice and invisible colleges.

Tonet and Paz [106] consider the knowledge sharing process as composed of four phases: initiation, which are actions that seek to identify and analyze the needs of knowledge, the existing opportunities to employ different or new knowledge, and the tracking of sources where this knowledge can be found; implementation, actions that seek to promote integration between sources and recipients of knowledge; support, actions that seek to create opportunities to exercise shared knowledge and that promote the guidance of its practice with the objective of assimilating knowledge and the development of the necessary skills for a proficient use; and incorporation, actions aimed at making that shared knowledge can flow freely among those who should use it.

7. Conclusion

Knowledge has been occupying the center of studies and research for some time, and this chapter seeks to contribute to the depth of this collection and aims to offer articles that deal with tacit knowledge and the sharing of that knowledge.

This work aimed to bring a review of the literature regarding TKS and to contribute with some definitions related to the topic. The observed understanding with this literature review is that organizations, and to a lesser degree, The Small Claims Federal Courts, are moving from a traditional model that is attached to physical and tangible assets to another, where knowledge, especially tacit, is treated as an organizational asset that, despite being immaterial, becomes valuable and a generator of wealth.

In this trajectory and encouraged by initial questioning, this chapter depicts the search for different references in which the sharing of tacit knowledge was related to a range of organizational vectors covering definitions that go from understanding it as an innovation factor to a strategic component for leveraging organizational performance. Seen by some as difficult to transfer, tacit knowledge is accepted by others as a component of communication flows, where dialogues, social networks, communities of practice, invisible colleges, and even 'TKS

^{&#}x27;In the literature 'not invented here' (nih) is described as an idea or innovation that was not developed within the company's workplace environment or that was not conducted by their own internal members, so the solution does not have the 'necessary characteristics or identity of the organization' and thus should not be considered.

points' cease to be mere expectations and become powerful channels and knowledge transmission arrangements. Although organizations find themselves inserted into a knowledge society, increasingly tacit, an unequal measure of attention seems to be given to information to the detriment of tacit knowledge.

Author details

José Roberto Ferretti^{1*} and Maria da Conceição L. Afonso²

*Address all correspondence to: ferretti.ci@gmail.com

1 Brazilian Federal Court-1st Region, Brazil

2 National Confederation of Industry Brazil, Brazil

References

- [1] Grant RM. The resource-based theory of competitive advantage: Implication for strategy formulation. In: Zack MH, editor. Knowledge and Strategy. New York: Routledge; 2011
- [2] Organization for Economic Cooperation and Development. Knowledge Management in the Learning Society: Education and Skills. Paris: OECD; 2000. 250 p
- [3] Inkpen AC. Creating knowledge through collaboration. California Management Reviews. 1999;**39**(1)
- [4] Lalande A. Vocabulário técnico e crítico da Filosofia. 3rd ed. São Paulo: Martins Fontes; 1999
- [5] Abbranano N. Dicionário de Filosofia. 5th ed. São Paulo: Martins Fontes; 2007
- [6] Nissen M, Levitt R. Dynamic Models of Knowledge-Flow Dynamics. Stanford: CIFE; 2002
- [7] Borgatti SP, Li X. On social network analysis in a supply chain context. Journal of Supply Chain Management. 2009;45(2):5-22
- [8] Nonaka I, Takeuchi H. Criação de conhecimento na empresa. Rio de Janeiro: Elsevier; 1997
- [9] Wolf MJP. Abstracting Reality: Art, Communication, and Cognition in the Digital Age. Lanham: University Press of America; 2000
- [10] Brody F. The medium is the memory. In: Lunegeld P, editor. The Digital Dialectic: New Essays on New Media. Cambridge: MIT Press; 1999
- [11] Lastres MM, Albagli S. Informação e globalização na era do conhecimento. Rio de Janeiro: Campus; 1999

- [12] Alvarenga Neto RCD. Gestão do conhecimento ou gestão de organizações da era do conhecimento? Um ensaio teórico-prático a partir de intervenções na realidade brasileira. Perspectivas em Ciência da Informação. 2007;12(1):5-24
- [13] Quéré M. On the Importance of Innovation Policy in Europe: A Micro-Macro Discussion About the Transition Towards a Knowledge-Based Society [Internet]. 2003. Available from: http://migre.me/t8Hko [Accessed: 20-09-2013]
- [14] Von Krogh G, Ichijo K, Nonaka I. Enabling Knowledge Creation: How to Unlock the Mistery of Tacit Knowledge and Release the Power of Innovation. New York: Oxford University Press; 2000
- [15] Balboni MR, Guerra M, Cristancho C, Sánchez M. Indicadores para la sociedad de la información en América Latina y el Caribe: avances y desafíos en la medición del acceso y uso de las tecnologías de información y comunicación. In: Proceedings of the Congreso Internacional del CLAD sobre la Reforma del Estado y de la Administración Pública; 2009; Salvador. Salvador: CLAD; 2009
- [16] Asian Productivity Organization (APO). Knowledge Management for the Public Sector. Japan: APO; 2013. 85 p
- [17] Castells M. Rise of the Network Society: The Information Age: Economy, Society and Culture. Oxford: Blackwell Publishing Ltd; 1999
- [18] Smith HA, Mckeen JD, Singh S. Tacit knowledge: Making it happen. Journal of Information Science and Technology. 2007;4(2)
- [19] Van Caenegem W. Inter-firm migration of tacit knowledge: Law and policy. Prometheus. 2005;23(3)
- [20] Ruschel AJ. Modelo de conhecimento para apoio ao juiz na fase processual trabalhista [thesis]. Florianópolis: Universidade Federal de Santa Catarina; 2012
- [21] Casselman RM, Samson D. Internal, collaborative and competitive knowledge capability. In: System Sciences (HICSS), Hawaii International Conference; 2011; Kauai. Kauai: HICSS; 2011
- [22] Gomes MVD. A Gestão do Conhecimento na Administração Pública: paradigma para o Judiciário fluminense [dissertation]. Rio de Janeiro: Fundação Getúlio Vargas; 2005. 54 p. Available from: http://www.tjrj.jus.br/c/document_library/get_file?uid=61cb1315-8247-454f-87b1-b04feba55775
- [23] Pires VS. Cultura organizacional e memória empresarial na indústria da moda: gestão de ativos do conhecimento como estratégia corporativa para a construção de identidades de marcas. E-COM. 2012;5(2):1-15
- [24] Cross J. Informal Learning. San Francisco: Pfeiffer; 2007
- [25] Alvares LMA. Telecentros de informação e negócios como veículo de educação corporativa nas microempresas e empresas de pequeno porte [dissertation]. Brasilia: Universidade de Brasília; 2010

- [26] Girard JP, McIntyre S. Knowledge management modeling in public sector organizations: A case study. International Journal of Public Sector Management. 2010;23(1):71-77
- [27] Kerckhove D. The Information Management Round Table: Connected. Information Management Round Table. 1999;12(May):1-4
- [28] Jelenic D. The importance of knowledge management in organizations with emphasis on the balanced scorecard learning and growth perspective. In: Management Knowledge and Learning. Celje: Make Learn; 2011. pp. 33-43
- [29] Peters T. Prosperando no caos. São Paulo: Harbra; 1989
- [30] Hair JF Jr, Black WC, Babin BJ, Aanderson RE, Tatham RL. Análise Multivariada de Dados. Porto Alegre: Bookman; 2009
- [31] Penrose E. A teoria do crescimento da firma. Campinas: Unicamp; 2006
- [32] Alvarenga Neto RCD, Souza RR, Neves JTR, Barbosa RR. Strategic knowledge management: In search of a knowledge-based organizational model. Comportamento Organizacional e Gestão. 1993;14(2):247-256
- [33] Polanyi M. The Tacit Dimension. Chicago: University of Chicago Press; 2009
- [34] Casini I. A evolução da Big Frango no contexto da avicultura [dissertation]. Curitiba: Universidade Federal do Paraná (UFPR); 2009. 107 p. Available from: http://acervodigital.ufpr.br/handle/1884/17934
- [35] Peteraf MA. The cornerstones of competitive advantage: A resource-based view. Strategic Management Journal. 1993;14(3):179-191
- [36] Bierly PE, Kessler EH, Christensen EW. Organizational learning, knowledge and wisdom. Journal of Organizational Change Management. 2000;13(6):595-618
- [37] Porter ME. Vantagem competitiva: criando e sustentando um desempenho superior. Rio de Janeiro: Campus; 1989
- [38] Porter ME. Competitive Strategy. New York: Free Press; 1980
- [39] Barney J. Firm resources and sustained competitive advantage. Journal of Management. 1991;17(1)
- [40] Grant RM. Toward a knowledge-based theory of the firm. Strategic Management Journal. 1996;17(S2):109-122
- [41] Zack MH. Knowledge and Strategy. New York: Routledge; 2011
- [42] Kogut B, Zander U. Knowledge of the firm and the evolutionary theory of the multinational corporation. Journal of International Business Studies. 1993;24(4):625-645
- [43] Kogut B, Zander U. What Firms Do? Coordination, Identity, and Learning. Organization Science. 1996;7(5):502-518. DOI: 10.1287/orsc.7.5.502
- [44] Associação para a Promoção e Desenvolvimento da Sociedade de Informação. Glossário da Sociedade da Informação. Caparica: APDSI; 2007. 150 p

- [45] Lima JLO, Alvares L. Organização e representação da informação e do conhecimento. In: Alvares L, editor. Organização da Informação e do Conhecimento. São Paulo: B4 Ed.; 2012. pp. 21-34
- [46] Kofman F. Metamanagement: o sucesso além do sucesso. Rio de Janeiro: Elsevier; 2004
- [47] Burke P. Uma história social do conhecimento: de Gutenberg a Diderot. Rio de Janeiro: Jorge Zahar Ed.; 2003
- [48] Taylor FW. Princípios de administração científica. 8th ed. São Paulo: Atlas; 1990
- [49] Stewart T, Ruckdeschel C. Intellectual capital: The new wealth of organizations. Performance Improvement. 1998;37(7):56-59
- [50] Haldin-Herrgard T. Diving Under the Surface of Tacit Knowledge [Internet]. [Updated: 2004]. Available from: http://www.coalescentknowledge.com/WP/diving.pdf [Accessed: 05-08-2016]
- [51] Davenport T, Prusak L. Conhecimento Empresarial. Rio de Janeiro: Campus; 1999
- [52] Collins W. Cobuild English Dictionary. London: HarperCollins; 1998
- [53] Leite FCL. O conhecimento científico tácito na dinâmica da pesquisa: alguns indícios. DataGramaZero. 2007;8(3)
- [54] Krishnaveni R, Sujatha R. Communities of practice: An influencing factor for effective knowledge transfer in organizations. The IUP Journal of Knowledge Management. 2012;10(1)
- [55] Knockaert M, Ucbasaran D, Wright M, Clarysse B. The relationship between knowledge transfer, top management team composition, and performance: The case of Science-Based Entrepreneurial Firms. Entrepreneurship Theory and Practice. 2011;35(4):777-803
- [56] Korhonen M. Tacit Knowledge and Weak Signals in Organizational Learning: Managing Knowledge to Create Sustainable Competitive Advantage Case Company Ltd [dissertation]. Finland: Turku University of Applied Sciences; 2014
- [57] Nonaka I, Konno N. The concept of "Ba": Building a foundation for knowledge creation. California Management Review. 1998;40(3)
- [58] Oliveira VP. O conhecimento tácito na transferência de conhecimento científico: Mr. Data aprende a dançar. Campinas: DPCT/IG/Unicamp; 2000
- [59] Maia VLH. A gestão do conhecimento integrada nos processos empresariais [dissertation]. Porto: Instituto Superior de Contabilidade e Administração do Porto; 2001
- [60] Goh SC. Managing effective knowledge transfer: An integrative framework and some practice implications. Journal of Knowledge Management. 2002;6(1):23-30
- [61] Oliveira VP. Uma informação tácita. DataGramaZero. 2005;6(3)
- [62] Alavi M, Leidner DE. Knowledge management and knowledge management systems: Conceptual foundations and research issues. Mis Quaterly. 2001;25(1)

- [63] Smith EA. The role of tacit and explicit knowledge in the workplace. Journal of Knowledge Management. 2001;5(4):311-321
- [64] Edivinsson L, Malone MS. El capital intelectual: cómo identificar y calcular el valor de los recursos intangibles de su empresa. Barcelona: Ediciones Gestión 2000 S. A.; 1999
- [65] Lemos B, Joia L. Fatores relevantes à transferência de conhecimento tácito em organizações: um estudo exploratório. Gestão & Produção. 2012;19(2):233-246
- [66] Saint-Onge H. Tacit knowledge: The key to the strategic alignment of intellectual capital. In: Zack MH, editor. Knowledge and Strategy. New York: Routledge; 2011. pp. 223-230
- [67] Carvalho ARS, Mascarenhas CC, Oliveira EAAQ. Ferramentas de disseminação do conhecimento em uma instituição de C, T & I de defesa nacional. Revista de Gestão da Tecnologia e Sistemas de Informação. 2006;3(2):77-92
- [68] Sveiby KE. The New Organizational Wealth: Managing and Measuring Knowledge-Based Assets. San Francisco: Berrett-Koehler Publishers Inc.; 1997
- [69] Martins LMP, Lima MF, Pessanha VLR. Gestão do conhecimento como instrumento de modernização do Poder Judiciário Fluminense [dissertation]. Rio de Janeiro: Universidade Federal do Rio de Janeiro; 2005
- [70] Stapleton L, Smith D, Murphy F. Systems engineering methodologies, tacit knowledge and communities of practice. AI & Soc. 2005;19(2):159-180
- [71] Patokorpi E. Low knowledge in cyberspace: Abduction, tacit knowledge, aura and the mobility of knowledge. Human Systems Management. 2006;25(3):211-220
- [72] Rezende Y. Informação para negócios: os novos agentes do conhecimento e a gestão do capital intelectual. Ciência da Informação. 2002;31(2):120-128
- [73] Razouk AA., Bayad M, Wannenmacher D. Strategic human resource management and tacit knowledge transfer: A case study. Human Systems Management. 2009;28(1-2):77-82
- [74] Leonard D, Sensiper S. The role of tacit knowledge in group innovation. California Management Review. 1998;40(3)
- [75] Dyer JH, Nobeoka K. Creating and Managing a High Performance Knowledge-Sharing Network: The Toyota Case [Internet]. 1998. Available from: http://dspace.mit.edu/bitstream/handle/1721.1/1441/147b.pdf [Accessed: 21-07-2016]
- [76] Decarolis DM, Deeds DL. The impact of stocks and flows of organizational knowledge on firm performance: An empirical investigation of the biotechnology industry. Strategic Management Journal. 1999;20(10):953-968
- [77] Hsu I. Knowledge sharing practices as a facilitating factor for improving organizational performance through human capital: A preliminary test. Expert Systems with Applications. 2008;**35**(3):1316-1326
- [78] Shoukat A. Investigating the correlation between knowledge assetmanagement and organizational performance. In: International Conference on Business and Economic Research; 2012; Bandung. Bandung: ICBER; 2012

- [79] Lee JCY, Nissen ME. Accelerating acculturation through tacit knowledge flows: Refining a grounded theory model. VINE Journal of Information and Knowledge Management Systems. 1971;40(3/4)
- [80] Mungai GCN. Tacit Knowledge Management in Public Institutions in Kenya: A Case of the Kenya Institute for Public Policy Research and Analysis (Kippra) Nairobi [dissertation]. Nairobi: KCA University; 2011. 82 p
- [81] Krogh G. Care in knowledge creation. California Management Review. 1998;40(3):133
- [82] Eraut M. Non-formal learning and tacit knowledge in professional work. British Journal of Educational Psychology. 2000;70(1):113-136. DOI: 10.1348/000709900158001
- [83] Chesher C, Howard S. Balancing knowledge management and knowledge mobility in the Connected University. In: Fenwick T, Farrell L, editors. Knowledge Mobilization and Educational Research Politics, Languages and Responsabilities. New York: Routledge; 2011
- [84] Bonache J. El valor estratégico de la movilidad global de puestos, personas y conocimientos em las multinacionales. Universia Business Review. 2010;3(Tercer trimester):14-29
- [85] Marsick VJ. Toward a unifying framework to support informal learning theory, research and practice. Journal of Workplace Learning. 2009;21(4)
- [86] Gassman O. Opening up the innovation process: Towards an agenda. R & D Management. 2006;36(3)
- [87] Liu CLE. Knowledge mobility in cross-border buyer-supplier relationships. Management International Review. 2012;**52**(2):275-291
- [88] Fang Y, Wade M, Delios A, Beamish PW. International diversification, subsidiary performance, and the mobility of knowledge resources. Strategic Management Journal. 2007;28(1053)
- [89] Vieira JLL. Um novo desafio para o Judiciário: o juiz líder. Porto Alegre: Tribunal de Justiça do Estado do Rio Grande do Sul; 2009
- [90] Ruschel AJ. Modelo de conhecimento para apoio ao juiz na fase processual trabalhista [thesis]. Florianópolis: Universidade Federal de Santa Catarina; 2012. 206 p
- [91] Instituto de Pesquisa Econômica Aplicada (IPEA). Acesso à Justiça Federal: dez anos de juizados especiais. Brasília: CJF; 2012
- [92] Taal S, Velde M, Langbroek P. Reducing Unwarranted Disparities: The Challenge of Managing Knowledge Sharing Between Judges. International Journal For Court Administration. 2014;6(2)
- [93] Serpell A. Administración de operaciones de construcion. Santiago: Universidad Catolica; 1993
- [94] Santos CMD, Castilho IV, Kilimnik ZMO. Juiz de Direito como gestor: competências necessárias para uma difícil conciliação de papéis. Revista Gestão & Tecnologia. 2006;6(2):1-15
- [95] Motta PR. Revisitando a controvérsia sobre a transmissibilidade intercultural do conhecimento administrativo. Cadernos Ebape. BR. 2014;**12**(3)
- [96] Stefanovitz JP. Criação de conhecimento e inovação na indústria de alta tecnologia: estudo e análise de casos em uma empresa do setor de automação industrial [dissertation]. São Carlos: Universidade de São Paulo (USP); 2006
- [97] Von Hippel E. The sources of Innovation. New York: Oxford University Press; 1988
- [98] Amabile TM. How to kill creativity. Harvard Business Review. 1998;7(5):76-87
- [99] Szulanski G. Sticky Knowledge: Barriers to Knowing in the Firm. Thousand Oaks: Sage Publications Inc.; 2003
- [100] Ardichvili A, Page V, Wentling T. Motivation and barriers to participation in virtual knowledge-sharing communities of practice. Journal of Knowledge Management. 2003;7(1)
- [101] Choo CW. A organização do conhecimento: como as organizações usam a informação para criar significado, construir conhecimento e tomar decisões. 3rd ed. São Paulo: Senac; 2003
- [102] Alberti A, Bertucci G. Innovation in governance and public administration: Key issues and perspective. In: United Nations, editor. Innovations in Governance in the Middle East, North Africa, and Western Balkans: Making Governments Work Better in the Mediterranean Region. New York: Department of Economic and Social Affairs; 2007
- [103] Tomaél MI, Alcará AR, Di Chiara IG. Das redes sociais à inovação. Ci. Inf. 2005; 34(2):93-104
- [104] Nonaka I, Toyama R. The knowledge-creating theory revisited: Knowledge creation as a synthesizing process. Knowledge Management Research & Practice. 2003;1(1):2-10
- [105] Wilbert JKW, Mallmann ML, Schmitz A, Souza JA. Comunidades de prática virtuais nos processos de inovação: aspectos e recomendações. In: Congresso Brasileiro de Engenharia de Produção; 2015; Ponta Grossa. Ponta Grossa: ConBRepro; 2015
- [106] Tonet HC, Paz MGT. Um modelo para o compartilhamento de conhecimento no trabalho. RAC. 2006;**10**(2):75-94
- [107] Valentim MLP. O processo de inteligência competitiva em organizações. DataGrama-Zero. 2003;4(3)
- [108] Miranda A, Simeão E. Transferência de informação e transferência de tecnologia no modelo de Comunicação Extensiva: a Babel.com. Información, Cultura y Sociedad. 2004;10:pp. 27-40
- [109] Heisig P. Harmonisation of knowledge management: Comparing 160 KM frameworks around the globe. Journal of Knowledge Management. 2009;13(4):4-31
- [110] Itami H. Mobilizing Invisible Assets. Cambridge: Harvard University Press; 1987
- [111] Christensen PH. Knowledge sharing: Moving away from the obsession with best practice. Journal of Knowledge Management. 2007;11(1):36-47

- [112] Instituto de Pesquisa Econômica Aplicada (IPEA). Gestão do conhecimento na administração pública: resultados da pesquisa IPEA 2014: práticas de gestão do conhecimento. Brasilia: IPEA; 2015
- [113] Longanezi T, Coutinho P, Bomtempo JVM. Um modelo referencial para a prática da inovação. Journal of Technology Management & Innovation. 2008;3(1)
- [114] Ferretti JRP. Variáveis do compartilhamento de conhecimento tácito nos Juizados Especiais Federais do Brasil: identificação e mensuração [thesis]. Brasilia: Universidade de Brasília; 2016. 200 p

Knowledge Sharing and Human Resource Development in Innovative Organizations

Michael Beyerlein, Rachele Collins, Shinhee Jeong, Christi Phillips, Suravee Sunalai and Lei Xie

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/intechopen.70067

Abstract

Global competition and rapid changes in technology demand more innovation in organizations. Such an increase in innovation depends on developing the capabilities of employees and providing them with knowledge management support that accelerates learning and discovery. Leading companies have been creating work environments that emphasize learning and knowledge management since the turn of the century. This chapter presents profiles of practices in five such companies and draws conclusions that result in a model that ties human resource development (HRD) practices to knowledge management practices as a guide for other organizations.

Keywords: innovation, human resource development, knowledge management, knowledge sharing, learning and development, collaboration, communities of practice

1. Introduction

Innovation is a key driver of economic growth. New and improved products, processes, and services account for the bulk of trade and industry advancement since the Industrial Revolution. Technology innovations receive a lot of the credit for economic growth. However, modifications in organizational design, infrastructure, policy, and other factors in the organizational ecosystems also deserve significant credit. All these changes depend on generation of new knowledge and its application. In recognition of the central role of knowledge in organization performance, the field of knowledge management (KM) emerged as a focus on improved understanding of the processes and their links to organizational performance



© 2017 The Author(s). Licensee InTech. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. in modern organizations [1, 2]. Simultaneously, the field of human resource development (HRD) emerged with a focus on improved understanding of the processes of learning and development (L&D) in organizations and their links to performance [3]. KM and HRD seem to intertwine naturally in their approaches to organizational effectiveness.

HRD processes change the competencies of employees at all levels in an organization and thereby play major roles in driving innovation and economic growth. People competencies and the way those competencies are leveraged across the organization due to changes in organizational design occur largely because of innovations in the HR, human resource management (HRM), and human resource development (HRD) departments—the people development functions in the organization. In that trio of special functions, HRD focuses on learning and development at the individual, team, and organizational levels. Innovation inside an organization occurs when employees have the knowledge, skills, and attitudes for learning, taking thoughtful risks, sharing knowledge, and thinking creatively, and the organization's systems provide needed support [4]. Ishak et al. [5] suggest KM is essential for developing the human resource. We further suggest HRD is essential for growing the knowledge that flows through the KM system.

1.1. Applicable literature

Although many definitions of innovation have been published, we will select the following: innovation is executing new ideas to create value [6]. Crossan and Apaydin [7] refer to innovation as both a process and an outcome. Using a complex adaptive systems framework, Harkema [8] linked learning and knowledge generation when he defined innovation as a knowledge process for generating new knowledge, so it drives incremental or radical change.

This chapter focuses on KM and HRD practices that enable innovation within some of the world's most innovative companies. Examples from these innovative organizations will facilitate an understanding of the way enablers and critical success factors of innovation, KM, and learning are implemented for competitive advantage.

A learning organization consists of formal and informal systems that facilitate learning by its members and rapid change in adapting to the environment—a culture where learning drives proactive change. Knowledge sharing benefits from developments in any part of the learning organization that improves the flow of knowledge. For example, HRD work in the organization can build up the communication-related competencies of employees, upgrade organizational design to reduce silo effects, and develop employees' ability to effectively utilize technology, all of which contribute to knowledge-sharing capability within the organization. HRD plays a pivotal role in leading innovative activities with responsive guidance [9]. HRD practitioners increase learning effectiveness from individual, group/team, and organizational levels. Through continuous learning and improving, HRD fosters a culture that favors innovation [10]. In addition, HRD offers diverse solutions to explore challenges by analyzing the organization from the organization development standpoint [11]. Finally, HRD creates reward mechanisms that motivate employees to challenge the status quo and embrace change.

We view KM as "an iterative process of handling actionable knowledge that results from individual, group, and organizational learning to improve organizational performance" [12]. In this context, knowledge consists of individuals' insights, experiences, know-how, and values that are justified through social interactions among participants [13] to make knowledge actionable and embedded in organizational repositories [14]. Consequently, KM functions so organizations can use their actionable knowledge effectively to improve their performance [12]. Generally, HRD can contribute to innovation by promoting organizational activities that better manage the flow of knowledge (i.e., to acquire, create, store, share, use, and assess) throughout an organization [15].

Formal, informal, and incidental learning drives innovation in industry [16–18]. Practitioners in HRD develop learning systems that include an emphasis on career development, training and development, and organizational development. The field of HRD has been changing rapidly over the past two decades and finds itself in a special role in companies of all sizes that make its tools and processes highly valuable in the rapidly changing organizations of the twenty-first century.

In this chapter, we profile five organizations where HRD provides the kind of support that makes learning and development central to the strategic development of knowledge-based organizations. Based on interviews of HRD staff members in these organizations and a study of their Websites, publications, and company reports, we present ways in which HRD creates conditions for enabling knowledge sharing and learning and the processes that enable innovation within these organizations. The five innovative organizations discussed in this chapter include Air Liquide based in France, Krungthai Card Public Company Limited (KTC) in Thailand, and Rockwell Collins, FMC Technologies Inc. (FMC Technologies), and Praxair in the USA (**Table 1**). This sample of innovative organizations was intentionally selected to represent a diverse set of countries and industries as a way of illustrating ways that HRD can promote learning and change inside organizations that stimulates their ability to innovate.

Sustainable organizations are complex, adaptive systems [19]. The challenges that organizations and their employees face today in turbulent market conditions demand an internal environment of continuous learning and knowledge creation, transfer, and sharing as keys to adaptation and innovation and a culture that stimulates a risk-taking mindset. The interrelated processes of learning and knowledge creation occur within a social and a cognitive system embedded in a hierarchy of complex systems. Many hurdles to effective learning and knowledge creation, transfer, and sharing in those systems impede the processes that promote problem solving and utilization of process outputs. Few aspects of the work environment effectively promote those processes. HRD professionals develop the required human capital and create a work environment that unleashes and leverages its human capital for continuous innovation. The same types of HRD tools and methods may be at use in several organizations, but their application depends on the local culture. The next section presents five profiles of innovative organizations using tools and methods of HRD to create more effective knowledge management systems (KMSs). The profiles are necessarily brief and so provide only a small slice of the dynamic processes of human capital development in these companies.

	Praxair		Rockwell Collins	FMC Technologies Inc.	Krungthai Card PCL.	
Industry	Industrial gas & chemicals	Industrial gas & Avionics & IT chemicals systems		Oil & gas	Consumer finance	
Country	USA	France	USA	USA	Thailand	
Year founded	1907	1902 2001		1928	1996	
Core values	 Safety; Governance; Energy and climate change; Eco portfolio; Sustainable productivity; People development 	Rely on operational excellence, selective investments, open innovation, a network organization	Access, connect, transcend	 Safety & sustainability; Collaboration; Innovation; Valuing people; Integrity; Customercentered; Quality 	 Modern; Dynamic; Professional; Simple; Fun 	
Corporate cultural areas of focus	Diversity, employee, community, and stakeholder engagement	Innovation, new product development; expand and develop new markets	Accelerate knowledge, create value enterprise-wide	Living corporate values; put customers and clients first	Mission-based projects; employee satisfaction	

Table 1. General company characteristics.

The companies were selected from the Fortune 500 list but also had to be recognized for their innovativeness. They represent a variety of industries and locations.

2. Case studies

Five case studies of twenty-first century organizations are presented below to illustrate how they have found ways to use HRD practices to improve innovation capability in their organizations. Each case shows unique features as the search for ways to increase innovation capability is tailored to the business context.

2.1. Praxair

Praxair, a Fortune 500 company based in the USA, was selected because of its ranking by *Forbes Magazine* on the list of the "World's Most Innovative Companies" [20] and the Achievers 50 Most Engaged Workplaces[™] in North America for 2016 [21]. Praxair is one of the largest industrial gas companies in the world. The organization's primary products are atmospheric and process gases for industrial customers in approximately 50 countries.

The vision for Praxair is "to be the best performing industrial gases company in the world" [21]. This performance culture is driven and reinforced by six priority factors: (1) safety; (2) governance, ethics and compliance; (3) energy and climate change; (4) eco portfolio (environmental innovation and product stewardship); (5) sustainable productivity; and (6) people development. The sixth priority involves four key performance indicators: diversity, employee engagement, community engagement, and stakeholder engagement. Additionally, Praxair's mission is "making our planet more productive," as the company serves customers and the planet for economic and environmental improvements. Success with that mission is reflected in Praxair's consistent placement on the Dow Jones Sustainability World Index for 13 consecutive years [22].

Development of human capital inside Praxair has involved continuous investment for decades. To create a culture based on the core values and achieve performance goals, senior leadership decided to have HR play a more strategic partnership role in 1999 [23]. The mandate for HR was to

- 1. develop the next generation of Praxair leadership,
- 2. accelerate employee development and new talent acquisition, and
- 3. improve integration of business and HR activities.

These three goals for HR operationalized the vision of the firm as it entered the twenty-first century.

The HR team chose eight avenues to pursue these goals:

- 1. Promote diversity and the core of talent management [24].
- 2. Create a global mindset.
- **3.** Develop leadership at all levels of the organization—"blended executives" as professionals with subject matter expertise [25].
- 4. Build a high level of employee engagement.
- **5.** Align HR process with the business via rewards and recognition, goal setting, and team design.
- **6.** Warehouse HR and customer information for data analytics to drive intelligent decision making [26].
- 7. Develop a more effective knowledge management system.
- 8. Create a more comprehensive approach to training and development.

Early success with the Praxair program was reflected by the inclusion of the company as the focus of one of 17 chapters in Carter et al.'s [27] book *Best Practices in Leadership Development and Organization Change* which profiled leading innovators in leadership and organization development. The programs for HRD have continuously evolved at Praxair as exemplified by the following sample of topics.

2.1.1. Job autonomy

Based on clear communications around corporate vision, goals, and mission, Praxair gives enough autonomy to all business units to run their own business. Compared to other companies in the chemical industry, Praxair has fewer controllers, commanders, or instructors who regulate or monitor employees' job-related behaviors. Praxair encourages employees to be creative thinkers, to accept and explore different or new perspectives to solve problems, and to expand conceptual frameworks. The flexibility and the absence of rigid job design can be also exemplified with these five principles representing careers at Praxair: "explore your possibilities; build your skills; enjoy your challenge; make your mark; and own your future" [21].

2.1.2. Learning and development

Learning and development (L&D) provides formally structured and informal learning opportunities to facilitate employees pushing the limits of knowledge and experience. Praxair regards every aspect of employees' actual job performance as a part of L&D. A special emphasis has placed knowledge sharing through mentoring at the center of the L&D process. The social support of learning emerges from mentoring relationships. Mentoring of women and by women leaders represents an important part of that emphasis. Praxair formally trains technical personnel in mentoring skills to accelerate the sharing of both technical and soft skills knowledge. The rapid development of new hires in both hard and soft skills prepares them for filling the gap created by retiring senior employees through mentoring relationships. Mid-career employees with advanced degrees may leave for other opportunities, so Praxair uses job shadowing and cross-training for all employees and internships, and job transfers for invited employees to achieve continuous development. Because of the risk involved in manufacturing processes, safety training and safety culture become a primary goal of L&D. The most knowledgeable employees in each work group earn the title of subject matter expert (SME) or corporate fellow [28]. Furthermore, Praxair puts a great emphasis on leadership development, not only for current leaders but also for the next generation of leaders. For example, Praxair uses simulations to develop leadership capability in technical staff with three levels of challenge–early career, mid-career, and longer service– where teams work together to solve problems of a fictional enterprise with ever-changing circumstances [29].

2.1.3. Knowledge management

Firmly grounded in the corporate mission/vision, Praxair operates productivity programs by sharing the best practices. Each employee is mandated to contribute to productivity goals that are measured in hard dollars. A sense of urgency is created to emphasize personal accountability balanced with the skills for teamwork. Furthermore, the company, especially the global engineering business unit, runs communities of practice (CoP) for knowledge creation and dissemination. Praxair launched an on-demand video system in 2010 and an ask-an-expert system in 2011 [30] to facilitate the sharing of knowledge between senior employees on the verge of retirement and less experienced employees. Viewing of the videos increased exponentially because of

leadership support. One key has been easy access to videos through such mechanisms as links embedded in E-mails. The technology for the ask-an-expert system has evolved from SharePoint to a customized system to increase effectiveness. Requiring every community of practice to participate with one or more questions per month increased participation and value [31].

2.1.4. Summary

Innovation involves the ideas and efforts of organizational employees. As their knowledge, empowerment, and engagement grow, so does innovation capability. Praxair has made a number of deliberate changes over recent years to build that capability.

2.2. Air Liquide

Headquartered in Paris, France, Air Liquide was founded in 1902 by a group of French scientists and engineers seeking to develop and commercialize a new process for air liquefaction (thus leading to the name of the company). During this period, Paris was considered a beacon of the arts, scientific research, instrument and tool manufacturing, banking, and venture capitalism and "an incubator for technological innovation and for the formation of high-tech start-up companies, not unlike what Boston/Cambridge and Silicon Valley would become in the United States in the late twentieth century" [32]. Smith refers to primary founder of Air Liquide, Georges Claude, as "Frances's leading industrial scientist of the last 100 years" (p. 51). Claude's inventions, creativity, and persistence were key drivers of innovation and success throughout the organization's early years.

Since its founding in the early 1900s, the organization has grown to be one of the world's top two industrial gas suppliers, providing oxygen, helium, nitrogen, carbon dioxide, and argon to business customers in the automotive, chemicals, food and beverage, semiconductor, and healthcare industries. Today, Air Liquide employs approximately 51,500 people across 80 countries and generated over \$18 billion in sales revenue in 2015 [33] (these figures do not include the May 2016 acquisition of Airgas).

2.2.1. The importance of innovation at Air Liquide

Innovation has been a key part of the corporate identity since the organization's founding. An excellent history of Air Liquide's early years of innovation can be found in Smith's [32] article "Product Innovation and the Growth of the Large Firm: The Case of Air Liquide, 1902–1930." New product development and innovation are important parts of Air Liquide's strategy as the organization endeavors to continuously improve its production technology, expand existing markets, and develop new markets.

2.2.2. Role of HRD and KM in facilitating innovation

Air Liquide leverages employee development programs, such as technical career pathing, mentoring, and high potential programs, along with more traditional KM approaches such as communities of practice, to facilitate knowledge sharing, transfer technical knowledge, and foster a culture of innovation throughout the organization.

From a developmental standpoint, human resources has established a Technical Career Leaders (TCL) dual career ladder, created in 2003, that serves to identify, develop, recognize, reward, and retain key technical expertise within the organization as well as facilitate the transfer of their knowledge. The TCL program includes six levels: two local levels and four international levels. The program offers structured, yet flexible, career tracks where technical talent has the option to develop and progress in their careers within either the technical or the managerial path, and the flexibility to move between career tracks during their careers. Each track includes mentoring, training, networking, and other developmental opportunities for participating employees.

TCL is a worldwide program covering all Air Liquide business lines. As of June 2016, there were approximately 2500 experts representing 67 nationalities participating in the TCL. Technical areas encompassed in the TCL program include electronics, engineering and technology, healthcare, industrial merchant (cylinder, bulk, and onsite; and applications and services), large industries, and research and development, and each of these areas has its own domain of expertise. Employees are selected to participate in the TCL program based on criteria such as their participation in knowledge transfer activities, their leadership and influence abilities, and their capacity to communicate and deliver on their technical vision and innovative ideas.

Within the Large Industries Business Line in North America, Air Liquide also engages select early- to mid-career high potential technical talent in a LEAP (learning, experiencing, and progressing) developmental program in order to accelerate time to competency of these employees and retain and engage this key employee population. An analysis of this technical population revealed that there was insufficient pipeline to fill anticipated gaps in technical roles due to attrition of technical experts. Therefore, the LEAP program was instituted in 2013 in order to provide early- to mid-career high potential technical talent with the practical, technical training and experiences that they need to quickly advance in their careers at the organization. The LEAP program began with 24 employees; today there are approximately 75 employees participating (45 protégés, and the reminder coaches, trainers, and managers). Each protégé is assigned a coach (some coaches have more than one protégé). The program consists of 10, three-day classes, supplemented with computer-based training, to cover technical content as well as soft skills training. Protégés are assigned a set of technical objectives at various points throughout the program to complete over the next 12 months, under the guidance of their coach. The program offers participating employees a variety of experiences and learning opportunities in an accelerated fashion in order to speed their developmental process, and help transfer technical knowledge from the coaches to their assigned protégés.

From a knowledge management perspective, Air Liquide has established technical communities of practice (CoPs) at a worldwide level, the business line level, as well as at the more regional hub level. For example, as of June 2016, the company has 7 worldwide technical CoPs established and 10 technical CoPs within the large industries business line in Europe. Hubs for the Americas, Asia Pacific, and Middle East Africa are in the process of establishing their own CoPs. The intent of the technical worldwide CoPs is to set standards, establish technical vision, and identify technical best practices. The hub CoPs, in turn, focus on implementing the best practices identified in the worldwide CoPs and people development. All communities facilitate knowledge sharing and transfer within the organization. Within the Americas zone, both formal and informal communities leverage Google Plus Sites as the enterprise platform to facilitate and enable virtual collaboration, content sharing, and Q&A amongst site members.

2.2.3. Summary

Programs such as TCL and LEAP facilitate innovation on an individual level by presenting employees with multiple and diverse opportunities for learning and growth and on a collective level by enabling employees to learn from one another and build on the ideas of others. Similarly, CoPs at Air Liquide provide forums for networking, knowledge sharing, and employee development, ultimately fostering individual and organizational learning and innovation.

2.3. Rockwell Collins

Rockwell Collins, Inc., headquartered in Cedar Rapids, Iowa, provides avionics and information technology systems and services to governmental agencies and aircraft manufacturers. The organization went public in 2001 as a spin-off from Rockwell Automation and has been experiencing tremendous growth over the past decade. Rockwell Collins runs operations in 60 locations with nearly 20,000 employees around the globe (Rockwell Collins, 2014). Its aircraft electronics are installed in the cockpits of nearly every airline in the world, and its airborne and ground-based communication systems transmit nearly 70% of all U.S. and allied military airborne communication. The organization also provides flight simulation and training, MRO (maintenance, repair, operations) services, navigation, and surveillance systems [31].

2.3.1. Innovation culture and knowledge management

Since 2001, Rockwell Collins has undergone a period of rapid growth. The unprecedented growth was a primary driver of its KM system. With the rapid expansion of the organization, the necessity to connect new employees with existing knowledge and expertise worldwide became increasingly urgent [31].

The history of Rockwell Collins' KM system can actually be traced back to 1999. In order to fulfill the mission of "access, connect, and transcend," corporate leaders developed a KM system to "accelerate knowledge, and create value" enterprise-wide [34]. Over the past 10 years, each department of Rockwell Collins has been closely connected by the KM system. Currently, many KM approaches and tools, such as communities of practice (organized groups for employees to share and learn), Epedia (companywide Wikipedia), Lessons Learned (a reflection tool in Epedia), and Enterprise Tools Integrated Forum (questions and answers forum) are used daily [35–37]. Rockwell Collins has promoted a learning culture by combining its formal training and development processes with these industry leading KM programs [34, 38]. The five to ten awards won by Rockwell Collins each year, include Blue Ribbon winner for Innovation by *Military Training Technology* magazine in 2014 [39].

2.3.2. HRD's role in innovation

There are several strategies that HRD leverages to support innovation at Rockwell Collins. For example, a virtual university (Rockwell Collins University) functions as the catalyst for innovation. The University employs highly developed e-learning systems to offer courses efficiently. Technology enables employees to access learning anytime and anywhere. Within the learning culture at Rockwell Collins University, e-learning is a richer source for learning beyond videotapes or online tools only. Employees seek answers actively instead of being offered solutions. For example, learning and development specialists regularly assess the need to retain the core knowledge that senior employees may take away when they retire. Learning and development specialists worked with the Performance Engineering Group (PEG) to invent "QuickLearns," computer-based tutorials featuring subject-matter experts performing key tasks [40].

At Rockwell Collins, a learning culture that emphasizes sharing has been formed with HRD's assistance. For example, HRD specialists facilitate ad hoc meetings with engineering-group leaders who have suggestions and feedback regarding training programs at Rockwell Collins. As a group, HRD specialists use such venues as one mechanism to identify training needs [40].

2.3.3. Summary

One pervasive risk to many technical organizations is employees leaving the organization with critical technical, cultural, and social knowledge that is difficult to replace. As a learning organization, Rockwell Collins leverages the HRD strategies and the well-established KM system described above to create a culture that recognizes inquiry, feedback, and creative thinking. The KM system helps the organization navigate both planned and unplanned organization change.

HRD at Rockwell Collins leverages training and development and knowledge sharing through KM tools to support organizational learning on each level (individual, team, and organizational). As a result of HRD's efforts to facilitate innovation, the organization has experienced several positive changes. First, the creation of a learning culture fosters innovation and out-of-the box thinking. Second, the implementation of KM programs facilitates knowledge sharing at the company more freely and efficiently. Finally, the company has created healthy recognition and reward mechanisms that encourage knowledge sharing and innovative behaviors by employees. In Rockwell Collins' case, formal and informal learning programs are complemented by a firmly established KM system, which makes the company a learning organization that encourages employees to think and work innovatively.

2.4. FMC Technologies, Inc.

Headquartered in Houston, Texas, FMC Technologies is a publicly traded oil and gas equipment services company, specializing in subsea, surface technology, and energy infrastructure. It was originally founded in 1884 as the Bean Spray Company, but then changed in 1928 to the Food Machinery Company, which is where the FMC initials originate. FMC Technologies was recognized by Forbes on its list of most innovative companies in 2013 [20]. At that time, FMC Technologies ranked 17th on Forbes' international list of innovative organizations, based on their innovation premium. The innovation premium is the difference between a company's market capitalization and a net present value (NPV) of cash flows from existing businesses [20]. Companies with a market cap above NPV of cash flows are ranked in order of the most market cap above NPV to the least. FMC Technologies had a market cap of \$12.4 billion USD.

Foss et al. [41] provide a solid definition of knowledge sharing, which is "...the provision or receipt of task information, know how, and feedback on a product or a procedure" (p. 458). The core beliefs espoused by early HRD practitioners and theorists include the notions that HRD itself is the process and practice of developing, harnessing, or releasing of human expertise through individual and organizational development in order to improve performance [42]. This is evidenced in these platforms of knowledge sharing at FMC Technologies: The Women's Leadership Program, a comprehensive program for the development of women as leaders; the Edge, a formal knowledge management system (KMS); WellLinked, a quarterly employee publication; and a business model of formalized internationally based centers of excellence (COEs). The HRD-specific function each of these platforms is discussed in the next section.

2.4.1. Knowledge sharing, innovation, and HRD

Knowledge sharing supports organizational productivity and also enriches its credibility with the marketplace and stakeholders [41]. HRD enhances knowledge exchange at FMC Technologies in three key internal areas that drive external value:

- 1. Leadership and talent development—Sustainment of success and mindfulness is critical in the leadership and talent development disciplines within FMC. The strategic purpose of the Women's Leadership Program is to support employee's potential and also to attract and retain top talent. Women learn about leadership roles, behaviors, styles, and leading teams, as well as focus on career development and leadership skills necessary for them to be successful. The company monitors the career advancement of the participants to ensure their success and that of the program [42].
- 2. Team learning and expertise development—A web-based application called the edge allows technical staff to post issues to a single location to receive electronic suggestions, responses, and answers from other knowledge workers within FMC Technologies. Technical staff make use of this global technology knowledge-sharing platform that enhances team learning, giving people opportunities to problem solve with others around the world. If a team member is having an issue or experiencing a challenge, another employee anywhere in the world can provide expertise from a virtual community with thousands of members. In addition to sharing expertise, employees learn about the various working conditions of their colleagues [42].

One of the most prominent features about this organization is consistent messaging about its internal organizational and operational expertise. FMC Technologies publishes a quarterly global employee publication called WellLinked highlighting the success of employees, teams, and organizational plans. The magazine's purpose is to inform all employees on the forward progress of the organization and any updates in organizational strategy [42].

(3) Networking—FMC boasts 50 online networks established to focus on different parts of the business. Any employee can read content shared within a network, but the people who work in the specific area of the operation run or manage the network. One of the networks, the New Hire Network, helps new hires get their questions answered. Team members welcome them, provide responses, and let them know how to find necessary resources [42].

2.4.2. Summary

HRD's role in innovation at FMC Technologies is to support collaboration and community building that contributes to performance through knowledge sharing. The work of HRD is an antecedent to this organization's ability to support its customers' success. FMC Technologies' stance on innovation is girded by formal organizational structures that ensure knowledge sharing, which makes their customers and the company, in turn, successful.

2.5. Krungthai Card PCL (KTC)

Krungthai Card Public Company Limited (KTC), a consumer finance service provider in Thailand founded in 1996, represents one of the Thai companies, recognized from the Stock Exchange of Thailand, as an innovative organization. KTC was the first Thai financial firm to reform itself into an innovative organization in order to make a difference in its corporate identity.

The company's core values (modern, dynamic, professional, simple, and fun) help KTC build a new corporate identity that differentiates it from its competitors [43]. To manage its identity, the company aligns the five core values with innovation infrastructure capability. The innovation infrastructure capability, which includes the organization structure, the organizational culture, and leadership style, supports the development of creativity and innovation. KTC's structure is characterized by low levels of hierarchy, high levels of cross-functional collaboration, and decentralization of decision making. The KTC corporate culture involves team mission-based projects, modern and dynamic working styles, and a workplace orientation focused on employee satisfaction. KTC leaders emphasize a people-oriented management approach and play an important role in managing organizational change related to creative and innovative capabilities.

2.5.1. Learning and innovation

Regarding KTC's corporate identity, the organization encourages employees to become dynamic and collaborative in order to cope with change in its business markets. KTC assumes that if the company sustains a relaxing learning environment, employees are able to generate a high degree of creativity and innovation.

At KTC, creativity in employees is developed through a lively and collaborative learning climate. The learning climate is shaped by the five core values that align with the organization's processes and practices. The company primarily focuses on collaborative learning in which employees share their knowledge and skills with their peers through a social community. KTC uses this collaborative learning through interactive working processes, such as knowledge sharing within a mission-based team and a specific learning program titled "KTC Knowledge Sharing." These learning activities are delivered via face-to-face and online media (i.e., KTC Mind Intranet) communications. Various employees acquire knowledge from each other when they are placed collectively in their communities.

Creativity and innovation are facilitated by a culture of the learning organization [10] and knowledge management [44]. This culture requires an alignment of organizational policies, procedures, and practices that foster a dynamic interaction of individuals, groups, and an organization that creates knowledge.

2.5.2. Role of HRD and knowledge management

HRD is a bridge between personal creativity and organizational performance through KTC policies, procedures, and practices. HRD strategically supports KTC's identity through the development of employee competencies that are aligned with creativity. KTC uses the knowledge-based organization concept to enhance individuals' competency development and create knowledge sharing within the firm [45]. Consequently, HRD at KTC contributes to innovation through a variety of learning perspectives: active training, social learning, as well as coaching and mentoring.

KTC facilitates many active training programs to enhance employees' competence through dynamic learning. The firm provides classroom training and e-learning programs to develop core, functional, and managerial competencies for its employees. These programs are designed with regard to a dynamic of learning activities and a degree of learners' participation. For example, the HR function implemented a grooming, acting, and voice training program for call center representatives to change their attitudes and make them aware that voice and facial expressions affect their performance when they talk with their clients on the phone [45]. The trainers designed a fun and relaxing environment through demonstration, practice, and a singing contest. KTC applied an aesthetic concept to design fun and dynamic training to develop individuals' creativity as well as to enhance morale and work engagement.

Social learning is about learning from others. The most recognized KTC social learning program is "KTC Knowledge Sharing." This program has succeeded with a large number of participants [46]. KTC employees were invited to be speakers who share their knowledge regarding their specialized areas. The knowledge shared focuses not only on the businessrelated content but also on the non-related business content (e.g., mind-map writing, health and safety, and fashion).

At KTC, coaching and mentoring are used as informal learning practices, not only to develop individuals' work skills, but also to help them in career and personal enhancement. The former CEO, Niwatt Chitalarn, believed that the most important responsibility in managing his employees was to build a leadership team at KTC [47]. His management philosophy was "my control and your control" by performing as a captain pilot who coaches co-pilots. A captain pilot must function as a leader to provide direction and coordination for his crew. At the right time, a captain pilot should allow co-pilots to make a decision. He believed that a leader had a responsibility, not only to oversee an organization's performance, but also to develop

the leadership team. In consistency with the previous CEO's philosophy, the current CEO, Srimongkol, coaches his executive teams by asking opinions, giving suggestions, then monitoring their performance, and providing feedback on their actions [45].

Under the knowledge-based organization concept, KTC focuses on learning and development to enhance individuals' capabilities and create practical knowledge sharing within the organization [46]. The firm creates a climate of team learning to move toward its corporate goals.

2.5.3. Summary

KTC uses a dynamic workplace approach in unleashing individuals' creativities in order to update its corporate identity to one that is more vibrant and creative. Senior leadership at KTC assumes that organizational creativity and innovation rely on learning. The knowledge, experiences, and cultural familiarity that each individual employee embodies can be delivered through social learning. The company creates and sustains a creative and innovative culture through leadership and management activities by leveraging actions across the organization. HRD at KTC encourages employees to improve their work through a lively learning environment and a climate of team work and knowledge sharing.

3. Conclusions

Throughout the chapter, examples were provided of organizations that are considered to be innovative. The conclusion of the chapter focuses on specifying commonalities between innovative organizations. We believe themes emerge in the five profiles highlighting linkages between HRD, KM, and innovation. The identification of similar practices and processes may not make it easier for organizations to be innovative, but it does facilitate understanding by practitioners and researchers of some organizational antecedents necessary for innovation to flourish.

We hope this chapter also accomplishes a secondary goal to solidify and clarify where the presence of HRD together with KM establishes conditions necessary for innovation. HRD looks to develop and unleash human expertise and creativity with its focus on career development, training, and organizational development. HRD at these five cases contribute to facilitating a learning culture, providing a broad spectrum of learning and development (L&D) opportunities, and providing employees a broad spectrum of opportunities and encouraging diversity of thought and new ideas. On the other hand, KM focuses on the development of communities of practices, organizational knowledge sharing through lesson learned process, and the utilization of technology to create infrastructure for storing and sharing expertise and best practices. These KM processes facilitate and sustain individual and collective activities to help manage the flow of knowledge (to acquire, create, store, share, use, and assess) throughout the organization. When and where there is expertise and creativity coupled with methods of harnessing, sharing, and growing knowledge and ideas, we believe an organizational environment or culture can exist for innovation. **Figure 1** captures the relation between HRD, KM, and innovative organization.

3.1. Themes of innovative organizations

Several themes emerged from examination of the five company profiles, which are depicted in **Figure 1**. The themes represent facets of organizing that enable the generation and sharing of new knowledge that relates to improved organizational performance. The themes are matched to the profiles of the five companies shown in **Table 2**.

3.1.1. Vision and core values

Many organizations begin with their vision statement as a starting point to motivate organizational performance. In their vision statements, Praxair and FMC Technologies insist on the long-term view of the organization as one of the global preeminence—a key to organizational sustainability. Core values occupy a supporting role in an organization's vision. With KTC its value-based perspective initiates the development of the resulting learning climate.

3.1.2. Autonomy

While organizational leadership may seem to exert control over its business, technology, and products, each organization demonstrates times when internal self-sufficiency provided flexibility for the organization [48]. Flexibility and agility are what allow organizational systems to



Figure 1. Relationship between HRD and KM practices.

	Praxair	Air Liquide	Rockwell Collins	FMC Technologies Inc.	Krungthai Card PCL.
Vision & core values in daily life	Make the planet more productive	Invent, design, produce high quality solutions for customers and patients	Establish a culture that recognizes inquiry, feedback, and creative thinking	Organizational focus on living the core values	Alignment of corporate policies and practices to enhance employee competency and organization culture
Autonomy	 Business- unit level autonomy flattens the hierarchy Flexible job 		Employees empowered to seek solutions	Team-based problem solving and solution implementation	
Collaboration	design Communities of practice	Communities of practice	 Communities of practice; Institutionalized cross-functional training 	Communities of practice for each business unit, and department	 Collaborative learning environment; Team-based knowledge sharing
Continuous learning	Simulation-based learning for at all career stages	Focus on continuous learning of succession planning	 Company-based Wikipedia Formalized lesson- leaned processes 	Employee- focused quarterly communication	Leader-directed corporate culture
Social learning	Rapid development of new hires			 International centers of excellence New hires inte- grated socially through online and face to face platforms 	Engagement- based learning
Talent development	 Engage HR for next gen- eration leader- ship, talent acquisition, and business integration; Women men- toring other women; Mid-career talent development 	 Technical career paths; Mentoring & coaching; Early & mid-career talent develop- ment 	Corporate university	 Corporate university Women's leadership development program 	Coaching & mentoring

Table 2. Company findings mapped to conceptual model.

adapt quickly within their markets and ever-changing business environments. Job autonomy and the self-rule provided in the communities of practice are examples of internal self-governance.

3.1.3. Collaboration

Across each of the companies profiled in this chapter, collaboration in team learning enhances competence by allowing employees to learn from one another. An effective KM system supports the collaborative learning.

3.1.4. Continuous learning

Within the communities of practice, constant dialogue by employees results in continuous new learning with processes improving and adaption occurring.

3.1.5. Social learning

As organizations are largely social systems, learning within their communities reinforces desired knowledge and behavior. These communities reflect learning and collaboration. Rockwell Collins and FMC Technologies communities of practice use technology to share knowledge across their internal teams.

3.1.6. Talent development

Whether the need for talent is within the technical sphere, such as with Air Liquide, Rockwell Collins, and Praxair, or in the leadership area, such as with FMC Technologies, internal programs whose focus is to hone specific competencies identify and guide both expertise and knowledge sharing within the organization. Early, middle, and end of career development supports internal career development and the life cycle of the organization. As talent and the organization mature, the needs of both individual and company are met.

Author details

Michael Beyerlein^{1*†}, Rachele Collins^{2†}, Shinhee Jeong^{1†}, Christi Phillips^{3†}, Suravee Sunalai^{4†} and Lei Xie^{1†}

- *Address all correspondence to: beyerlein@tamu.edu
- 1 College Station, Texas A&M University, USA
- 2 American Productivity and Quality Center, Houston, USA
- 3 CHI St. Luke's Health, Houston, USA
- 4 Dhurakij Pundit University, Bangkok, Thailand
- ⁺ These authors contributed equally

References

- Nguyen TNQ, Neck PE, Nguyen TH. The critical role of knowledge management in achieving and sustaining organizational competitive advantage. International Business Research. 2009;2. DOI: 10.5171/2010.322246. http://ro.uow.edu.au/commpapers/1721/
- [2] Spender JC. Knowledge management: Origins, history, and development. In: Advances in Knowledge Management. Springer International Publishing, New York, NY; 2015. pp. 3-23
- [3] Hughes C, Gosney MW. The History of Human Resource Development: Understanding the Unexplored Philosophies, Theories, and Methodologies. Springer, New York, NY; 2016
- [4] Wang Z, Wang N. Knowledge sharing, innovation and firm performance. Expert Systems with Applications. 2012;39:8899-8908. DOI:10.1016/j.eswa.2012.02.017 DOI:10.1016/j. eswa.2012.02.017#doilink
- [5] Ishak NB, Eze UC, Ling LS. Integrating knowledge management and human resource management for sustainable performance. Journal of Organizational Knowledge Management. 2010:2010:1-13. DOI:10.5171/2010.322246. http://ibimapublishing.com/articles/ JOKM/2010/322246/
- [6] Kastelle T. The Problem of Defining Innovation [Internet]. 2010. Available from: http://timkastelle.org/blog/2010/08/an-innovation-definition
- [7] Crossan M, Apaydin M. A multi-dimensional framework of organizational innovation: A systematic review of the literature. Journal of Management Studies. 2010;47(6):1154-1191. DOI: 10.1111/j.1467-6486.2009.00880.x
- [8] Harkema S. A complex adaptive perspective on learning within innovation projects. The Learning Organization. 2003;**10**:340-346. doi:10.1108/09696470310497177
- [9] Gibb S, Waight CL. Connecting HRD and creativity: From fragmentary insights to strategic significance. Advances in Developing Human Resources. 2005;7:271-286. DOI: 10.1177/1523422305274530
- [10] McLean LD. Organizational culture's influence on creativity and innovation: A review of the literature and implications for human resource development. Advances in Developing Human Resources. 2005;7:226-246. DOI: 10.1177/1523422305274528
- [11] Taylor MA, Callahan JL. Bringing creativity into being: Underlying assumptions that influence methods of studying organizational creativity. Advances in Developing Human Resources. 2005;7:247-270. DOI: 10.1177/1523422305274529
- [12] Sunalai S. Knowledge Management Systems in Higher Education Institutions in Thailand: A Holistic Model of Enablers, Processes, and Outcomes [dissertation]. Texas A&M University; 2015
- [13] Nonaka I, Takeuchi H. Knowledge creation and dialectics. In: Takeuchi H, Nonaka I, editors. Hitotsubashi on Knowledge Management. Singapore: John Wiley & Sons (Asia); 2004. pp. 1-28

- [14] Davenport TH, Prusak L. Working Knowledge: How Organizations Manage What They Know. Boston: Harvard Business School Press; 1998
- [15] Sunalai S, Beyerlein M. Exploring knowledge management in higher education institutions: Processes, influences, and outcomes. Academy of Educational Leadership Journal. 2015;19:289-308. Available from: http://search.proquest.com/docview/1768629497?accou ntid=44522
- [16] Callahan MHW. Case study of an advanced technology business incubator as a learning environment [dissertation]. The University of Georgia; 1999
- [17] Marsick VJ, Watkins K. Informal and Incidental Learning in the Workplace. London: Routledge; 1990
- [18] Marsick VJ, Watkins KE, Callahan MW, Volpe M. Reviewing Theory and Research on Informal and Incidental Learning [Internet]. 2006. Available from: http://eric. ed.gov/?id=ED492754
- [19] Sherif K, Xing B. Adaptive processes for knowledge creation in complex systems: The case of a global IT consulting firm. Information & Management. 2006;43:530-540. DOI: 10.1016/j. im.2005.12.003
- [20] Forbes. How We Rank the World's Most Innovative Companies 2013 [Internet]. 2013. Available from: http://www.forbes.com/sites/innovatorsdna/2013/08/14/howwe-rank-the-worlds-most-innovative-companies-2013/
- [21] Achievers. 50 Most Engaged Workplaces[™] in North America for 2016 [Internet]. 2016. Available from: http://www.achievers.com/engaged/winners/2016
- [22] Angel S. 2015 Annual Report Shareholders Letter [Internet]. 2016. Available from: http:// www.praxair.com/investor-relations/2015-annual-report-shareholders-letter
- [23] Harris BR, Huselid MA, Becker BE. Strategic human resource management at Praxair. Human Resource Management. 1999;38:315-320. DOI: 10.1002/(SICI)1099-050X(199924) 38:4<315</p>
- [24] Praxair. Values and Vision [Internet]. 2016. Available from: http://www.praxair.com/ our-company/vision-and-values
- [25] Heller M. Developing blended executives [Internet]. 2014. Available from: http://www. cio.com/article/2453321/cio-role/developing-blended-executives/cio-role/cio-role/developing-blended-executives.html
- [26] Overby S. HR Departments Invaded by Data Scientists [Internet]. 2013. Available from: http://www.cio.com/article/2383195/business-intelligence/hr-departments-invaded-bydata-scientists.html
- [27] Carter L, Ulrich D, Goldsmith M, editors. Best Practices in Leadership Development and Organization Change: How the Best Companies Ensure Meaningful Change and Sustainable Leadership. Wiley, New York, NY; 2005

- [28] Bourgeois T, Jordan B, Kuberka K, Nestle M, Vasquez A. Praxair Inc. Case Study. In: APQC 2015 Knowledge Management Conference. Center holds an annual conference; 2015; Houston, TX. https://www.apqc.org/apqcs-2015-knowledge-management-conference
- [29] Chief Learning Officer. How Praxair Grooms Tech Talent for Business Roles [Internet]. 2013. Available from: http://www.clomedia.com/2013/11/22/how-praxair-grooms-techtalent-for-business-roles/ [Accessed: November 22, 2013]
- [30] Bourgeois T. Using Video on Demand and Ask an Expert to identify and develop experts. APQC Webinar. [Internet]. 2015. Available from: https://sites.google.com/a/ki-network. org/announcements-site/announcements/1441109740023
- [31] APQC. Enabling knowledge capture and transfer with video and Ask-An-Expert systems at Praxair. Session Overview. In: APQC 2015 Knowledge Management Conference. Center holds an annual conference; 2015; Houston, TX. https://www.apqc. org/apqcs-2015-knowledge-management-conference
- [32] Smith M. Product innovation and the growth of the large firm: The case of Air Liquide, 1902-1930. Essays in Economic & Business History; 1999;17:49-61. http://ebhsoc.org/journal/index.php/journal/issue/view/13 [Accessed: June 30, 2017]
- [33] Hoovers. Air Liquide Company Information [Internet]. 2016. Available from: www. hoovers.com
- [34] APQC. Knowledge Management at Rockwell Collins [Internet]. 2012. Available from: http://www.apqc.org/knowledge-base/documents/knowledge-management-rockwellcollins [Accessed: October 12, 2012]
- [35] APQC. Knowledge Management at Rockwell Collins: Accelerate Knowledge. Create Value [Internet]. 2011. Available from: http://www.apqc.org/knowledge-base/documents/ knowledge-management-rockwell-collins-accelerate-knowledge-create-value [Accessed: July 03, 2011]
- [36] Jorgensen R. Creating an environment for knowledge sharing: corporate knowledge sharing at Rockwell Collins. International Council on Systems Engineering. 2011;14:22-25. DOI: 10.1002/inst.201114125
- [37] Bhandari R, Wendt T. Using agile techniques to enable the flow of knowledge at Rockwell Collins. In: APQC 2016 Knowledge Management Conference. Center holds an annual conference.; March 2016; Houston, TX. https://www.apqc.org/ apqcs-2016-knowledge-management-conference
- [38] Rockwell Collins. 2012 Corporate Responsibility Report Executive Summary [Internet]. 2012. Available from: http://www.rockwellcollins.com/Our_Company/~/media/FCD0 D1A8174B446D8E257C56706EF7F0.ashx
- [39] Rockwell Collins. 2014 Rockwell Collins Annual Report [Internet]. 2014. Available from: http://www3.rockwellcollins.com/annualreport/2014/_images/global/RockwellCollins2 014.pdf

- [40] Purington C, Butler C, Gale, S F. Built to Learn: The Inside Story of How Rockwell Collins Became a True Learning Organization. New York: AMACOM, American Management Association; 2003
- [41] Foss NJ, Husted K, Michailova S. Governing knowledge sharing in organizations: Levels of analysis, governance mechanisms, and research directions. Journal of Management Studies. 2010;47:455-482. DOI: 10.1111/j.1467-6486.2009.00870.x
- [42] Phillips CA. Role of HRD in innovative organizations FMC Technologies Inc. Poster In: the 2014 Academy of Human Resource Development Conference; 2014; Houston, TX. http://www.ahrd.org/?13_americas_archive#ahrd14
- [43] Thongsuk P. Creative organization. In: HROD Talk; The National Institute of Development Administration, Bangkok, Thailand; April 2009
- [44] Loewenberger P. The role of HRD in stimulating, supporting, and sustaining creativity and innovation. Human Resource Development Review. 2013;12:422-455. DOI: 10.1177/1534484313494088
- [45] Adisornmongkol K. Rathian Srimongkol Writes the KTC Story [Internet]. 2012. Available from: http://www.brandage.com [Accessed: April 01, 2012]
- [46] Krungthai Card. KTC Annual Report 2014. Bangkok: 2015. https://www.ktc.co.th/ KTCEBook/AnnualReport/2015_EN.pdf
- [47] Chiamrotchananon W. Leadership Team @ KTC [Internet]. 2008. Available from: http:// www.bangkokbiznews.com [Accessed: July 23, 2008]
- [48] Pasmore B. Leading Continuous Change: Navigating Churn in the Real World. Berrett-Koehler Publishers, San Francisco: CA; 2015

Network Market Orientation, Knowledge Management and Born Globals' Competitiveness

Diego Monferrer Tirado and Marta Estrada Guillén

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/intechopen.70064

Abstract

This chapter studies the influence of network market orientation (NMO) on Spanish Born Globals' (BGs) competitiveness. The study analyses the contribution of a specific knowledge-based dynamic capability, namely absorptive capability, as a key mediator variable between NMO and competitiveness, finally influencing the international performance achieved by this particular type of firm. Results from testing the structural equation model proposed confirm that NMO facilitates the development of an absorptive capability in BGs and that this capability, in turn, influences the firms' capacity to exploit the new relational knowledge, thus improving their competitiveness and international performance.

Keywords: Born Globals, network market orientation, absorptive capability, competitiveness, international performance

1. Introduction

Muchof the previous research on internationalisation behaviour concludes that firms become involved in international markets gradually [1]. However, in the last 20 years, a new kind of business has emerged that does not follow a gradual internationalisation pattern, but is characterised by rapid commitment to international activity. Oviatt and Dougall [2] called these businesses 'Born Globals' (BGs), and they have given rise to a new line of research known as international entrepreneurship centred on the study of how new ventures are committed to developing the processes of 'discovery, enactment, evaluation, and exploitation of opportunities across national borders to create future goods and services' ([3], p.7). Past research in this area has centred on determining the factors that might explain the exceptional speed with



© 2017 The Author(s). Licensee InTech. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. which certain new ventures can internationalise [3–10]. However, several authors consider that to be able to advance our understanding of BGs, we need to analyse the factors that can affect their competitiveness after their creation, particularly from a marketing perspective [11–17].

Traditionally, lack of market knowledge has been regarded as one of the biggest impediments small businesses have to overcome in their internationalisation process [18–20] in that it constitutes a key variable for proactively seeking international opportunities [21, 22]. McNaugthon [23] shows that companies concerned to expand their knowledge of foreign markets have a broader perspective and there is a greater chance that they will seize the opportunities that appear in them.

If we attempt to identify the main source of this knowledge, we observe that traditional gradualist models attribute a fundamental role in knowledge generation—and, therefore, in the internationalisation process—to the firm's experience (at home and abroad) [1]. However, in the case of BGs, knowledge based on experience cannot be considered the only source of knowledge about foreign markets that these firms use, as their experience is minimal. For this reason, it becomes necessary to investigate how BGs manage to acquire and interpret information about markets and how they translate it into specific actions that affect the development of skills that help them to remain dynamic in international markets [22, 24, 25].

Many marketing scholars argue that market orientation plays a determining role in the international success of new businesses due to its contribution to effective knowledge management [26–30]. In this line, Knight and Cavusgil [31] point out that the market orientation of new international ventures enhances their understanding of their customers' present and future needs and optimises the development of distinctive actions that can meet these needs by providing a more valuable offer than that of their rivals.

In addition, based on the social perspective, the study of entrepreneurship has also highlighted the importance of the links entrepreneurs develop with members of their networks as they facilitate rapid access to the information and knowledge associated with the latest trends in the new markets, and with certain key resources that are not initially available to these firms [32–36].

However, although from the theoretical viewpoint there seems to be a strong complementarity between market orientation and membership of business networks, as shown in certain studies [13, 37, 38], little is known about the way these factors combine. In this vein, Loane and Bell [18] highlight the need for research attention to the mechanisms and routines that enable new international ventures to generate and manage knowledge through relationships they establish in networks. In turn, Evanschitzky [38] highlights the importance of studying their influence on competitiveness and firm performance, in light of the scarce knowledge to date.

In this respect, this study covers the mutually complementary nature of both factors in a single construct known as network market orientation (NMO). Thus, although previous studies have emphasised the importance of MO in the context of relationships between firms, they have not analysed it as an inter-business phenomenon in itself, but rather as the sum of the market orientations of the various individual firms [22, 37, 39, 40]. The present work analyses

the role of NMO differently, understanding it as an orientation established jointly by the different members making up the network.

However, the effect of any organisational factor is usually subject to significant causal ambiguity regarding the way firms maintain their capacity to use the knowledge generated in order to continue seizing the market opportunities that arise [41]. Adopting an NMO that promotes a cognitive effort and collective approach to learning can help to overcome this ambiguity by improving understanding of the new and changing links between action and performance and, in this way, recognising the need to face changes in existing routines [22, 40, 42, 43]. In line with this idea, it is essential that BGs complement their adoption of an NMO with the generation of dynamic absorptive capability in order to ensure continuing commitment to construct new resources and capabilities and reconfigure existing ones, thereby adapting to new market demands [22, 44–48].

Specifically, this absorptive capability is related to the assimilation of external information into the firm's internal knowledge base [4, 49]. It is crucial for each firm to develop this capability individually, since it enables the external knowledge generated in the network context to be perfectly understood and assimilated internally by the firm and put to use successfully [50–54].

Ultimately, this study aims to contribute to the existing literature by analysing the way in which BGs start by adopting an NMO in order to access and systematically manage information from foreign markets and the way this orientation contributes to generating a dynamic absorptive capability that helps them to capitalise on the knowledge generated, thus facilitating their consolidation abroad.

In order to meet this objective, we first present the theoretical frame used to formulate hypotheses on the influence of NMO on BGs' international competitiveness and performance. The empirical study carried out to test the hypotheses is then reported, with an analysis of the results obtained and the main conclusions. Finally, limitations and future research lines associated with this study are described.

2. Theoretical development

2.1. The relevance of NMO for BGs

The international entrepreneurship literature has increasingly emphasised the role of business networks in the process of learning and knowledge generation in BGs [36, 55–57]. The relationships an entrepreneur builds with the other network members (family members, customers, distributors and providers) can be crucial in (1) generating more differential and valuable knowledge on new clients' needs and the business environment conditions, (2) knowing how to exploit the positive conditions and avoid the negative ones and (3) accessing the capabilities and resources required to do it [34]. Networks, therefore, offer BGs a way of compensating for their limitations of newness [58], smallness [59] and foreignness [60] by providing the opportunity to access valuable supplementary knowledge on aspects such as technology, distribution

channels and customer bases [61] and to improve other knowledge-related aspects such as information exchange and coordination [62] and the speed of know-how and technology transfer [63]. Thus, networks are key for BGs to develop a broader knowledge base beyond what they could achieve alone [64], abandoning the idea that the generation of new knowledge is a purely internal process [19, 25, 56, 65]. This is particularly so if we bear in mind that BGs are characterised by their lack of necessary resources.

The processes that contribute to relational information management in adopting an NMO are presented in the theoretical proposal of Helfert et al. [37]. These authors define four relational knowledge management mechanisms developed jointly by the members of a network: knowledge exchange processes, in order to satisfy clients' requirements; coordination processes, aimed at synchronising the network ties through formal and informal routines; coupling processes, necessary to adapt the particular features of a network member and conflict resolution mechanisms designed to solve unexpected contexts. The work of Helfert et al. [37] represents a significant step forward in the study of this construct. In fact, previous research was limited to considering the sum of market orientations in individual companies [66] or adapting the original market orientation models (behavioural and cultural) to the network setting.

Following the work of Helfert et al. [37], Monferrer et al. [22:p.388] defined NMO as 'a strategic orientation established jointly by the different members in the business relations network. This strategic orientation involves, in a climate of trust, collaboration and commitment, engaging in certain activities and fundamental shared behaviours (adaptation, coordination, conflict resolution and exchange) based on the generation of an extended intellectual capital'. These activities seek to increase the competitiveness of the network and its individual members in an attempt to provide superior value to end customers by satisfying their needs.

2.2. NMO and absorptive capability in BGs

Absorptive dynamic capability is related to the processes developed in the company to seek new information, internalise it and integrate it into the firm's existing knowledge base [49]. Developing this capability is vitally important for BGs' survival, since their business opportunity has not yet been consolidated. Furthermore, given that BGs compete in a global market segment [2, 3], the nature of the factors that influence the conditions of their business environment come from a greater diversity of sources. This variety of sources has the effect of increasing the complexity associated with the mechanisms BGs need to generate in order to manage and internally integrate knowledge from their external markets. For this reason, before developing new knowledge search and integrative mechanisms, firms must stimulate the use of processes that allow original knowledge from different sources to be managed efficiently [22, 48].

BGs are recently created companies and therefore they assume limited resource availability [5, 6, 8, 14, 17, 67]. In these circumstances, networks facilitate the development of absorptive capability in BGs [68]. Nahapiet and Ghoshal [69] indicate that to access the different sources of knowledge from their external markets, firms need to generate communication processes in order to guarantee the capture and integration of real and potential capabilities and resources associated to the relationships an individual unit builds through its network.

Similarly, Cohen and Levinthal [49] suggest that absorptive capability is built on communication structures that cross firm boundaries.

Belonging to a market-oriented network will therefore give BGs some relevant advantages, first by improving their capacity to develop an agreed perception of their markets based on multiple agents and sources of knowledge; and second, by incorporating coordination processes to interpret and understand that knowledge and integrate it into the internal firm's knowledge base [22, 37, 40]. In sum, a market-oriented network facilitates the development of BGs' absorptive capability. We therefore posit that:

H1: Participation of BGs in market-oriented networks stimulates their absorptive capability.

2.3. Absorptive capability and competitive advantages in BGs

To specify these competitive advantages, we are guided by the work of Porter [70], who considers that the different meanings used to refer to competitive advantages can be condensed into two general ones which cover all the rest [13, 71–73]: product differentiation-based and cost-based competitive advantages. The former refers to factors such as quality, design and other attributes that distinguish firms' offers of value from those of their rivals [70–72], as well as advantages linked to services such as delivery speed, reliability and managing additional services [40, 72, 73]. In turn, cost-based advantages lie in the domain of manufacturing, administration and commercialisation costs. They give the producer value in the form of lower costs and offer the consumer the lowest price [40, 70, 72, 73].

Regarding the influence that absorptive capability has on these advantages, Cohen and Levinthal [49] argue that establishing practices which encourage the assimilation of external knowledge creates a positive incentive to invest in R&D and thus improve the firm's possibilities of achieving superior competitive advantages. According to Ref. [74], improving the differential nature of a new company requires (1) the internal application of certain valuable knowledge-based resources and capabilities; (2) the skill to associate them with other external ones; (3) the capacity to integrate the knowledge derived from this association internally and (4) the capability to apply this knowledge to potentially successful business aims. As Laursen and Salter ([75], p.146) manifest 'the lack of openness of firms to their external environment may reflect an organizational myopia, indicating that managers may overemphasize internal sources and under emphasize external sources'. An excessively internal vision may therefore negatively influence the competitiveness of the firm [49]. To encourage firm competitiveness, ideally there should be a balance between the concentration of internal capabilities and openness to the exterior [76]. In this regard, Vinding [77] extols the role of absorptive capability, as it enables the firm's internal capability and its external collaboration to complement each other. On the basis of this capability firms manage to capture, absorb and use external knowledge, thereby facilitating the achievement of competitive advantages [22, 78-80].

BGs tend to gradually define the specifications of their products or services by taking into account the particularities of their customers and the conditions that characterise their competitive environment [22]. BGs thus need to avoid internal short-sightedness when specifying market trends in differential and low cost products or services [3, 22, 81]. Therefore, BGs use the information and knowledge generated on the basis of their absorptive capability to

introduce the valuable specifications and features required in their products or services to guarantee the feasibility of their business project. These arguments lead us to propose that:

H2: Greater absorptive capability in BGs leads to greater development of differentiation-based competitive advantages.

H3: Greater absorptive capability in BGs leads to greater development of cost-based competitive advantages.

2.4. Competitive advantages and international performance in BGs

A review of the conceptualisations of the competitive advantage concept shows that, by definition, a firm has a competitive advantage over another firm when it obtains a superior performance [82]. Therefore, the competitive advantage can be seen as a direct antecedent of a BG's international performance because the superiority arising from its attempt to provide value determines the purchasing performance of its target market [55], and consequently its performance [40, 72, 83]. When a firm achieves competitive advantages (differentiation-based and/or cost-based), it has a higher capacity to generate value for its clients and, in turn, greater levels of client satisfaction, business volume and market share and lower customer service costs [30, 84, 85]. As Snoj et al. [83] point out, sustaining competitive advantages is the basis for achieving superior business performance, survival and development. Therefore, the following hypotheses are proposed:

H4: BGs' achievement of differentiation-based competitive advantages contributes to higher levels of international performance.

H5. BGs' achievement of cost-based competitive advantages contributes to higher levels of international performance.

Figure 1 summarises the model of effects for the study.



Figure 1. Model of effects.

3. Methodology

3.1. Sample selection and data collection

We started with 2012 Spanish firms, all under 7 years old and operating internationally, taken from the Dun and Bradstreet and SABI databases. To refine the sample and ensure that the firms selected were BGs, a total of six selection criteria were applied on two different occasions: prior to and during the field study. Thus, based on the filters available in the databases, the following criteria were applied: (1) the firms should be no more than 7 years old, thus guaranteeing that they are new firms; (2) they should make their own strategic decisions, thus excluding subsidiary or affiliated firms; (3) they should have a minimum of three employees and a maximum of 250, ruling out micro-firms, self-employed individuals and big firms. This process of refinement resulted in a population of 1023 firms.

The questionnaire was drawn up, based on an in-depth bibliographical review, including the three remaining selection criteria (not available without direct consultation with the firm): (4) their international activity should have started in the first three years following their creation, thus showing their immediate incorporation into foreign markets; (5) they should generate a minimum of 25% of their annual sales abroad, thus providing evidence of a consolidated international presence and (6) they should form a part of a network of firms with a minimum of three members. Regarding this last criterion, in order to ensure that interviewees focused their responses on their main business network, as an introduction the following definition was included, in which business networks are understood as *'the set of relationships that the firm maintains with other agents such as customers, suppliers, competitors, consultants, government agencies, universities, research centres, market research firms, advertising agencies and sales or distribution agents with the aim of obtaining knowledge, information, technology, resources or skills' ([86], p.24).*

The questionnaire was then pre-tested with a pilot sample of 25 firms to ensure it would be correctly understood. An electronic version was then prepared for the field work, of which 303 valid responses were returned (29.62% of the total).

An analysis of the primary data revealed the principal characteristics of the sample (**Table 1**). The firms are SMEs, mostly from the industrial sector (61.2% of the total), notably the agrifood, metal and textile sectors. Commerce is represented by 31.6%, including firms that export and import products related to the aforementioned industrial sectors. Finally, less than 7.2% of firms come from the services sector, including mostly financial, tourism and communication agencies. The firms have on average 28.55 employees, 41.50% of their commercialisation work is abroad, and their average age is 3.90 years.

Regarding the networks to which the firms belong (**Table 2**), almost all of them became part of the network on their creation (96.4% after their first year of life), they are usually networks with marketing aims (in 92.5% of cases) and are composed on average of 5.81 firms, located near the firm (62.19% of the networks have a regional scope).

Years of international experience		Total number of employees		International activities in	Activity sector		
Years elapsed	%	Employees	%	Activity	Average%	Sector	%
0	76.6	3–5	26.4	Manufacturing process	11.58	Industrial	61.2
1	19.8	6–11	25.4	Research and development	7.40		
2	2.1	12–25	23.8	Commercialisation	41.50	Commercial	31.6
3	1.6	Over 25	24.4	Advertising and promotion	11.18		
Average years elapsed = 0.28		Average employees = 28.55		After-sales service	16.51	Services	7.2
				Average international act			

Note: 'Figures expressed as a percentage of total responses.

Table 1. General characteristics of the firms.

Years since creation to entrance in the network		Network size		Geographical scope of the network ¹		Type of network		
Years elapsed	%	Firms	%	Scope	%	Туре	%	
0	82.4	3	55.6	Self-governing	62.19	Social network	2.5	
1	13.5	4–5	19.4	region		Technological network	14.9	
2	1.4	6–10	11.1	National	15.42	Institutional network	2.0	
3	2.7	>10	13.9			Infrastructure network	5.5	
Average years		Average	e firms = 5.81	International	22.39	Marketing network	92.5	
elapsed = 0.12					Market network	15.4		

¹According to the location of most network members.

Table 2. General characteristics of the main networks of the studied firms.

3.2. Measurement instruments

To measure the market orientation of the network we adapted the scale proposed by Helfert et al. [37]. These authors move away from the idea defended in previous studies of measuring the network's market orientation by simply adapting the dimensions used in the seminal scales of MO. Specifically, this scale includes a total of 12 items on the four dimensions that reflect the relational processes management of NMO: coupling (2 items), coordination (3 items), conflict resolution (3 items) and exchange (4 items).

To measure absorptive capability, we use the three-item scale of Ref. [87] that evaluates the degree to which the firm's management systems encourage the ability to acquire, assimilate, transform and exploit knowledge.

Competitive advantages are measured with the scale developed by Ling-Yee and Ogunmokun [71] and the proposals made by Morgan et al. [72] and Ha-Brookshire and Dyer [73]. This scale reflects the firm's position in relation to competition in terms of differentiation and costs when adapting its offer to international market needs.

Finally, we use an adaptation of the scale proposed by Jantunen et al. [81] to measure international performance. Specifically, we asked BG managers about their degree of satisfaction with their activity in terms of turnover, profitability, market share and global satisfaction.

3.3. Validity and scale reliability

To refine the scales, a confirmatory factor analysis was performed using structural equations models. The analyses guarantee a measurement model consistent with the theoretical proposals, supported by scales that are reliable, valid and present a certain degree of unidimensionality.

Based on the recommendations of Jöreskog and Söbom [88], we first examined the estimation parameters. We removed those indicators with standardised coefficients (λ) under 0.7, significance of the Student *t* statistic under 2.58 (*P*=0.01) and *R*² under 0.49, thus ensuring that the strong and weak convergence conditions were met [89]. This process led to the removal of the indicators EXCH.3 from the NMO scale, ABS.1 from the absorption capability scale, and CACOS.3, CADIF.3 and CADIF.4 from the competitive advantages based on costs and differentiation, respectively. Several tests were then performed to verify whether or not the process of refinement of the scales had altered their level of reliability. We used Cronbach's alpha [90] to analyse internal consistency. Other complementary tests of reliability were carried out: the composite reliability of the construct and the analysis of the average variance extracted (see **Table 3**).

Finally, the convergent and discriminant validity were analysed. With reference to the former, it was sufficient to observe that the estimated value of the correlations between the dimensions configuring the scales was high and significant. The confidence interval test was performed to examine discriminant validity, verifying that '1' was not found in the intervals estimated for the correlations between each pair of dimensions (**Table 4**). The measurement model proposed is therefore reliable and valid for use in the testing of hypotheses.

Further tests were also carried out. First, we checked for signs of multicollinearity by testing the variance inflation factor (VIF) among latent variables in our proposed overall model. Values were below 10 [91], suggesting multicollinearity was not an issue in our study. Second, a *t*-test of independent means was performed on the dimensions of the variables in the proposed model. This test was conducted using the first 45 and last 45 respondents. No significant differences were found between these respondents at the 0.05 level, indicating an absence of non-response bias [92]. Third, various ANOVA were run to confirm that sample characteristics do not affect the model constructs. The following control variables were used, based on the data gathered in the questionnaire: sector of activity, international consolidation, age, international seniority, size, seniority in the network (all firm-related variables) and size of the network. Results revealed no significant differences in any of the analyses. Finally, the possibility of common method variance bias was tested with Harman's test, concluding that the bias caused by the method used was not a problem for the validity of the results obtained in the subsequent testing of the hypotheses [93, 94].

Scale	1st order								2nd order
	PERF	ABS	CACOS	CADIF	COUP	COOR	CONF	EXCH	NMO
λ	0.74–0.87	0.72–0.84	0.85–0.86	0.71-0.82	0.81-0.91	0.73–0.90	0.77–0.91	0.72-0.87	0.76-0.95
α	0.887	0.749	0.841	0.732	0.845	0.871	0.887	0.847	NP
CR	0.887	0.753	0.842	0.738	0.850	0.876	0.891	0.857	0.915
AVE	0.612	0.606	0.727	0.586	0.741	0.704	0.732	0.667	0.731
Significant	loads		All <i>t</i> > 2.58			R^2	All $R^2 > 0$.49	
χ^2/df	RMSEA	SRMR	GFI	AGF	I NF	[N	NFI	IFI	CFI
1.726	0.049	0.050	0.894	0.864	0.90	01 0.	947	0.956	0.955

Table 3. Summary of the results after the definitive factor analysis.

Pair of variables	CACOS-ABS	CACOS-CADIF	CACOS-NMO	ABS-CADIF	ABS-NMO
I.C.	[0.143; 0.415]	[0.145; 0.421]	[0.124; 0.376]	[0.148; 0.436]	[0.198; 0.454]
Pair of variables	CADIF-NMO	CACOS-PERF	ABS-PERF	CADIF-PERF	NMO-PERF
I.C.	[0.036; 0.312]	[0.117; 0.369]	[0.049; 0.321]	[0.154; 0.418]	[0.279; 0.503]

Table 4. Discriminant validity analysis using the confidence interval (CI) test.

4. Results

Research in business-related sciences has gradually been enriched by the introduction of more sophisticated methodologies. This greater degree of development has enabled researchers to design and test increasingly complex models addressed to explaining the business reality.

Structural equations models have emerged from the evolution of multi-equation modelling developed chiefly in econometrics and combined with the principles of measurement from psychology and sociology, aimed at overcoming the limitation shared by other multivariate techniques such as multiple regression, factor analysis, multivariate analysis of variance or discriminant analysis, all of which can examine just one relation at a time [91]. Unlike the above-mentioned techniques, a structural equations model (SEM)-based analysis is able to explore a series of dependent relations simultaneously [91]. Therefore, this technique is particularly useful when a dependent variable becomes an independent variable in subsequent dependency relations.

For this reason, as with the scale validation, SEM was used to test the hypotheses, since they have proved to be highly useful in a non-experimental situation when the study aims to uncover the causal influence from one factor to another [88].

The hypotheses that constitute the model of effects were tested using EQS 6.1 software. **Table 5** shows that none of the proposed hypotheses were refuted. First, as expected, NMO

presented a positive and significant effect on BGs' absorptive capability (H1: $\lambda = 0.355$; t = 4.931). Moreover, BGs' absorptive capability turned out to be a key variable in determining their achievement of competitive advantages, including differentiation-based competitive advantages (H2: $\lambda = 0.222$; t = 2.620) and cost-based competitive advantages (H3: $\lambda = 0.330$; t = 4.355). Finally each one of these variants of competitive advantage (differentiation-based and cost-based) showed a positive and significant influence on BGs' international results (H4: $\lambda = 0.208$; t = 3.050 and H5: $\lambda = 0.128$; t = 2.093, respectively).

5. Conclusions

This research was motivated by the large number of studies pointing to the important contribution BGs make to the progress of global economies. This circumstance has identified the study of the particularities that define BGs and that characterise their internationalisation process as a topic of major interest for academics and researchers [5, 6, 14, 17, 22]. Recent work in the international entrepreneurship field has urged researchers to contribute to the current understanding of firms' rapid internationalisation processes through richer empirical and theoretical studies [7, 8, 17, 25]. In this line, the Journal of International Business Studies will dedicate a special issue in 2017 to the determination of the factors that facilitate the creation and capture of entrepreneurial opportunities across national borders.

In general, our work adds to the growing support in the recent literature to the consideration of market knowledge as a key factor that determines the speed with which new companies adopt a proactive approach in crossing their national borders to seize new opportunities in

Relationshi	ip				Weight	t	Нур.	Result
$NMO \rightarrow BO$	Gs' absorptiv	ve capability	•	0.355	4.931***	H	Not refused	
BGs' absorj on BGs' dif	ptive capabi ferentiation	lity → Comp	oetitive adva	0.222	2.620**	H_2	Not refused	
BGs' absorj on BGs' cos	ptive capabi sts	lity →Comp	etitive adva	0.330	4.355***	H_3	Not refused	
Competitive advantages based on BGs' differentiation \rightarrow BGs' international performance					0.208	3.050**	H_4	Not refused
Competitive advantages based on BGs' differentiation \rightarrow BGs' international costs					0.128	2.093*	H_5	Not refused
Measureme	ents of quali	ty of fit						
χ^2 (gl)	RMSEA	SRMR	GFI	AGFI	NFI	NNFI	IFI	CFI
0.569 (231) 0.043 0.062 0.919 0.895					0.926	0.967	0.972	0.972
* = <i>p</i> <0.05.								
** = $p < 0.01$								
*** = <i>p</i> < 0.00)1.							

Table 5. Results of the estimation of standardised parameters for the model of effects.

other foreign markets [19, 56, 95–98]. As Javalgi et al. [99: p.15] point out, in current dynamic contexts BGs 'must listen to and correctly interpret the voice of the market. [...] Firms that lose touch with the market, that either ignore or misinterpret its signals, will fail in hyper-competitive environments'.

Specifically, our findings contribute to the specialised literature with various relevant theoretical and empirical implications. First, gradualist models emphasise experience as an essential factor in the firm's internationalisation process by centring on determining the most relevant sources of external market knowledge [1]. However, the experience derived from physical presence in foreign markets is minimal in the case of BGs due to their recent creation. Experience cannot therefore be considered as the main source of knowledge BGs use in their entry into foreign markets. For this reason, a growing number of scholars highlight the role of vicarious knowledge from relations in networks as key source of external information and knowledge in BGs [56, 57]. As Lin and Chen [65: p.160] note, 'relationships between firms in a network context constitute an inter-organisational scenario for the exchange, compilation, integration and development of resources and valuable complementary knowledge that originally come from individual agents'. The links a BG builds into its network from its creation give it access to external information, knowledge, experiences and resources that it can apply internally [22]. In line with these arguments, the construct of NMO can be seen as a reference for determining how BGs generate and manage vicarious knowledge.

Second, based on ref. [37], our research enables theoretical identification and empirical testing of the behaviours and basic mechanisms that characterise the development of a coordinated market orientation among firms in BG networks. Our study responds to previous calls emphasising the need to complete the assumptions of individual market orientation with a relational view [43, 100, 101]. These findings show that the market orientation construct must be applied at individual and also network level [101], thereby supplementing the existing literature on the importance of individual market orientation in BGs [102, 103].

Furthermore, our results show that the knowledge these firms derive from their marketoriented networks helps them to develop an absorptive dynamic capability in order to act sustainably in their international markets. Therefore, BGs' development of absorptive capability should not be understood as a completely internal process, but as a relational process. This finding contributes to the discussion around the lack of attention paid to identifying the main variables beyond BG competitiveness [13, 71–73]. Indeed, our study responds to some authors' remarks that, despite the growing interest shown in recent years as a result of the intense process of globalisation in the markets, the theoretical and empirical knowledge about the internal and external antecedents of the international performance of new companies remains insufficient [13, 72, 73, 104, 105]. Our study complements previous literature in which analysis of these variables has tended to adopt an individual approach, highlighting the relevance for BGs of developing a market orientation [102, 103], an entrepreneurial orientation [3, 106, 107] or specific capabilities such as marketing and managerial capabilities [7, 22, 108, 109].
In sum, it is empirically demonstrated that adopting an orientation grounded in market knowledge and social networks (NMO) promotes the development by BGs of a dynamic absorptive capability that contributes to the systematic and effective adaptation to the dynamic and changing markets in which they operate, and the exploitation of the valuable knowledge generated. Our study integrates theoretical approaches from international business and entrepreneurship, providing a new framework that improves our understanding of the central role of knowledge in the way BGs capture and create opportunities across national borders.

6. Limitations and future research

The interpretation of the conclusions derived from this study should take into account some limitations, which lead us to propose certain future lines of research.

Regarding the theoretical limitations, our study proposes one particular model of effects to which other new relationships may be added, taking into account additional variables to complete the explanation of the achievement of competitive advantages and international performance by BGs.

Additionally, considering that our study complements the results from previous works that demonstrate from an individual perspective the influence of BGs' market orientation on their competitiveness, future research could explore the relationship between an NMO and a particular market orientation.

Moreover, taking into account that BGs' main internationalisation decisions are related to the speed and the mode they enter new foreign markets, future studies could analyse the influence of an NMO on different levels of speed and modes of entry.

Methodological limitations include the use of a single nation sample, since our findings may not be generalised to other national contexts. Future research could replicate and contrast the hypotheses presented in our work in other countries. We also used a multi-sector sample. Considering that networks may differ in their behaviour, structure and performance, future research could explore sectors, facilitating a larger homogeneity of firms and networks.

Another methodological limitation concerns the use of a single interviewee response per network and firm. This raises two questions: (1) Can the manager of an individual firm respond to questions on how a network of companies works as a whole? and (2) Similarly, can a single interviewee represent an entire company in his or her responses?

Further doubts may arise on the question of who actually responded to the questionnaire, bearing in mind that it was distributed online.

Finally, to make causal inferences using cross-sectional data can limit the value of the results. New research could usefully analyse the proposed model of effects with longitudinal data.

Author details

Diego Monferrer Tirado* and Marta Estrada Guillén *Address all correspondence to: dmonferr@uji.es Universitat Jaume I de Castellón, Spain

References

- Johanson J, Vahlne JE. The internationalization process of the firm: A model of knowledge development and increasing foreign market commitment. Journal of International Business Studies. 1977;8(1):23-32
- [2] Oviatt BM, McDougall PP. Toward a theory of international new ventures. Journal of International Business Studies. 1994;25(1):45-64
- [3] Oviatt BM, McDougall PP. The internationalization of entrepreneurship. Journal of International Business Studies. 2005;**36**(1):2-8
- [4] Zahra SA, George G. Absorptive capacity: A review, reconceptualization, and extension. Academy of Management Review. 2002;**27**(2):185-203
- [5] Jones MV, Coviello N, Tang YK. International entrepreneurship research (1989-2009): A domain ontology and thematic analysis. Journal of Business Venturing. 2011;26(6):632-659
- [6] Mainela T, Puhakka V, Servais P. The concept of international opportunity in international entrepreneurship: A review and research agenda. International Journal of Management Reviews. 2014;16(1):105-129
- [7] Cavusgil ST, Knight G. The born global firm: An entrepreneurial and capabilities perspective on early and rapid internationalization. Journal of International Business Studies. 2015;46(1):3-16
- [8] Coviello N. Re-thinking research on born globals. Journal of International Business Studies. 2015;46(1):17-26
- [9] Lin S, Mercier C, Salloum C. The Chinese born globals of the Zhejiang Province: A study on the key factors for their rapid internationalization. Journal of International Entrepreneurship. 2016;14(1):75-95
- [10] Knight GA, Liesch PW. Internationalization: From incremental to born global. Journal of World Business. 2016;51(1):93-102
- [11] Autio E. Creative tension: The significance of Ben Oviatt's and Patricia McDougall's article: Toward a theory of international new ventures. Journal of International Business Studies. 2005;**36**:9-19
- [12] Zahra SA. A theory of international new ventures: A decade of research. Journal of International Business Studies. 2005;36(1):20-28

- [13] Monferrer D. La orientación al mercado de la red como determinante de la competitividad de las nuevas empresas internacionales. El efecto mediador de las capacidades dinámicas [doctoral dissertation]. Spain: Universitat Jaume I; 2011
- [14] Sarasvathy S, Kumar K, York JG, Bhagavatula S. An effectual approach to international entrepreneurship: Overlaps, challenges, and provocative possibilities. Entrepreneurship Theory & Practice. 2014;38(1):71-93
- [15] Gerschewski S, Rose EL, Lindsay VJ. Understanding the drivers of international performance for born global firms: An integrated perspective. Journal of World Business. 2015;50(3):558-575
- [16] Knight G. Born global firms: Evolution of a contemporary phenomenon. In: Zou S, Xu H, Shi LH, editors. Entrepreneurship in International Marketing. Emerald Group Publishing Limited, Bingley, United Kingdom; 2015. pp. 3-19
- [17] Zander I, McDougall P, Rose EL. Born globals and international business: Evolution of a field of research. Journal of International Business Studies. 2015;46(1):27-35
- [18] Loane S, Bell J. Rapid internationalisation among entrepreneurial firms in Australia, Canada, Ireland and New Zealand: An extension to the network approach. International Marketing Review. 2006;23(5):467-485
- [19] Grégoire DA, Barr PS, Shepherd DA. Cognitive process of opportunity recognition: The role of structural alignment. Organization Science. 2010;21(2):413-431
- [20] Alvarez SA, Barney JB, Anderson P. Forming and exploiting opportunities: The implications of discovery and creation processes for entrepreneurial and organizational research. Organization Science. 2013;24(1):301-317
- [21] Autio E, Sapienza HJ, Almeida JG. Effects of age at entry, knowledge intensity, and imitability on international growth. Academy of Management Journal. 2000;43:909-924
- [22] Monferrer D, Blesa A, Ripollés M. Catching dynamic capabilities through market-oriented networks. European Journal of International Management. 2015;9(3):384-408
- [23] McNaugthon RB. The number of export markets that a firm serves: Process models versus the born-global phenomenon. Journal of International Entrepreneurship. 2003;1:297-311
- [24] Knight GA, Liesch PW. Information internalisation in internationalising the firm. Journal of Business Research. 2002;55:981-995
- [25] Jones MV, Casulli L. International entrepreneurship: Exploring the logic and utility of individual experience through comparative reasoning approaches. Entrepreneurship Theory & Practice. 2014;38(1):45-69
- [26] Bhuian SN, Menguc B, Bell SJ. Just entrepreneurial enough: The moderating effect of entrepreneurship on the relationship between market orientation and performance. Journal of Business Research. 2005;58:9-17

- [27] Cadogan JW, Cui CC, Morgan RE, Story VM. Factors facilitating and impeding the development of export market-oriented behavior: A study of Hong Kong Manufacturing Exporters. Industrial Marketing Management. 2006;35:634-647
- [28] Qiu T. Scanning for competitive intelligence: A managerial perspective. European Journal of Marketing. 2008;**42**(8):814-835
- [29] Odorici V, Presutti M. The entrepreneurial experience and strategic orientation of hightech born global start-ups: An analysis of novice and habitual entrepreneurs. Journal of International Entrepreneurship. 2013;11(3):268-291
- [30] Tan O, Sousa CMP. Leveraging marketing capabilities into competitive advantage and export performance. International Marketing Review. 2015;**32**(1):78-102
- [31] Knight GA, Cavusgil T. Innovation, organizational capabilities, and the born-global firm. Journal of International Business Studies. 2004;35:124-141
- [32] Johanson J, Vahlne JE. Building a model of firm internationalisation. In: Blomstermo A, Sharma D, editors. Learning in the Internationalisation Process of Firms. Cheltenham: Edward Elgar; 2003. pp. 3-15
- [33] Chen M, Wang M. Social network and a new venture's innovative capability: The role of trust within entrepreneurial teams. R&D Management. 2008;38(3):253-264
- [34] Perks K, Hughes M. Entrepreneurial decision-making in internationalization: Propositions from mid-size firms. International Business Review. 2008;17(3):310-330
- [35] Ojala A. Internationalization of knowledge-intensive SMEs: The role of network relationships in the entry to a psychically distant market. International Business Review. 2009;18(1):50-59
- [36] Zahra S, Newey L, Li Y. On the frontiers: The implications of social entrepreneurship for international entrepreneurship. Entrepreneurship Theory & Practice. 2014;38(1):137-158
- [37] Helfert G, Ritter T, Walter A. Redefining market orientation from a relationship perspective. Theoretical considerations and empirical results. European Journal of Marketing. 2002;36(9/10):1119-1139
- [38] Evanschitzky H. Market orientation of service networks: Direct and indirect effects on sustained competitive advantage. Journal of Strategic Marketing. 2007;15:349-368
- [39] Siguaw JA, Simpson PM, Baker TL. Effects of supplier market orientation in distributor market orientation and the channel relationship: The distributor perspective. Journal of Marketing. 1998;62:99-111
- [40] Monferrer D, Blesa A, Ripollés M, Kuster I, Vila N. Effects of network market orientation on new ventures' international performance. International Journal of Business Environment. 2013;5(3):268-298
- [41] Lippman SA, Rumelt, RP. Uncertain imitability: An analysis of interfirm differences in efficiency under competition. Bell Journal of Economics. 1982;13:418-438

- [42] Zollo M, Winter SG. Deliberate learning and the evolution of dynamic capabilities. Organization Science. 2002;13:339-351
- [43] Behyan M. Analyzing correlation between internationalization orientation and social network. International Economics Studies. 2014;43(2):47-56
- [44] Teece DJ, Pisano G, Shuen A. Dynamic capabilities and strategic management. Strategic Management Journal. 1997;18(7):509-533
- [45] Eisenhardt KM, Martin JA. Dynamic capabilities: What are they? Strategic Management Journal. 2000;21:1105-1121
- [46] O'Reilly CA, Tushman ML. Ambidexterity as a dynamic capability: Resolving the innovator's dilemma. Research in Organizational Behavior. 2008;28:185
- [47] Zhou KZ, Li CB. How strategic orientations influence the building of dynamic capability in emerging economies. Journal of Business Research. 2009;63(3):224-231
- [48] Lichtenthaler U. Determinants of absorptive capacity: The value of technology and market orientation for external knowledge acquisition. Journal of Business & Industrial Marketing. 2016;31(5):600-610
- [49] Cohen WM, Levinthal DA. Absorptive capacity. A new perspective on learning and innovation. Administrative Science Quarterly. 1990;35(1):128-152
- [50] Tsai W. Knowledge transfer in intraorganizational networks: Effects of network position and absorptive capacity on business unit innovation and performance. Academy of Management Journal. 2001;44(5):996-1004
- [51] Lenox M, King A. Prospects for developing absorptive capacity through internal information provision. Strategic Management Journal. 2004;25(4):331-345
- [52] Wang C, Ahmed PK. Dynamic capabilities: A review and research agenda. International Journal of Management Reviews. 2007;9(1):31-51
- [53] Nätti S, Hurmelinna P, Johnston WJ. Absorptive capacity and network orchestration in innovation communities—Promoting service innovation. Journal of Business & Industrial Marketing. 2014;29(2):173-184
- [54] Belso JA, Expósito M, Tomás JV. Knowledge network dynamics in clusters: Past performance and absorptive capacity. Baltic Journal of Management. 2016;11(3):310-327
- [55] De Clercq D, Castañer X, Belausteguigoitia I. Entrepreneurial initiative selling within organizations: Towards a more comprehensive motivational framework. Journal of Management Studies. 2011;48:1269-1290
- [56] Bell VA, Cooper SY. Acquisition of knowledge in networking for internationalisation. In: Groen A, Cook G, Van Der Sijde P, editors. New Technology-Based Firms in the New Millennium. Emerald Group Publishing Limited, Bingley, United Kingdom; 2015. pp. 29-53
- [57] Oxtorp LA, Elg U. The Alliance capability of technology-based born globals. In: Zou S, Xu H, Shi LH, editors. Entrepreneurship in International Marketing. Emerald Group Publishing Limited, Bingley, United Kingdom; 2015. pp. 73-107

- [58] Stinchcombe AL. Organizations and social structure. In: March JG, editor. Handbook of Organizations. Chicago: Rand McNally; 1965. pp. 142-193
- [59] Baum JAC. Organizational ecology. In: Clegg S, Hardy C, Nord W, editors. Handbook of Organization Studies. London: Sage; 1996. pp. 77-114
- [60] Zaheer S. Overcoming the liability of foreignness. Academy of Management Journal. 1995;38(2):341-363
- [61] Baucus DA, Baucus MS, Human SE. Consensus in franchise organizations: A cooperative arrangement among entrepreneurs. Journal of Business Venturing. 1996;11(5):359-378
- [62] Larson A. Partner networks: Leveraging external ties to improve entrepreneurial performance. Journal of Business Venturing. 1991;6(3):173-188
- [63] Ramachandran K, Ramnarayan S. Entrepreneurial orientation and networking: Some Indian evidence. Journal of Business Venturing. 1993;8(6):513-524
- [64] Johanson J, Vahlne JE. The Uppsala internationalization process model revisited: From liability of foreignness to liability of outsidership. Journal of International Business Studies. 2009;40:1411-1431
- [65] Lin BW, Chen CJ. Featuring product innovation in industry networks: The mediating role of knowledge integration. International Journal of Human Resource Management. 2006;17(1):155-173
- [66] Chung J, Jin B, Sternquist B. The role of market orientation in channel relationships when channel power is imbalanced. The International Review of Retail, Distribution and Consumer Research. 2007;**17**(2):159-176
- [67] Arregle JL, Naldi L, Nordqvist M, Hitt MA. Internationalization of family-controlled firms: A study of the effects of external involvement in governance. Entrepreneurship Theory & Practice. 2012;36(6):1115-1143
- [68] Beverland MB, Lindgreen A. Implementing market orientation in industrial firms: A multiple case study. Industrial Marketing Management. 2007;36(4):430-442
- [69] Nahapiet J, Ghoshal S. Social capital, intellectual capital, and the organizational advantage. Academy of Management Review. 1998;23(2):242-266
- [70] Porter ME. Competitive Advantage: Creating and Sustaining Superior Performance. New York: Free Press; 1985
- [71] Ling-Yee L, Ogunmokun GO. The influence of interfirm relational capabilities on export advantage and performance: An empirical analysis. International Business Review. 2001;10:399-420
- [72] Morgan N, Kaleka A, Katsikeas C. Antecedents of export venture performance: A theoretical model and empirical assessment. Journal of Marketing. 2004;68:90-108
- [73] Ha-Brookshire JA, Dyer B. The impact of firm capabilities and competitive advantages on import intermediary performance. Journal of Global Marketing. 2009;22(1):5-19

- [74] Lynskey MJ. Determinants of innovative activity in Japanese technology-based start-up firms. International Small Business Journal. 2004;**22**(2):159-196
- [75] Laursen K, Salter A. Open for Innovation: The role of openness in explaining innovative performance among U.K. manufacturing firms. Strategic Management Journal. 2006;27:131-150
- [76] Roper S, Youtie J, Shapira P, Fernández A. Knowledge, capabilities and manufacturing innovation: A USA-Europe comparison. Regional Studies. 2010;44(3):253-279
- [77] Vinding AL. Absorptive capacity and innovative performance: A human capital approach. Economics of Innovation & New Technology. 2006;**15**(4/5):507-517
- [78] Fosfuri A, Tribó JA. Exploring the antecedents of potential absorptive capacity and its impact on innovative performance. Omega. 2008;36:173-187
- [79] Chuang MY, Chen CJ, Lin MJ. The impact of social capital on competitive advantage: The mediating effects of collective learning and absorptive capacity. Management Decision. 2016;54(6):1443-1463
- [80] Jurksiene L, Pundziene A. The relationship between dynamic capabilities and firm competitive advantage: The mediating role of organizational ambidexterity. European Business Review. 2016;28(4):431-448
- [81] Jantunen A, Nummela N, Puumalinen K, Saarenketo S. Strategic orientation of born globals—Do they really matter? Journal of World Business. 2008;43(2):158-170
- [82] Grant RM. Toward a knowledge-based theory of the firm. Strategic Management Journal. 1996;17:109-122
- [83] Snoj B, Milfelner B, Gabrijan V. An examination of the relationships among market orientation, innovation resources, reputational resources, and company performance in the transitional economy of Slovenia. Canadian Journal of Administrative Sciences. 2007;24(3):151-164
- [84] Mizik N, Jacobson R. Trading off between value creation and value appropriation: The financial implications of shifts in strategic emphasis. Journal of Marketing. 2003;67(January):63-76
- [85] Kamukama N, Ahiauzu A, Ntayi JM. Competitive advantage: Mediator of intellectual capital and performance. Journal of Intellectual Capital. 2011;12(1):152-164
- [86] Monferrer D, Blesa A, Ripollés M. Born globals trough knowledge-based dynamic capabilities and network market orientation. Business Research Quarterly. 2015;18(1):18-36
- [87] Chen YS, Lin MJJ, Chang CH. The positive effects of relationship learning and absorptive capability on innovation performance and competitive advantage in industrial markets. Industrial Marketing Management. 2009;38:152-158
- [88] Jöreskog K, Söbom D. LISREL 8: Structural Equation Modeling with the SIMPLIS Command Language. Chicago, Illinois: Scientific Software International; 1993

- [89] Steenkamp JBEM, van Trijp HCM. The use of lisrel in validating marketing constructs. International Journal of Research in Marketing. 1991;8(4):283-299
- [90] Nunnaly J. Psychometric Theory. Nueva York: McGraw-Hill; 1979
- [91] Hair JF, Black WC, Babin BJ, Anderson RE, Tatham RL. Multivariate Data Analysis. 6th ed. Upper Saddle River, N.J: Pearson Education Inc; 2006
- [92] Armstrong JS, Overton TS. Estimating non-response bias in mail surveys. Journal of Marketing Research. 1977;16(August):396-402
- [93] Podsakoff PM, MacKenzie SB, Lee J, Podsakoff N. Common method biases in behavioral research: A critical review of the literature and recommended remedies. Journal of Applied Psychology. 2003;88:879-903
- [94] Friedrich TL, Byrne CL, Mumford MD. Methodological and theoretical considerations in survey research. The Leadership Quarterly. 2009;20:57-60
- [95] Saarenketo S, Puumalainen K, Kyläheiko K, Kuivalainen O. Linking knowledge and internationalization in ICT SMEs. Technovation. 2008;28:591-601
- [96] Brennan L, Garvey D. The role of knowledge in internationalization. Research in International Business and Finance. 2009;**23**(2):120-133
- [97] Liao SH, Chang WJ, Wu CC. An integrated model of learning organization with strategic view: Benchmarking in the knowledge-intensive industry. Expert Systems with Applications. 2010;37(5):3792-3798
- [98] Fuerst S, Zettinig P. Knowledge creation dynamics within the international new venture. European Business Review. 2015;27(2):182-213
- [99] Javalgi RG, Martin LC, Young RB. Marketing research, market orientation and customer relationship management; a framework and implications for service providers. Journal of Service Marketing. 2006;20(1):12-24
- [100] Tuominen M, Rajala A, Möller K. Market-driving versus market-driven: Divergent roles of market orientation in business relationships. Industrial Marketing Management. 2004;33:207-217
- [101] Racela O, Chaikittisilpa C, Thoumrungroje A. Market orientation, international business relationships and perceived export performance. International Marketing Review. 2007;24(2):144-163
- [102] Ripollés M, Blesa A, Monferrer D. Factors enhancing the choice of higher resource commitment entry modes in international new ventures. International Business Review. 2012;21:648-666
- [103] Madsen TK, Sørensen HE, Torres R. The market orientation of domestic and international new ventures. In: Zou S, Xu H, Shi LH, editors. Entrepreneurship in International Marketing. Emerald Group Publishing Limited, Bingley, United Kingdom; 2015. pp. 21-44

- [104] Baronchelli G, Cassia F. Exploring the antecedents of born-global companies' international development. International Entrepreneurship and Management Journal. 2014;10(1):67-79
- [105] Omri W, Becuwe A. Managerial characteristics and entrepreneurial internationalization: A study of Tunisian SMEs. Journal of International Entrepreneurship. 2014;**12**(1):8-42
- [106] Kocak A, Abimbola T. The effects of entrepreneurial marketing on born global performance. International Marketing Review. 2009;26(4/5):439-452
- [107] Kumar N, Yakhlef A. The effects of entrepreneurial marketing strategies on the longterm competitive sustenance of born global firms: Examples from the Indian knowledge-intensive services industry. In: Zou S, Xu H, Shi LH, editors. Entrepreneurship in International Marketing. Emerald Group Publishing Limited, Bingley, United Kingdom; 2015. pp. 45-72
- [108] Fernhaber SA, McDougall PP. New venture growth in international markets: The role of strategic adaptation and networking capabilities. In: Shepherd DA, Katz J, editors. Advances in Entrepreneurship, Firm Emergence and Growth. Vol. 8. Greenwich, CT: JAI Press; 2005
- [109] Kumar N, Yakhlef A. How capabilities evolve in a born global firm?: A case study of an Indian knowledge-intensive service born global firm. Journal of Entrepreneurship in Emerging Economies. 2014;6(3):223-242

Knowledge Management Hybrid Strategy with People, Technology and Process Pillars

Ivy Chan

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/intechopen.70072

Abstract

The importance of knowledge management (KM) contributes organizational competitiveness, which is widely addressed and became a central topic of management agenda in the last decade. This chapter examines three major KM pillars, including people, process, and technology, for effective KM deployment. Based on a questionnaire-based survey, the study investigates the perceived importance of three KM pillars that influence organizational inclinations of KM strategies and ultimately affect organizational performance. Quantitative findings are sought from 44 key informants in organizations. The results show a hybrid strategy that balances the importance of people, process, and technology pillars brings desirable impact on organizational performance, comparing with the KM strategy inclined to a particular KM pillar. Recommendations of KM endeavors on three KM pillars are provided to suggest the joint efforts from both management and employees.

Keywords: knowledge management, process, people, technology, hybrid strategy

1. Introduction

Knowledge management (KM) has been recognized as an imperative element for leveraging organizational effectiveness and performance. Organizations practicing effective KM methods achieve positive results in their organizational competitiveness, particularly innovation and creativity [1–3]. Despite the positive effects of KM, research on the KM pillars remains diverse. This study examines the interplay of the three major KM pillars acting as the platform for effective KM instigation.

On the basis of interviews with KM-intensive organizations, their good practices were categorized through a thematic analysis. Thereafter, an exploratory quantitative study from a sample



of 44 respondents in different organizations was conducted to examine the relationship of the three major KM pillars. The survey results were further examined to evaluate the effects on organizational performance. Organizations harnessing a hybrid strategy to balance deployment of people, process, and technology gain positive results in their performance. The findings can provide direction for future studies to facilitate management in the deployment and integration of the KM pillars for attaining desirable organizational outcomes.

2. Knowledge management pillars

To understand effective KM practices, interviews were conducted with the senior management from two locally renowned KM-intensive organizations; the interviewees serve as key informants who steer and propagate KM. The qualitative data from these organizations were analyzed and thematized into three essential KM pillars, namely, people, process, and technology, which constitute organizational performance (**Figure 1**) [4–7].

2.1. Pillar of people

KM is embedded in working processes and practiced by each individual at different organizational levels, spanning from the senior-most to junior-most personnel. Typically, top managers champion the instigation of KM programs, provide vision to align the organizational strategy with the KM programs, and oversee the diffusion of KM initiatives throughout their organizations [8]. Members at different organizational levels act as knowledge workers who harness and utilize the knowledge assets residing in their cognitive repositories [9]. Through collective inquiry, sharing, and assimilation of knowledge, innovations and desirable organizational outcomes are boosted [1, 10]. Therefore, people are considered the heart of leveraging and creating knowledge for organizational competitiveness. Their cognitive minds influence their



Figure 1. Three knowledge management pillars.

approach to the processes of accommodating, validating, and creating different ideas solicited from different sources.

2.2. Pillar of technology

Using communication and collaboration technologies to support knowledge management is ubiquitous. Its effect is universally discussed as an indispensable means to facilitate the acquiring, codifying, indexing, updating, and disseminating of knowledge among employees [10–12]. Organizations invest in KM technology, such as document management systems and yellow pages, which enables the presentation of captured knowledge in readily available forms for different users.

Equipped with collaborative-oriented KM technology, employees can connect to one another within (e.g., through organization-specific intranets) and outside organizations (e.g., through the Internet) for rapid knowledge flows with enhanced time value. Employees using KM system (KMS) with learning and creation intention aim to capitalize knowledge assets through social networking and collective inquiries [12, 13]. With trust and reciprocity of exchange, employees share resembling identity over communities of practices to explore or exploit more new ideas and collaboration. User-oriented KM systems (KMS) supports, such as training workshops and forums, may further assist the adoption of KM processes in daily operations.

2.3. Pillar of process

Knowledge is mostly characterized by its tacit and intricate nature, and it resides in the mind of individuals [14–17]. Through individual inquiry and exploration, knowledge is activated from one's repository and externalized in different formats. By engaging through group dialogue, interaction, and exchange, knowledge can be pooled from different sources and created into different explicit stances. KM processes can generally be defined as an array of designated practices facilitating the flow and added value of knowledge. These processes not only help organizations identify and acquire knowledge from multiple sources but also allow their employees to explicate and disseminate knowledge in comprehensible formats. Valuable knowledge, skills, and competencies are documented and stored in knowledge repositories assuming various forms (e.g., minds of employees). Structured and planned documentations enable employees to share and retrieve relevant knowledge for implementing associated tasks. Employees can also assimilate new knowledge input to reconfigure existing knowledge and create new ideas for enhancing organizational goals [18, 19].

3. Organizational performance

A central tenet of harmonizing the three KM pillars is their association with organizational performance. Prior research indicates that effective human resources deployment, such as organizing employees to work as a team to leverage collective expertise, can be conducive to innovative activities [13, 20]. Management should also integrate KM processes allowing employees to harness, access, share, use, and create knowledge at different stances [21].

Equipped with KM technology, employees can leverage personal knowledge and improve skills through sharing and collaborative learning [9].

In this study, management who can utilize knowledge can leverage their capabilities to improve business profitability, streamline working processes, and influence coordination of efforts as well as responsiveness to market-changing innovation [16, 18, 22], thus contributing to desirable organizational performance.

4. Perceived importance, practice, and performance

Prior study explained that although most companies find KM promising, they can only capitalize on a few processes [19]. They further asserted that action is vital to turn knowledge into practice, which, in turn, allows knowledge workers to learn from mistakes and move on to the next stage. As an exploratory examination, the current study focuses on the practice of the three KM pillars in organizations. The relative importance and hierarchical position of the three KM pillars are then examined. The perceived importance of KM pillar is construed to influence the way management steers the KM program. Accordingly, the congruence between perception and the KM orientation is investigated. Given the different KM strategies and mix of the KM pillars, organizational performance is expected to vary. The notions are illustrated in **Figure 2**.



Figure 2. Framework of perceived importance, practice, and performance.

5. Methodology and data collection

This exploratory study employed questionnaire-based survey for data collection. Pilot tests with one professor and one business practitioner were conducted in order to solicit feedback on the structure, readability, and completeness of the questionnaire. In 2013, the revised questionnaires were distributed to 93 study informants, with a cover letter to depict the major objective of the study. To improve the understanding of information from respondents with conception and experience of KM, study informants who engage, steer, or participate in organizational KM were primarily solicited to participate in the survey. The data collection period lasted about 5 months, with 44 valid questionnaires were returned for analysis.

In order to minimize the social desirability bias, anonymity was stated explicitly to all study informants in the cover letter. The study mainly examined the constructs of perceived

importance of the KM pillars, practice of KM pillars, and organizational performance. To assess the interplay of the three pillars and their effects on organizational performance, the study also conducted tests of the correlations of the three pillars with the organizational performance. Each pillar was measured with multiple questions. Except for questions regarding the demographic background of respondents and the company information, all questions adopted a five-point Likert scale (**Table 1**).

Characteristics	Number of respondents	Percentage of respondents
Gender		
Male	29	65.91
Female	15	34.09
Age group		
Under 25	2	4.54
25–34	10	22.73
35–44	30	68.19
45–54	2	4.54
Business sector		
Banking and insurance	6	13.63
Engineering	10	22.73
IT and telecommunication	10	22.73
Manufacturing	5	11.36
Wholesale, retail, export, and import	13	29.55

Table 1. Profile of respondents.

6. Results

To assess the interrelationships among the three KM pillars, descriptive statistics and correlation coefficients were derived with the aid of Statistical Package for the Social Sciences (SPSS). The perceptions of respondents were also examined to evaluate whether the three pillars influence how they harness knowledge assets.

6.1. Perceived importance

All 44 respondents expressed a unanimous agreement toward the importance of the three KM pillars, namely, people, technology, and process, to organizational growth. The awareness and recognition toward the three pillars are presumed to influence the KM agenda and endeavors in their organizations. Upon their consensus, the respondents were asked to rank the order of the three pillars in their organizations according to importance. Two diverse views were identified from the respondents: (1) the three pillars are conceived as equally important and (2) a specific KM pillar is more crucial than the other two KM pillars. In **Figure 3**, nearly half of the respondents (45.4%, 20) explicated that people, process, and technology are inseparable and valued equally significant in their organizations. The rest of the respondents (54.6%, 24) perceived their organizations to have dissimilar emphasis over the three KM pillars. This dissimilarity accounted for the diversity in their organizational profile, history, competitive edges, and environment.

The 24 respondents were further asked to reflect their views toward the most important and rudimentary pillar in their organization and rank the three pillars accordingly (from the most to the least important pillar). The result is illustrated in **Figure 4**; 11 respondents (25.0%) perceived "people" as the most important pillar, followed by eight respondents (18.2%) for "technology," and finally five respondents (11.4%) for "process."

Apart from the ranking order of KM pillars, the 24 respondents were asked to reveal their perception toward the degree of importance of the KM pillars. The study employed a 5-point Likert scale (ranging from 5 = most important to 1 = least important) and computed the mean scores accordingly. The higher the mean, the higher the perceived importance of the particular pillar toward the organizational performance. Among the three KM pillars, "people" was rated with a mean score of 4.19, which was higher than "technology" and "process" pillars with respective scores of 3.88 and 3.71.

6.2. Practice

Emanating from theory of action advocated by Argyris and Donald [23], individuals are encompassed with a disparity between their "espoused" theory and theory in use. For example, participants in community of practice clearly know the benefits of knowledge sharing. However, in practice, employees may not explicate or externalize their knowledge continually because of different private agendas held or reciprocity toward community members.



Figure 3. Perception of the most important knowledge management pillar.

Knowledge Management Hybrid Strategy with People, Technology and Process Pillars 75 http://dx.doi.org/10.5772/intechopen.70072



Figure 4. Perceived importance and inclination of knowledge management pillars.

A similar assertion is found in the current study. Despite the slight disparity, the perceived importance of KM pillars communicated to others is realized to be only partially congruent to the respective KM pillar deployment. To illustrate this point, all 44 respondents were further asked to evaluate the inclination of their KM strategy. The findings (**Figure 4**) presented four major KM strategies adopted in the organizations, namely, hybrid, people-oriented, technology-oriented, and process-oriented [24, 25].

In connection with the previous finding, 18 of the 20 respondents revealing the equal importance of the three KM pillars asserted that a "hybrid" strategy of KM practice is deployed in their organizations. Their KM plans incorporated and assimilated the three KM pillars to leverage people to engage in various KM processes, with the aid of KM-related technology to drive innovation and organizational improvement. They emphasized the interdependence and indispensability of people, process, and technology enabling organizational members to explore and exploit different types of knowledge.

Likewise, 10 of 11 respondents discerning "people" as the most important KM pillar, asserted that people-oriented KM strategy is carried out in their organizations. They emphasized that sources of innovation and new ideas are primarily instigated from people, given that most of the knowledge are tacit in nature and deeply residing in the mind of individuals. The organizations deploy diverse groups of KM people to articulate, interpret, and share knowledge among one another.

Concerning the supremacy of the ubiquitous technology in organizations, 11 respondents asserted that the technology-oriented strategy is adopted in their organizations, whereas only 8 respondents conceived "technology" as the most important pillar in the previous session.

The 11 respondents explained that technology is extensively used in their workplace to connect, communicate, and collaborate with parties in and outside the organization. Technology facilitates the integrative (e.g., new and old knowledge combination) and interactive flow (e.g., different knowledge workers exchange knowledge) of knowledge assets.

The five respondents valuing "process" pillar mostly concurred that KM strategies are primarily process oriented. The process-oriented strategy is characterized as a deliberated series of KM activities, including acquiring, storing, retrieving, reusing, applying, and creating knowledge, sequentially integrating with other organizational operations through the knowledge workers or technology-enhanced platforms.

Other than the congruence of the KM perception and KM strategy, understanding the effectiveness of their KM practices is important to evaluate organizational performance. The general results of organizational performance presented in **Table 2** showed that organizations adopting a "hybrid" strategy attained better scores than those organizations adopting KM strategies driven by a particular KM pillar. The 18 organizations demonstrated and experienced the highest organizational performance (mean = 4.32); the three KM pillars were welladjusted and developed, resulting in moderately high scores of 4.28, 4.08, and 4.11 for people, technology, and process, respectively.

Organizations with KM inclinations showed interesting findings with regard to KM effectiveness. The results from people-oriented organizations revealed that the pillar of people performed the best with the mean score of 3.87, followed by process and technology with mean scores of 2.97 and 2.90, respectively. Technology-oriented organizations deployed efforts and realized highest effectiveness in the pillar of technology when compared with the results of other two pillars (technology = 4.36, process = 3.12, people = 2.33). The KM effectiveness of process- and people-oriented organizations demonstrated a similar pattern. In process-oriented organizations, the most effective KM pillar is process (mean = 4.0), followed by people and technology, which shared the same mean value of 3.40.

Recognizing the organizational performance of other non-hybrid organizations with less favorable results is necessary (**Figures 5–7**): people-oriented, process-oriented, and technol-ogy-oriented organizations obtained a mean of 3.47, 3.4, and 3.06, respectively. Although the inclination toward a particular KM pillar enables organizations to exploit their KM resources, the inattentive practice or under-utilization of other KM pillars may hinder their long-term

		Strategy				
		Hybrid	People	Technology	Process	
Pillar	People	4.28	3.87	2.33	3.40	
	Technology	4.08	2.90	4.36	3.40	
	Process	4.11	2.97	3.12	4.00	
	Organizational performance	4.32	3.47	3.06	3.40	

Table 2. Correlation between knowledge management pillars and knowledge management strategy.

Knowledge Management Hybrid Strategy with People, Technology and Process Pillars 77 http://dx.doi.org/10.5772/intechopen.70072



Figure 5. Organizational performance by hybrid strategy and people-oriented strategy.

growth in today's dynamic environment. Given the intricate nature of knowledge process, high mobility of the knowledge workers and swift change in advanced technology as well as support and championship from management are paramount for encouraging organizational members to explore the current knowledge sources in organizations or to acquire the pillars externally (e.g., recruitment of quality staff).



Figure 6. Organizational performance by hybrid strategy and technology-oriented strategy.



Figure 7. Organizational performance by hybrid strategy and process-oriented strategy.

The descriptive statistical results indicated that the gap between espoused theory (regarding their perceived importance) and theory in use (regarding the inclination practice) was further evaluated. **Figure 8** illustrates the correlation coefficients of the three major pillars. The perceived importance of KM pillars showed a relatively strong relationship with the KM practice (0.80). Most of the organizations are consistent with what they believe and communicate to others in regard to their KM strategies. No obvious disparity exists between their degree of championing KM and the degree they engage in KM. Similarly, the results demonstrated a strong relationship between the KM practice and organizational performance, with a correlation coefficient of 0.87. The KM strategy steered by management is important in promoting the synergistic coordination of different organizational resources to achieve desirable organizational results.



Figure 8. Correlation of perceived importance, practice, and organizational performance.

7. Discussion and KM in the future

The preliminary findings reveal that management and employees have variable perceptions toward the relative importance of the three KM pillars. Their inclined perceptions influence the degree of KM pillar being practiced and exploited in KM programs. The current study also shows that organizations propagating hybrid KM strategy result in relatively higher effectiveness in organizational performance than those non-hybrid organizations do. The skewed emphasis on a particular KM pillar, such as technology orientation, seems like a one-legged chair that is rigidly stacked at a particular point.

Given the dynamic change of business environment, organizations not only have to maximize the effectiveness of organizational routines but also harness changes and develop new competitive strengths. The findings from current study encourage joint efforts from management and employees to configure a hybrid KM strategy, that is beyond the two-dimensional classification of KM strategies, either personalization or codification [26, 27]. The following section provides recommendation to management for courses of action to explore more on KM pillars that may be least attended or emphasized in current stage, ultimately to attain a desirable organizational performance.

7.1. Pillar of people

The attitude of the top management toward KM serves as an explicit gauge for an organization-wide KM activity. Steered by the top management, a KM task force can be created to symbolize their dedication and enthusiasm toward their employees [8]. The key values and affirmative perception toward KM, such as knowledge is a fluid and growing asset; knowledge is not confined as personal power; and knowledge is best valued if it is shared, can be cultivated, and institutionalized to all employees. The vision to embody the significance of KM can attract additional dedication from employees when KM initiative is in its infant stage.

Centered as the frontrunner in KM community, the top management can stretch KM boundary and embody the KM significance to different departments. They can identify early adopters with enthusiasm for KM processes and involve them in propagating the KM vision through connections and interactions. Thereafter, the community can be further extended to diverse work groups/departments and encourage members to bring in a constructive disruption toward the status quo and stimulate other new ideas. The bond among people can be strengthened through the participation of knowledge workers characterized by different roles (e.g., novice, regulars, and experts) in the community. Regular meetings to exchange ideas or articulate competent skills to members are beneficial to peer learning.

Within the community, KM activists, including the top management and employees, can advance the KM perception as a spiritual KM culture. Fostering a knowledge-friendly culture with unwritten norms and beliefs is crucial because turning KM visible to all organizational members requires time. Organizations can encourage people to experiment with different ideas to develop a new working process that is in parallel with their autonomy in task. Such working atmosphere can facilitate open communication channels and knowledge-sharing sessions at both formal and informal setups, such as conducting a bimonthly good work practice sharing allowing employees at all levels to explicate or solicit feedback of their work practices.

7.2. Pillar of process

KM processes involve both formal and informal dynamics, ranging from casual conversations in the hallway to socialization regarding work processes over departmental meals to pre-arranged, focused discussion sessions held in meeting rooms. Organizations can devote extra efforts to engage major processes, including acquisition and capture, conversion and organization, storage and dissemination as well as creation and usage [27]. Further actions are required to interweave the KM processes holistically because such knowledge assets can be an added value to facilitate informed decision making and strategy.

Knowledge is intricate with its multiple facets [1, 2, 28]. Management must identify its paucity and presence at individual, group, and organization contexts. Therefore, employees are encouraged to tap in diverse sources, acquire the critical knowledge, such as core competencies and know-how residing in a particular employee's mind, or capture the knowledge embedded within a particular organization process, or deduce good practices adopted from outside organizations.

Efficiently and effectively conversing and organizing knowledge into the appropriate format for easy access and retrieval are crucial in the organizational KM agenda. Explicit knowledge, such as text-based reports and procedural manuals, is relatively communicable to others. Thus, experts can explicate their knowledge and codify them in a structured form. By contrast, tacit knowledge, such as capability to understand and to read the cues from customers' propensity to purchase, cannot be verbalized and articulated entirely to others. Organizations may have to devote extra resources to convert them into demonstrative video or narrative good practices that can be learned by knowledge seekers through a different mode.

Sharing is one of the most challenging processes if knowledge is confined as a personal asset rather than a social capital in the organization. Equipped with a sound incentive system (covering both monetary and non-monetary recognition), the infant KM stage may progress with a mandatory sharing from experts or experienced staff. During the growth stage of the KM program, sharing exercise can be regularly conducted with the active participation from the top management. Further sharing can be boosted on a voluntary basis, with many employees sharing and exchanging their good practices reciprocally through an informal setup, such as social conversation, or through a formal route, such as documented publications [28].

Knowledge creation is occasionally a spontaneous process, where innovative ideas are not confined to the domain expertise or experienced colleagues. It can also be an emergence of a novel idea or one that adds value to reconfigure a current idea or enhance working practices, which, in turn, can be applied in new contexts. Management can encourage employees to explore their ideas playfully through trial-and-errors for invention. Furthermore, management can provide extra "time" resources to support creation, such as releasing employees from work to cross-fertilize ideas with colleagues from different disciplines.

7.3. Pillar of technology

In some organizations, knowledge management is closely associated with sophisticated systems, enormous database, and advanced infrastructure. Management must realize that the presence of KMS offers no guarantee that the users will automatically come nor hoard knowledge and skills from the repositories. KMS is capitalized as an effective and useful conduit when users interact, learn, and collaborate with others and harness reusable knowledge for their own work situations and processes [18, 29].

The perceptions held by the KMS users affect the extent of system usage, such as perceived ease of use and usefulness. Therefore, management can designate the IT staff for involving end users to participate in the KMS design and development process. The genuine needs in knowledge representation, expectation of feature-rich interface, and potential problems in the evolving knowledge repositories can be directly solicited. The involvement from end users reinforces the commitment and satisfaction when the system is launched because KMS is developed "for" them.

Undoubtedly, KMS enables organizations to be more agile and fluid because skills, competencies, and ideas can be stored, accessed, retrieved, and disseminated to the right people, at the right time, and at the right place [11, 12]. It presumably connects employees on a potentially fruitful platform that enables them to access, integrate, and generate knowledge. Therefore, management must be cautious to avoid putting excessive efforts in preserving and storing knowledge assets as stock in the repositories. Employees are encouraged to keep knowledge "alive" through a continuous review, updating, disposal, and sharing. Regular evaluations of system effectiveness, such as portal design and relevance of knowledge content for decision making are required.

8. Limitation and research directions

This study is an exploratory attempt to examine the KM pillars and impacts on organizational performance. Further studies should be conducted to overcome the limitations of the present study. First, the findings were drawn from individual study informants who engage in their organizational KM programs. Using the samples from multiple respondents of a work unit may shed new insights in KM, such as the degree of unanimity on KM pillars moderated by culture, private agenda, or work commitment. Second, the future research can collect more samples to generate additional statistical power and allow added robustness to the model testing. Third, in-depth examination of the hybrid strategy could be conducted to enable organizations to obtain a clear picture of their KM status. For example, future activity can be extended to the behavioral traits of knowledge workers, impact of different KM processes on sustaining competitive advantages, and complementary and substitutability roles of IT in KM process.

9. Conclusions

Knowledge management has become one of the most important activities across different organizations. Management is struggling with the efforts to embark on KM initiatives and the minimal return in competitiveness. This exploratory study identified three KM pillars acting as fundamental constituents driving KM programs to attain a desirable organizational performance. The findings showed that organizations perceived the KM pillars differently,

which influence their practices to implement KM strategy. The orientation toward different KM pillars resulted in varying effectiveness of organizational performance. Adapting a hybrid strategy is concluded to yield better results. Accordingly, actions are recommended to enable organizations to re-examine their current status and adopt changes for balanced KM programs.

Acknowledgements

The work described in this chapter was substantially supported by a grant from the College of Professional and Continuing Education, an affiliate of The Hong Kong Polytechnic University.

Author details

Ivy Chan

Address all correspondence to: ccivy@hkcc-polyu.edu.hk

College of Professional and Continuing Education, The Hong Kong Polytechnic University, Kowloon, Hong Kong

References

- Nonaka I. A dynamic theory of organizational knowledge creation. Organization Science. 1994;5(1):14-37
- [2] Ruggles R. The state of the notion: Knowledge management in practice. California Management Review. 1998;40(3):80-89
- [3] Spiegler I. Knowledge management: A new idea or a recycled concept? Communications of the Association for Information Systems. 2000;**3**(14):1-24
- [4] Ainissyifa H. The vital pillar of the implementation of knowledge management on high schools. International Journal of Scientific and Technology Research. 2012;1(4): 132-138
- [5] Chan I, Chao KCK. Knowledge management in small and medium-sized enterprises: Capabilities and way ahead. Communications of the Association for Information Systems. 2008;51(4):83-88
- [6] Edwards J. A process view of knowledge management: It ain't what you do, it's the easy that you do it. Electronic Journal of Knowledge Management. 2011;9:(4)297-306
- [7] Huang LS, Quaddus M, Rowe AL, Lai CP. An investigation into the factors affecting knowledge management adoption and practice in the life insurance business. Knowledge Management Research and Practice. 2011;9(1):58-72

- [8] Jones NB, Herschel RT, Moesel DD. Using knowledge champions to facilitate knowledge management. Journal of Knowledge Management. 2003;7(1):49-63
- [9] Sutanto J, Jiang Q. Knowledge seekers' and contributors' reactions to recommendation mechanisms in knowledge management systems. Information and Management. 2013; 50(5):258-263
- [10] Kulkarni U, Ravindran S, Freeze R. A knowledge management success model: Theoretical development and empirical validation. Journal of Management Information Systems. 2007;23(3):309-348
- [11] Malhotra Y. Integrating knowledge management technologies in organizational business processes: Getting real time enterprises to deliver real business performance. Journal of Knowledge Management. 2005;9(1):7-28
- [12] Janicot C, Mignon S. Knowledge codification in audit and consulting firms. Knowledge Management Research and Practice. 2012;10(1):4-15
- [13] Kim SK, Trimi S. IT for KM in the management consulting industry. Journal of Knowledge Management. 2007;11(3):145-155
- [14] Durst S, Edvardsson IR. Knowledge management in SMEs: A literature review. Journal of Knowledge Management. 2012;16(6):879-903
- [15] Ekweozor U, Theodoulidis B. Who owns the knowledge that I share? In: Proceedings of 16th Americas Conference on Information Systems. Association for Information Systems, Georgia, 12-15 August 2010; Peru. pp.1361-1376
- [16] Holsapple CW, Jones KG. Toward an elaboration of the knowledge chain model. In: Proceedings of the Ninth Americas Conference on Information Systems. Association for Information Systems, Georgia, 4-6 August 2003; Florida. pp. 2524-2528
- [17] Mishra B, Bhaskar AU. Knowledge management process in two leading organisations. Journal of Knowledge Management. 2011;15(2):344-359
- [18] Bhatt GD. Knowledge management in organizations: Examining the interaction between technologies, techniques, and people. Journal of Knowledge Management. 2001;5(1):68-75
- [19] Pfeffer J, Sutton RI. The Knowing-Doing Gap: How Smart Companies Turn Knowledge into Action. Harvard Business School Press; Boston, 2000
- [20] Chen MY, Huang MJ, Cheng YC. Measuring knowledge management performance using a competitive perspective: An empirical study. Expert Systems with Applications. 2009;36(4):8449-8459
- [21] Lee H, Choi B. Knowledge management enablers, processes, and organizational performance: An integrative view and empirical examination. Journal of Management Information Systems. 2003;20(1):179-228
- [22] Gold AH, Segars AH, Malhotra A. Knowledge management: an organizational capabilities perspective. Journal of Management Information Systems. 2001;18(1):185-214

- [23] Argyris C, Donald AS. Organizational Learning: A Theory of Action Perspective. Addison-Wesley; Boston, 1978
- [24] Apostolou D, Abecker A, Mentzas G. Harmonizing codification and socialisation in knowledge management. Knowledge Management Research and Practice. 2007;5(4):271-285
- [25] Pinho I, Rego A, Cunha MP. Improving knowledge management processes: A hybrid positive approach. Journal of Knowledge Management. 2012;16(2):215-242
- [26] Kim TH, Lee JH, Chun JU, Benbasat I. Understanding the effect of knowledge management strategies on knowledge management performance: A contingency perspective. Information and Management. 2014;51(4):398-416
- [27] Fink K, Ploder C. Balanced system for knowledge process management in SMEs. Journal of Enterprise Information Management. 2009;22(1):36-50
- [28] Teng J T C, Song S. An exploratory examination of knowledge sharing behaviors: Solicited and voluntary. Journal of Knowledge Management. 2011;**15**(1):104-117
- [29] Edenius M, Borgerson J. To manage knowledge by intranet. Journal of Knowledge Management. 2003;7(5):124-136

Knowledge Management for Informally Structured Domains: Challenges and Proposals

Karla Olmos-Sánchez and Jorge Rodas-Osollo

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/intechopen.70071

Abstract

Eliciting requirements of products or solutions in informally structured domains is a highly creative and complex activity due to the inherent characteristics of these domains, such as the great quantities of tacit knowledge used by domain specialists, the dynamic interaction between domain specialists and their environment in order to solve problems, the necessity of these solutions of products to be developed by teams of specialists and the asymmetry of knowledge between domain specialists and requirements engineers. The knowledge management discipline promotes an integrated approach in order to face these challenges; therefore, a strategy for addressing requirements elicitation that incorporates techniques and methods of this discipline has been proposed as a serious approach to deal with those challenges. The valuable results of the application of the strategy in real cases prove empirical insights about its utility.

Keywords: knowledge management, informally structured domains, requirements elicitation, tacit knowledge, knowledge creation spiral

1. Introduction

Requirements elicitation (RE) is a valuable process for the identification of *solution* requirements according to the need of clients of users [1]. In this chapter, the concept of *solutions* includes products, such as software systems or intangible solutions, such as data analysis. According to several authors, application domain knowledge is essential to obtain the correct and appropriate requirements. The application domain is an area where a solution is or will be used. Consequently, requirements engineers must understand, as soon as possible, the structure, the processes and the restrictions of a domain in which they are generally



© 2017 The Author(s). Licensee InTech. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. neophytes. This knowledge belongs to domain specialists, any person possessing application domain knowledge and/or having a role in the domain. Therefore, requirements engineers must elicit the application domain knowledge from domain specialists in order to include it into a set of solution requirements. It is a complex and highly creative activity that involves intensive cognitive activities, especially when the application domain has a high degree of informality where knowledge is informally stated, partially complete, implicitly assumed, tacit and unstructured [2].

This phenomenon is presented in many disciplines such as *intelligent tailored solutions for ill-structured domains, software for complex domains, intelligent tutoring systems, knowledge based systems, industrial design, among others.* In general, every necessity that requires a complex, highly creative solution, in which the requirements engineers are not a part of the application domain and need eliciting sufficient high-quality knowledge to understand the clients' need and expectations, faces this challenge [2]. Therefore, instead of focusing on the challenges of developing a requirements elicitation proposal for each of these complex areas, we have expanded the vision and generalized these domains as *informally structured domains* (ISD) [3], which is widely explained in Section 2.

In addition, solutions in ISD usually respond to clients and users' specific needs. As a result, they are diverse, consensus and unverifiable, and there are not fully defined processes to develop them. Therefore, these solutions or products must be developed according to the experience of domain specialists. These characteristics hamper the requirements elicitation process because the implications of knowledge transfer and transformation, the appropriate management of tacit knowledge and the issues of knowledge exchange must be considered.

In this context, we assume that a perspective of requirements elicitation that emphasizes the importance of knowledge management (KM) is a useful approach for addressing ISD inherent problems. KM is a discipline with the aim of enhancing an organization by sharing and managing knowledge flow among the people, taking advantage of information technologies [4]. Regarding KM in requirements elicitation is not new, but only few efforts offer a full knowledge management perspective [5].

The knowledge management strategy for requirements engineering (KMoS-RE[®]) [6] is a highlevel plan oriented to the transfer or transformation of knowledge. The strategy has the aim of eliciting, structuring and creating knowledge that can be incorporated into a specification closest to the needs and expectations of clients. It is especially design from a full KM perspective in order to be applied in the context of ISD. The goal of this chapter is to describe the challenges of ISD and make a critical analysis of the KMoS-RE[®] strategy as a serious requirements elicitation proposal to face them. The analysis is based on the experience of applying the strategy in several ISD real cases. According the valuable results, the KMoS-RE[®] strategy promises to be a useful tool in the requirements elicitation of solution or products, especially in disciplines that share ISD characteristics [8].

The remainder of this chapter is organized as follows: Section 2 presents a characterization of ISD in order to explain the challenges of eliciting requirements in these domains. This section also includes a wide explanation of tacit knowledge. Section 3 describes fundamental concepts of KM

in requirements elicitation. Section 4 discusses the utilization of KMoS-RE[®] as a serial proposal to face the challenges in ISD. Finally, in Section 5, the conclusions and future works are presented.

2. Informally structured domains

2.1. Tacit knowledge

As discussed above, a key element in a successful requirements elicitation process in ISD is *knowledge*. But, *what is knowledge*? Despite the widely recognised importance of knowledge as the main asset in today's society, defining it is an unresolved issue. In order to establish a baseline, this work supports the idea that knowledge has a subjective and personal quality. This view is based on the traditional definition of knowledge as *justified true belief*. However, as in Ref. [9], the focus is on the *justified* rather than the *true* aspect of belief. The *justified* view of knowledge makes it as dynamic, context-specific, humanistic, deeply rooted in individuals' value system and created in social interactions among individuals as opposed to the *true* view in which knowledge is absolute, static and non-human.

According to Ryle, knowledge can be classified in *knowing-that* and *knowing-how*. *Knowing-that* means storing and recalling facts. *Knowing-how* is a practical knowledge. This distinction carries through Polanyi's theory of personal knowledge, which classifies knowledge in *explicit* and *tacit* [10]. *Explicit knowledge* is transmitted through any language or formal representation: from text written in natural language to complex formalism as ontologies. On the other hand, *tacit knowledge* is personal and context-specific, generated by experience and therefore difficult to communicate and formalize.

Polanyi was interested in '... to show that complete objectivity, as usually attributed to the exact science, is a delusion and is in fact a false idea'. Thus, he examined how individuals gain and share knowledge. He concluded that knowledge is highly personal and questioned the commonly held view of the dispassionate objective scientist. He also emphasized that people can often know how to do things without either consciously knowing, or being able to articulate to others why what they do works.

According to Polanyi, *tacitness* is something personal, usually abilities or skills that people use to solve a problem or to do something valuable. *Tacitness* depends on people's experiences and learning. Polanyi suggested that all knowledge has a tacit component and discussed the process of how the tacit cooperates with the explicit. He also argued that language is a vital tool that people use to share knowledge, and that with the appropriate use of it, much, but not all, of this knowledge can be transmitted among individuals who share a mutually agreed language. When *tacitness* predominates, this articulation is not possible. However, it does not prevent knowledge from being transmitted by other means, such as observation or task repetition. This is what people do when learning to ride a bicycle or when an art master transfers knowledge to his or her apprentices. We should keep in mind that Polanyi's theory was generated in the field of psychology and his work was addressed towards perception. Thus, from Polanyi's perspective, any attempt to convert tacit knowledge to explicit will be unfruitful because it cannot be articulated at all.

Grant [11] provides a graphical representation of knowledge degradation as it is expressed by Polanyi's work (**Figure 1**). The bar represents how the knowledge is flowing in a continuum between tacit and explicit. The continuum ranges from knowledge inherently tacit to knowledge that can be easily expressed by words. The knowledge that can be expressed by words ranges from explicit to experts to explicit to most people. The knowledge explicit to experts requires specialized language. Most of this knowledge is also implicit, i.e. knowledge that can be expressed by words, but that for some reason it has not made explicit. The tacit knowledge ranges from ineffable to highly personal. Much of this knowledge is related to the use of instruments, such as playing piano or using a specialized machine.

To Gourlay [12], Polanyi's work has been misunderstood. He argued that some tacit knowledge does become amenable to analysis and decomposition, allowing recording it in an explicit form. Likewise, tacit knowledge in requirements elicitation has been misused. For example, Janik [12] has identified that the concept of tacit knowledge is used in two ways:

- Concerning to knowledge that can be expressed, but for some reasons, it remains hidden. Janik identified three reasons why knowledge tends to remain tacit: (1) concern for secrecy and power, (2) because no one has bothered to recognize it or tried to explain it and (3) because it concerns presuppositions, we all generally hold. These situations can be aware, as the first one, or unaware, as the second and third ones. However, there are noninsuperable barriers to make this knowledge explicit.
- **2.** Concerning to knowledge gained through familiarity and practice, which is inexpressible in words, or knowledge gained by perception as sight, smell or know-how. A wine taster or identifying an instrument when listening to a sound, are some examples of this knowledge.

What is really important in requirements elicitation is making the most possible quantity of knowledge explicit. Whether it is tacit, implicit or that for some reason remains hidden, even because nobody asks.



Figure 1. Granularity of knowledge.

The problem of tacit knowledge in requirements elicitation is not new. Goguen [13] did an extensive analysis of the term from a social perspective. He analysed several methods to elicit requirements such as introspection, questionnaires, interviews, focus group and even protocol analysis. He argued that these methods have limitations to manage tacit knowledge. To Goguen, it is indispensable considering a social perspective to attend this problem; thus, he suggests using combinations of these methods besides including discourse, conversations and interactions analysis.

Later, Nuseibeh [14] emphasized the importance of tacit knowledge and how it may affect the requirements of elicitation process. For him, the responses of the domain specialists to direct questions about their domain of expertise do not reflect, neither their current behaviour nor the reality, for the large amounts of tacit knowledge that is handled by them. Thus, product developers or solution solvers should consider theoretical and practical techniques of *cognitive psychology, anthropology, sociology* and *linguistics* to have better results.

The importance of sharing tacit knowledge to improve the problem-solving processes or as a strategy to gain competitive advantage in organizations is undeniable. For example, Wyatt [15] argues that much of the medical progress in modern times has been attributed to an evolution from tacit to explicit knowledge. Despite that, nowadays, tacit knowledge remains as an ambiguous and inconsistent concept. We are aware that not all knowledge of specialists is susceptible to becoming explicit; however, it is essential trying this transformation with a well-founded strategy for the requirements as close as possible to the reality of the application domain.

2.2. Formality and informality

Intuitively, a domain is a well-defined area of human activity with formal and informal issues in which a *universe of discourse* occurs. According to Webster's Dictionary, 'formal' means definite, orderly and methodical. In computer science, to be formal does not necessarily require the use of formal logic, or even mathematics, but the use of a formal notation to represent system models. Everything that computers do is formal because the syntactic structures in a program are manipulated according to well-defined rules [13]. In domains with a significant social context, much of the information is embedded in the social world of domain specialists; it is informal (not susceptible to be formalized or not yet formalized) and depends on the context for its interpretation [16]. These kinds of domains share characteristics such as informally defined concepts and lack of absolute verification of the processes; therefore, the domains specialists should use a great quantity of tacit knowledge to solve everyday situations.

Every domain is susceptible to be formalized to a certain level, but there will always be issues that remain informal. If a domain is mainly formal, the domain specialists can build, in a relative easy way, formal structures to solve problems. On the other hand, if a domain is mainly informal, it does not mean that domain specialists cannot build a structure; definitely they do. In some way, it is possible to solve diagnostic or design problems; however, these structures are informal, i.e. they depend on the context and the domain specialists' experience and knowledge. When informal characteristics prevail, the process and effort to solve problems can be extremely costly and time consuming. Nguyen and Shanks [17] describe requirements elicitation as an ill-structured problem due to its openness, its context poorly understood and the existence of multiple domains. For Nguyen, solving problems in requirements elicitation requires a complex and dynamic social interaction between domain specialists and developers. The knowledge of both actors evolves as the project advances: the domain specialists get involved with software-solution and the developers with the organizational structure and business processes, i.e. the application domain. According to Nguyen, to solve ill-structured problems, understanding the problem and the structure of the solution are interleaved. The problem solvers, i.e. the requirements engineers, must explore different areas of the problem to find a solution. To accomplish this task, they communicate with the diverse actors who have domain knowledge or another perspective of the possible solution. By performing this task, their domain knowledge increases and they can return to previous stages of the problem, but with additional knowledge that allows them to explore new possibilities of solution. Therefore, the knowledge of the problem and its solution gradually evolves as the requirements engineers gain more knowledge of the domain, mainly due to social interactions and involvement of business processes.

We go further and assume that the grade of informality of the domain application influences the complexity of the process, as **Figure 2** depicts. Our focus is on the requirements elicitation process for domains with a high degree of informality, where knowledge is informally stated, partially complete, implicitly assumed, tacit and unstructured.

2.3. Characteristics of ISD

In order to effectively deal with ISD, we assume that they are located in the intersection of knowledge engineering and requirements engineering, and have the following characteristics [2]:

 Presence of multiple *domain specialists* who have different backgrounds, perspectives, interests and expectations, and whose knowledge, either tacit or explicit, of the application domain varies depending on their experience and their role in the domain. Usually, domain specialists are not aware of the details of the product or solution and only have a vague idea of its general functionality.



Figure 2. Complexity of domains.

- Presence of a group of *requirements engineers*, responsible for eliciting the requirements, who generally are not involved in the application domain. They have general technical knowledge about the development of the product or solution; however, they must elicit the application domain knowledge in order to understand the details of it and derive the correct and appropriate solution requirements.
- The solution solves or addresses a particular and unrepeatable situation. Thus, it must have its own design. However, there could be an infinite number of alternative solutions. In addition, the solution could be a tangible or intangible product and must be developed according to a requirements specification.
- A *requirements specification* is a document that contains the set of solution requirements. A requirement is a natural language statement to be enforced by the solution, possibly in cooperation with other system components, and formulated in terms of the application domain. The development of the requirements specification requires eliciting, synthesizing, validating, sharing and creating great quantities of application domain knowledge and solution knowledge in order to reach an acceptable solution. In addition, in order to develop the requirements specification, a dialectical thinking is necessary among all involved in the project.

Figure 3 depicts the characteristics described above; the figure represents *explicit knowledge* by puzzle pieces and *tacit knowledge* by clouds. The *requirements specification* is formed by pieces of knowledge of the domain specialists and requirements engineers. The requirements engineers must make the greatest possible amount of tacit knowledge explicit, synthesize the



Figure 3. Informally structured domains.

disperse knowledge and reconcile the diverse beliefs and necessities of the domain specialists. In addition, they need to incorporate their own technical knowledge in order to generate the set of requirements of the solution. In the figure, this process is represented by the solved puzzle. The cloud in the solved puzzle means that there will always be knowledge that is not susceptible to be formalized.

2.4. Challenges in ISD

Some challenges of ISD are described as follows:

2.4.1. Tacit knowledge

Tacit knowledge can cause critical knowledge, goals, expectations or assumptions to remain hidden. In consequence, the emergent requirements will appear incomplete and inappropriate, which can cause poor systems or costly effects [18].

2.4.2. Situatedness

Situated actions involve a dynamic interaction with the actor and its environment; they only acquire meaning through interpretation in a specific context [19]. Situated actions involve conscious reference to the context and choice of action. An action is not situated if it takes the form of a prescribed response or if it is an unconscious automatic response. In ISD, situated actions occur frequently; in consequence, requirements are mostly situated and depend on a process of negotiation. In this situation, domain knowledge is fundamental in order to understand the rationality behind requirements, facilitate the negotiation process and propose technological aspects of the solution according to the real necessities of the domain specialists.

2.4.3. Disperse knowledge

Products or solutions in ISD are so complex that the human knowledge required to develop them generally is vastly larger than the maximum individual human capacities [20]. Therefore, organized teams formed of specialists must develop them. In order to cooperate in the solution, domain specialists must share some knowledge about the domain. However, they always have different backgrounds, perspectives, interests and expectations, and their knowledge and experience vary depending on their own practice and role in the domain. Sometimes even inconsistent and incompatible beliefs can exist. Product developers or solution-solvers should reconcile and prioritize the diverse beliefs and knowledge about the application domain in order to incorporate it to the solution.

2.4.4. Asymmetry of knowledge

Domain knowledge is the knowledge of the area to which a set of theoretical concepts is applied. In ISD, the concept of domain knowledge has two meanings. Firstly, solution domain knowledge corresponds to methods, techniques and tools that form the basis for the development of the product or solution. Secondly, those products or solutions are developed to necessities of real-word problems that exist in an application domain. Thus, both solution domain knowledge and application domain knowledge are necessary to develop the product or solution [20].

Asymmetry of knowledge, or symmetry of the ignorance, refers to the knowledge gap that exists between domain specialists, owners of the application domain knowledge, and requirements engineers, owners of the solution domain knowledge [5]. In ISDs, this phenomenon is increased because of the large amount of tacit knowledge involved. When the gap is big, there is not cognitive empathy and the communication process is not effective. Therefore, requirements engineers could produce models that do not represent the reality.

3. Knowledge management

Knowledge management (KM) is a discipline with the aim of enhancing an organization by sharing and managing the knowledge flow among the people [5]. KM is much more than just the use of information technology to manage knowledge. Due to the complexity of deal with knowledge, this discipline has developed theoretical concepts in order to explain and face the underlying problem of elicitation, creation, exchange and validation of knowledge. According to Pilat and Kaindl [5], there are three fundamental concepts in KM closely related to requirements elicitation such as the knowledge transfer and transformation process, the distinction between explicit and tacit knowledge and the issues of knowledge exchange. In addition, we consider that a creation knowledge process, where the knowledge of all involved in the project evolves, is also present in ISD.

3.1. Knowledge transfer and transformation process

The knowledge transfer process is carried out when the knowledge of a person is transformed into natural language, and in non-verbal channels of human communication in order to be transferred to another person, who then decodes this knowledge according to their own interpretation. Any transfer of knowledge is inherently bound to acknowledge transformation, so there will always be some degree of ambiguity. Ambiguity affects the elicitation of correct requirements because people involved in the project could build different and possibly incompatible interpretations of the concepts, relations and processes of the domain. Linguists point to several sources of ambiguity such as lexical, syntactic, semantic and pragmatic. ISD produce an additional kind of ambiguity named nocuous when two people mutually ignore that they have their own different interpretation. In this situation, they end up talking about different concepts while they think that they are talking about the same topic. According to Gacitua et al. [18], any person involved in the process can be aware of this phenomenon, because they do not have access to the tacit knowledge of each other.

3.2. Conversion of tacit knowledge to explicit

Nonaka and Takeuchi [7] propose a model of conversion of knowledge in organizations based on Polany's theory of tacit knowledge. For them, knowledge creation in an organization is the result of social interaction where tacit and explicit knowledge is transferred. The model postulates four iterative conversion modes such as socialization, externalization, combination and internalization (SECI) which are described as follows:

- **Socialization** is the process of transferring tacit knowledge among individuals by sharing mental models and technical skills.
- **Externalization** is the process of converting tacit knowledge to explicit through the development of models, protocols and guidelines.
- **Combination** is the process of recombining or reconfiguring existing bodies of explicit knowledge to create new explicit knowledge.
- **Internalization** is the process of learning by repetition of tasks that apply explicit knowledge. Individuals will absorb the knowledge as tacit knowledge again.

According to Nonaka, if this cycle is done consciously, looping through this knowledge spiral may evolve the overall knowledge held collectively. The spiral of knowledge can be applied to requirements elicitation in order to face the inherent knowledge management challenges of the process. Despite that, we found just a few researches that explore this possibility. Wan et al. [21] proposed a model of knowledge conversion to the requirements elicitation process with the aim to minimize the symmetry of ignorance between developers and domain specialists. The authors base their model on the SECI model and consider the knowledge flowing between domain specialists and developers. They introduced a new agent in the process: the requirements specialist. This person would act as an intermediary between the domain specialists and the developers, so he or she must earn the trust of those involved in the process. The authors use their model to analyse a requirements elicitation process of a real software development project. In conclusion, the authors argue that the proposed model can reduce the symmetry of ignorance and facilitate the elicitation of tacit requirements. Nevertheless, to be successful in the process they suggest that the requirement specialists must have enough domain knowledge. We consider that it is difficult, if not impossible, that a person knows about every domain, so the incorporation of this new agent could hinder the, complex by itself, elicitation process. On other hand, Vásquez-Bravo et al. [22] proposed a classification of elicitation techniques to facilitate their selection in an RE process based on the phases of Nonaka's model. However, they do not propose how to use these techniques and how to elicit tacit knowledge.

3.3. Knowledge sharing

In order to implement the knowledge spiral property, it is crucial to facilitate the exchange of knowledge among all involved in the project [16]. It implies focusing on the knowledge holders, especially in ISD where knowledge is mostly tacit. This task may become difficult to handle because requirements engineers can be confronted with several persons whom they do not know. KM offers the concept of knowledge map [5], an artefact that points to knowledge but does not contain it. The artefact could be a table or a matrix indicating which person has what knowledge. The knowledge map should be created and initialized at the beginning of the process and continually be updated as the spiral of knowledge evolves. A knowledge map is also useful to discover for which knowledge a knowledge holder might be missing. In
ISD, we assume that a knowledge map would also be useful for indicating the tacitness level of the knowledge holders. Thus, we propose the piece of knowledge (PoK) matrix, an artefact to fulfil the functions mentioned above and to be used in the KMoS-RE[®] strategy.

The PoK matrix is a data structure that stores the relation of every individual (solution solver or domain specialist) involved in the project with every piece of knowledge about the domain. A piece of knowledge can be a concept, a relationship or behaviour. The PoK matrix is used as a reference to figure out which concepts, relationships or behaviours had been made explicit and which of them remain tacit. The aim of the KMoS-RE[®] strategy is to look for the transformation, from 0 to 1, of the most possible values in the PoK matrix. This is in order to make explicit the most possible quantity of tacit knowledge. It would be ideal if the requirements engineers could make explicit all pieces of knowledge. However, there will always be knowledge that it cannot be converted to explicit; therefore, the requirements engineers must propose the most suitable solution with the explicit knowledge obtained at a particular moment.

3.4. Knowledge evolution spiral

As was mentioned above, in ISD, understanding the problem and the structure of the solution are intertwined. The problem solvers, that is, the requirements engineers, must explore different areas of the problem to find a solution. In order to accomplish this task, they should dialogue with the domain specialists, who have their own domain knowledge and perspective of the possible solution. By performing this task, the knowledge of the problem solvers about the application domain evolves. If were necessary, they can return to previous states of the project, but their knowledge is not the same, they will have additional knowledge that allows them to explore new possibilities of solution. In summary, the knowledge of the problem and its solution gradually evolves as requirements engineers gain more knowledge of the application domain due to social interaction and their involvement with the business processes. In order to model that behaviour, the knowledge evolution model for requirements elicitation (KEM-RE) was developed based on the SECI model. The KEM-RE is an iterative cycle (**Figure 4**) that consists of four stages that include the four kinds of knowledge processes in the innovation of complex problem solving:

- Knowledge elicitation (KE) stage. The requirements engineers elicit knowledge from domain specialists and vice versa. The socialization mode predominates.
- Knowledge integration and application (KI & A) stage. The requirements engineers integrate the acquired knowledge and their own experience into models. This is a complex activity in which combination and externalization modes are presented. In addition, as the requirement engineers develop models they internalize the domain knowledge.
- Knowledge sharing and exchange (KS & E) stage. The models developed by requirements engineers will be shared with the domain specialists. This phase takes place through socialization.
- Knowledge validation (KV) stage. The domain specialists validate the models. In order to develop this activity, they must internalize the knowledge behind the models through a cognitive dialogue. This process leads to the elicitation of new knowledge. Then the cycle starts again.



Figure 4. Knowledge evolution model for requirements elicitation.

4. KMoS-RE[®]: an approach from knowledge management discipline

The KMoS-RE[®] strategy [6] is a high-level plan to achieve a set of requirements of a solution or product through the eliciting, structuring and creating of knowledge. Following the work of [24], the strategy consists of three phases: *domain modelling* (*DM*), *system modelling* (*SM*) and *specification developing* (*SD*) and structures its flow of activities according the KEM-RE. Furthermore, it includes transversal activities to identify and make explicit the most possible quantity of tacit knowledge. Those activities are conducted by the identification of presuppositions [18] and classification of verbs according the Blooms' taxonomy [23]. The strategy also includes artefacts to facilitate the sharing knowledge: a record of wrong beliefs and the PoK matrix (Section 4.3). A brief explanation of each phase is provided as follows:

- Domain modelling phase (DM). In this phase, the terms, i.e. the concepts, attributes and relationships, and the basic integrity restrictions are formalized through a consensus, in order to understand the application domain without worry about the solution. The terms are recorded in the Knowledge of Domain on an Extended Lexicon (KDEL); a lexical that classifies them into objects, subjects and verbs. The KDEL is used to facilitate the building of a graphical conceptual model. The externalization of this knowledge will enable achievement a consensus among the stakeholders; hence to minimize the symmetry of ignorance. The concepts and relationships identified in this phase will generate the first version of the piece of knowledge (PoK) matrix. In addition, a graphical conceptual model is required in order to facilitate the cognitive dialogue with the domain specialists. Requirements engineers will decide what kind of conceptual model use, from entity-relationship model to ontologies, depending on the characteristics of the domain.
- System modelling phase (SM). In this phase, the current and future system processes are formalized. The current system corresponds to the system, as it exists at present. The future system represents the system after the deployment of a solution or product. The Use Cases technique was selected to model the system, both current and future, because its usefulness

has been demonstrated through the time. The system model is obtained from the KDEL and the conceptual model. The behaviours identified in this phase will also change the values of the PoK matrix.

• **Specification development phase (SD)**. In this phase, the requirements are derived from the Uses Cases' scenarios of the future system and incorporated into the solution requirements specification (SIRS).

Figure 5 depicts a general view of the KMoS-RE[®] strategy in a unified modelling language (UML) activity diagram. Every activity of the strategy corresponds to one stage of the KEM-RE: model validations (MV) is related to knowledge validation (KV), knowledge elicitation (KE) is related with the stage of the same name, model discussion (MD) corresponds to knowledge sharing and exchange (KS & E) and domain modelling (DM), system modelling (SM) and specification development (SD) correspond to knowledge integration and application (KI & A). The swim lanes in the figure represent the activities developed by each type of actor.



Figure 5. Knowledge management on a strategy for RE.

4.1. KMoS-RE[©] analysis

According to Maalej and Thurimella [24], managing requirements knowledge is about efficiently identifying, accessing, externalizing and sharing domain and technical knowledge by and to all involved in the project, including analysts, developers, and domain specialists, which is closely related to a full perspective of KM. The rationality of the KMoS-RE[®] strategy is based on the fundamental issues described as follows:

- The flow of activities in the KMoS-RE[®] strategy is based on the knowledge evolution model KEM-RE, which is based on the knowledge evolution spiral proposed by Nonaka. The evolution spiral knowledge has the aim of facilitating the conversion of tacit knowledge to explicit. In addition, the incorporation of techniques such as the identification of presuppositions and the classification of verbs according the Bloom's taxonomy make easieridentifying knowledge that could be tacit, and hence hidden.
- Representing requirements knowledge targets an efficient information access and artefact reuse within and between projects. The KMoS-RE[®] strategy proposes several artefacts in order to represent different views of the system. They can be accessed and shared by all involved in the project. Although several requirements elicitation proposals use lexical, conceptual models, use cases models and scenarios, few of them combine those techniques in a strategy. Besides, the KMoS-RE[®] strategy proposes two innovative artefacts: the record of belief and the PoK matrix.
- Sharing requirements knowledge improves the collaboration among all involved in the project and ensures that their experiences do not get lost. The knowledge spiral in which the activities of the KMoS-RE[®] strategy are based compels to sharing the knowledge among solution-solvers and domain specialists through socialization.
- Reasoning about requirements and their interdependencies aims at detecting inconsistencies and deriving new knowledge. Externalizing the knowledge through the development of the different artefacts let the solution-solver reason and internalized the domain knowledge.

4.2. KMoS-RE[®] applied to real ISD cases

The KMoS-RE[®] strategy has been applied in the development of solution of several real ISD cases:

- Software development for complex domains. This is a complex and creative activity in which software developers should understand, as soon as possible, the knowledge of a domain in which generally are neophytes. Then, combine this knowledge with their own technical knowledge in order to reach a solution that meets clients' expectations. The KMoS-RE[®] strategy has been used to develop a cognitive rehabilitation system for sclerosis multiple patients [6].
- **Soft computing**. This artificial intelligence (AI) subarea includes several techniques that are suitable for solutions in ISD, since it is tolerant of imprecision, uncertainty, partial truth and approximation. A complex problem in soft computing is how to elicit the knowledge of specialists in order to incorporate it in an appropriate representation and to reach correct solutions. A case-based reasoning system to support heating ventilation and air conditioning (HVAC) design decisions was developed using the KMoS-RE[®] strategy [8].

- Intelligent tutoring systems. Over the past decade, intelligent tutoring systems have become increasingly accepted as viable learning tools in academia and industry. However, most of these solutions had been developed for well-defined domains. Informally structured domains, such as computer programming, laws and ethics, present a number of unique challenges for researchers in intelligent tutoring, especially to represent and evaluate the knowledge. We are currently exploring the adaptation of the KMoS-RE[®] strategy with the aim of getting a method to develop Bayesian Networks for evaluation in intelligent tutoring systems in the context of ISD.
- Industrial design. The KMoS-RE[®] strategy had been used as a HVAC requirements process in a real company [8]. The HVAC design is a difficult task because the information necessary for solving the problem is incomplete and vague. This knowledge belongs to the domain specialists, generally a set of specialists from different fields, such as mechanical engineers, control engineers, electrical engineers and architects. In addition, there could be multiple and controversial solutions and the criteria that determine the best design solution is complex and imprecise.

4.3. Discussion

Nowadays, the negative effects of inappropriate, incorrect and ambiguous requirements have been widely studied and are well known. Despite the vast quantity of proposals, methods, techniques and tools, requirements elicitation is still an open problem, as shown by many projects that do not fulfil clients' expectations or that exceed the development time due to bad elicited requirements. Thus, there are still clear opportunities to improve.

The application of the KMoS-RE[®] strategy in real ISD cases has showed that its characteristics are clear contributions to the requirements elicitation area, as it is described as follows:

- Emphasis on application domain knowledge. The importance of the application domain knowledge in order to improve the requirements elicitation process is widely accepted. However, currently the most of the methods and tools of requirements elicitation are designed for general problem domains, where problem-specific domain knowledge is not completely necessary [25]. The KMoS-RE[®] strategy emphasizes the importance of the application domain knowledge, either tacit or explicit, besides of proposing techniques and methods to facilitate its discovery, representation, sharing and appropriation among all involved in the project. Thus, the strategy can be applied in knowledge-intensive projects.
- Generality and adaptability. The theoretical concepts of knowledge management, in which the KMoS-RE[®] strategy is supported, allow it to be applied to domains with different levels of informality. It also has the advantage of being considered as a high-level plan; therefore, the requirements engineers have the authority to decide which phases are necessary. In some cases, they can even choose between different techniques and methods.
- **Evolutive**. The strategy is not limited to the proposed so far. The model allows the incorporation of new proposals from methods and techniques to knowledge management models or perspectives of elicitation of requirements. For example, we are currently analysing the

adaptation that is based on goals elicitation approach [26], the knowledge audit model [25] and the social network analysis [27].

• Algorithmic. Despite its generality and adaptability, the KMoS-RE[®] strategy is algorithmic in the sense that the process of its implementation is well defined and limited, so the requirements engineers do not need a deep knowledge about the theoretical concepts of knowledge management. Most of the cases showed in the previous section were led by undergraduate engineering students [8]. Although a process of awareness of the issues of informal structured domains is always recommended.

Finally, we would like to emphasize that the most important contribution of the KMoS-RE[®] strategy is that it does not try to work against human nature; it recognizes its capabilities and limitations and builds the best proposal based on that. Thus, according to the below and the valuable results of the application of the KMoS-RE[®] strategy in several and different contexts, it can already be considered as a serious approach for requirements elicitation knowledge.

5. Conclusions and future works

The KMoS-RE[®] strategy is a novel approach from KM in order to face the challenges of eliciting knowledge and creatively transform it into a set of requirements of a product or solution in order to satisfy the needs and fulfil whole expectations of clients and users. The strategy is focused on dealing with ISD. Due to the characteristics of these domains, the strategy has a full KM perspective, i.e. it incorporates knowledge engineering techniques in order to properly manage tacit knowledge. The *domain modelling phase* handles the problem of formalizing the concepts and relationships; at least a consensus about it is reached. The *system modelling phase* deals with the problem of structuring the processes in the domain. Thus, the problem of handling tacit knowledge has addressed properly by KMoS-RE[®] strategy.

The strategy was applied to several ISD real cases in different and diverse areas. The solutions achieved provide evidence about the usefulness, the value and the generalization of it. Therefore, the application of the KMoS-RE[®] strategy in several real cases shows that it is a useful approach in order to elicit the requirement of solutions or products especially in ISD.

Finally, the challenge of managing the tacit knowledge requires to analyze more cases in order to improve all the KM approaches including the KMoS-RE[®] strategy.

Author details

Karla Olmos-Sánchez* and Jorge Rodas-Osollo

*Address all correspondence to: kolmos@uacj.mx

Autonomous University of Ciudad Juárez, Ciudad Juárez, Chihuahua, México

References

- Hansen S, Berente N, Lyytinen K. Requirements in the 21st century: Current practice and emerging trends. In: Design Requirements Engineering: A Ten-Year Perspective. Springer Berlin, Heidelberg; 2009. pp. 44-87
- [2] Olmos-Sánchez K, Rodas-Osollo J. A strategy of requirements engineering for informally structured domains. The International Journal of Combinatorial Optimization Problems and Informatics. 2016;7(2):49-56
- [3] Olmos K, Rodas J. Requirements engineering process model for informal structural domains. International Journal of Computers Communications & Control. 2013;2(1):75-77
- [4] Chen L, Fong P. Evaluation of knowledge management performance: An organic approach. Information and Management. 2015;**52**(4):431-453
- [5] Pilat L, Kaindl H. A knowledge management perspective of requirements engineering. In: Proceedings of the Fifth International Conference on Research Challenges in Information Science (RCIS), 2011; 19-21 May 2011; Gosier, France. IEEE; 2011. pp. 1-12
- [6] Olmos K, Rodas J. KMoS-RE knowledge management on a strategy to requirements engineering. Special Issue Requirements Engineering Software Product Line, Requirements Engineering Jornal 2014;19(4):421-440
- [7] Nonaka I, Takeuchi H. The knowledge-creating company: How Japanese companies foster creativity and innovation for competitive advantage. Harvard Business Review. 1995;69(6):96-104
- [8] Olmos-Sánchez K. Knowledge Management on a Strategy for Requirements Engineering. Universidad Autónoma de Ciudad Juárez; 2015
- [9] Nonaka I, Toyama R, Konno N. SECI, Ba and leadership: A unified model of dynamic knowledge creation. Long Range Planning. 2000;**33**(1):5-34
- [10] Polanyi M. The Tacit Dimension. Chicago, United States of America: University of Chicago Press; 1966
- [11] Grant KA. Tacit knowledge revisited We can still learn from Polanyi. Electronic Journal of Knowledge Management. 2007;5(2):173-180
- [12] Gourlay S. Towards conceptual clarity for 'tacit knowledge': A review of empirical studies. Knowledge Management Research & Practice. 2006;4(1):60-69
- [13] Goguen JA. Formality and informality in requirements engineering. In: Proceedings of the 4th International Conference on Requirements Engineering; IEEE Computer Society, Washington, DC, USA 1996, pp. 102-108

- [14] Ma L, Nuseibeh B, Piwek P, De Roeck A, Willis A. On presuppositions in requirements. In: Proceedings of the Second International Workshop on Managing Requirements Knowledge (MARK), 2009; 1 Sept. 2009; Atlanta, GA, USA. IEEE; 2009. pp. 27-31
- [15] Wyatt JC. Management of explicit and tacit knowledge. Journal of the Royal Society of Medicine. 2001;94(1):6
- [16] Sutcliffe A, Sawyer P. Requirements elicitation: Towards the unknown unknowns. In: 2013 21st IEEE International Requirements Engineering Conference, RE 2013— Proceedings; 15-19 July 2013; Rio de Janeiro, Brazil. IEEE; 2013. pp. 92-104
- [17] Nguyen L, Shanks G. A framework for understanding creativity in requirements engineering. Information and Software Technology. 2009;51(3):655-662
- [18] Gacitua R, Ma L, Nuseibeh B, Piwek P, de Roeck AN, Rouncefield M, Sawyer P, Willis A, Yang H. Making tacit requirements explicit. In: Proceedings of the Second International Workshop on Managing Requirements Knowledge (MARK) 2009; 1-1 Sept. 2009; Atlanta, GA, USA. IEEE; 2009. pp. 40-44
- [19] Fernandes KJ. Interactive situation modeling in knowledge-intensive domains. International Journal of Business Information Systems. 2009;4(1):25-46
- [20] Vessey, I. The Effect of the Application Domain in IS Problem Solving: A Theoretical Analysis. In: Information Systems Foundations: Theory, Representation and Reality, edited by Hart Dennis N. and Gregor Shirley D., 25-48. ANU Press, 2007
- [21] Wan J, Zhang H, Wan D, Huang D. Research on knowledge creation in software requirement development. Journal of Software Engineering and Applications. 2010;3(5):487-494
- [22] Vásquez-Bravo D-M, Sánchez-Segura M-I, Medina-Domínguez F, Amescua A. Combining software engineering elicitation technique with the knowledge management lifecycle. International Journal of Knowledge Society Research. 2012;3(1):1-13
- [23] Bloom, BS. Taxonomy of educational objectives. Vol. 1: Cognitive domain. New York: McKay. 1956:20-4
- [24] Maalej W, Thurimella AK. An introduction to requirements knowledge. In: Managing Requirements Knowledge. Berlin, Heidelberg: Springer; 2013. pp. 1-20
- [25] Taheri L, Che Pa N, Abdullah R, Abdullah S. A knowledge audit model to assess the knowledge in requirement elicitation process. In: 2015 9th Malaysian Software Engineering Conference (MySEC); 16-17 Dec. 2015; Kuala Lumpur, Malaysia. IEEE; 2015. pp. 106-111
- [26] Franch X, López L, Cares C, Colomer D. The i* framework for Goal-Oriented modeling. In: Domain-Specific Conceptual Modeling. Cham: Springer International Publishing; 2016. pp. 485-506
- [27] Lim SL, Ncube C. Social networks and crowdsourcing for stakeholder analysis in system of systems projects. In: 2013 8th International Conference on System of Systems Engineering; 2 Jun-06 Jun 2013; Wailea-Makena, HI, USA. 2013. pp. 13-18

Performance Management by Causal Mapping: An Application Field of Knowledge Management

Sarah Kölbel, Wolfgang Ossadnik and Stefan Gergeleit

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/intechopen.70297

Abstract

As implied by the performance management (PM) concept, modern corporate management has to focus on cause-and-effect relationships underlying a firm's financial performance generation. To determine the causes of financially desirable effects, subject-bound experiences and knowledge of employees, called tacit knowledge, should be realised. For this, knowledge management (KM) offers various elicitation techniques to reveal corporate-specific success factors (SFs) of financial performance generation from the corporate experts' implicit knowledge. The identified factors have to be organised within a network of cause-and-effect relationships. In this framework, PM can apply the instrument of mapping to structure the individually revealed knowledge, to aggregate and visualise it for the entire company. For a valid representation of the causal relationships, the subjective bias arising within the mentioned process has to be minimised. In the literature, a variety of mapping methods can be found that differ in their approaches and their level of significance. As such a method, causal mapping will be presented in this paper. For providing intersubjectivity, the decision-making trail and evaluation laboratory (DEMATEL) as a multi-criteria approach will be debated in the context of mapping as a research field.

Keywords: causal mapping, knowledge management, performance management, implicit knowledge, explicit knowledge, success factors, DEMATEL, subjectivity, intersubjectivity

1. Introduction

In today's information age, companies face high competition and pressure while trying to perform successfully in the long term. To meet the dynamic competitive environment and the globalisation of markets accessible, companies are forced to hunt for competitiveness resulting



© 2017 The Author(s). Licensee InTech. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. **(c) By** from the efficient use of general and specific knowledge. Accordingly, knowledge is becoming a competitive factor and provides an essential cause for company success. Therefore, it is important not only to consider obviously accessible knowledge but also to directly accomplish a performance-related use of the specific, implicit knowledge of a company. These individual experiences and knowledge such as fundamental components of human capital inhere a huge chance to improve the steering and control processes of performance generation and hence to master competition successfully. In the context of such a performance management (PM), expert knowledge is indispensably focused on relations between causes and effects to generate financial performance. By considering cause-and-effect relationships underlying the financial performance generation process the traditional perspective of measuring value realisation is extended to causally ambitioned value generation management. As a consequence, such a causal knowledge reveals options for actions influencing the financially as well as nonfinancially dimensioned causes, which are linked to future financial performance. Thus, PM provides relevant starting points to control the financial performance generation process.

In reality, companies comprise many departments with multiple environmental factors and, as a consequence, there exist many latent or manifest interdependently structured characteristics relevant for performance generation. Without knowledge of such relations, the management cannot efficiently control desirable effects by their causes. A map of causal relationships could care for more transparency in this respect. However, expert knowledge on success factors (SFs) and their causal relations is usually not available in the explicit form of a graphical representation. Instead, subjectively based knowledge, stemming from individuals' observations and experiences, which are called implicit or tacit knowledge, might be identifiable and pending to be elicited.

Knowledge management (KM) recognised as a subdiscipline of PM can be applied to convert this implicit knowledge into explicit subjective knowledge on causal relationships, which can be identified and depicted by construction of a tailor-made causal map. Through the construction of the causal linkages during the mapping process, a subjective judgment bias can arise. With regard to this problem of subjectivity, specific methods of the multi-criteria research field, in particular the decision-making trail and evaluation laboratory (DEMATEL), can be used in the mapping context. DEMATEL provides a reduction of a potential personal bias when applying one of the common mapping methods. For this, a fictional case study will be presented adopting the target of achieving intersubjectivity.

2. Performance management and performance measurement

PM and performance measurement can generally be associated with a strategic management and control to focus on long-term financial success. To realise this objective, to implement strategies promisingly and for the alignment of the entire organisation to a consistent development of success potentials, the strategic factors relevant for financial success need to be identified. Through the control of the critical SFs via measureable key performance indicators (KPIs), the company's financial performance can be influenced beneficially [1, 2]. This requires the identification of upstream causes of the financial success, which can be dimensioned financially or non-financially. With the subsequent consideration of non-financially dimensioned SFs, which often are deeply rooted in the intuitive knowledge of managers and employees, PM offers a concept of steering and control extending the restrictions of traditional control procedures entirely based on logical decompositionally constructed financial ratios [3–6].

According to Neely et al. [7], performance measurement can be described as the 'process of quantifying the efficiency and effectiveness of action'. An evaluation of the efficiency and effectiveness of action always requires the reflection of the corporate strategic objectives [3]. Thus, performance measurement enables a holistic assessment of the company's performance. For analysing the whole performance generation process, it necessitates the additional identification of cause-and-effect relationships. The specific SFs of a company are integrated into a cause-and-effect network. Each factor can directly or indirectly be linked with the company's financial performance [8–10]. A cause-and-effect relationship will only be defined as causal, if a strong correlation between cause(s) and effect(s) exists and the cause of action temporally precedes the effect. Additionally, the relationship has to be plausibly explicable [11]. With an accurate specification of the causal relationships among the factors, a comprehensive understanding of the performance generation is provided. Moreover, in decision-making situations, the consequences of chosen actions on the company's success can be estimated [5, 9].

Within this context, the 'performance' is determined as a multi-criteria and therefore multidimensional construct, which—in a causal sense—cannot only be measured ex post, but also ex ante. Performance measurement thereby is classified in a superordinate control context of PM as an integral part of it. At this point PM describes the process of planning, implementing and executing the corporate strategy by a coherent system of actions. In linkage with a strategy, it determines values of measurable KPIs respectively operating numbers. If, due to the permanent monitoring of the operating numbers, a significant variance is identified, the PM can initiate appropriate (feedback- or feedforward-directed) strategic action [1, 6]. In conclusion, for a comprehensive analysis and control of the performance generation, SFs and their company-specific causal relationships need to carefully be identified and examined in depth. Therefore, PM captures KM as a complementary discipline.

3. Knowledge management: explicit and implicit knowledge

KM implies the acquisition, development, transfer, storage and use of knowledge in a company [12–15]. A condition for knowledge generation is generic information like scientific theories and models. The information is based on data defined as simple facts or events etc., which are not systematised for a specific context. Thereby, the information forms a structured, meaningful summary of explicit data points in order to take a conclusion or prediction. In particular, knowledge represents the generic skill to connect and use recent information with previously collected information in several new application fields. As a consequence, the knowledge carriers evolve a new understanding or subjective perception of the actual situation, which generates a new knowledge basis [13, 16]. Here, knowledge can generally differ in explicit and implicit one [17].



Figure 1. Degree of tacitness and the steps to objectivisation.

Explicit knowledge is communicable and thus not exclusively available to the person who possesses and uses it. It declares the relevant know-what [18, 19]. Knowledge concerning company-specific cause-and-effect dependencies can furthermore be drawn on the subjectbound, intuitive experience-based knowledge of competent employees and managers. This as 'tacit' specified implicit knowledge is difficult or impossible to verbalise as well as to formalise in contrast to explicit knowledge [18]. It is understood as individual specific know-how.

According to Ambrosini and Bowman [20], knowledge can be graded in relation to the degree of tacitness as shown in **Figure 1**.

Between the explicit knowledge (A) and deep-rooted tacit knowledge (D), which cannot generally be revealed, the communicable knowledge (B and C) have to be specified. One specification comprises the implicit knowledge (B), which can be appropriately articulated and revealed. But, this knowledge becomes less obvious over time, because the knowledge carriers have not been mentally concerned with it and no third party has demanded for it. Additionally, there exists tacit knowledge (C), which can be articulated only incompletely. Although it is possible to get access to this knowledge, it is not describable by general language use [20]. The implicit knowledge of type B and C is of special importance for a company in order to discover the performance-relevant SFs and develop their hypothetical causal relationships in a map.

The experts' tacit knowledge has to be externalised by applying adequate elicitation techniques in the context of KM [15, 21, 22]. For this purpose, three groups are basically distinguished in the literature under the term 'knowledge elicitation techniques': observations and interviews, process tracing and conceptual techniques [23]. It generally cannot be defined that one method is more appropriate than another. The choice of a technique for extracting performance-relevant knowledge should be taken case specifically. However, for the development of causal hypotheses, there exists the experience that interview techniques as most commonly applied methods generate more information on company-specific connections than other approaches [22–24].

Subsequently, the externalised implicit knowledge of the performance-related SFs is causally systematised into a more generic and easily comprehensible form by formulating a causal

map [25, 26]. Depending on the chosen mapping method, the causal relationships base on purely subjective judgments. This subjective knowledge stands for the relationship of an individual to its environment. Thus, it is not objective. Subjectivity can be seen as an error source in the current subject, although it offers an epistemic value [27]. Increasing comparability and transparency of individual subjective evaluations about causal relationships generate a degree of intersubjectivity [28]. Thus, intersubjectivity is achievable, if only more than one individual can clearly comprehend the formulation and structuring process of causal relations among SFs.

However, strategic forecasts about the future performance developments are only possible to a certain extent or cannot even be performed by application of subjectively and intersubjectively based maps. (Intersections of subjective maps would deliver intersubjectively based ones.) But, only a statistical validation of the causal map generates an objective understanding of the causal relationships, which thus are directly empirically verifiable [5, 29]. As a consequence, valid predication of the performance generation can be given. Initially, the subjective mapping methods are considered more closely in the following section.

4. Subjective mapping

Mapping methods are used to depict company-specific explicit and implicit knowledge, such as control-relevant factors and their causal relationships [20]. There are several definitions and names of mapping methods in the literature [10, 30–32], which can be distinguished according to their type and to their concept of construction.

4.1. Types of maps

Two types of maps with relevance for the current subject are the 'cognitive mapping' and the 'causal mapping'.

Cognitive maps can be seen as a summary of different concepts of mapping that rely on the beliefs of an individual about a specific topic [26]. In its core, a cognitive map refers to how an individual person can explain its environment and to what extent it is able to understand it. It visualises the individual perception of the reality and thus represents person-specific knowledge [28, 33]. This knowledge is needed for a comprehensive assessment of corporate performance because it captures experiences and know-how about corporate-specific internal and external factors in a detailed way [31].

Causal maps generally illustrate the individual understanding about linkage of events occurring at a certain time [26]. In the context of PM, the instrument of causal mapping is suitable for displaying company-specific explicit as well as implicit information describing the influences of performance relevant causes on the top objective of financial performance [20]. Causal maps consist, on the one hand, of nodes, which can represent control-relevant SFs and, on the other hand, of arrows, which are used to represent the cause-and-effect relationships between these nodes [34]. The node, from where the path of an arrow begins, is interpreted as the cause of the consequently influenced effect. The effect is depicted through the node where the arrow finally ends. The direction of an arrow implies the assumed causality. Thus, a causal map can be interpreted as a cognitive map, which describes the process of performances in a company [8]. But, a cognitive map is always constructed from a single individual, whereas a causal map can also represent the cause-and-effect relationships as an aggregated result of several individuals [35]. **Figure 2** gives an example of a causal map [22]. The contained factors might be measured directly and would be manifest in that case. Otherwise, they are latent and can be operationalised by one or more selected measure(s). Measurable data are transferable into an indicator system of strategic success generation.

4.2. Development and participation in causal mapping approaches

A causal map based on local tacit knowledge can be formed by a group of experts itself [36] or by aggregating the individual maps of the group members [37, 38]. After the development of individual causal maps, it might be a scientific objective to measure the differences between these maps [24, 39]. But, in the related literature, approaches are most favoured that aim for a specific form of an aggregation of individual maps. The aggregation can follow specific 'counting rules' of factors and relations depicted as arrows [8, 34]. Moreover, the finalisation of an aggregated group map is widely spread via group discussions and workshops [31, 38]. Such a group aggregation process can also be computer-supported [40–42]. In order to realise the advantages of causal mapping, it is absolutely necessary to involve a sufficient number of experts in the mapping process [43].

To construct a causal map, one of the elicitation techniques mentioned in section three has to be applied. Afterwards, the mapping process can be conducted by an interviewed expert itself, by the support of qualitative software, solely by an external researcher, a consultant or a team —so-called ethnographical protocol interpretation—or by the interaction of external persons and company experts [8, 20, 44]. The most relevant and applied mapping techniques that can be distinguished from each other and contain essential attributes are the ethnographical protocol interpretation and the interactive mapping.



Figure 2. Example of a causal map.

In addition, there is another approach developing a causal map by group discussion without applying any elicitation technique advance. According to Akkermans and van Helden [45], experts are asked to collectively form one causal map. Herein, the objective basically is to construct a unified view of a group of experts through their discussion. By group discussion, the different individual perceptions are summarised and structured to finally achieve a common understanding of the problem.

When reviewing the mapping procedures in the related literature, it is obvious that the epistemological perspective is far from a comprehensive as well as general approach. The individually conducted steps differ from case to case. A mixture of several techniques is always conceivable and a clear distinction between the documented techniques is difficult to specify. The question 'how to map?' generally depends on the preferences and objectives of internal and external experts that are involved in the process of causal mapping. Notwithstanding the construction process of a causal map, there are advantages and disadvantages provided by causal mapping.

4.3. Advantages and disadvantages of causal mapping

The advantages of causal mapping are apparently associated with a corporate's financial success and the implementation of a strategy: causal mapping enforces (a) the elicitation, (b) the visualisation and (c) the communicability of performance-relevant knowledge.

Already in the starting phase of elicitation, involved individuals develop a more extensive understanding of the corporate performance and its causes. They are invited to reflect all processes in their company and, therefore, will be able to distinguish between performancerelevant factors and those which have less importance. Furthermore, concerned individuals start to reflect their daily operation in a critical manner and may generate an alignment of their work to the principles of PM and performance measurement. Involving a sufficient number of experts from all departments of a company as participants in the mapping process amplifies the acceptance of the respective system. During the implementation of this system, employees do not only provide their causal knowledge but are also motivated to scrutinise it. They develop as well as apply the respective indicator system in a reflective manner and adjust their decisions and chosen actions to this system [46]. Due to this reflexion, learning effects emerge. Besides, the visualisation by causal mapping provokes a focus on those factors that have the largest influence onto the financial performance objective. It induces different people within a company to reflect about it. Moreover, the visualisation creates an extensive comprehension about the effects of certain actions as causes. The existing cause-and-effect chains to achieve a better (or even a worse) performance become obvious [10]. At least, the management of a company is equipped with a mapping tool that enhances the communication of a vision, of strategies as well as objectives and measures based on a common understanding of the performance generation. By causal mapping, the employees communicate on causal relations and become more aware of them. This contributes to an efficient management of the company [10].

Since the cause-and-effect relationships are primarily derived from the experience and knowledge of employees, they are categorised as subjective. The experts from different functional areas may have an unequal perception of processes. During the amalgamation of explicated assessments of cause-and-effect relationships from different subjective perspectives, inconsistent results can

occur. Therefore, it will be necessary to aggregate or to synthesise these partial perspectives in a sufficiently complex overall model of causal relationships [47].

Nevertheless, every aggregation of subjective statements can generate biases because involved managers and employees are specialised on their area of responsibility and herein collect their experiences. The subjectivity of the statements might be driven by factors like organisational blindness, vanity, satisfaction as well as dissatisfaction or the degree of motivation. Further, in group discussion, participants might answer strategically in the way to not annoy others [10, 40, 41]. As a consequence, it is not sure that the most important causal relationships among factors will be detected. Instead, it might be the case that less relevant SFs and relationships will be determined. All these challenges have to be overcome and a corrective against the biases resulting from subjective statements has to be offered.

Therefore, the multi-criteria DEMATEL method can be introduced as a technique that is able to decrease the amount of subjectivity in constructing a causal map. Thus, it enables to achieve an intersubjective validity by providing a transparent and replicable process of mapping among all participants. The technique is more appropriate to get an equilibrated and balanced causal map for the purpose of all employees. Group discussions and aggregation approaches cannot meet the requirements of unifying the variety of different individual opinions. DEMATEL, as presented in Section 5, collects the individual opinions in a more unbiased way.

5. Intersubjective mapping

Between 1972 and 1976, Fontela and Gabus have developed the DEMATEL approach for structuring and solving multi-criteria problems in a multi-personal context [48, 49]. DEMATEL can represent an algebraic method of analysis, which aggregates the collected individual implicit knowledge to identify and quantify the causal interdependencies between the detected SFs [50]. Furthermore, it strictly structures the given SFs according to their relevance in performance generation [51]. Finally, the determined causal relationships of the performance-related SFs are illustrated in an appropriate causal map, described as impact relation map (IRM) [50].

5.1. DEMATEL

In this section, some essentials of the DEMATEL approach are briefly described (Figure 3) [52].

In the first step, an $n \times n$ individual evaluation matrix X^k of each expert k (k = 1, ..., H) is determined as follows [52, 53]:

$$X^{k} = \begin{pmatrix} x_{11}^{k} & \cdots & x_{1j}^{k} & \cdots & x_{1n}^{k} \\ \vdots & \ddots & \vdots & \ddots & \vdots \\ x_{i1}^{k} & \cdots & x_{ij}^{k} & \cdots & x_{in}^{k} \\ \vdots & \ddots & \vdots & \ddots & \vdots \\ x_{n1}^{k} & \cdots & x_{nj}^{k} & \cdots & x_{nn}^{k} \end{pmatrix} = \left[x_{ij}^{k} \right]_{n \times n}$$
(1)

Performance Management by Causal Mapping: An Application Field of Knowledge Management 111 http://dx.doi.org/10.5772/intechopen.70297



Figure 3. Procedure steps of DEMATEL.

For this purpose, *H* skilled employees pairwise compare the given factors i(i = 1, ..., n) and j(j = 1, ..., n) on a Likert scale from 0 to 4 (with 0 = no effect, 1 = very small effect, 2 = small effect, 3 = strong effect, 4 = very strong effect) to identify how strong the factor *i* directly influences the factor *j*. The results are described by the matrix elements x_{ij}^k . In addition, for all cases i = j, each x_{ij}^k takes the value 0, since the factors are compared to themselves [52]. Hence, it can be formulated the assumption that a cause cannot be its effect at the same time.

According to Eq. (2), the direct relation matrix *A* is calculated by the aggregation of all individual evaluation matrices. The numerical value a_{ij} illustrates the group perception about the direct causal relationship between the factors *i* and *j*. If the condition $a_{ij} \le 1$ is fulfilled, no cause-and-effect relationship exists [52].

$$A = \left[a_{ij}\right]_{n \times n} = \frac{1}{H} \sum_{k=1}^{H} \left[x_{ij}^{k}\right]_{n \times n}$$
⁽²⁾

In the second step, the direct relation matrix A is normalised to the matrix D as follows [52, 54]:

$$D = \frac{A}{s} = \left[d_{ij} \right]_{n \times n} \tag{3}$$

$$s = \max\left\{\max_{1 \le i \le n} \sum_{j=1}^{n} a_{ij}, \max_{1 \le j \le n} \sum_{i=1}^{n} a_{ij}\right\}$$
(4)

Here, the normalisation value *s* can be specified as the maximum value of the set of maximal column and row sum of the matrix *A*. Besides, the column sum $\sum_{i=1}^{n} a_{ij}$ of the matrix *A* represents the total direct effect, which all factors *i* exert on the factor *j*. Compared with this, the total direct impact of factor *i* on all other factors *j* is described by the row sum $\sum_{j=1}^{n} a_{ij}$ of the matrix *A* [53, 55].

To determine the direct and indirect interdependent relationships of the SFs, the total relation matrix T has to be calculated in the subsequent step [52]. For generating indirect convergent effects, the potentiation of matrix D needs to convert to infinite as follows [50]:

$$T = \lim_{m \to \infty} (D + D^2 + \dots + D^m)$$
(5)

According to Eq. (6), the total relation matrix *T* is calculated under consideration of the normalised matrix *D* as well as the $n \times n$ identity matrix *I* [52]:

$$T = D(I - D)^{-1} = [t_{ij}]_{n \times n}$$
(6)

Before transferring the identified causal relationships of the SFs in an IRM, a threshold α as average influence intensity has to be specified in the fourth step. The threshold α is determined as the quotient from the sum of all values t_{ij} divided by the number of elements N of matrix T and follows the formula [56, 57]:

$$\alpha = \frac{\sum_{i=1}^{n} \sum_{j=1}^{n} [t_{ij}]}{N}$$
(7)

For a further reduction of complexity and to develop a clearly structured and manageable map, only the elements t_{ij} of the matrix T, which exceed the stated threshold α , are transferred in the map. The cause-and-effect relationship values t_{ij} , that satisfy the condition $t_{ij} > \alpha$, are classified as sufficiently significant and thus as performance-relevant influences [53].

In the last step of the DEMATEL approach, the identified SFs and their performance-relevant relationships are depicted in a causal IRM. Furthermore, the factors can be classified into causes and receivers [14]. For this purpose, the row sum $r_i = \sum_{j=1}^{n} t_{ij}$, as well as the column sum $c_j = \sum_{i=1}^{n} t_{ij}$, of the total relation matrix *T* have to be calculated [52]. The column sum c_j describes the total direct and indirect effect that all factors *i* exert on the factor *j* (called as degree of receiving). Assumed a high degree of receiving, minor changes of the factors *i* already lead to strong alteration of the factor *j*. However, the row sum r_i represents in which extent the factor *i* has an effect on all other factors *j* (called as degree of causing). A high degree of causing means that a small change of factor *i* causes great alterations of the other factors *j*. Moreover, in the case of i = j, the total of the row and column sum $(r_i + c_j)$ illustrates the accumulated outgoing and received effects of a factor. The higher the determined influence intensity, the higher the relevance of this factor for the corporate management will be [53, 54].

By forming the difference of the row and column sum $(r_i - c_j)$ for the case i = j, the factors will be specified as causes or receivers according to its resulting net effect. If $c_j < r_i$, then the factor will be defined as a cause, because its impact on the other factors is higher than the other factors' influence on it. But assumed $c_j < r_i$, the factor is mostly influenced by other factors and thus will be assigned to the group of receivers [53, 54].

Finally, all identified SFs and only their performance-relevant causal relationships will be visualised in an IRM. This causal map is framed as kind of coordinate system, of which the

abscissa represents the values of the full effects $(r_i + c_j)$ and the ordinate axis is scaled to the net effect values $(r_i - c_j)$ [53, 54]. In the following section, the approach of DEMATEL will be illustrated in a fictional case study example.

5.2. Case study as an application example

The example of a causal map is demonstrated for a typical company and its PM. For this propose, the financially and non-financially dimensioned SFs are identified and their causal relationships are analysed as well as visualised in a causal map. To construct the map in a manner to achieve intersubjectivity, the DEMATEL method is applied. By conduction of semistructured interviews with 15 experts from the company and in the following group discussion between an external research team and expert group, a pool of eight strategically relevant factors can be developed. These identified SFs are mentioned as follows: financial success (FC), competitive environment (CE), structural circumstances (SC), product range (PR), product quality (PQ), pricing (PRI), image (IM) as well as ability to supply (AS).

5.2.1. Identification of causes and receivers

First of all, the 15 experts pairwise evaluate the causal relationship structure among SFs on a Likert scale from 0 (no effect) to 4 (very strong effect). The individual evaluation matrices X^k (with k = 1, ith) are described by Eq. (1) and are aggregated to the direct relation matrix A by Eq. (2). After that, the matrix A is normalised to matrix D according to Eqs. (3) and (4). With the help of Eqs. (5) and (6), the final total relation matrix D can be calculated. **Table 1** shows the results which describe the direct influence intensity that SF *i* exerts on a SF *j*:

For more clarity in the causal structure, only these influence relationships between the SFs are considered in the IRM, of which the influence intensity is greater than the calculated threshold of $\alpha = 0.1426$ {Eq. (7)}. In **Table 1**, the sufficiently significant results are marked in bold. Consequently, nearly half of the amount of causal relationships among the factors is specified as above-average causal interdependent and thus can be determined as very performance-relevant relation.

Factors	FS	CE	SC	PR	PQ	PRI	IM	AS
FS	0.1157	0.1825	0.0345	0.1235	0.1014	0.1548	0.1535	0.0785
CE	0.2940	0.1303	0.1015	0.2041	0.1380	0.2142	0.1944	0.1132
SC	0.2905	0.1508	0.0260	0.1467	0.0891	0.1786	0.1166	0.0464
PR	0.2808	0.2318	0.0404	0.0769	0.1031	0.1755	0.2556	0.1438
PQ	0.2995	0.2257	0.0414	0.0880	0.0700	0.2018	0.2881	0.1083
PRI	0.2507	0.2019	0.0796	0.0705	0.1106	0.0911	0.2215	0.0636
IM	0.2658	0.2067	0.0294	0.1257	0.1187	0.1688	0.1074	0.0940
AS	0.2019	0.1630	0.0232	0.0727	0.0674	0.1433	0.2019	0.0390

Table 1. Total relation matrix.

Factors	r _i	C _i	$r_i + c_i$	$r_i - c_i$	Characteristic
Financial success	0.9445	1.9988	2.9433	-1.0544	Receiver
Competitive environment	1.3897	1.4927	2.8824	-0.1030	Receiver
Structural circumstances	1.0446	0.3761	1.4207	0.6685	Cause
Product range	1.3080	0.9082	2.2163	0.3998	Cause
Product quality	1.3229	0.7980	2.1209	0.5248	Cause
Pricing	1.0896	1.3280	2.4176	-0.2385	Receiver
Image	1.1164	1.5391	2.6555	-0.4227	Receiver
Ability to supply	0.9121	0.6868	1.5989	0.2253	Cause

Table 2. Classification of the SFs as causes and receivers.

To organise the SFs in the groups 'causes' and 'receivers', the row sum r_i and column sum c_i as well as their difference have to be calculated in the subsequent step as follows:

According to **Table 2**, the SFs 'structural circumstances', 'product range', 'product quality' and 'ability to supply' are identified as causes, under the condition that $r_i > c_j$ and thus the results of their difference are positive. However, the SFs 'financial success', 'competitive environment', 'pricing' as well as 'image' fulfil the condition $c_j > r_i$. As a result, the calculated difference between the row and column sum is negative and thus the factors are specified as receivers. Moreover, within the groups 'causes' and 'receivers' the SFs can be clearly ranked by their total influence intensity ($r_i + c_i$) in respect of their significance for the performance generation. It can be realised for the group 'causes' that the SF 'product range' is the most influencing factor, which largely determines all other SFs. In contrast the SF 'structural circumstances' has the lowest impact on the whole system. Considering the group 'receivers', the SF 'financial success' is mostly influenced by the other SFs compared to the SF 'pricing', which is less determined by the other ones.

5.2.2. Tailor-made impact-relation map

In this section the identified SFs and only their above-average calculated influence intensity from **Table 1** as well as the group specification of receivers and causes from **Table 2** is finally visualised in an appropriate IRM (**Figure 4**).

According to the coordinate system in **Figure 4**, the ordinate represents the difference between received and outgoing effects of a factor. Factors that can be characterised as causes, for example 'ability to supply', are always pictured in the positive range. Whereas, receivers like the factor 'pricing' are depicted in the negative value range. Furthermore, the abscissa displays the overall intensity of the influence relationship of an individual factor. The further away a factor is located from the coordinate origin, the greater its total influence intensity in the whole system is. Following **Figure 4**, the factor 'financial success' is the most performance-relevant factor in relation to the others.

Performance Management by Causal Mapping: An Application Field of Knowledge Management 115 http://dx.doi.org/10.5772/intechopen.70297



Figure 4. Impact-relation map.

Generally, through the construction of an IRM, a better comprehension of the relevant direct and indirect causal relationships can be developed. Besides, the IRM underlines which SFs are most important for the corporate management and the focus should lay on them. Compared to a qualitative causal mapping process DEMATEL strictly distinguishes the SFs between causes and receivers and quantifies their cause-and-effect relationships [53, 54]. However, because the individual evaluations are ordinal a cardinal interpretation of the SFs' causal relations cannot be provided. Only a systematisation by building a hierarchy among SFs is possible.

6. Conclusion

Causal knowledge on SFs underlying financial performance generation is an important prerequisite for an effective PM. For this purpose, important parts of the PM have to be drawn on the subjective experiences and knowledge of the employees. It is the current task for KM to extract the subject-bound tacit knowledge and make it explicitly available for the management of an organisation. Subsequently, by application of a convenient mapping method revealed tacit knowledge has to be aggregated, structured as well as systematised in a more general and for the employees' applicable manner. In addition, the complex financial performance generation process will be represented and analysed as for performance relevance of the SFs and their causal relations. In this way, a general and clarified understanding of the performance generation is achievable among the employees. The concept of causal mapping and the multi-criteria DEMATEL method illustrate approaches how to construct a causal map from the base of externalised tacit knowledge. Both methods differ in procedures and in results. Causal mapping offers a low quality of the identified causal structures of SFs because of the lack of quantitative assessment and the highly subjective aggregation of the implicit knowledge. However, applying DEMATEL in the mapping context, the subjective bias can be minimised by a systematic and transparent pairwise evaluation of the SFs. Because of its replicability it achieves intersubjectivity. But since the discovered causal relationships among the factors are only interpretable on an ordinal scale, strategic forecasts of future performance developments are only possible to a limited extent.

To achieve an objective validity, the existence of adequate data and the use of suitably selected statistical procedures are necessary. If for all variables of the causal map manifest time series data are available, the validation of causal relationships can be done by using a multivariate time series model. When the variables of the causal map are not directly observable, but can be operationalised as latent variables with appropriate factors, structural equation modelling can be used to validate the cause-and-effect relationships among the SFs [29]. The statistical validation of the causal relationship network objectifies the previous ordinal data in metric forms to achieve relative comparability and clear predictability. So, the significance of the map is optimised compared to the one constructed by DEMATEL. Finally, in the context of PM, the performance realisation and generation can be represented and analysed qualitatively as well as quantitatively by the validated map in a comprehensible way.

Author details

Sarah Kölbel, Wolfgang Ossadnik* and Stefan Gergeleit

*Address all correspondence to: wolfgang.ossadnik@uni-osnabrueck.de

Department of Management Science/Management Accounting and Control, University of Osnabrück, Osnabrück, Germany

References

- [1] Kaplan RS, Norton DP. Strategy Maps: Converting Intangible Assets into Tangible Outcomes. 1st ed. Boston, MA: Harvard Business School Press; 2004. p. 454
- [2] van Veen-Dirks P, Wijn M. Strategic control: Meshing critical success factors with the balanced scorecard. Long Range Planning. 2002;35(4):407-427. DOI: 10.1016/S0024-6301 (02)00066-3
- [3] Lebas MJ. Performance measurement and performance management. International Journal of Production Economics. 1995;**41**(1-3):23-35. DOI: 10.1016/0925-5273(95)00081-X

- [4] Malmi T. Balanced scorecards in Finnish companies: A research note. Management Accounting Research. 2001;12(2):207-220. DOI: 10.1006/mare.2000.0154
- [5] Ittner CD, Larcker DF. Extending the boundaries: Nonfinancial performance measures. In: Chapman CS, Hopwood AG, Shields MD, editors. Handbook of Management Accounting Research. 3rd ed. Amsterdam: Elsevier; 2009. pp. 1235-1251. DOI: 10.1016/ S1751-3243(07)03002-7
- [6] Ossadnik W. Performance management of internationally acting firms as a challenge for management accounting and control and stimulant of their further development [Performance Management international agierender Unternehmen als Herausforderung für die Controllinglehre und Stimulanz für deren Weiterentwicklung]. In: Funk W, Rossmanith J, editors. International Accounting and International Controlling [Internationale Rechnungslegung und Internationales Controlling]. 3rd ed. Wiesbaden: Gabler; 2016. pp. 1-31
- [7] Neely A, Gregory M, Platts K. Performance measurement system design: A literature review and research agenda. International Journal of Operations & Production Management. 2005;25(12):1228-1263. DOI: 10.1108/01443570510633639
- [8] Abernethy MA, Horne M, Lillis AM, Malina MA, Selto FH. A multi-method approach to building causal performance maps from expert knowledge. Management Accounting Research. 2005;16(2):135-155. DOI: 10.1016/j.mar.2005.03.003
- [9] Aranda C, Arellano J. Consensus and link structure in strategic performance measurement systems: A field study. Journal of Management Accounting Research. 2010;22(1):271-299. DOI: 10.2308/jmar.2010.22.1.271
- [10] Franco-Santos M, Lucianetti L, Bourne M. Contemporary performance measurement systems: A review of their consequences and a framework for research. Management Accounting Research. 2012;23(2):79-119. DOI: 10.1016/j.mar.2012.04.001
- [11] Edwards JR, Bagozzi RP. On the nature and direction of relationships between constructs and measures. Psychological Methods. 2000;5(2):155-174. DOI: 10.1037//1082-989X.5.2,15
- [12] Gold AH, Malhotra A, Segars AH. Knowledge management: An organizational capabilities perspective. Journal of Management Information System. 2001;18(1):185-214. DOI: 10.1080/07421222.2001.11045669
- [13] Bolloju N, Khalifa M, Turban E. Integrating knowledge management into enterprise environments for the next generation decision support. Decision Support Systems. 2002;33(2):163-176. DOI: 10.1016/S0167-9236(01)00142-7
- [14] Wu WW. Choosing knowledge management strategies by using a combined ANP and DEMATEL approach. Expert Systems with Applications. 2008;35(3):828-835. DOI: 10.1016/ j.eswa.2007.07.025
- [15] Rowe BJ, Widener SK. Where performance measurement and knowledge management meet: Evaluating and managing corporate knowledge. Journal of Accounting and Finance. 2011;11(2):91-106

- [16] French S, Maule J, Papamichail N. Decision Behaviour, Analysis and Support. Cambridge: Cambridge University Press; 2009. p. 472
- [17] Polanyi M. The logic of tacit inference. Philosophy. 1966;41(155):1-18
- [18] Nonaka I. A dynamic theory of organizational knowledge creation. Organization Science. 1994;5(1):14-37. DOI: 10.1287/orsc.5.1.14
- [19] Lubit R. Tacit knowledge and knowledge management: The keys to sustainable competitive advantage. Organizational Dynamics. 2001;29(4):164-178. DOI: 10.1016/S0090-2616 (01)00026-2
- [20] Ambrosini V, Bowman C. Tacit knowledge: Some suggestions for operationalization. Journal of Management Studies. 2001;38(6):811-829. DOI: 10.1111/1467-6486.00260
- [21] Zack M, McKeen J, Singh S. Knowledge management and organizational performance: an exploratory analysis. Journal of Knowledge Management. 2009;13(6):392-409. DOI: 10.1108/13673270910997088
- [22] Ossadnik W, Kaspar RH, Schinke SM. Constructing a tailor-made performance management system supported by knowledge elicitation tools and dynamic modeling. International Journal of Business Research and Management. 2013;4(4):75-98
- [23] Cooke NJ. Varieties of knowledge elicitation techniques. International Journal of Human-Computer Studies. 1994;41(6):801-849. DOI: 10.1006/ijhc.1994.1083
- [24] Markóczy L, Goldberg J. A method for eliciting and comparing causal maps. Journal of Management. 1995;21(2):305-333. DOI: 10.1177/014920639502100207
- [25] Eden C, Ackermann F, Cropper S. The analysis of cause maps. Journal of Management Studies. 1992;29(3):309-324. DOI: 10.1111/j.1467-6486.1992.tb00667.x
- [26] Huff AS. Mapping strategic thought. In: Huff AS, editor. Mapping Strategic Thought. Chichester: Wiley; 1990. pp. 11-49
- [27] Ferreira A, Otley D. The design and use of performance management systems: An extended framework for analysis. Management Accounting Research. 2009;20(4):263-282. DOI: 10.1016/j.mar.2009.07.003
- [28] Eden C, Jones S, Sims D, Smithin T. The intersubjectivity of issues and issues of intersubjectivity. Journal of Management Studies. 1981;18(1):37-47. DOI: 10.1111/j.1467-6486.1981.b00090.x
- [29] Malina MA, Norreklit HSO, Selto FH. Relations among measures, climate of control, and performance measurement models. Contemporary Accounting Research. 2007;24(3):935-982. DOI: 10.1506/car.24.3.10
- [30] Fiol CM. Maps for managers: Where are we? where do we go from here? Journal of Management Studies. 1992;29(3):267-285. DOI: 10.1111/j.1467-6486.1992.tb00665.x
- [31] Langfield-Smith K. Exploring the need for a shared cognitive map. Journal of Management Studies. 1992;29(3):349-368. DOI: 10.1111/j.1467-6486.1992.tb00669.x

- [32] Luft J, Shields MD. Mapping management accounting: Graphics and guidelines for theory-consistent empirical research. Accounting, Organizations and Society. 2003;28(2-3): 169-249. DOI: 10.1016/S0361-3682(02)00026-0
- [33] Eden C. Strategic thinking with computers. Long Range Planning. 1990;23(6):35-43. DOI: 10.1016/0024-6301(90)90100-I
- [34] Laukkanen M. Comparative cause mapping of organizational cognitions. Organization Science. 1994;5(3):322-343. DOI: http://dx.doi.org/10.1287/orsc.5.3.322
- [35] Eden C. Analyzing cognitive maps to help structure issues or problems. European Journal of Operational Research. 2004;**159**(3):673-686. DOI: 10.1016/S0377-2217(03)00431-4
- [36] Nelson RE, Mathews KM. Cause maps and social network analysis in organizational diagnosis. Journal of Applied Behavioral Science. 1991;27(3):379-397. DOI: 10.1177/0021886391273009
- [37] Weick KE, Bougon MG. Organizations as cognitive maps: Charting ways to success and failure. In: Sims HP, Gioia DA, editors. The Thinking Organization: Dynamics of Organizational Social Cognition. 1st ed. San Francisco: Jossey-Bass; 1986. pp. 102-135
- [38] Eden C. Cognitive mapping. European Journal of Operational Research. 1988;36(1):1-13. DOI: 10.1016/0377-2217(88)90002-1
- [39] Montazemi AR, Conrath DW. The use of cognitive mapping for information requirements analysis. MIS Quarterly. 1986;10(1):45-56. DOI: 10.2307/248879
- [40] Tegarden DP, Sheetz SD. Group cognitive mapping: A methodology and system for capturing and evaluating managerial and organizational cognition. Omega. 2003;31(2):113-125. DOI: 10.1016/S0305-0483(03)00018-5
- [41] Ackermann F, Eden C. Using causal mapping with group support systems to elicit an understanding of failure in complex projects: Some implications for organizational research. Group Decision and Negotiation. 2005;14(5):355-376. DOI: 10.1007/s10726-005-8917-6
- [42] Scavarda AJ, Bouzdine-Chameeva T, Goldstein SM, Hays JM, Hill AV. A methodology for constructing collective causal maps. Decision Sciences. 2006;37(2):263-283. DOI: 10.1111/ j.1540-5915.2006.00124.x
- [43] Wouters M, Wilderom C. Developing performance-measurement systems as enabling formalization: A longitudinal field study of a logistics department. Accounting, Organizations and Society. 2008;33(4-5):488-516. DOI: 10.1016/j.aos.2007.05.002
- [44] Hodgkinson GP, Maule AJ, Bown NJ. Causal cognitive mapping in the organizational strategy field: A comparison of alternative elicitation procedures. Organizational Research Methods. 2004;7(1):3-26. DOI: 10.1177/1094428103259556
- [45] Akkermans H, van Helden K. Vicious and virtuous cycles in ERP implementation: A case study of interrelations between critical success factors. European Journal of Information Systems. 2002;11(1):35-46. DOI: 10.1057/palgrave.ejis.3000418

- [46] Wouters M. A developmental approach to performance measures—Results from a longitudinal case study. European Management Journal. 2009;27(1):64-78. DOI: 10.1016/j. emj.2008.06.006
- [47] Ossadnik W, Schinke SM, Kaspar RH. Group aggregation techniques for analytic hierarchy process and analytic network process: A comparative analysis. Group Decision and Negotiation. 2016;25(2):421-457. DOI: 10.1007/s10726-015-9448-4
- [48] Fontela E, Gabus A. Current perceptions of the world problematique. In: Churchman CW, Mason RO, editors. World Modeling—A Dialogue. Amsterdam: North-Holland Publisher; 1976. pp. 81-88
- [49] Tsai WH, Chou YW, Lee KC, Lin WR, Hwang E. Combining decision making trial and evaluation laboratory with analytic network process to perform an investigation of information technology auditing and risk control in an enterprise resource planning environment. Systems Research and Behavioral Science. 2013;30(2):176-193. DOI: 10.1002/sres.2129
- [50] Lin CL, Tzeng GH. A value-created system of science (technology) park by using DEMATEL. Expert Systems with Applications. 2009;36(6):9683-9697. DOI: 10.1016/j.eswa. 2008.11.040
- [51] Li Y, Duan W, Sun Y, Zhang Q. A group DEMATEL approach based on interval estimation. Journal of Convergence Information Technology. 2013;8(10):292-299. DOI: 10.4156/ jcit.vol8.issue10.36
- [52] Yang JL, Tzeng GH. An integrated MCDM technique combined with DEMATEL for a novel cluster-weighted with ANP method. Expert Systems with Applications. 2011;38(3):1417-1424. DOI: 10.1016/j.eswa.2010.07.048
- [53] Chen FH, Hsu TS, Tzeng GH. A balanced scorecard approach to establish a performance evaluation and relationship model for hot spring hotels based on a hybrid MCDM model combining DEMATEL and ANP. International Journal of Hospitality Management. 2011; 30(4):908-932. DOI: 10.1016/j.ijhm.2011.02.001
- [54] Wu HY. Constructing a strategy map for banking institutions with key performance indicators of the balanced scorecard. Evaluation and Program Planning. 2012;35(3):303-320. DOI: 10.1016/j.evalprogplan.2011.11.009
- [55] Miao R, Xu F, Zhang K, Jiang Z. Development of a multi-scale model for customer perceived value of electric vehicles. International Journal of Production Research. 2014;52(16):4820-4834. DOI: 10.1080/00207543.2014.890757
- [56] Tzeng GH, Chen WH, Yu R, Shih ML. Fuzzy decision maps: A generalization of the DEMATEL methods. Soft Computing. 2010;14(11):1141-1150. DOI: 10.1007/s00500-009-0507-0
- [57] Behera PK, Mukherjee K. Application of DEMATEL and MMDE for analyzing key influencing factors relevant to selection of supply chain coordination schemes. International Journal of Information Systems and Supply Chain Management. 2015;8(2):49-69. DOI: 10.4018/ IJISSCM.2015040104

Intelligent Knowledge Retrieval from Industrial Repositories

Antonio Martin, Mauricio Burbano and Carlos León

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/intechopen.70724

Abstract

Actually, a large amount of information is stored in the industrial repositories. Accessing this information is complicated, and the techniques currently used in metadata and the material chosen by the user do not scale efficiently in large collections. The semantic Web provides a frame of reference that allows sharing and reusing knowledge efficiently. In our work, we present a focus for discovering information in digital repositories based on the application of expert system technologies, and we show a conceptual architecture for a semantic search engine. We used case-based reasoning methodology to create a prototype that supports efficient retrieval knowledge from digital repositories. OntoEnter is a collaborative effort that proposes a new form of interaction between users and digital enterprise repositories, where the latter are adapted to users and their surroundings.

Keywords: artificial intelligence, ontology, semantic web, experts systems, CBR, interoperability

1. Introduction

Nowadays, industrial information provides effective knowledge of existing resources through databases and repositories, which provide details on the hosted equipment, including information on their capacity, performance, start/stop dates, turbine and generators models, etc. All of these properties and information are stored in digital repositories, digital files, and business websites. To collect, contribute, and share the knowledge about the resources installed in the industrial area, online databases named digital industry repository (DIR) is used. Therefore, the way in which information and knowledge stored in digital repositories is retrieved is of vital importance. DIRs provide centralized hosting and access to content, establish permissions, and



© 2017 The Author(s). Licensee InTech. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. controls for access to content, the ability to share digital objects or files, to protect the intellectual, and integrity property rights of content owners and creators, etc.

So far, traditional search engines treat the information as an ordinary database that manages the content inefficiently. Current search engines retrieve the information by comparing the contents of the database with searched patterns. The generated result is a list of data that contain this patter. Although search engines are becoming more effectively, information overload hinders search and accurate knowledge retrieval. Consequently, it is necessary to develop new semantic and intelligent models that contribute new possibilities. The presented work offers a new approach to the information retrieval based on semantics and intelligent models. For this, the case-based reasoning (CBR) technique is applied contributing to the goal of improving knowledge recovery in the industrial field.

A significant number of researchers have already investigated the application of intelligence and semantic techniques, but just a few from the point of view of full integrated of both technologies an industrial environment. There are researchers and related field works that include ontology retrieval methods such as [1] present a system that uses an ontology query model to analyze the usefulness of ontologies in effectively performing document searches and proposes an algorithm to refine ontologies for information retrieval tasks with preliminary positive results. In this paper [2], real-time image capture was achieved by using digital camera technology and image processing technology. By extracting the glue line curve from image, thinning glue curve by morphological method, and extracting the frame information, the closure and quality of the glue curve can be detected. Results of test show that the effect is satisfactory and the method is effective. The major contribution of [3] is a novel semantic query expansion technique that combines association rules with ontologies and Natural Language Processing techniques, which utilizes the explicit semantics as well as other linguistic properties of unstructured text corpus. It makes use of contextual properties of important terms discovered by association rules, and ontology entries are added to the query by disambiguating word senses.

Semantic Web utilizes concepts, taxonomic relations, and nontaxonomic relations in a given domain ontology to capture knowledge efficiently. For example, [4] describes one component of a knowledge management platform with a multiagent search module (MASH), which employs domain ontology to search for Web pages that contain relevant information to each concept in the domain of interest. The search is then constrained to a specific domain to avoid as much as possible the analysis of irrelevant information. Ref. [5] expounds the function of each layer and analyses the implementation of this system from the knowledge organization and expression and knowledge retrieval and proposes a framework of knowledge management system based on ontology. This management system establishes a sharable ontology that can be understood both by human and computer, which people can found more relations of different concepts through a better circumstance of knowledge retrieval interface. The work of [6] proposes an ontology-based user model, called user ontology, for providing personalized information service in the Semantic Web which utilizes concepts, taxonomic relations, and nontaxonomic relations in a given domain ontology to capture the users' interests. The research of [7] presents a semantics-based digital project which provides faceted search and represents a novel

approach to Digital Libraries, integrating social Web and multimedia elements in a semantically annotated repository. In other investigation, Ref. [8] describes the architecture of the dynamic retrieval analysis and semantic metadata management system (DREAM) designed to automatically and intelligently index huge repositories of special effects video clips, based on their semantic content, using a network of scalable ontologies to enable intelligent retrieval. Ref. [9] presents an information search and retrieval framework based on the semantically annotated multifacet product family ontology. The major contribution of [10] is an innovative comprehensive semantic search model, which extends the classic information retrieval (IR) model, addresses the challenges of the massive and heterogeneous Web environment.

There are a lot of researchers on applying these new technologies into current information retrieval systems, but no research addresses artificial intelligence (AI) and semantic issues from the whole life cycle and architecture point of view [11]. This article analyses the search methods efficiency in a distributed data space such as industrial information repositories. The paper presents an intelligent proposal to optimize search engines in a specific industrial domain and to focus our discussion on the indexing and retrieval strategies of cases and provides the application technical aspects. This paper describes the current problems of semantic interoperability and proposes an intelligent method to address them. To do these, technologies based on metadata and intelligent techniques are used. The main proposal goal is the intelligent search management in decentralized industrial repositories, where no global information scheme exists. The most important novel introduced by this proposal is that contextual user profiles are built based on ontologies and metadata facilitating the ontological search using expert systems technologies. The objective has focused on creating technologically complex environments industrial domain and incorporates Semantic Web and AI technologies to enable precise location of industrial resources [12]. For this reason, we are improving representation by incorporating more metadata from within the information.

We propose a new paradigm to achieve efficient knowledge retrieval from digital repositories. This paper presents an intelligent search engine for industrial process, especially for resources repositories, and proposes an intelligent agent–based personalized model. One major research area is intelligent systems, with the general intention to replace human operators with intelligent agents. We have used CBR methodology to develop a prototype for supporting efficient retrieval knowledge from DIR.

In the following sections, we review the CBR framework and its features for implementing the reasoning process over ontologies. Section 2 presents a general overview about the industrial domain and technology infrastructure, analyzing its failures and discovering the needs that push us toward new intelligent paradigms. Section 3 analyses ontology requirements and proposes the design criteria to guide the development of ontologies for knowledge-sharing purposes. Then, we show the methodology followed to conduct this research, and we describe the semantic-based management system in DIRs environment. Next section concerns the design of a prototype system for semantic search framework, in order to verify that our proposed approach is an applicable solution. Moreover, the functional requirements of the engine and the knowledge base are described in detail. Finally, we present the results of our work on the adaptation of the framework, and we outline the future works.

2. Importance of applying ontology in industrial sector

Our objective here is thus to contribute to a better knowledge retrieval in the industry repositories field. Ontologies are being developed to facilitate knowledge sharing and reuse. In this section, we explain more formally what ontologies are and what problems can arise from knowledge sharing in industrial area. We have proposed a method to efficiently search the target information on a digital repository network with multiple independent information sources [13]. The use of AI and ontologies as a knowledge representation formalism offers many advantages in information retrieval [14]. In this chapter we analyzed the relationship between both factors ontologies and expert systems.

Currently, the electronic search is based mainly on matching keywords introduced by the users with sought data Web pages containing those keywords. The ambiguity of word blends and phrases and the poor linguistic features of Web content indexing mechanisms greatly affect the results obtained from Web resource searches. The efficiency of the results search obtained can vary depending on the quality of the search query from a limited set of results to a too large number of irrelevant results. For certain cases specifying a couple of keywords can be enough, if they are really specific and no ambiguity is possible. In another way, many Web users search for information that cannot be described easily by a set of keywords, and this is due to the wideness of expected results, which cannot be retrieved from existing search engines just with one search query [15].

Industrial repositories contain a large volume of digital information, generally focusing on making their knowledge resources to improve associate decision-support systems. Within a pool of heterogeneous and distributed information resources, users take site-by-site searching. Thus, considerable effort is required in creating meaningful metadata, organizing and annotating digital documents, and making them accessible. The presentation of semantic-enabled resources introduces some benefits of the Semantic Web technology as a possibility to perform a semantic search, integration of heterogeneous data, and use of semantically annotated search results by software. This work concerns applications of the Semantic Web technology for improving existing information search systems by adding semantically enabled extensions that enhance information retrieval from information systems.

A recent comprehensive document covering the main aspects of ontologies in AI research is the technical roadmap of the ontology field in Europe and worldwide produced by the OntoWeb project [16]. In this chapter, we want to emphasize that the first step toward real portability between systems is ontologies. Ontologies can be effectively used to address the problem of global and general models construction between similar domains. Furthermore, it is possible to instantiate and adapt ontology with a specific configuration to automatically build and validate new models [17]. With respect to the research involved in this study, ontologies can provide:

- Share and common understanding of the knowledge domain that can be communicated among agents and application systems.
- Explicit conceptualization that describes the semantics of the data.

The proposal is based on the principle that information items are abstracted to a characterization by introducing metadata, which is used and processed by search engines. This principle is based on a vocabulary/ontology that is shared in order to access the relevant sources of information. This creates new challenges for the research community and motivates scientists to look a recovery approach based on ontologies and intelligent information that automatically search and filter information based on a higher level of understanding required. In this sense, in the present work, we make an effort of investigating techniques that use ontologies to improve the effectiveness in knowledge retrieval. Thus, ontologies are key elements for the definition of the semantic Web [18].

To achieve these objectives, we must consider the interoperability of information. In other words, the ability of different information systems, platforms, and services to share, communicate, and exchange data, information and knowledge effectively and accurately, as well as integrate with other systems, applications, and services in order to deliver new electronic services and products.

Initiatives in this sense such as interoperability between different industrial domains require the establishment of collaborative semantic repositories between private and public sector organizations. Especially, semantic interoperability is necessary, which has a special relevance within the program to support the implementation of distributed services.

3. Challenging the interoperability between systems

Industry and companies are seeking to gain maximum business value from their investments in information and communications technologies. The industry has recognized the everincreasing importance of systems and software interoperability to enable business processe/ government service development and the integration of systems and business processes. In the business case, it expands to include the ability of two or more business processes, or services, to easily or automatically work together [19]. In order to reduce costs of industrial integration and inefficiencies, increase business agility, and allow the adoption of new and emerging technologies, the ability to interoperate between systems is key issue. For two systems to be interoperable, they must be able to exchange data and subsequently present that data such that a user can understand it. Interoperability describes the extent to which systems and devices can exchange data and interpret that shared data. Connectivity and interoperation among computers and entities and among software components can increase the flexibility and agility of industrial systems, thus reducing administrative and software costs for industry.

In June 2002, European heads of state adopted the Europe Action Plan 2005 at the Seville summit. It calls on the European Commission to issue an agreed interoperability framework to support the delivery of European digital services to enterprises [20].

This document recommends policies and technical specifications for linking public administration information systems across the EU. This research is based on open standards and the use of open source software. These are the pillars to support the European provision of digital services in the recently adopted European Framework of Interoperability (EIF) [21] and its Spanish equivalent [22]. This document is the reference for the interoperability of the new Interoperable Program of Digital PanEuropean Services Provision to Public Administrations, Companies, and Citizens (IDAbc). European institutions and bodies should use the EIF for their operations with each other and with the citizens, businesses, and administrations of the respective EU Member States (IEF, 2014). Member States' administrations should use the guidance provided by the EIF to complement its national interoperability frameworks with a pan-European dimension and thus enable pan-European interoperability.

In this context, interoperability is the ability of information and communication technology systems and the business processes they support to exchange data and enable the exchange of information and knowledge. The ISO/IEC 2382 Information Technology Vocabulary defines interoperability as the ability to communicate, execute programs, or transfer data between several functional units in a way that requires the user to have little or no knowledge of the unique characteristics of those units. Interoperability can be considered on very different abstraction levels, and the distinctions to be made in this respect cut across all the other matrix dimensions. An interoperability framework can be described as a set of standards and guidelines, which describe the way in which organizations have agreed, or should agree, to interact with each other.

At the level of technical infrastructure, the industry is approaching interoperability through standards and in many cases conceptualizes those standards through stacks of technology. Technology stacks are conceptual layers of software and software functionality that interoperate between layers within stacks and between stacks in the same conceptual layer [23]. Within a continuum rank from a very concrete to a very abstract perspective, it is possible to distinguish three layers as shown in next **Figure 1**.

The main semantic interoperability objective is to improve communication on industrial knowledge related both between machines and between humans. To achieve this, a two-fold approach is necessary to achieve a unified ontology and tackle specific and clearly delineated issues. Inside semantic interoperability, various dimensions, such as medial/administrative or human/machines levels, can be distinguished. Organizational interoperability is defined as the state in which the organizational components of the industrial system are able to function perfectly together. The goal is an integrated industrial system, which provides efficient, effective, and holistic. The functional objective is to allow data to be exchanged between different platform in various corporations using different software, hardware, equipment, etc. from multiple manufacturers. Technical interoperability allows communication and iteration between systems from different manufacturers. Technical data, uniform presentation of data, uniform user controls, uniform safeguarding data security and integrity, uniform protection of industrial confidentiality, and uniform assurance of a common degree of service quality.

Numerous efforts are being leveraged by many standard efforts to address semantic and organizational interoperability and are proving to be a model for addressing semantic and organizational interoperability such as ebXML, RosettaNet, the new CEFACT/UN work to align their global work process standards with Web services, etc.



Figure 1. Conceptual interoperability layers.

The achievement of semantic and organizational interoperability requires strictly agreeing on the meaning of information and aligning business processes between companies/governments. At one level, general interprofessional frameworks and software infrastructure approaches can and are being developed for semantics and business processes. For example, the general semantics of major business transactions, such as purchase orders and invoices, is described through standards such as Universal Business Language (UBL), CEFACT Core Components, and Open Applications Group Integration Standard.

4. Intelligent and semantic architecture

The goal is achieved from a search perspective, with possible intelligent infrastructures to construct decentralized industrial repositories, where no global schema exists. This goal implies the application of CBR technique [24]. The prototype is the main tool to verify that the proposed architecture is an applicable solution, and this work attempts to achieve such verification by documenting in the proposal solution. In order to support the semantic of retrieval knowledge in industrial repositories, we develop a prototype named OntoEnter based on ontologies and expert system technologies. Obviously, our system is a prototype; nevertheless, it gives a good picture of the on-going activities in this new and important field. The architecture of our system is shown in **Figure 2**, which mainly includes three parts: the search engine, ontology knowledge base, and intelligent user interface.

The proposed architecture relies on the approach to efficiently retrieve information through metadata characterization and the inclusion of domain ontology. It involves using ontology

as a vocabulary to define complex and multi-relational case structures to support CBR processes. Our system works by comparing objects that can be retrieved through heterogeneous repositories and capturing a semantic view of the independent world of data representation.

4.1. The case-based reasoning engine

Keeping in mind that our final goal is to reformulate queries in the ontology to queries in another with least loss of semantics, we come to a process for addressing complex relations between ontologies. CBR is widely discussed in the literature as a technology for building information systems to support knowledge management, where metadata descriptions are used to characterize knowledge elements. CBR is a paradigm of problem solving that solves a new problem, in our case a new search, remembering a similar previous situation and reusing information and knowledge of that situation. Recovering one solves a new problem or more previously experienced cases, reusing the case, reviewing, and retaining. This approach when a description of the current problem is input to the system, the reasoning cycle may be described by the following processes **Figure 3**.

The system retrieves the closest matching cases stored in a case base. Reuse a complete design where case-based and slot-based adaptation can be hooked, provided. If appropriate, the validated solution is added to the case for use in solving future problems. Review the proposed solution, if necessary. Given that the proposed result may be inadequate, this process can correct the first proposed solution. Keep the new solution as part of a new case. This process allows CBR to learn and create a new solution. The solution is validated by comments from the user or the environment.

In our CBR application, problems are described by metadata concerning desired characteristics of an industry resource, and the solution to the problem is a pointer to a resource described



Figure 2. System architecture of OntoEnter.



Figure 3. Case-based reasoning cycle in OntoEnter.

by metadata [25]. The development of a quite simple CBR application already involves a number of sttif, such as collecting case and background knowledge, modeling a suitable case representation, defining an accurate similarity measure, implementing retrieval functionality, and implementing user interfaces. CBR case data could be considered as a portion of the knowledge (metadata) about an OntoEnter object [26]. Every case contains both description problem and the associated solution.

Compared to other AI approaches, CBR reduces the effort required to acquire knowledge and representation significantly, which is undoubtedly one of the main reasons for the commercial success of CBR applications. However, implementing a CBR application from scratch remains a time-consuming software engineering process and requires a lot of specific experience beyond pure programming skills [27]. In this work, we have chosen framework jColibri to develop the intelligent search.

JColibri is a java-based configuration that supports the development of knowledge intensive CBR applications and helps in the integration of ontology in them [28]. The metadata descriptions of the resources and objects (cases) are abstracted from the details of their physical representation and are stored in the case base. This way the same methods can operate over different types of information repositories. The mapping between the two layers is done by connectors that read the values of the columns and the ontology of the database and return them to the application. That is to say, assign these values to different attributes of the case. Based on the same idea, the case base implements a common interface for similarity methods

to evaluate cases. This includes the generation of case representations, the definition of similarity measures, the testing of retrieval and use of explanation functionality, and finally, the implementation of stand-alone applications. The main focus of methods in this category is to find similarity between cases.

The use of structured representations of cases requires approaches for the evaluation of similarities that allow to compare two objects structured in different ways, in particular, objects belonging to different classes of objects. An important advantage of the similarity box recovery is that if there is no case that exactly matches the user's requirements, this may show cases that are more similar to your query. The use of structured representations of cases requires approaches for similarity assessment that allow to compares two differently structured objects, in particular, objects belonging to different object classes. The retrieval strategy used in our system is correlation technique. Correlation is a bivariate analysis that measures the strengths of association between two variables. A line that runs through all the data points and has a positive slope represents a perfect correlation between the two objects. In statistics, the value of the correlation coefficient varies between +1 and -1. When the value of the correlation coefficient is about ± 1 , then it is said to be a perfect degree of association between the two variables [29]. As the value of the correlation coefficient goes toward 0, the relationship between the two variables will be weaker, **Figure 4**.

We measure correlations with the Pearson correlation method. The Pearson coefficient is a more complex and sophisticated approach to finding similarity. This best fit line is generated by the Pearson coefficient, which is the similarity score. The following formula is used to calculate the Pearson r correlation:

$$\rho_{x,y} = \frac{\operatorname{cov}(X, Y)}{\sigma_x \sigma_y} = \frac{E\left[(X - \mu_x)(Y - \mu_y)\right]}{\sigma_x \sigma_y}$$

Where:

r = Pearson r correlation coefficient.

N = number of value in each data set.

 $\sum xy = sum of products of paired scores.$

 $\sum x = \text{sum of } x \text{ scores.}$

 $\sum y = sum of y scores.$

 $\sum x^2$ = sum of squared x scores.

 $\sum y^2$ = sum of squared y scores.

Pearson r correlation is widely used in statistics to measure the degree of the relationship between linear related variables. For example, in the industrial stocks and storage, if we want to measure how two commodities are related to each other, Pearson r correlation is used to measure the degree of relationship between the two commodities. The coefficient is found by dividing the covariance by the product of the standard deviations of the


Figure 4. The method generates a best-fit line between attributes in two data objects.

attributes of two data objects. The advantage of the Pearson Coefficient over other techniques is that it is more robust against data that is not standardized. For example, if one person gives the rank for movies "a", "b", and "c" with scores of 1, 2, and 3, respectively, has a perfect correlation to someone who ranked the same movies with 4, 5, and 6. The following python code implements the Pearson coefficient for the same data described previously.

```
#Return the Pearson Coefficient for person1 and person2
def similarity_pearson(preferences, person1, person2):
 #Get the list of mutually rated items
 shared_items={}
 for item in preferences[person1]:
   if item in preferences[person2]: shared items[item]=1
 #Find the number of elements
 num shared = len(shared items)
 #If there are no ratings in common, return 0
 if num_shared == 0: return 0
 #Add up all the preferences
 sum1 = sum([preferences[person1][item] for item in shared items])
 sum2 = sum([preferences[person2][item] for item in shared items])
 #Sum up the squares
 sum1_squares = sum([pow(preferences[person1][item],2) for item in shared_items])
 sum2_squares = sum([pow(preferences[person2][item],2) for item in shared_items])
 #Sum up the products
 prod sum = sum([preferences[person1][item]*preferences[person2][item] for item in shared items])
 #Calculate the Pearson score
 num = prod sum - (sum1*sum2/num shared)
 den = sqrt((sum1_squares-pow(sum1,2)/num_shared)*(sum2_squares - pow(sum2,2)/num_shared))
 if den == 0: return 0
 r = num/den
 return r
```

The open source jColibri system provides a framework for building CBR systems based on state-of-the-art software engineering techniques. The reason for choosing the jColibri framework is based on a comparative analysis between it and other frameworks, designed to facilitate the development of CBR applications. jColibri enriches the other shells CBR: CATCBR, CBR * Tools, IUCBRF, Orenge, in several aspects:

- Availability: open source framework.
- Implementation: Java implementation is one of our main requirements with respect to easy integration into the OntoEnter system, which is implemented in the J2EE environment.

Another decision criterion for our choice is related to the fact that jColibri offers the opportunity to incorporate the ontology in the CBR application to use it for the representation of cases and methods of reasoning based on content to evaluate the similarity between them. Providing easy-to-use model generation, data import, similarity modeling, explanation, and test functionality along with convenient graphical user interfaces, the tool allows even CBR beginners to quickly create their first CBR applications. However, at the same time, it ensures sufficient flexibility to enable expert users to implement advanced CBR applications.

4.2. Ontology knowledge base

Semantic modeling can help define the data and relationships between these entities. An information model provides the ability to abstract different types of data and provides an understanding of how data elements are related. A semantic model is a type of information model that supports the modeling of entities and their relationships. The total set of entities in our semantic model comprises the class taxonomy that we use in our model to represent the real world. Together, these ideas are represented by an ontology—the semantic model vocabulary that provides the basis upon which user-defined model queries are formed. The model supports the representation of entities and their relationships and can withstand the constraints on those relationships and entities. This provides the semantic composition of the information model [30].

Semantic models allow users to ask questions about what is happening in a modeled system in a more natural way. As an example, an oil production company could consist of five geographic regions, each region containing three to five drilling rigs and each drilling rig controlled by various control systems, each with a different purpose. One of these control systems could control the temperature of the extracted oil, while another could control the vibration in a pump. A semantic model will allow the user to ask a question like "What is the temperature of the oil extracted in Platform 3?" Without having to understand details such as, what specific control system supervises that information or what physical sensor is reporting temperature of oil on that platform.

The understanding provided through semantic models is fundamental to be able to correctly drive the correct ideas of supervised instrumentation that can ultimately lead to optimize business processes or, in this case, city services. As a result, semantic models can greatly improve the usefulness of the information obtained through operations integration solutions.

Ontology models can be used to relate the physical world, to the real world, in the line-ofbusiness and decision makers. In the physical world, a control point such a valve or temperature sensor is known by its identifier in a particular control system, possibly through a tag name like 14-WW13. This could be one of several thousand identifiers within any given control system, and there could be many similar control systems across an enterprise. To further complicate the problem of information referencing and aggregation, other data points of interest could be managed through databases, files, applications, or component services with each having its own interface method and naming conventions for data accessing. A key value of the semantic model then is to provide access of information in the context of the real world in a consistent way. Within a semantic model implementation, this information is identified using "triples" of the "subject-predicate-object" form; for example:

Tank1 <has temperature> Sensor 7 Tank 1 <is part of> Platform 4 Platform 4 <is part of> Plant1

These triples, taken together, constitute the Plant1 ontology and can be stored on a model server, as is described in more detail later in this article. This information, then, can be easily traversed using the model query language to answer questions such as "What is the temperature of tank 1 on Platform 4," more easily than the case without a semantic model.

We concentrate on the critical issue of metadata/ontology-based search and expert system technology. The main objective of the system is to improve the modeling of a semantic coherence to allow the interoperability of different modules of environments dedicated to the industrial area. We have proposed to use ontology together with CBR in acquiring expert knowledge in the specific domain. The primary information managed in the OntoEnter domain is metadata about industrial resources, such as guides, digital services, alarms, information, etc. We need a vocabulary of concepts, resources, and services for our information system described in the scenario that requires definitions about the relationships between objects of discourse and their attributes [31]. OntoEnter project contains a collection of codes, visualization tools, computing resources, and data sets distributed across the grids, for which we have developed a well-defined ontology using RDF language. Our ontology can be regarded as quaternion OntoEnter:={caller, resources, properties, relation}, where profiles represent the user kinds, collection contains all the services and resources of the institutional repository, the matter cover the different information sources: electronic services, web pages, BB.DD., guides, etc., and a set of relationships intended primarily for standardization across ontologies.

We integrate three essential sources for the system: electronic resources, the catalog of documents, and the personal database. The W3C defines standards that can be used to design an ontology [32]. We write the description of these classes and properties in RDF semantic markup language. We chose Protégé as our ontology editor, which supports the acquisition of knowledge and the development of knowledge bases. Protégé provides an environment for the creation and development of semantic knowledge structures-ontologies and semantically annotated Web services. Protégé organizes these elements as a dynamic workflow [33]. For the construction of the ontology of our system, we follow the sttif detailed below.

- Determine the domain and scope of the ontology. This should provide the location of different online resources. These are included from different sources: Catalog of Publications, Websites, Electronic Resources, etc.
- Enumerate important terms in the ontology. It is useful to write down a list of all terms we would like either to make statements about or to explain to a user.
- Define the classes and the class hierarchy. When designing the ontology, we need first group together related resources of the institutional repositories. There are three major groups of resources: users, services, and resources. IA detailed picture of our effort in designing this ontology is available in **Figure 5**. This shows the high level classification of classes to group together OntoEnter resources as well as things that are related with these resources.
- Generating the ontology instances with SW languages. To provide a conversational CBR system to retrieve the requested metadata satisfying a user query, we need to add enough initial instances and item instances to the knowledge base.

After designing the ontology, we write the description of these classes and properties in RDF semantic markup language. Then the domain expert, in this case, the administrative staff fills the blank units of the instance according to the knowledge of the domain. A total



Figure 5. Class hierarchy for the OntoEnter ontology.

of 13,000 cases were collected for user profiles and their different resources and services. Each case contains a set of attributes related to both metadata and knowledge. As a plus, domain specific rules defined by domain experts can infer more complex high-level semantic descriptions, for example, by combining low-level features in local repositories. Considering that our final objective is to reformulate queries in the ontology to queries in another with less loss of semantics, we arrive at a process to approach complex relations between two ontologies. As mentioned in previous sections, the relationships between ontologies can be compounded as a form of declarative rules, which can be handled in inference engines.

In addition, our system combines ontologies with rules into the ontology-based search due to the description limitations in current ontology languages. As mentioned in previous sections, relations among ontologies can be composed as a form of declarative rules, which can be further handled by inference engines.

4.3. Intelligent user Interface

OntoEnter is software, which is an intermediate link between users and search engine. By using OntoEnter user can tune the query in accordance with his needs. Advanced conversational user interface interacts with users to solve a query, defined as the set of questions selected and answered by the user during a conversation. The real way to get an individualized interaction between a user and a website is to present the user with a variety of options and allow the user to choose what is of interest at that specific time. In our system, the user interacts with the system to fill the gaps to retrieve the correct cases.

The algorithms used allow to give intelligent advice on improving the search query to obtain more relevant results to a narrow number of documents obtained or, conversely, to extend it. But the main task of OntoEnter is to specify in which exact word, the word is used and to formulate the "question" to the search engine, excluding answers from an inappropriate domain and adding semantically similar results [34].

For the distributed retrieval of learning resources, we use profile users, which are used for personalized searches according to user specifications. The methodology was based on incremental user profiling, which assumes mapping of a user's keywords to the concepts of the domain ontology according to the presented transformation rules. Transformation algorithm was implemented in the research prototype as the combined capability of the query transformation agent and the ontology agent of the intelligent multi-agent information retrieval mediator. The user interface helps user to build a particular profile that contains his interest search areas in the industry repositories domain. In an intelligence profile setting, people are surrounded by intelligent interfaces merged, thus creating a computing-capable environment with intelligent communication and processing available to the user by means of a simple, natural, and effortless human-system interaction. The objective of profile intelligence has focused on creating of user profiles: Plan Managers, Assistants, Operators, and Engineers. If the information space is well designed, then this option is easy, and the user achieves optimal information through the use of natural intelligence, that is, the options are easy to understand so users know what they will see if they click in a link, and what they negate by not following other links, Figure 6.

Or	ntoEnter - DIRs Ontology	Search ITo obtain enhanced result search yo must provide initial information!
	Plant manager Acsistant O Coperators O E	ngineers O Dominium:
Learning	and Research	Country
🔄 Guides a	nd tutorials	Wireles connection
Departm	ents and Services	Recommendations for Acquisition
Exerning	Resources Center and fulloring	RefWorks. Web-based station manager
Undergra	aduate Research	Access from Home
Our Colle	Clicers	inda.
] Introduct	ory and Specialized courses	Cpening hours
Alumni P	ublications	FAQ: frequently Asked Questions
2) Librery C	Cetalogue FAMA	Higrant Student Services
2) Electroni	c Resources	Contact Us
Others		
Festura	e Center for Students with Disabilities	 European Space of University Studies
New Stur	dent Programa	Pre-college programs
Pattern	Search. (The maximum length of the field is 255 character	*)

Figure 6. User profiles, graphical user interface.

The profile agent is an environment, in which software agents can be executed to retrieve E-learning resources and which is wrapped by a Web service. This configuration contains the user requirements that typically described the relative needs, tasks, and goals of the user for an individual search. Profile agents help students with the search, according to the specifications they made. Search parameters in a profile, the initiation of a search, or access to the list of retrieved learning objects can be controlled by invoking appropriate search operations that extract learning resource metadata. Ideally, profile agents learn from their experiments, communicate and cooperate with other agents, around in DL. A profile agent uses a registry to locate learning searches. The agent compares the metadata and the search keywords for possible matches and presents the search results to the user. For this, a statistical analysis has been done to determine the importance values and establishing specified user requirements.

5. Experimental evaluation

In order to validate the approach, we have developed intelligent control platform in an electrical power system. This system integrates management knowledge into network resource specifications. We study an example of alarm detection and intelligent resolution of incidents related to a private network. We have used a telecommunications network belonging to a company in the electricity sector Sevillana-Endesa (SE), a Spanish electricity company. OntoEnter is used to optimize the operation of hundreds of connected sensors currently installed. Many of these sensors are wireless because they can be installed more quickly and at less cost than their wired equivalents, often with no required downtime. These low-cost wireless sensors and accompanying analytics can dramatically improve plant performance, increase safety, and pay for themselves within months. The Spanish electricity grid has a wireless network in the regional high voltage grid. Part of the long distance traffic in this network is controlled by a wireless intelligent system distributed through this private network. The use of knowledge integration in agents can help the system administrator to use the maximum capabilities of the intelligent network management platform without having to use another specification language to customize the application [35].

The intelligent development of the system must meet the following requirements: it must be robust, and the management activity should not interfere with the normal operations of the network and should only intervene when necessary. We will use the SCADA system due to management limitations of the network communication equipment. SCADA consists of the following subsystems **Figure 7**.

- Remote terminal units (RTUs) are connected to sensors in the process, the conversion of sensor signals to digital data and the sending of digital data to the monitoring system.
- Communication infrastructure that connects the monitoring system to RTUs.
- A monitoring system (computer), collecting (acquiring) data about the process and sending commands (control) to the process, which is our IA.

SCADA systems are configured around standard basic functions such as data acquisition, monitoring and event processing, archiving and data storage analysis, etc. The RTU encodes sensor inputs in protocol format and sends them to the SCADA master. The fundamental role of an RTU is the acquisition of various types of power process data, accumulation, packaging,



Figure 7. Elements of the prototype.

and data conversion in a form that can be communicated back to the master, interpretation and output of the commands received from the master, local filtering performance, calculation, and processes to allow specific functions to be performed locally. The supervision below and RTU includes all network devices and substation and feeder levels like circuit breakers, reclosers, autosectionalizers, the local automation distributed at these devices, and the communications infrastructure [36].

OntoEnter can monitor, in real time, the network's main parameters, making use of the information supplied by the SCADA, placed on the main company building, and the RTUs are installed at different stations. From the information provided, the operator can take action to solve any errors that may arise or send a technician to repair the station equipment. OntoEnter allows the operator to search for information, alarms, or digital and analog parameters of measurement, registered in each IA or RTU. The system has the ability to select the IA that is best suited to satisfy the client's requirements, without the client being aware of the details about the agent. In addition, the AI is able to communicate and negotiate with the other IAs. Collaborative IAs are useful, especially when a task involves several systems in the network.

6. Evaluation and proofs

When we perform a search on a search engine, we are looking to find the most relevant material, while minimizing the junk that is retrieved. This is the basic objective of any search engine. Get important information while avoiding junk is difficult, if not impossible to accomplish. The experiments carried out, in order to evaluate the effectiveness of the assignment of runtime ontology. The main objective has been to verify if the agent-assisted query formulation mechanism provides a suitable tool to increase the number of significant documents extracted from the DIRs to be stored in the CBR. For our experiments, we included about 50 users with different profiles. It set a context for users, they were asked to at least start their essay before issuing any query to the system. They were also asked to look through all results returned by OntoEnter before clicking on any result [37].

We compared the top 10 search results for each keyword phrase per search engine. Our application recorded the results they clicked on, which we used as a form of implicit user relevance in our analysis. We must consider that the relevance of recovered documents is subjective. That's different people can assign different relevance values to the same document. In our study, we have agreed different values to measure the quality of recovered documents, excellent, good, acceptable and poor, as can be seen in **Table 1**. After the data were collected, we had a record of queries with an average of 5 queries per user. From these queries, some of

	Excellent	Good	Acceptable	Poor
OntoEnter	7.5%	42.3%	35.1%	14.4%
Traditional SE	1.4%	25.7%	31.5%	21.3%

Table 1. Analysis of retrieved document relevance for select queries.

them had to be deleted, either because multiple results were clicked, no results were clicked or no information was available for that particular query.

In each experiment, we report the average rank of the user-clicked result for our baseline system, another search engine, and for our search engine OntoEnter. Thus basically, we can define two set-based measures: precision and recall.

$$precision = \frac{|\{relevan t_{document}\} \cap \{retrieve d_{documents}\}|}{|\{retrieve d_{documents}\}|}$$
$$recall = \frac{|\{relevan t_{documents}\} \cap \{retrieve d_{documents}\}|}{|\{relevan t_{documents}\}|}$$

It is possible to measure how well a search performed with respect to these two parameters. For each such set, precision and recall values can be plotted to give a precision-recall curve. We need these measures, if we are to evaluate the ranked retrieval results for search engines. These measures are computed using unordered sets of documents. The remaining queries were analyzed and evaluated **Table 2**.

It is easy to compare several classifiers in the precision graph. Curves near the perfect precision-recall curve have a better performance level than those closest to the baseline. In other words, a curve above the other curve has a better performance level (**Figure 8**).

Precision and retrieve are inversely related, i.e., as precision increases recall falls and vice-versa. When a relevant document is not retrieved at all, the precision value in the above equation is taken to be 0. A balance between these two needs to be achieved by the search engine that to achieve this and to compare performance, the precision-recall curves come in the practice.

Recall	OntoEnter Precision	Traditional SE Precision	
0.10	0.90	0.60	
0.20	0.85	0.56	
0.30	0.79	0.43	
0.40	0.68	0.35	
0.50	0.61	0.25	
0.60	0.52	0.05	
0.70	0.39	_	
0.80	0.20	_	
0.90	0.05	_	
1.00	-	_	

Table 2. Precision and recall values.



Figure 8. Performance OntoEnter & Traditional Search Engine (TSE).

This trade-off between precision and recall can be observed using the precision-recall curve and an appropriate balance between the two obtained. The precision-recall curves for two algorithms are shown. Depending on the requirement of high precision at the cost of the recall, or high recall with lower precision, an appropriate algorithm can be chosen. In our case, we choose the appropriate system depending on the high precision and data with false positives allowed. Two precision-recall curves represent the performance levels of the two search engines. The search engine OntoEnter clearly outperforms TSE in this domain example. Our system performs satisfactorily with about a 95.2% rate of success in real cases.

Another important aspect of the design and implementation of OntoEnter is the determination of the degree of speed in the answer that the system provides. During experimentation, heuristics and measures that are commonly adopted in information retrieval have been used. A statistical analysis was performed to determine the importance values in the results. While users were performing these searches, an application continued to run in the background on the server and captured the content of the written queries and search results. We can establish that OntoEnter speed in our domain improves the answer time and the average of the traditional search engine. **Figure 9** shows a graphic of these parameters that was collected as a part of the experiment.

We can establish that the speed in the system improves the procedure time and the average of the traditional search engine. Results for OntoEnter are 15.1% better than procedure time and 19.5% better than running time/sec searches on traditional search engines.

Intelligent Knowledge Retrieval from Industrial Repositories 141 http://dx.doi.org/10.5772/intechopen.70724



Figure 9. OntoEnter search analysis report.

7. Conclusion and future works

In this chapter, we investigated how the semantic technologies can be used to provide additional semantics from existing resources in industrial repositories. For this purpose, we presented a system based on ontology and AI architecture for knowledge management in industrial repositories. We describe an effort to design and develop a prototype to manage resources in a repository such as the OntoEnter project and exploit them to help users as they select resources. Our study addresses the main aspects of a Semantic Web Information Retrieval System architecture attempting to respond to the requirements of the next generation of Semantic Web users. This scheme is based on the following principle: knowledge elements are abstracted from a characterization by a metadata description that is used for further processing.

In this chapter, we offer different possibilities, which the semantic Web opens for the industry. An important goal is to study appropriate industrial cases, compile arguments, launch industrial projects, and develop prototypes for industrial companies that not only create with us but also benefit from the semantic Web.

As described here, semantic models play a key role in the evolving solution architectures that support the business goal of obtaining the complete view of "what is happening" within operations and then deriving business insights from that view. Semantic models based on industry standards take that one step further, especially as application vendors adopt those standards (which, as always, will happen more rapidly through pressure from the user community). This study addresses the main aspects of a semantic and intelligent information retrieval system architecture trying to answer the requirements of the next-generation semantic search engine. We have investigated how the semantic technologies can be used to provide additional semantics from existing resources in institutional repositories.

This scheme is based on the principle of knowledge elements that are abstracted from a metadata description characterization that is used for further processing. We have proposed to use ontology together with CBR in acquiring expert knowledge in the industry specific domain. We have developed the domain ontology, and we have studied how the content-based similarity between the concepts typed attributes could be assessed in CBR system. The study analyses the implementation results and evaluates the viability of our approaches in enabling search in intelligent-based digital repositories. It introduced a prototype Web-based CBR retrieval system, which operates on an RDF file store. With this, characteristic of the model ability of an individual will be increased to learn through collective searches experience. Furthermore, an IA was illustrated for assisting the user by suggesting improved ways to query the system on the ground of the resources in industry repositories according to his own preferences, which come to represent his interests. We have used all the profile agents effectively to generate relevant and recommended personalized profile for the different users.

OntoEnter can be part of a bigger framework of interacting global information networks including other DIRs, scientific repositories, commercial providers, and relies as much as possible on standards and existing building blocks as well as be based on Web standards. The combination of effective information retrieval techniques and IAs continues to show promising results in improving the performance of the information that is being extracted from the online repositories for users. Our findings suggest that IA is the central manager in the knowledge transfer process. Their mediation is essential to help adapt the knowledge produced by academics and makes it easier to adopt and use by the educational community. We conclude pointing out an important aspect of the obtained integration: improving representation by incorporating more metadata from within the information and intelligent techniques into the retrieval process, the effectiveness of the knowledge retrieval is enhanced. The model has good characteristics in providing preference to the users with a novel approach of finding nearby meaning of query and user can also recommend result pages by their opinion.

Future work will address the exploitation of information from other institutional repositories and digital services and refine the suggested queries, expand the system to provide other support, and refine and evaluate the system through user testing.

Future work will focus on the design of distributed and self-managed services based on the Web and services, which are:

- Able to examine and filter information based on semantic similarity and closeness
- Able to handle heterogeneous data/knowledge /intelligence sources.
- Able to discover, compose, and integrate heterogeneous components automatically.
- Able to create, deploy, and exploit linked data.
- Able to perform automated and user-driven application/service orchestration and choreography, etc.

Author details

Antonio Martin, Mauricio Burbano and Carlos León*

*Address all correspondence to: cleon@us.es

Technology Electronic Department, Seville University, Spain

References

- Jimeno-Yepes A, Berlanga-Llavori R, Rebholz-Schuhmann D. Ontology refinement for improved information retrieval. Information Processing & Management. 2010;46(4): Semantic Annotations in Information Retrieval
- [2] Yang Z, An Y, Sun Y, Zhang J. Research on intelligent glue-coating robot based on visual servo. Physics Procedia. 2012;24(Part C):2165-2171
- [3] Song M, Song I, Hu X, Allen R. Integration of association rules and ontologies for semantic query expansion. Data & Knowledge Engineering. October 2007;63(1):63-75
- [4] Bañares-Alcántara R, Jiménez L, Aldea A. Multi-agent systems for ontology-based information retrieval. In: Puigjaner L, Espuña A, editors. Computer Aided Chemical Engineering. Vol. 20. Elsevier; 2005. p. 1549-1554
- [5] Zhang J, Zhao W, Xie G, Chen H. Ontology- based knowledge management system and application. Procedia Engineering. 2011;**15**:1021-1029
- [6] Jiang X, Tan A. Learning and inferencing in user ontology for personalized semantic web search. Information Sciences. 2009;**179**(16):2794-2808 ISSN
- [7] García-Crespo A et al. Digital libraries and web 3.0. The CallimachusDL approach. Computers in Human Behavior. 2011;**27**(4):1424-1430
- [8] Badii a et al. Semi-automatic knowledge extraction, representation and context-sensitive intelligent retrieval of video content using collateral context modelling with scalable ontological networks. Signal Processing: Image Communication. 2009;24(9):759-773
- [9] Lim SCJ, Liu Y, Lee WB. Multi-facet product information search and retrieval using semantically annotated product family ontology. Information Processing & Management. 2010;46(4, Semantic Annotations in Information Retrieval):479-493
- [10] Fernandez M et al. OnlineSemantically Enhanced Information Retrieval: An Ontology-Based Approach. Web Semantics: Science, Services and Agents on the World Wide Web; 2016 In Press, Corrected Proof
- [11] Govedarova, D., Stoyanov, S., and Popchev, I. (2008) "An ontology based CBR architecture for knowledge management in BULCHINO catalogue," in Proc. International Conference on Computer Systems and Technologies (CompSysTech).

- [12] Stuckenschmidt, H., and Harmelen, F.V. (2011). "Ontology-based metadata generation from semi-structured information," K-CAP, pp. 163-170, ACM.
- [13] Ding H. Towards the metadata integration issues in peer-to-peer based digital libraries. In: Jin GCCH, Pan Y, Xiao N, Sun J, editors. LNCS. Vol. 3251. Berlin, Germany: Springer; 2004
- [14] Guha R, McCool R, Miller E. Semantic search. In: Proceedings of WWW2003; 2003
- [15] Martín Antonio, Mauricio Burbano, Iñigo Monedero, Joaquín Luque. (2016). "Semantic reasoning method to troubleshoot in the industrial domain", the fifth international conference on Intelligent systems and applications, pp.89.
- [16] OntoWeb project. OntoWeb Technical Roadmap v2.0. Deliv- erable 1.1.2 of IST Project IST-2000-29243 OntoWeb, November 2002
- [17] Ceccaroni L, Cortés U. OntoWEDSS: Augmenting environmental decision-support systems with ontologies. Environmental Modelling and Software, pp. 2003;3:
- [18] Martín A. Intelligent search engine to a semantic knowledge retrieval in the digital repositories. International Journal on Advances in Intelligent Systems. 2015;8:67
- [19] Segaran T. Programming Collective Intelligence: Building Smart Web 2.0 Applications. O'Reilly Media; 2007a
- [20] SEC. Commission Staff Working Paper: linking up Europe, the importance of interoperability for egovernment services [Online 2016]. Available from: http://europa.eu.int/ ISPO/ida/export/files/en/1523.pdf, 2016.05.3
- [21] EIF. European Interoperability Framework Version 2. [Online 2016]. Available from: http://ec.europa.eu/isa/strategy/doc/annex_ii_eif_en.pdf, 19 April 2015.
- [22] MAP. Aplicaciones utilizadas para el ejercicio de potestades. Criterios de Seguridad, Normalización y Conservación. Ministerio de Administraciones Públicas. [Online 2016]. Available from: http://www.csi.map.es/csi/criterios/index.html. 5 March 2016;
- [23] CompTIA. European Interoperability ICT Industry Recommendations; 2004. p. 13
- [24] Toussaint J, Cheng K. Web-based CBR (case-based reasoning) as a tool with the application to tooling selection. International Journal of Advanced Manufacturing Technology. 2006;29(1):24-34
- [25] Sure Y, Studer R. Semantic web technologies for digital libraries. Library Management Journal, Emerald. 2005;26:190-195
- [26] Bechhofer S, Harmelen FV, Hendler J, Horrocks I. OWL web ontology language reference. W3C recommendation. 2004;10(February): Publisher W3C
- [27] Martin A, León C, Expert Knowledge Management Based on Ontology in a Digital Library. Departamento de Tecnología Electrónica: Universidad de Sevilla; 2010. p. 4
- [28] GAIA-Group for Artificial Intelligence Applications. jCOLIBRI project-Distribution of the development environment [Online 2016]. Available from: http://gaia.fdi.ucm.es/ research/colibri/jcolibri/25 April 2016

- [29] Segaran T, "Programming Collective Intelligence: Building Smart Web 2.0 Applications, Published by O'Reilly Media, August 23rd 2007.
- [30] Hanis T, Noller D. The role of semantic models is smart industrial operations", developer works. IBM. 2011;2
- [31] Taniar D, Rahayu JW. Web Semantics and Ontology. Hershey, PA: Idea Group; 2006
- [32] W3C. RDF Vocabulary Description Language 1.0: RDF Schema. [Online 2016]. Available from: http://www.w3.org/TR/rdf-schema/, 10 August 2016.
- [33] PROTÉGÉ. The Protégé Ontology Editor and Knowledge Acquisition System. [Online 2016]. Available from: http://protege.stanford.edu/, 5 July 2016.
- [34] Terziyan V. Semantic Search Facilitator: Concept and Current State of Development. Industrial ontologies group; 2004
- [35] Martín A, León C. Intelligent management experience on efficient electric power system. ICREPQ. 2014;12:842
- [36] Warren P. Applying Semantic Technologies to a Digital Library: A Case Study. Library Management Journal: Emerald; 2005
- [37] Amerland D. Google Semantic Search: Search Engine Optimization (SEO) Techniques that Get your Company More Traffic, Increase Brand Impact and Amplify your Online Presence. Que Publishing Kindle Edition; 2013

Affective Technology Acceptance Model: Extending Technology Acceptance Model with Positive and Negative Affect

Angela Lee Siew Hoong, Lip Sam Thi and Mei-Hua Lin

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/intechopen.70351

Abstract

Research works on TAM, TAM2, TAM3 and UTAUT has always focused on cognitive aspect of technology acceptance in the past two decades. Acceptance of technologies such as eCommerce, Mobile and ERP that considered emotion and affect are still less. This creates a gap in the technology acceptance research, which consider the role of affect into technology acceptance model. This study considers the role of affect of a knowledge worker that work in Multimedia Super Corridor (MSC)-status organizations in Malaysia on their behavioural intention to use knowledge sharing tools (KS tools) in their day-to-day tasks. Hence, Affective Technology Acceptance (A.T.A) model has been proposed. The behavioural intention on the acceptance of KS tools will be hypothesize in the Affective Technology Acceptance (A.T.A) model. Positive (PA) and Negative (NA) affect as the role of affect construct were introduce in this model to investigate its influence on KS tools usefulness and ease of use among employees in Multimedia Super Corridor organizations. The findings of this study highlighted that NA has no impact on perceive usefulness. The findings also showed that PA has very significant positive influence on PU, PEOU and BI with impact on PEOU being the greatest.

Keywords: positive affect, negative affect, TAM, knowledge sharing tools, knowledge workers, affective technology acceptance model

1. Introduction

In the past few decades, works on technology acceptance research have always focused on cognitive instead of emotional factors to predict acceptance of technologies. The role of affect influence, state of mind, and feelings are not comprehended as comprehension



and wordings utilized as a part of this range has dependably been utilized conversely by specialists. Numerous conflicting reports and clashing discoveries from past investigations that consider influence have brought about modest number of research endeavors here. Nonetheless, inquiry about them has demonstrated that reflexes, social judgment, discernment, and conduct [1, 2] are impacted by influence, mind-set, and feeling that constitute the major parts of individuals.

In the information systems' (IS) area, client assessment or client acknowledgment of information technology (IT) is considered as a volitional conduct [3] and has been examined basically with an intellectual introduction [4–6]. Research in this area has dependably been vigorously affected by the insight state of mind conduct models, from Theory of Reasoned Action and the Theory of Planned Behavior [7]. Even though some works on affect, affectivity, playfulness, enjoyment, and emotion have been studied, the affective aspects are less central in most of these studies, with some exceptions, such as studies on aesthetics [8], computer playfulness [9], flow [10], and users' experiences in technology acceptance [11]. Therefore, if the roles of affect indeed play a role in technology acceptance, what aspect of study should be examined and in what relationships of role of affect toward other constructs in the technology acceptance model.

2. Research gaps

Due to conflicting findings and inconsistent terminologies used in the research that considers affect, moods, emotions, and feelings, and the role of affect has been very much ignored by researchers in general. However, recent research has found that the inclusion of affective constructs is able to explain attitude and behavior more extensively in their models. Nevertheless, research that examines role of affect from the perception of the knowledge workers on the KS tools' characteristics in terms of features and functions to induce positive or negative affective (PA and NA) states is lacking. This study extends technology acceptance model (TAM) with PA and NA on perceived ease of use (PEOU), perceived usefulness (PU), and BI to predict the behavioral intention to use KS tools by knowledge workers in MSC-status organizations.

3. Literature review

3.1. Related works on technology acceptance

Davis [4] develops technology acceptance model (**Figure 1**) to determine factors that influence the acceptance of technology. Two most important individual beliefs about using information technology are perceived usefulness (PU) and perceived ease of use (PEOU) that are able to explain individual's intention to use the technology. Davis [4] concluded that



Figure 1. Technology acceptance model [4].

perceived usefulness was the strongest predictor to one's intention to use an information technology.

In TAM, the goal is to utilize the primary determinant of use to accept or not to accept a new tool. The intention to utilize is controlled by the individual's personality toward utilizing a specific tool. Perceived usefulness (PU) and perceived ease of use (PEOU) impact a person's state of mind toward utilizing a specific tool. Perceived usefulness (PU) is characterized as how much individuals trust that utilizing a specific tool would improve his or her task execution [4]. Perceived usefulness is the key determinant that emphatically influences users' convictions and expectation to utilize the innovation. Perceived ease of use (PEOU) is characterized as how much the user utilizes a specific tool, and it is free of effort [4]. Past research has demonstrated that perceived ease of use (PEOU) impacts aim in two ways: direct and indirect impact through usefulness of the tool [4]. As indicated by Davis [4], PEOU has no critical impact on behavioral expectation to utilize in light of the fact that PU intervened its impact. PEOU does not affect straightforwardly on user's behavioral goal since it affects behavioral expectation through PU.

Venkatesh and Davis [12] extended TAM by calling it TAM2 with social influence and cognitive processes on the Perceived Usefulness and intention usage (**Figure 2**). In TAM2, subjective norm [7] is hypothesized to have a direct effect on the intention of an individual to choose to perform a certain behavior even if he/she is not favorable toward that behavior, but due to other referents think he/she should; hence, the individual complies with these referents. In mandatory system usage settings, subjective norms were found to have direct effect on intention over PU and PEOU. The model posits voluntariness as a moderating variable to distinguish between mandatory versus voluntary. Nevertheless, subjective norms can influence intention indirectly through perceived usefulness that is called internalization. Therefore, according to TAM2, the direct compliance-based effect of subjective norm on intention over PU and PEOU will occur in mandatory but not voluntary system usage settings [12]. Job relevancy, output quality, and result demonstrability are determinants for cognitive instruments on PU.



Figure 2. Extended technology acceptance model (TAM 2) [12].

TAM2 proposes that individuals rely on the fit between their job and the performance outcomes of using the system. This will determine their perceived usefulness of the system based on the job relevancy. It was defined as an individual's perception regarding the degree to which the target system is applicable to his or her job. Output quality is the quality of the end result produced by the system to the individual. An individual will take into account on how well the system performs those tasks. If the system does not produce any desirable output to enhance individual performance, it is deemed to believe that the user acceptance rate will drop. Therefore, TAM2 theorizes that result demonstrability defined by Moore and Benbasat [13] as "tangibility of the results of using the innovation" will directly influence perceived usefulness. TAM3 [12] is an extension of TAM where anchors and adjustments are hypothesized to influence PEOU in the model. Anchors are the degree to have general beliefs about computers and its usage, whereas adjustments are the degree of belief that is shaped based on direct experience with the target technology. The results indicate that there are strong correlations for these variables to PEOU. The antecedents for perceived ease of use include computer self-efficacy, perceptions of external control, computer anxiety, computer playfulness, perceived enjoyment, and objective usability. Unified theory of acceptance and use of technology (UTAUT) was introduced by Venkatesh in 2003. UTAUT was developed through the consolidation of various construct of eight models applied to IS usage behavior. These eight models are TAM, TRA, TPB, motivational model, integration of TAM and TPB, PC utilization model, innovation diffusion theory, and social cognitive theory. Behavioral intention and usage behavior were the two dependent variables. On the other hand, eight independent variables include performance expectancy, effort expectancy, social influence, facilitating condition, gender, age, experience, and voluntariness of use. Three main constructs are the determinants of the intention to use and behavior usage (Figure 3): performance expectancy, effort expectancy, and social influence. Performance expectancy was the strongest predictor among the eight factors. UTAUT theorizes that social influence holds significance only in mandatory technology use of situations.

Affective Technology Acceptance Model: Extending Technology Acceptance Model with Positive... 151 http://dx.doi.org/10.5772/intechopen.70351



Figure 3. The unified theory of Acceptance and Use of Technology [6].

3.2. Affect, mood, emotion, sentiment, and feeling

Every single sociology shares an interest to attempt, to clarify, and to foresee individual's behaviors, where these behaviors are impacted by subjective procedures. Most theories derived from behavioral often ignore role of affect factors. Role of affect refers to one's feeling or how an individual feels when performing tasks [14, 15]. Affect also refers to one's emotions, moods, and feelings, and they are used interchangeably [2, 16–18].

Dispositional affect is defined as a person's affective predisposition toward perceiving the world around him or herself either positively or negatively [17, 19]. It has strong influences on individual behavior [20, 21]. Many related information systems research uses different terms to represent the role of affect such as "anxiety" when using computers, "computer playfulness," "affect" toward computers, the influence of emotions toward users' attitudes, and use of specific IT [9, 22–24]. Mood is an intra-individual change, generally nonintentional which is not associated with explicit intentions to act [25, 26]. Lazarus [17] defined mood as an affective state that comes and goes depending on particular conditions. Mood is low intensity, diffuse feeling states that usually do not have a clear antecedent [27]. Mood can be characterized as relatively unstable short term intra-individual changes [28]. Mood can be evoked by both dispositional affect and emotions. Unlike emotions, people may not realize that they are experiencing a "mood" and may also not realize that mood is influencing their behavior [27]. Emotions differ from both dispositional affect and mood. Emotions have a clear cause or object, usually are shorter in duration and more focused and intense [29]. Emotions are more likely to change beliefs than mood [30, 31]. Emotions are more likely to disrupt activity [17]. It is also said as an intense feeling; a complex and usually strong subjective response that typically accompanied by physiological and behavioral changes in body [32]. Emotions can occur during the impact period (i.e., when the new information technology (IT) had been deployed and was being used). In this period, emotions are generated based on individuals' perceptions on the features of the new technology and on their usage of the new technology resources. Individuals will assess whether the technology constitutes a threat or an opportunity and how it can adapt into their daily tasks by changing their working behaviors [33]. Some specific emotion terms such as pleasure, arousal, and enjoyment are used to relate users' attitude toward actual use of a technology [6]. Feelings are sensations perceived by the sense of touch; an affective state of consciousness that resulted from emotions, sentiments, or desires. On the other hand, cognition arises on the human beings' perception toward using technologies [16, 34]. Behavioral aspect would be from the individual's reactions toward using the information technologies [11]. Emotional Intelligence is a variable with a multifactor individual difference [35] that meets the traditional standards of intelligence. Being emotionally intelligent involves being actively able to identify, understand, process, and influence one's own emotions and those of other to guide feeling, thinking, and action. Sentiments are valence appraisals of an object that involves evaluation of whether something is liked or disliked. These evaluations were evoked by phenomena. It can come from previous experience with the object or situation or through social learning [29]. Satisfaction has been the most widely studied sentiment. Most of the work conducted has focused on satisfaction at the individual level either because of workplace events or as a predictor of workplace outcomes [19].

Zhang and Li [36] examined the affects of emotional assessments of IT on IT utilized choices. Refer to Zhang and Li [36], two protest based full of feeling assessment builds: recognition on IT's ability to incite positive affect and impression of the IT's capacity to prompt negative affect able to influence. Their investigation demonstrated that positive affect and negative affect are particular ideas that affect perceived usefulness (PU), perceived ease of use (PEOU), and attitude toward utilizing IT tools. These impacts remain constant amid individuals in using and utilizing IT tools (ATT). Positive affect impacts PU, PEOU, and ATT, yet it turns out to be less critical to PU after some time, and positive affect just impacts PEOU; however, it turns out to be more vital to PEOU over the long run. Therefore, Zhang and Li [37] presumed that affect influence a key part in individuals' connections in using IT tools.

Loiacono and Djamasbi [15] also found that positive mood played a significant role in the adoption of a new technology. Their study looked at the effects of positive mood, and to understand how individual's characteristics affect an individual's cognition and behavior on the acceptance of a Decision Support System. The objective of their research is to investigate how affect can be a vital component for technology acceptance to make rational decision making. Based on Isen et al. [38], qualities of task characteristics impact one's certain state of mind particularly on tolerating another innovation, for example, Decision Support Systems (DSS) which requires subjective capacities to deal with troublesome/complex task. Association can control one's state of mind by encouraging positive temperament inside the association, and it can enhance association's results [16]. From their findings, Loiacono and Djamasbi [15] reported that positive mood could bring improvements in new technology acceptance.

4. The proposed affective technology acceptance (ATA) model

Based on the literature findings, affective technology acceptance model involves PA and NA that were used to induce positive and negative affect states on the individuals who uses the technology were proposed. Zhang and Li [36] adapted these constructs and defined them as the perception of IT's capability to induce these feelings. It was said that the technology functions and features are capable of inducing these feelings in the individuals. Therefore, this research proposed an extension on the technology acceptance model by including this two affect states on the use of KS tools by the knowledge workers in the MSC-status organizations in Malaysia. Indication of the respondent's feelings was recorded at eight different points in times on the instrument to gather the different affective states of the knowledge workers on using the knowledge sharing tools. The measurement scale was adapted from Perlusz [39]. Two groups of undergraduate students were used to validate the scale, and it was found that the technology affect scale were consistent and valid in Perlusz studies.

In this research, PA and NA are defined as the perception on KS tools' characteristics in terms of features and functions to induce positive or negative affective states [36, 37, 40]. PA and NA were adapted from Zhang and Li [36], where they defined PA and NA as the perception of an IT's capability to induce positive or negative affect. It is an individual's perception or evaluation that an IT has the features and functions to induce positive or negative affect in him or her. In this study, the external stimulus is KS tools used by the knowledge workers in the MSC-status organizations in Malaysia. The respondents were asked to indicate the extent of how he/she feels on the usefulness, ease of use, and intention to use the KS tools in eight different points in times in the instrument. The different affective states of the knowledge workers were self-reported on the survey form. The measurement scale for PA and NA is adopted from Technology Affect Scale [39] where Perlusz [39] adapted the 10-item scale from Watson and Tellegen [41]. The scale was validated using two groups of undergraduate students who were exposed to several types of affects before interacting with mobile technologies. The Technology Affect Scale is found to be consistent and valid in his experiments.

4.1. PU, PEOU, ATT, and BI in technology acceptance model

The relationships among PU, PEOU, ATT, and BI are consistent with the literature. TAM originally included attitude as a mediator between the personal beliefs constructs, and behavioral intention [4]. Individual's actual usage of the technology is dictated by behavioral goal, which is determined by perceived usefulness and perceived ease of use. The value of perceived usefulness is the degree to which an individual trusts that utilizing the innovation will upgrade his or her employment performance, and perceived ease of use is the degree to which individual trusts that utilizing the technology will be free of effort [4].

H7: There is a significant relationship between PEOU and PU.

H8: There is a significant relationship between PU and ATT.

H9: There is a significant relationship between PEOU and ATT.

H10: There is a significant relationship between ATT and BI.

4.2. PA and NA on the perceived usefulness, perceived ease of use, and behavioral intention to accept KS tools

This research considers PA and NA based on evidences obtained by Zhang and Li [36]. They found that PA strongly influences PEOU, PU, and ATT, while NA only influences PEOU at the initial stage of usage. In their work, BI is mediated by ATT, but the direct influence of PA and NA on BI was not being investigated (Figure 4). Isen [42] presented his findings by stating that positive affect state such as joy and elation will lead a person to be creative, playful, and explore innovative ideas and think broadly. Another piece of work conducted by Isen et al. [38] using four experiments on positive and negative affects induced by a series of activities such as watching comedy films for few minutes, receiving a small bag of candy, or showing film of unpleasant feelings. They found that positive affect induced by a comedy film or a small gift of candy facilitates creativity on tasks given. At the same time, activities that designed to induce negative affect using primitive arousal devoid of any affective tone (exercise) had no effect on these measures. In their findings, negative affect neither facilitates nor impairs creativity. However, they pointed out that one of their experiments showed that negative affect was only induced by showing subjects film that induces unpleasant feelings. The proposed work in this research hypothesized the extent of how a person feels in his perception on the KS tools' features and functions (or characteristics) in their day-to-day tasks that induce positive or negative affect. This research fills the gap by examining the relationship of PA and NA and the behavioral intention to accept KS tools in the organizations. The affect induced by the perception toward how knowledge workers evaluate KS tools' affective quality is believed to be able to influence an individual's behavior and intention to accept a



Figure 4. ATA model.

tool. Affect construct included in the proposed Affective Technology Acceptance Model, the hypothesis is as follow:

H1: There is a significant relationship between PA and BI.

H2: There is a significant relationship between PA and PU.

H3: There is a significant relationship between PA and PEOU.

H4: There is a significant relationship between NA and PU.

H5: There is a significant relationship between NA and PEOU.

H6: There is a significant relationship between NA and BI.

5. Methods

5.1. Study contexts and samples

The population where sample will be drawn in this research consists of knowledge workers who work in MSC-status organizations in Malaysia. The samples are individuals who deal with information, require developing, or using knowledge to solve problems in their jobs. About 2500 MSC-status organizations from the MSC directory (http://www.mscmalaysia.my/company_ directory) have been invited to participate in this research. Two thousands and five invitations were sent out to these organizations and 300 forms were received. Two hundred ninety-five forms were usable. A response rate of 11.87% was gathered from the self-administered questionnaire. **Figure 5** illustrates KS tools that are highly utilized by the respondents, and **Figure 6** presents those that are not used at all in the activities that are carried out by the respondents.



Figure 5. KS tools with high usage frequency.



Comparative Analysis on Frequency of different KS tools usage - Not at All

Figure 6. KS tools not used at all among knowledge workers in MSC organizations.

Most of these organizations are in their initial stages of tool implementation or tools have been implemented but with very minimum tools usage among knowledge workers in these organizations. This rationalized the importance to carry out a research on their intention to use KS tools in their day-to-day jobs.

5.2. Instrument and measures

All the constructs in the research model were measured with items adapted from prior research. All the items in the questionnaire used a five-point Likert-type scale ranging from "Strongly Agree" to "Strongly Disagree" (**Table 1**).

Sections	No. of items
General information	9
KS tools behavioral intention	30
Role of affect	24

Table 1. Instrument.

Respondents were asked to indicate to what extend one feels when one evaluates the KS tools when interacting and using the tools. The respondents were asked to record their feelings that were induced by the tools when they interacted with them. The positive and negative affect (PA and NA) are hypothesized to have an impact on PEOU, PU, and BI in the research model. Items for positive and negative affect (PANAS) were adapted from Perlusz [39], Tellegen [28], Watson [43], and Watson et al. [44]. Pre-test and pilot test were carried out before actual survey took place. A total of 30 respondents participated in these tests.

5.3. Data analysis

The proposed model and hypothesis testing was carried out using SmartPLS 3.0 software. The measurement and structural model analysis follows methodology described in Hair

et al. [45]. In this research, a sample size of 295 is sufficient using G*Power software to examine the predictive accuracy of constructs and path coefficients of relations in the proposed model. The analysis on the years of experience of the knowledge workers in these MSC-status organizations highlighted that most of them have been in their industry more than 10 years. The preliminary analysis highlights that long service in the industry gives knowledge workers more domain knowledge where majority of them possess an undergraduate degree.

6. Results

6.1. Measurement model analysis

This section discusses the measurement model, which consists of several analyses. **Table 2** illustrated the composite reliability on the results. Based on the analysis, it was shown that PU, PEOU, ATT, and BI all achieved the value of composite reliability higher than 0.90, which satisfy the threshold of composite reliability.

Convergent validity looks at the extent to which a measure correlates positively with alternative measures of the same construct. In **Table 2**, it was found that all the constructs' AVE are significant with at least 0.50 and above.

Discriminant validity is the extent to which a construct is truly distinct from other constructs by empirical standards. This means a construct has captured the phenomena not represented by other constructs in a model. Cross loading and Fornell-Larcker criterion discriminant validities are used in this analysis. The analysis indicates that AVE of all constructs has high correlation. Formative measurement analysis conducts a separate set of validity and consistency test. The formative constructs in the proposed model consist of positive and negative affect for PU (NA_PU and PA_PU), positive and negative affect for PEOU (NA_PEOU and PA_PEOU), and positive and negative affect for BI (NA_BI and PA_BI). Convergence validity is the extent to which a measure correlates positively with other measures or indicators of the same construct. A redundancy analysis is used to perform the convergence validity test by

Constructs	Composite reliability (CR)	Cronbach's alpha	AVE (Convergent validity)	Discriminant validity	
				Cross loading	Fornell-Larcker criterion
Perceived usefulness (PU_I)	0.945	0.932	0.710	Y	Y
Perceived ease of use (PEOU_H)	0.928	0.913	0.591	Y	Y
Attribute toward use of KS tools (ATT_G)	0.925	0.906	0.639	Y	Y
Behavioral intention to use KS tools (BI_J)	0.937	0.922	0.682	Y	Y

Table 2. Composite reliability for reflective constructs.

evaluating formative measurement models; one must test whether the formatively measured construct is highly correlated with a reflective measure of the same construct. A global indicator is designed for this test. To conduct the convergent validity, a separate model is created with the global indicator for each formative construct. From the outcomes of the redundancy analysis, negative affect on perceive usefulness, negative affect on perceive ease of use, negative affect on behavioral intention, positive affect on perceive ease of use and positive affect on behavioral intention are 0.8 or above. Their formative indicators are significant enough to capture content that these constructs want to capture.

For collinearity assessment, when collinearity has high correlations between two formative indicators in a formative construct, it is a problematic indicator and it is unwanted for a formative construct. VIF is used to assess collinearity. Once the collinearity of formative indicators has been treated, outer weights in formative measurement models can then be interpreted. All formative indicators satisfy the requirement of VIF values uniformly with values below the threshold value of 5. There are five items used to test PA \rightarrow PU, PA \rightarrow PEOU, and PA \rightarrow BI. Similarly, five items were designed for NA \rightarrow PU, NA \rightarrow PEOU, and NA \rightarrow BI. The items that are labeled as AA1A..E, BB1A..E, and CC1A..E are positive affect items whereas AA1F..I, BB1F..I and CC1F..I are negative affect items. There is no collinearity problem found in the model except for the items AA1G and AA1H from negative affect on perceived usefulness, BB1G and BB1H from negative affect on perceived ease of use and CC1F and CC1J from negative affect on behavioral intention. Based on items AA1G, AA1H, BB1G, BB1H, BB1J, CC1F, and CC1J being important questions to measure negative affect on perceived usefulness, perceived ease of use, and behavioral intention in the instrument, therefore, these items will be retained. A formative indicator of its relevance is analyzed based on the values of its outer weight as it is compared with others to determine its relative contribution to the construct.

To determine whether an indicator is significant or not, each indicator's *t*-value must fulfill the critical value of 1.65 for two-tailed tests at a significant level = 10%. The indicator significance level analysis for positive affect and negative affect on perceived usefulness is significant. Positive and negative affect on perceived ease of use has ten formative items that measure the construct. Five items were chosen to measure positive and negative affect, respectively. One negative affect item is not significant. Its *t*-value is less than 1.65, and outer weight and outer loading do not fulfill the criteria. However, BB1J was not considered to be deleted from this construct because this item has been validated and tested in the previous instrument. Positive and negative affect on behavioral intention has ten formative items that measure the construct. Five items were chosen to measure positive and negative affect, respectively. Negative affect has one item that is not significant. The outer loading value of 0.482 for CC1J is rounded up to be 0.5. Hence, all items are significant. In short, based on the theoretical model and measurement scale used for the proposed research model, the existing items for each construct will be kept.

6.2. Structural model analysis

Collinearity assessment on a structural model involves examination of each set of predictor constructs for each part of the structural model. Collinearity is assessed based on those constructs that have tolerance levels below 2.0 or VIF above 0.50. If such collinearity exists, one should consider eliminating the constructs, merging predictors into a single construct, or creating higher-order constructs to treat collinearity problem. **Table 3** shows that there is no collinearity problem encountered in the research model.

Structural model is used to calculate the estimates of the structural model relationships (path coefficient) that are represented as the hypothesized relationships among the constructs. For this research, we choose to take a significant level of 10% with a critical value of 1.65. Besides examining *t*-values, *p*-values are considered in this analysis. To obtain the *t*-values, a bootstrapping procedure with 5000 resamples was applied. Based on the analysis results, the hypothesis testing results are summarized as follows in **Table 4**.

Another important measure is the total effect of each path. Direct effect for each path may not be very significant in some cases; hence, Total Effect is to assess the significant of paths in the model. The coefficient of determination is a measure of the model's predictive accuracy using R^2 where it represents the exogenous latent variables' combined effects on the endogenous latent variables. R^2 also represents the amount of variance in the endogenous constructs explained by all the exogenous constructs linked to it. In scholarly research that focuses on marketing issues R^2 values of 0.75, 0.50, and 0.25 for endogenous latent variables can be described as substantial, moderate, and weak, respectively.

Attitude toward using KS tools can predict with an accuracy that is close to value 1. Followed by behavioral intention to use KS tools with a R^2 value of 0.625 and Task Category-KS tools Fit of a R^2 value of 0.593. As for perceived usefulness, it has a R^2 value of 0.45 and perceived ease of use has the smallest R^2 value of 0.360. By examining *t*-values based on the critical values 1.65 for two-tailed tests at a significant level = 10%, all the *t*-values in the table are significant. Hence, all the predictive accuracy values are significant (**Table 5**). Hence, ATT_G and BI_I are substantial and PU_I and PEOU_H are moderate endogenous latent variables in the proposed model.

Constructs	ATT_G	BI_J	PU_I	PEOU_H
ATT_G		2.326		
NA_BI		1.075		
PA_PU			1.307	
NA_PU			1.062	
PEOU_H	1.421		1.492	
PU_I	1.421			
PA_BI		1.371		
PA_PEOU				1.211
NA_PEOU				1.051

Table 3. Summary of VIF for collinearity analysis.

No	Hypothesis	Results
1.	There is a significant relationship between PA and BI.	Supported
2.	There is a significant relationship between PA and PU.	Supported
3.	There is a significant relationship between PA and PEOU.	Supported
4.	There is a significant relationship between NA and PU.	Not supported
5.	There is a significant relationship between NA and PEOU.	Supported
6.	There is a significant relationship between NA and BI.	Supported
7.	There is a significant relationship between PEOU and PU.	Supported
8.	There is a significant relationship between PU and ATT.	Supported
9.	There is a significant relationship between PEOU and ATT.	Supported
10.	There is a significant relationship between ATT and BI.	Supported

Table 4. Summary of the hypothesis testing results.

Endogenous latent variable	R^2	t-values	
ATT_G	0.628542	14.3111	
BI_J	0.625851	16.30823	
PU_I	0.450406	6.329501	
PEOU_H	0.360258	6.716182	

Table 5. R^2 .

7. Discussion and conclusion

Role of affect has been measured in this study to investigate on its relationship between positive and negative affect with knowledge workers' behavioral intention to use knowledge-sharing tools. The results show that these two constructs have significant relationship on behavioral intention to use and this was also supported by Zhang and Li's findings in 2007. Zhang and Li [36] pointed out that negative affect is easier to measure and investigate compared to positive affect. It was believed that negative affect creates impacts on knowledge worker's performance and their intention to use a technology. However, this study outcome produces a new finding where the results show that negative affect has no impact on individuals' perceived usefulness on the knowledge-sharing tools, whereas the results for attitude toward knowledge-sharing tools usage aligned with past literature works [4, 6]. Negative affect has a strong impact on behavioral intention usage of KS tools in the ATA model. From the findings, we know that knowledge workers use knowledge-sharing tools to perform their daily task in work, and the results gather their affect states after interacting with the tools. From the study, we conclude that if individuals have a strong negative influence on their behavioral intention, they will show less interest in the use of knowledge-sharing tools. Therefore, negative affect has the strongest influence on perceived ease of use and perceived usefulness of the tools. On the other hand, positive affect shows significant impact on perceived ease of use, perceived usefulness and behavioral intention. This implies that positive affect plays a great role in behavioral intention to use a technology; therefore, constant improvement on the tools is needed to induce positive affect on the individuals in using the tools.

Positive and negative affects are a major determining factor in technology acceptance. Past review works in information systems show that the outcomes were inconclusive and inconstant. Operationalization of affect and its related terms has been ambiguous. Many terms have been used interchangeably in a number of research; however, this is a growing field and interest among many researchers [12, 28, 37, 39, 40, 41, 46–56]. From the past related works, it has shown empirical evidence on the role of affect and it was used to explain better in a behavioral study. The proposed ATA model has hypothesized positive affect and negative affect in influencing PEOU, PU, and BI to use knowledge-sharing tools. Past research works has shown that affect has no significant influence on the usefulness of the tools. However, in this study, we found that positive affect has significant influence on the usefulness and ease of use on knowledge-sharing tools and subsequently influence individuals' behavioral intention to use. These findings are a new contribution to the theoretical aspect of affect relating technology acceptance area of study.

Practitioners and business operators can then maximize these findings by knowing what to do to their tools in order to encourage more usage of the tools. Software designers can consider affect element as they design new tools for knowledge workers. Unpleasant and uncomfortable feelings at the first glance on the tools may induce negative affect on the individuals, hence, impact of affect must be considered while designing user interface of a technology. Technology that induces positive affect will increase the tools acceptance. This study brings new implications to the top management on the factors that impact the knowledge workers' intention to use KS tools in their works. This study suggests that organizations should focus on knowledge workers' affective aspects besides other factors before implementing any KS tools in their organizations. The affective aspects of knowledge workers induced by the tools are found to be significant in this research. Therefore, top management should pay attention while formulating their knowledge-sharing tools implementation strategies in their organizations.

Author details

Angela Lee Siew Hoong^{1*}, Lip Sam Thi² and Mei-Hua Lin³

*Address all correspondence to: angelal@sunway.edu.my

1 School of Science and Technology, Department of Computing and Information Systems, Sunway University, Malaysia

2 School of Business Management, Universiti Utara Malaysia, Malaysia

3 School of Science and Technology, Department of Psychology, Sunway University, Malaysia

References

- Forgas JP, George JM. Affective influences on judgments and behavior in organizations: An information processing perspective. Organizational Behavior and Human Decision Processes. 2001;86(1):3-34
- [2] Russell JA. Core affect and the psychological construction of emotion. Psychological Review. Jan 2003;**110**(1):145-172
- [3] Bagozzi RP. A field investigation of causal relations among cognitions, affect, intentions and behavior. Journal of Marketing Research. 1982;10:562-584
- [4] Davis F. Perceived usefulness, perceived ease of use, and user acceptance of information technology. MIS Quarterly. September 1989;**13**(3):319-340
- [5] Goodhue DL. Understanding User Evaluations of Information Systems. Management Science. 1995;41(12):1827-1844
- [6] Venkatesh V, Morris MG, Davis GB, Davis FD. User acceptance of information technology: Toward a unified view, MIS Quarterly. 2003;27(3):425-478
- [7] Fishbein M, Ajzen I. Belief, Attitude, Intention and Behavior: An Introduction to Theory and Research. Reading, MA: Addison-Wesley; 1975
- [8] Tractinsky N, Katz AS, Ikar D. What is beautiful is usable. Interacting with Computers. 2000;13:127-145
- [9] Webster J, Martocchio JJ. Microcomputer playfulness: Development of a measure with workplace implications. MIS Quarterly. 1992;**16**(1):201-226
- [10] Ghani J. Flow in human computer interactions: Test of a model, In: Carey J, editor. Human Factors in Information Systems: Emerging Theoretical Bases. New Jersey: Ablex Publishing Corp. 1995
- [11] Agarwal R, Karahanna E. Time flies when you're having fun: Cognitive absorption and beliefs about information technology usage. MIS Quarterly. 2000;**24**(2):665-694
- [12] Venkatesh V, Davis FD. A theoretical extension of the technology acceptance model: Four longitudinal field studies. Management Science. 2000;46(2):186-204
- [13] Moore GC, Benbasat I. Development of an instrument to measure the perceptions of adopting an information technology innovation. Information Systems Research. 1991;**2**(3)192-222
- [14] George JM, Jones GR. The experience of work and turnover intentions: interactive effects of value attainment, job satisfaction, and positive mood. Journal of Applied Psychology. 1996;81(3):318
- [15] Loiacono E, Djamasbi S. Moods and their relevance to systems usage models within organizations: An extended framework. AIS Transaction on Human-Computer Interaction. 2010;2(2):55-72

- [16] Fredrickson BL. Positive emotions and upward spirals in organizations. In Cameron KS, Dutton JE, Quinn RE. Editors. Positive Organizational Scholarship: Foundations of a New Discipline. San Francisco, CA: Berrett-Koehler. pp. 163-175
- [17] Lazarus RS. Progress on a cognitive-motivational-relational theory of emotion. American Psychologist. 1991;46(8):819
- [18] Moore BS, Isen AM. Affect and Social Behavior. UK: Cambridge University Press; 1990
- [19] Staw BM. Organizational psychology and the pursuit of the happy/productive worker. California Management Review. 1986;28(4):40-53
- [20] Isen AM, Baron RA. Positive affect as a factor in organizational-behavior. Research in Organizational Behavior. 1991;13:1-53
- [21] Staw BM, Sutton RI, Pelled LH. Employee positive emotion and favorable outcomes at the workplace. Organization Science. 1994;5(1):51-71
- [22] Compeau DR, Higgins CA. Computer self-efficacy: Development of a measure and initial test. MIS Quarterly. 1995;19(2):189-211
- [23] Compeau D, Higgins CA, Huff S. Social cognitive theory and individual reactions to computing technology: A longitudinal study. MIS Quarterly. 1999;23(2):145-158
- [24] Todman J, Monaghan E. Qualitative differences in computer experience, computer anxiety, and students' use of computers: A path model. Computers in Human Behavior. 1995;10(4):529-539
- [25] Bagozzi RP, Gopinath M, Nyer PU. The role of emotions in marketing. Journal of the Academy of Marketing Science. 1999;27(2):184-206
- [26] Frijda NH. The Emotions: Studies in Emotion and Social Interaction. Edition de la. 1986
- [27] Forgas JP. Affect in social judgments and decisions: A multiprocess model. Advances in experimental. Social Psychology. 1992;25:227-275
- [28] Tellegen A. Structures of mood and personality and their relevance to assessing anxiety, with an emphasis on self-report. In: Tuma AH, Master JD (editors). Anxiety and the Anxiety Disorders. Hillsdale, NJ: Erlbaum; 1985. pp. 681-706
- [29] Frijda NH. Emotions are Functional, Most of the Time. New York: Oxford University Press; 1994
- [30] Schwarz N, Bless H, Bohner G. Mood and persuasion: Affective states influence the processing of persuasive communications. In: Zanna MP (editor), Advances in Experimental Social Psychology. Orlando, FL: Academic Press; 1991
- [31] Nunally JC, Bernstein IH. Psychonometric Theory. 3rd Edition, New York: McGraw-Hill; 1994
- [32] Mulligan K, Scherer KR. Toward a working definition of emotion. Emotion Review. 2012;4(4):345-357

- [33] Beaudry A, Pinsonneault A. Understanding user responses to information technology: A coping model of user adaptation. MIS Quarterly. 2005:493-524
- [34] Benbasat I, Todd P. The effects of decision support and task contingencies on model formulation: A cognitive perspective. Decision Support System. 1996;17:241-252
- [35] Mayer JD, Caruso DR, Salovey P. Emotional intelligence meets traditional standards for an intelligence. Intelligence. 1999;27(4):267-298
- [36] Zhang P, Li N. Positive and negative affect in IT evaluation: A longitudinal study. Proceedings of the Sixth Annual Workshop on HCI Research in MIS, Montreal, December 8; 2007. pp. 67-71
- [37] Zhang P, Li N. Love at first sight or sustained effect? The role of perceived affective quality on users' cognitive reactions to IT. Paper presented at International Conference on Information Systems. Washington, DC. 2004
- [38] Isen AM, Daubman KA, Nowicki GP. Positive affect facilitates creative problem solving. Journal of Personality and Social Psychology. June 1987;52(6):1122-1131
- [39] Perlusz S. Emotions and technology acceptance: development and validation of a technology affect scale. In: Proceedings: 2004 IEEE International Engineering Management Conference, 2004. Vol. 2. IEEE; October 2004. pp. 845-847
- [40] Zhang P. The affective response model: A theoretical framework of affective concepts and their relationships in the ICT context. Journal MIS Quarterly. March 2013;37(1):247-274
- [41] Watson D, Tellegen A. Toward a consensual strucutre of mood. Psychological Bulletin. 1985;98:219-235
- [42] Isen AM. Some perspectives on positive affect and self-regulation. Psychological Inquiry. 2000;11(3):184-187
- [43] Watson D. Intraindividual and interindividual analyses of positive and negative affect: their relation to health complaints, perceived stress, and daily activities. Journal of Personality and Social Psychology. 1988;54(6):1020
- [44] Watson D, Clark LA, Tellegen A. Development and validation of brief measures of positive and negative affect: The PANAS scales. Journal of Personality and Social Psychology. 1988;54(6):1063
- [45] Hair JF, Sarstedt M, Ringle CM, Mena JA. An assessment of the use of partial least squares structural equation modelling in marketing research. Journal of the Academy of Marketing Science. 2012;40(3):414-433
- [46] Zhang P, Li N. The importance of affective quality. Communications of the ACM. 2005;48(9):105-108
- [47] Sun H, Zhang P. The role of affect in IS research: A critical survey and a research model in human-computer interaction and management information systems. In: Zhang P, Galletta D (editors), Foundations. Armonck, NY: M. E. Sharpe. 2006

- [48] Russell JA. A circumplex model of affect. Journal of Personality and Social Psychology. 1980;39(6):1161-1178
- [49] Russell B. Analysis of Mind. Routledge; 2005
- [50] Posner J, Russell JA, Peterson BS. The circumplex model of affect: An integrative approach to affective neuroscience, cognitive development, and psychopathology. Development and Psychopathology. 2005;17(3):715-734
- [51] Feldman LA. Valence focus and arousal focus: Individual differences in the structure of affective experience. Journal of Personality and Social Psychology. 1995;69(1):153-166
- [52] Zelenski JM, Larsen RJ. Predicting the future: How affect-related personality traits influence likelihood judgments of future events. Personality and Social Psychology Bulletin. 2002;28(7);1000-1010
- [53] Barrett LF, Russell JA. The structure of current affect controversies and emerging consensus. Current Directions in Psychological Science. 1999;8(1):10-14
- [54] Yik M, Russell JA, Steiger JH. A 12-point circumplex structure of core affect. Journal of Emotion. 2011;11(14): 705-731
- [55] Ekkekakis P. The Measurement of Affect, Mood and Emotion: A Guide for Healthbehavioral Research. UK: Cambridge University Press; 2013
- [56] Watson D, Wiese D, Vaidya J. The two general activation systems of affect: Structural findings, evolutionary considerations, and psychobiological evidence. Journal of Personality and Social Psychology. 1999;76(5):820-838
Knowledge Management Applications

The Impact of the Internet of Things to Value Added in Knowledge-Intensive Organizations

Maja Meško, Jana Suklan and Vasja Roblek

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/intechopen.70070

Abstract

This chapter has focused on the importance and influence of Industry 4.0 and consequently the Internet connected technologies' creation of the value added for knowledgeintensive organizations and society. The chapter considers the development of knowledge management through different periods; it has focused on its importance during the use of social media and the rise of the Internet of Things. The goal of this chapter is to achieve a better understanding of the impact of knowledge management processes to the value of knowledge that is established by the extent of knowledge sharing within the organization and with suppliers and customers (knowledge sharing outside the organization).

Keywords: knowledge organization, Internet of Things, innovative economy, value added chain, knowledge management, organizational changes

1. Introduction

In the last 25 years, as a rapidly developing discipline, knowledge management (KM) has become the utmost important source for organizations that aim to raise efficiency, innovation, and hence the competitive abilities [1, 2]. During this period, the theory of knowledge management has also been developing [3] due to the recognition of knowledge as a fundament for competitiveness and the key to business success [4].

Handzic [5] defined the fundamental purpose of knowledge management as collecting, organizing, and processing knowledge into a form that is useful to all employees. An important role of knowledge management is shown in applications that exist in the organizations, such as intranet, enterprise resource planning systems (ERP), and customer relationship management systems (CRM). Such applications store transactions and customer data for business



decision-making processes [6]. With launching the Web 2.0 and social media, these processes are running outside the organization [7]. According to the fast adoption of the social media platforms and mobile applications in the last 10 years, organizations have faster (in real time) and easier access to the knowledge that is created in human living environment [2]. The Internet transformation of the digital industry is still in progress; the artificial intelligence, big data, and connectivity indicate the certainty of a new round of digital revolution [8].

The concept of Industry 4.0 has launched the fourth technological revolution, which is based on the concepts and technologies that include cyber-physical systems, the Internet of Things (IoT) and the Internet of Services (IoS). Industry 4.0 is based on perpetual communication via Internet that allows a continuous interaction and exchange of information, not only between humans and machines, but also between the machines themselves [9]. This communication interaction influences the establishment knowledge management 4.0. The trend of the social customer relationship management (social CRM) for an emergent concept that integrates classical customer relationship management (CRM) and social media is leading toward the establishment of a communication channel for the continuous exchange of information about the needs and individual situations in real time to e-retailers, healthcare, manufacture, housekeeping, coworkers, customers themselves, energy suppliers, etc. [10].

This chapter has focused on better understanding of the impact of the Internet of Things on organizational changes in knowledge-intensive organizations. The contribution of the article is mainly conceptual.

2. Transformation from knowledge economy to innovative economy

The transformation of "traditional heavy industry" into the technological development-oriented economy proceeded in the middle of 1980s in twentieth century [11]. The economic period of knowledge economy has influenced organizational changes in strategies, structures, and management styles [12].

The emergence of the Internet in the early 1990s of the twentieth century (period of the Internet or digital economy) has influenced the rise of the third wave of capitalism. The Internet launched the new institutional revolution in the globe. Consequences have been seen in increasing the importance of networks and crowds in relation to market institutions. Internet revolution is based on a population size (N), the resources feeding supply sub-system (F) encompassing both the natural resources and food system, and accumulated technological and scientific knowledge sub-system (K). The key factors N-F-K have an important influence on human development and economic growth [13].

The technological development forces organizations that are based on the Taylors paradigm hierarchy into the changes of the business processes that consequently lead to the changes of the organizational structures. Organizations must have in mind that their role in a new "smart technological capitalistic" era will be not only to provide the development of a hightech information and communication technologies, but also to support the social innovative projects and thus have an impact on social and economic development of the society [14]. During the time of economic crisis (2008–2009), the first signs of changes in business models occurred. Innovation companies gave an answer how to increase the economic growth in last few years. Entrepreneurs have begun to open up a start-up technology companies related to the development of online platforms and digital technologies [15]. The launching of a new business models based on the Internet and digital technologies affected in last few years due to the changes in the consumer behavior. The fast Internet connections and mobile technologies enable the development of the trading and sharing platforms (e.g., Airbnb, Backfeed, Fairmondo, Uber, Tapaz, etc.). Digital platforms enable individuals and business operating in real time via the Internet, which increases the range of potential customers and due to the nature of the business reduces the cost of marketing and business resulting in lower prices for goods and services [16].

Managers and leaders should be aware also of the knowledge negative aspects, because it can cause economic crisis. Control of the knowledge resources and prevent sharing of knowledge and information have a negative impact on the development of innovative society. Cause is the large gap between developed and emerging countries [17]. Pagano and Rossi [18] claim that the cause of the crisis lies at the monopoly of developed countries over intellectual rights. International agreements on trade-related aspects of intellectual property have caused a rise in the cost of investments in countries that had neither abundant inexpensive labor nor high amounts of intellectual property. The authors believe the solution to the crisis, besides changes in monetary policy, financial regulation, and standards of Keynesian economic policy, also needs a measure that will reduce the intellectual monopolization of the economy.

The innovative economy that succeeded knowledge economy introduces new approaches for the development of business models in the process of organizational evolution. The innovative smart Internet connected technologies become a system for creating and distributing the knowledge and as such they have characterized firms and many other social systems [19]. The main factor for the development of the entrepreneur idea is not any more the financial capital but a social capital, because products or services are a result of knowledge that is based on innovative ideas by either individuals or teams [20]. Knowledge and information have become a key factor for success in the knowledge economy as an innovative economy [21].

It is necessary for the development of an innovative environment to facilitate access to venture capital (e.g., crowdsourcing, kickstar, venture capital funds, etc.) and move on to the new forms of cooperation that are based on short term relations for which the duration depends on the project timeframe, provide tax relief for R&D, enhancement of intellectual property rights (patents and licenses), and facilitation of the networking organization that facilitates cooperation among businesses (e.g., clustering).

2.1. Industry 4.0 and rise of Internet of Things

A new concept of German economic policy was formed in Germany in 2011, based on high-tech strategies named as Industry 4.0 [22]. That concept raised the fourth technological revolution, which is based on the concepts and technologies that include cyber-physical systems, the Internet of Things (IoT), and the Internet of Services (IoS) [23, 24], based on perpetual communication via Internet that allows a continuous interaction and exchange of information, not only between humans (C2C) and human and machine (C2M), but also between the machines themselves (M2M) [25].

The rise and expansion of Industry 4.0 with its current fundamental concepts (**Table 1**) is based on the assumption of increasing global urbanization [26]. Demographic changes are becoming a challenge for urban renewal and development, which will have to enable the infrastructure of residents for ensuring their quality of life and sustainable orientation [7, 24]. What actually presents the phenomenon of Industry 4.0, and in which parts of the economy and the human environment it is expanding, is probably most evident from the expressions with which it is associated. These fundamental concepts of Industry 4.0 and the explanation of their contents are shown in **Table 1**.

As shown in **Table 1**, in order to create a smart project, smart technologies and devices have been used. A critical component for the success of urbanization and societal development will be a smart technology. It is predicted that the purpose of the technology will be aimed at collecting and analyzing data from the human environment in order to design a circular economy, increase revenues, lower capital spending, improve services, and mobility [9, 27].

An organization needs a developed business-support infrastructure as a result of initiating partnerships, which allows it access to information, advice, and education. In this model, information is changing and in such a rapidly changing environment, information is not so much important [4].

Expression/fundamental concept	Explanation
Smart factory, smart manufacturing, intelligent factory, factory of the future	The smart factory will be more intelligent, flexible, and dynamic. Manufacturing will be equipped with sensors, actors, and autonomous systems. Machines and equipment will have the ability to improve processes through self- optimization and autonomous decision-making
New systems in the development of products and services	Product and service development will be individualized. In this context, approaches of open innovation and product intelligence as well as product memory are of outstanding importance
Self-organization	In manufacturing, processes change in the entire supply and manufacturing chains. These changes will have an impact on changing processes from suppliers to logistics and to the life cycle management of a product. Along with all these changes manufacturing processes will be closely connected across corporate boundaries These changes in supply and manufacturing chains require greater decentralization from existing manufacturing systems. This fits with a decomposition of the classic production hierarchy and a change toward decentralized self-organization
Smart product	Products inserted with sensors and microchips that allow communication via the Internet of Things, with each other and with human beings. Cars, t-shirts, watches, washing powder, etc., are set to become "smart," as their makers attach sensors to their packaging that can detect when the product is being used, and that can communicate with smartphones when scanned. Smart products are eliciting the question of invasion of privacy and consequently personal safety

Expression/fundamental concept	Explanation	
New systems in distribution and procurement	Distribution and procurement will increasingly be individualized	
Adaptation to human needs	New manufacturing and retailers' systems should be designed to follow human needs instead of the reverse. It is suggested that these systems may well be a combination of robotic-like tools such as personal intelligent agents such as Siri, Viv, Cortana, Google Now, and others, which will be combined with the Internet of Things. That can become the dominant model of the interaction between buyers and sellers	
Cyber-physical systems	Systems will integrate the computation, networking, and physical processes. Embedded computers and networks will monitor and control the physical processes, with feedback loops where physical processes affect computations and vice versa. An example is the control of vital human functions that allow urgent health care through mobile applications, sensors in clothing, sensors, and surveillance cameras in flats	
Smart city	Smart city is defined as a city that comprises six factors in its development policy: smart economy, smart mobility, smart environment, smart people, smart living, and smart governance. It is the product of accelerated development of the new generation information technology and knowledge-based economy, based on the network combination of the Internet, telecommunications network, broadcast network, wireless broadband network, and other sensors networks where IoT as its core	
Digital sustainability	Sustainability and resource-efficiency are increasingly in the focus of the design of smart cities and smart factories. It is necessary to respect ethical rules when using private information. These factors are fundamental framework conditions for successful products	
Source: Ref. [2].		

Table 1. Fundamental concept of the Industry 4.0 and implementations of IoT.

If management wishes that the new digital business models have an impact on increasing the added value of organizational knowledge, it is necessary to focus on:

- Definition of adoption of human communication methods in the external environment (terms of quality, distinctiveness, and ownership of data, information, and knowledge).
- How people account for the risk of sharing their content with strangers. Are people receptive to the issues of ownership and transfer of data between strangers (the problem of transmission of information from researchers and developers with external experts to help create the so-called open innovation)?
- How it affects the takeover or merger of the development organization that owns the new technologies developed by its members to communicate with experts in acquiring.

It must be realized that today the company due to globalization changes is exposed to more rapid changes in society and nature, which as such have the greatest impact on product development, operations, and ultimately to the success of the company [28]. This change caused the growth of importance of reaction in interaction with the environment in which the entity operates (headquarters, production), as well as the markets in which it sells. This means that the massive classical information passes into custom-made information known user or target groups of users with high added value [27]. Such information is transformed into specialist advice and continuous monitoring of users, while education is increasingly necessary in the fields of encouraging companies for corporations, informing about the types of cooperation and management of partnership development networks.

3. Changes in knowledge management and in the knowledge-intensive organizations

3.1. Knowledge-intensive organizations

In the case of the contents described in this chapter, the objective is to contribute to the understanding of the structures and contexts of utilization of knowledge-based resources; the question is how complex implementation of the Internet of Things as a technological instrument effects on organizational changes and adaptation of knowledge processes; and to the understanding of the relationship between innovations of IoT technologies and organizational performance. According to Porter [29], a value chain represents a key source of competitive advantage. In this case, cost advantages and successful differentiations are tracked more effectively by considering the chain of activities that organization performs with the goal to deliver value to its customers. The most products of the knowledge-intensive organizations are hidden assets. These invisible assets have presented knowledge about what the organization produces, how an organization is organized, and for what reasons [30].

Organization is an open system that is affected by the external environment with emphasis on the global nature and constantly changing requirements of customers, suppliers, and market conditions (nature conditions, the legislation, customs, taxes, and religion). The value of external system is featured as the ability to solve problems and needs of our customers. According to the information about customer behavior, organization can establish a policy for the customer relationship management (CRM) and suppliers develop a marketing strategy that provides market positioning and design of brand loyalty [3]. Web platforms become an important source of knowledge about the customers and with help from information getting from web platforms and Internet connected products, organizations have created the content value chain. This is achieved by linking complementary organizations and respective organizations with their distributors and customers [31].

The external effects on the organization are evident in organizational effectiveness and relevance of the activities in the field of R&D of individuals and groups within the organization and the organization's contribution to society [7]. The purpose of networking between organizations is the tendency to develop and implement technology solutions and processes that will enhance the organizational added value and bring added value to the customer in the form of utility value. Linked organizations that constitute the value chain have to reach decisions on strategies to increase the added value (e.g., acquisition, accumulation, and divestment) with a consensus with partners. In order for successful participation to occur in the value chain, organizations have to identify common goals, be complementary, and trust each other [32].

Internet saturation is influenced on changing the Internet architecture (architecture principles named Kondratieff cycle that began in 1970). The sixth Kondratieff cycle is started with new Internet technologies and it is predicted that IoT and big data will change the KM in next few years [33].

Drucker et al. [34] used the term "knowledge work" such as the fact that knowledge is a key resource in most organizations, not capital land or labor. Knowledge combined with experience helps in decision-making and innovation. The economy, which is variable and uncertain knowledge, appears as nonstatic phenomenon.

Drucker [35] had, at the end of the 1980s, pointed to the emergence of a new organization based on knowledge, consisting of experts, subject to share their experiences (and thus gain quick response) with their colleagues, customers, and the leadership of the organization. He named it as the information-oriented organization.

Zack [36] defines knowledge-intensive organizations by their process, place, purpose, and perspective, which will be discussed briefly and including the case of smart health services.

- *Process*: This includes application of existing knowledge and creation of knowledge. Knowledge sharing within organization provides benefit based on past knowledge, increase of teamwork, creation of new knowledge, and emergence of opportunities for experimentation and learning.
- *Place*: The knowledge-intensive organization, people, and supporting resources are creating and applying knowledge by continued interaction. The organization is seeking for knowledge wherever and connects with anyone who can help it to learn.
- *Purpose*: The organization should establish a corporate culture that will enable sharing, exploitation, and use of internal organizational knowledge in everyday working processes. Organization uses information and information technologies in order to ensure performance of these processes. The knowledge-intensive organization focuses on who needs to work with whom instead for whom.
- *Perspective*: The organization is holding its knowledge-oriented image regardless of the service or a product that is provided by it. Knowledge and learning are the primary criteria for evaluating organizational working processes, location, and relationships with customers. It is what, and how it learns from the customers. The cost of learning is investment in future innovations, not a cost.

Mahesh and Suresh [8] and Gummesson [9] identified three types of knowledge-intensive organizations, based on knowledge perspective:

- Manufacturing organizations: knowledge-intensive work is limited to the research and development departments. A narrow group of employees is tied to creating knowledgeintensive decisions and creation of a value-added chain within this context. Cognitive and social distances are lower among the employees in departments and higher at the entire level of organization.
- Consulting firms, legal firms, and pure play technology organizations: knowledge constitutes a major part of the output. The value chain depends on the ability of processing specialist knowledge of employees in the firm. Social and cognitive distance is smaller in all parts of the organization.
- Information, communication, and consumer-oriented technology organizations: these organizations have a diverse business network, producing and marketing a wide range of customized business and technology products and services. Investment in high technology and innovations represent key factors for their existence. This causes a perspective distance between the middle and the bottom level of hierarchy of the organizations. The consequences are visible in slower responses to market conditions. The measures cannot be thought cross the organizational hierarchy quickly. This has led to problems in achieving set goals of management at the business process level and the transfer of responses of organizational changes from the lower to a higher management level.

An example of a knowledge-intensive organization and its features in the period of the fourth industry revolution is presented in case of the digital-health companies and processes of a smart home health care.

In the case of a smart home, health care is going for the system where embedded computers and networks are monitoring and controlling the physical processes with feedback loops where physical processes affect computations and vice versa. All information is stored in clouds and digital-health companies enable the exchange of health data [37].

The concept of the smart home health care enables ill people to stay home as much as possible and continue living a quality life. Such digital healthcare platform is not only influenced by organizational changes of a health-service but also by decreasing costs of this service. McKinsey research showed that implementation of patient self-services with using digital channels instead of the direct physician visit, or patient self-management solutions can produce net economic benefits of 7–11% of total healthcare spending [38].

Control of vital human functions can be presented as a case study; this allows urgent health care through mobile applications, sensors in clothing, sensors, and surveillance cameras in flats (e.g., integrated in the house or smart phone application) monitoring the patient and sending information to physicians [10, 3].

In the case of a process of a smart home health care, the integration and sharing information are the key concept. It is going for the process of knowledge management 4.0 whose characteristics are defined by the following [39]:

- Big data are acquired directly from the things and users. Information is analyzed and saved in clouds;
- Information is shared in real time. Content is available online. There are no limitations for sharing information between people or things. The development of Internet technologies resulted in creation of knowledge and its easier sharing in external organizational environment.
- In the case of personal data, it is necessary to allow access only to authorized personnel. High level of privacy and strong data protection should be provided.
- Information sharing and collaboration is going on via wireless communications between people, between people and things, and between things.

3.2. Creation and the importance of knowledge management

Marshall et al. [33] defined KM as a process in which we are trying to figure out what kind of human capital lies in the minds of individuals and the shift in organizational capital, to which he is able to access a large group of individuals, each of which depends on the development of the organization.

Nonaka [26] claims that the production of knowledge at the individual level, which is available to others, is the pivotal activity of the organization, based on knowledge-creating company. The author emphasizes that it is a permanent process that takes place at all the levels within the organization, which may, in certain moments, occur in unexpected forms. He claims that the quantitative data acquisition of new knowledge—increase efficiency, reduction of costs and improvement of return on investment—is not sufficient for knowledge management. It recalls the importance of rapid response to customer needs, creating new markets, and innovative products.

Rowley [40] identified KM as a definition, acquisition, use, maintenance, and disposal of the assets of knowledge, for the purpose of increasing the value and benefits of all stakeholders.

Epistemology is claimed on what knowledge is valid in research, how that knowledge is presented, and what kind of knowledge is found in knowledge-intensive organizations, society, environment, and specially with using and analyzing the information obtained from social media and Internet of Things in the virtual world (e.g., big data clouds storage) [15].

The creation of new knowledge is not just a matter of processing objective information. The point is that the creation of new knowledge depends on the exploitation of tacit knowledge and often highly subjective knowledge of each employee within the organization [20]. In addition to the tacit knowledge, it is necessary to be vigilant even in explicit knowledge (knowledge of official records, such as patents) and the importance of their mutual interaction in the development of new products or services [7].

Kogut and Zander [23] share knowledge in the organization on information and "know-how." They believe that knowledge is based on the competencies of individuals and the principles of the organization, through which relations between individuals, groups, and members of the business connections are arranged and coordinated. The authors present the conclusion that organizations exist only because they are better able to transmit and share knowledge as a market. This conclusion follows from the premise that organizations are able to process and store large amount of information than individuals [41].

The development of knowledge management can be divided into three phases (**Table 2**). First phase appeared in the mid-1990s, and it can be classified as a classical knowledge processes development phase. The main feature of this period is that knowledge is sought outside the organization. The task of the organization is to capture, decode, and share the knowledge in organization and its environment. The procedures of knowledge management begin after the creation of knowledge, and the next processes are knowledge development and its transformation into practice. Organizations were highlighted in this first phase of the integration of knowledge in business processes [42].

In the first period of development of knowledge management, it was lead to the conclusion that knowledge has to be produced in a social environment. It has been established that the knowledge is generated through the processes of individuals. The creators of knowledge are obliged to care for its accuracy. In the process of cooperation between individuals both within the organization and in the external environment, it comes to sharing knowledge [42, 43].

The second phase started around 2003 with the launch of the Internet 2.0-based knowledge processes.

The second phase of knowledge management is characterized by the awareness of the importance of external information and knowledge to the organization. Providing access to external information and knowledge, including their involvement in the value chain creation is of utmost importance. Important characteristics of knowledge organizations are in their advantageous utilization of superior information technology and highly skilled employees (knowledge workers) that is able to implement its innovative activities into realization [44, 45].

Classical knowledge processes	Internet 2.0-based knowledge processes	IoT knowledge processes	
Knowledge based on the data acquired from the intranet, CRM. Data is saved in local servers	Information is accessed and stored via clouds and platforms like Google and Facebook	Big data acquired directly from the things and customers. Analyzed and saved in clouds	
Local time and personal limited access	Business or private content is available on any device, any place, and any time	Real time. Content is available online. No limitations for sharing information between people or things	
Organization limited networking; information sharing and discussion via email or intranet	Internet 2.0 provides online relations between the customer and supplier. The discussion is limited to the matter of content and physical data entry	Information sharing and collaboration via wireless communications between people, between people and things, and between things	

Source: Ref. [39].

Table 2. Differences between classical knowledge processes and the IoT knowledge processes.

The emergence of the Web 2.0 has had a significant impact on the development of the third stage of KM. During the period after 2005, with the development of social media, web portals were integrated [7]. Knowledge thus became available outside the organization and management, which is one of the critical factors of business success [46]. Von Krogh [45] proposed the theory that the Web 2.0 applications are not necessarily included in the context of KM, but simply a means of enabling access to knowledge. By integrating information and communication technology (ICT) into products, the von Krogh theory is a basis for the understanding of next third phase of knowledge management development. The IoT has influenced the development of new knowledge management generation (**Figure 1**) that is arising from the phase of integration between people and people with documents and passes to the phase of connection between devices. KM processes are also located between the consumer and the manufacturer or service provider.



Figure 1. Implementation of the knowledge management system in small and medium company. Source: Authors.

Industry 4.0 is based on mobile computing, cloud computing, and big data. The importance of cloud computing and mobile computing for Industry 4.0 lies in the provision of services, which can be accessed globally via the Internet. Services can easily be integrated and used [2].

For the establishment of the IoT and the course of the processes of knowledge management, companies set up a circuit between product and service: (i) radio frequency identification (RFID); (ii) wireless sensor networks (WSN); (iii) middleware, (iv) cloud computing, and (v) IoT application software [37]. This system can operate both with the help of people and artificial intelligence. The data collected with the help of these systems is saved in clouds. Products integrated with cloud computing in the field can provide data that enable a predictive maintenance, and to provide information about optimization possibilities in production. The use of integrated networking and integration of products into Internet data will allow for far reaching possibilities to collect data [47]. Instead of single data points or short intervals, a continuous stream of data is now available. The huge amounts of data available can now be used to continuously analyze and optimize production. This enables fostering of predictive analytics [48]. Such system is based on big data analytics, which enables, for example, the informing of vehicles that are driven in a column, the distances between them, events on the road in front of them, weather conditions, etc. The drivers of vehicles-thus not only based on geostationary information – know the route, but with the help of sensors and connectivity between vehicles, receive notice of current dangers on the road [37, 15].

3.2.1. Processes of knowledge management in organizations

Organization must develop an organizational culture to the level of employee awareness that knowledge sharing is presenting a fundamental concept for the firm, which enables further growth of the organization. The process includes the identification of the knowledge, sharing knowledge, knowledge acquisition, collection, and storage of knowledge and refresher [49].

Grower and Davenport [50] consider that the processes of KM are located between the information and the source of income of the firm (services rendered or products sold). The process is based on three sub processes: acquisition of knowledge, a definition (record) of knowledge, and the transfer of knowledge and realization. The general concept model of the process stems from the fact that by coordinating, coordinated and introduced operational processes occur. Together, these processes form the system of KM as important operational processes so representing [26]:

- Need for skills: determine which skills the organization needs. This process also includes the sharing of the existing knowledge, because the purpose is to figure out which is the already known knowledge that could be helpful for the organization;
- Acquisition and knowledge creation: new knowledge is often generated also by combining existing knowledge of the transfer;
- Documenting and storing knowledge: the results of the newly acquired knowledge that is documented and retained.

3.2.2. Organizational knowledge sources and their value

Knowledge has become a force, a power, which is providing to the organization a competitive advantage. In doing so, the organization shall establish all levers (for example, the establishment of corporate culture) that enable it to exploit and use internal organizational knowledge in daily operations [51, 30]. Knowledge, which is used in business processes, is called organizational knowledge [30]. Organizational knowledge can be found in databases, business papers, archive organization, and minds of individuals. Knowledge is in newly emerging companies typically fragmented, and it can be found in chunks throughout the organization. When an organization moves the initial period of operation and begins to grow, to make all these fragmented pieces integrated into the system KM the purpose of which is to collect, organize, and process the knowledge into a usable form, applicable to all employees (for example, in this end, the organization set up an intranet, ERP systems, CRM systems, etc.) [14].

An advantage of using the KM system is that it allows the conversion of intangible assets into useful business resources. In practice, for example, it shows in the event that the organization develops a commercial solution (technology, processes, designs presentations, etc.). In addition, hereinafter referred to license to offer in the form of advisory services (in the case of a consulting company) or in the form of licenses and franchises. An organization that purchases a license can use the existing knowledge to solve similar problems and create sources of revenue for the license vendor.

Organizations often include a KM system into the marketing strategy, especially when they want to reach out to their customer, and they are entrusted with the execution of a specific project (for example, the establishment of CRM and other software solutions to customer needs) by an external service provider. Through the system of KM as subscriber's they see what employees know about the project, which will be implemented, and what solutions have already been developed in the past.

In fact, KM as system has to ensure a more accurate communication between employees and thereby accelerate the settlement procedures and search, and providing solutions, which gives employees the ability to make better decisions [42]. It is necessary to understand that all of the obsolete knowledge that can be found inside or inside the organization is in various forms, which have to be physically moved into the KM system.

Organizational knowledge is being created on the basis of resources [9, 52]:

- 1. Explicit knowledge: official documents, procedural instructions, and patent;
- **2.** Tacit knowledge: unstructured knowledge, usually personal knowledge, based on the experience of an individual;
- **3.** Strategic knowledge: the knowledge to which access is restricted (marketing strategy, financial analysis, business contracts, software code, etc.).

The system KM is formed on the basis of the organizational structure and the company's own needs. All databases connected into a whole have represented the KM as system (**Figure 1**).

4. Conclusions

The chapter represents an important theoretical contribution to the understanding of the influence of the IoT to value added in knowledge-intensive organizations. Managers should be aware that innovative information and communication technologies are influencing the business performance in the twenty-first century.

The trend of KM 4.0 is leading toward the establishment of a communication channel for the continuous exchange of information, in most cases between machines themselves. The purpose of such automation is the individual customer-oriented adaptation of products and services that will increase value added for organizations and customers [14, 53]. Therefore, the IoT technology enables the creation of completely new products, services, and business models, which promise gains in virtually all industries [54, 55].

The findings are based on the literature review. A particular limitation of the text is that no research was made. Further researches should be focused on the effect of this direction of technology on the ecosystem. Deeper investigation of this topic could include case study with elements of implementation, testing business benefits, and social and environmental benefits with real data.

Author details

Maja Meško^{1,2,3}, Jana Suklan^{1,2,3} and Vasja Roblek^{1,2,3*}

- *Address all correspondence to: vasja.roblek@gmail.com
- 1 Faculty of Primorska, University of Primorska, Koper, Slovenia
- 2 School of Advanced Social Studies, Nova Gorica, Slovenia
- 3 Fizioterapevtika, Institution of Higher Education, Medvode, Slovenia

References

- Drucker FP. Management and the world's work. Harvard Business Review. 1988;66(5): 65-76
- [2] Roblek V, Meško M, Krapež A. A complex view of industry 4.0. SAGE Open. 2016;6(2):1-12
- [3] Drucker PF, Wilson G. The Essential Drucker. Oxford: Butterworth-Heinemann; 2001
- [4] Ning H, Liu H. Cyber-physical-social-thinking space based science and technology framework for the Internet of Things. Science China Information Sciences. 2015;58(3): 1-19

- [5] Handzic M. Evaluating KMS effectiveness for decision support: Preliminary results. In: King WR, editor. Knowledge Management and Organizational Learning. New York: Springer; 2009. pp. 237-252
- [6] Tennis JT. Epistemology, theory, and methodology in knowledge organization: Toward a classification, metatheory, and research framework. Knowledge Organization. 2008; 35(2/3):102-112
- [7] Hanna R, Rohm A, Crittenden VL. We're all connected: The power of the social media ecosystem. Business Horizons. 2011;54(3):265-273
- [8] Mahesh K, Suresh JK. Knowledge criteria for organization design. Journal of Knowledge Management. 2009;13(4):41-51
- [9] Gummesson E. Qualitative Methods in Management Research. Newbury Park: Sage; 2000
- [10] Argrys C. Teaching smart people how to learn. In: Drucker, P, editor. Harvard Business Review on Knowledge Management. Boston: Harvard Business School Press; 1998. pp. 81-108
- [11] Alavi M, Leidner ED. Review knowledge management and knowledge management systems: Conceptual foundations and research issues. MIS Quarterly. 2001;25(1):107-136
- [12] Aue G, Biesdorf S, Henke N. How healthcare system can become digital-health leaders [Internet]. 2016. Available from: http://www.mckinsey.com industries/ healthcare-systems-and-services/our-insights/how-healthcare-systems-can-becomedigital-health-leaders [Accessed: 4.6.2016]
- [13] Bach MP, Lojpur A, Peković S, Stanovčić T. The influence of different information sources on innovation performance: Evidence from France, The Netherlands and Croatia. South East European Journal of Economics and Business. 2015;10(2):89-101
- [14] Kanevsky V, Housel T. The learning-knowledge-value cycle. In: von Krogh G, Roos J, Kleine D, editors. Knowing in Firms: Understanding, Managing and Measuring Knowledge. London: Sage; 1998. pp. 269-289
- [15] Hessman T. The down of a smart factory. Industry Week. 2013. Available from: http:// www.industry.usa.siemens.com/topics/us/en/ManufacturinRenaissance/Documents/ ManufacturingRenaissance-SmartFactory.pdf [Accessed: 6.6.2016]
- [16] Ivanov D, Dolgui A, Sokolov B, Werner F, Ivanova M. A dynamic model and an algorithm for short-term supply chain scheduling in the smart factory industry 4.0. International Journal of Production Research. 2016;54(2):386-402
- [17] Von Osten M. Such views miss the decisive point...the dilemma of knowledge-based economy and its opponents. 2004. Available from: http://biennale2004.werkleitz.de/ html_en/index_e.html [Accessed: 4.7.2017]
- [18] Pagano U, Rossi AM. The crash of the knowledge economy. Cambridge Journal of Economics. 2009;33(4):665-683

- [19] Roblek V, Pejić Bach M, Meško M, Bertoncelj A. The impact of social media to value added in knowledge-based industries. Kybernetes. 2013;42(4):554-568
- [20] Alexander CP. The new economy. Time. May 30, 1983:62-70
- [21] Baskerville R, Dulipovici A. The theoretical foundations of knowledge management. Knowledge Management Research & Practice. 2006;4(4):83-105
- [22] Mosconi F. The New European Industrial Policy: Global Competitiveness and the Manufacturing Renaissance. London: Routledge; 2015
- [23] Kogut B, Zander U. Knowledge and the speed of the transfer and imitation of organizational capabilities: An empirical test. Organization Science; 1995:6(1):76-92
- [24] Nachtigal M, Bertoncelj A. Ownership restructuring in transition societies from historic perspective. Acta Histrae. 2013;21(3):449-466
- [25] Davenport HT, Prusak L. Working Knowledge: How Organisations Manage What They Know. Boston: Harvard Business School Press; 2000
- [26] Nonaka I. The knowledge creating company. In Drucker PF, editor. Harvard Business Review on Knowledge Management. Boston: Harvard Business School Press; 1998. pp. 21-46
- [27] Smither R, Houston J, McIntire S. Organization Development: Strategies for Changing Environments. New York, NY: Routledge; 2016
- [28] Teece DJ. The foundations of enterprise performance: Dynamic and ordinary capabilities in an (economic) theory of firms. The Academy of Management Perspectives. 2014;28(4):328-352
- [29] Porter M. Competitive Advantage: Creating and Sustaining Superior Performance. New York, NY: The Free Press; 1985
- [30] Kogut B, Zander U. Knowledge of the firm, combinative capabilities, and the replication of technology. Organization Science. 1992;3(3):383-397
- [31] Pejić Bach M, Zoroja J, Bosilj Vukšić V. Review of Corporate Digital Divide Research: A Decadal Analysis (2003-2012). International Journal of Information Systems and Project Management. 2013;1(4):41-55
- [32] Miranda LCM, Lima CAS. Trends and cycles of the internet evolution and worldwide impacts. Technological Forecasting and Social Change. 2012;79(4):744-765
- [33] Marshall C, Prusak L, Spilberg D. Financial risk and the need for superior knowledge management. California Management Review. 1996;38(3):76-101
- [34] Drucker FP, Dyson E, Handy C, Saffo P, Senge MP. Looking ahead: Implications of the present. Harvard Business Review. 1997;75(6):18-32
- [35] Drucker P. The coming of the new organization. In: Drucker FP, editor. Harvard Business Review on Knowledge Management. Boston: Harvard Business School Press; 1998. pp. 1-19

- [36] Zack HM. Rethinking the knowledge-based organization. MIT Sloan Management Review. 2003;44(4):67-71
- [37] Dominici G, Roblek V, Abbate T, Tani M. Click and drive: Consumer attitude to product development: Towards future transformations of the driving experience. Business Process Management Journal. 2016;22(2):420-434
- [38] Antonelli C. The Economics of Innovation, New Technologies, and Structural Change. London: Routledge; 2003
- [39] Pemberton DJ, Stonehouse HG. Organisational learning and knowledge assets an essential partnership. The Learning Organization: An International Journal. 2000;7(4):184-194
- [40] Rowley J. What is knowledge management? Library Management. 1999;20(8):416-419
- [41] Croteau D, Hoynes W. Media/society: Industries, Images, and Audiences. 5th ed. Los Angeles: Sage Publications; 2013
- [42] McElroy MW. The New Knowledge Management: Complexity, Learning and Sustainable Innovation. Amsterdam, Boston: Butterworth-Heinemann; 2003
- [43] Nonaka I, Hirotaka T. The Knowledge-Creating Company: How Japanese Companies Create the Dynamics of Innovation. Oxford: Oxford University Press; 1995
- [44] McInerney CR, Koenig ME. Knowledge management (KM) processes in organizations: Theoretical foundations and practice. Synthesis Lectures on Information Concepts, Retrieval, and Services. 2011;3(1):1-96
- [45] Von Krogh G. How does social software change knowledge management? Toward a strategic research agenda. The Journal of Strategic Information Systems. 2012;21(2):154-164
- [46] Pang Z, Zhengb L, Tianb J, Walterc-Kao S, Dubrovab E, Chen Q. Design of a terminal solution for integration of in-home health care devices and services towards the Internetof-Things. Enterprise Information Systems. 2015;9(1):86-116
- [47] Cooper J, James A. Challenges for database management in the internet of things. IETE Technical Review. 2009;26(5):320-329
- [48] Abbott A. The System Of Professions: An Essay on the Division of Expert Labor. Chicago: University of Chicago Press; 2014
- [49] Mee LY, Huei CT. A profile of the internet shoppers: Evidence from nine countries. Telematics and Informatics. 2015;**32**(2):344-354
- [50] Grower V, Davenport HT. General perspectives on knowledge management: Fostering a research agenda. Journal of Management Informations System. 2001;18(1):5-21
- [51] Autry WC, Goldsby JT, Bell EJ. Global macro trends and their impact on supply chain management. New Jersey: Pearson Education; 2013
- [52] Vaidyanathan R, Aggarwal P. The impact of shopping agents on small business e-commerce strategy. Journal of Small Business Strategy. 2015;13(1):62-79

- [53] Spender CJ. Making knowledge the basis of a dynamic theory of the firm. Strategic Management Journal. 1996;17(Special winter issue):3-9
- [54] Chen Y, Hu H. Internet of intelligent things and robot as service. Simulation Modelling Practice and Theory. 2013;**34**:159-171
- [55] Teece JD. Capturing value from knowledge assets: The new economy, markets for knowhow, and intangible assets. California Management Review. 1998;**40**(3):55-79

Knowledge Management Trends in Biotechnology in Brazil

Maria de Fátima Ebole Santana

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/intechopen.70156

Abstract

This chapter presents a study on knowledge management and innovation in biotechnology area through analysis of scientific and technological advances of biotechnology trends in Brazil, providing an overview of the science profile as well as regional development and its relation to issues on topics based on the analysis of scientific publications for the last 20 years. Given these promising prospects, the monitoring and searching of scientific advances and trends in this area of knowledge have become essential for searching opportunities in research and development and also for potential innovations and business opportunities, both in the developed countries as well as in countries of emerging economies such as Brazil. The research was realized using database Web of Science with 60 terms selected in Biotechnology area and 73,125 documents have been organized. Scientific indicators were produced using data/text mining tools. A greater number of scientific publications were found in areas such as biochemistry and molecular biology; genetics and heredity showing a greater frequency in these terms: vaccine, PCR, and genome. Results pointed out the US as the main foreign partner-country of scientific publications followed by the UK, France, and Germany. It was possible to verify cooperation network with others Latin American countries.

Keywords: knowledge management, innovation, scientific trends, biotechnology, Brazil

1. Introduction

Technological advancement has been the dominant driving force in modern society leading to a widespread diffusion of products from the rational activity as well as of scientific, technological, and administrative sources, which in turn requires new types of management [1].



© 2017 The Author(s). Licensee InTech. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. This technological progress has induced a new paradigm based on the development of a set of intensive knowledge on scientific technologies, which represent many applications of scientific discoveries, whose core is to develop an increasing competence to manage information and knowledge. In this contemporary setting, the intangibles assets such as knowledge (know how), patents, and intellectual capital become increasing value strategic elements to be the center of contemporary forms of capital accumulation [2].

Biotechnology as a highly scientific sector is one of the most knowledge-intensive activities in the contemporary economy, having a direct and indirect impact on regional and national economies. The phenomenon of growth in the production of biotechnological products is relatively new, and in the long term, the growth potential is probably associated with the greater diffusion and use of your products and processes due to its convergence with nanotechnologies, information technologies, and other applied sciences [3].

In this context, biotechnology is considered as one of the most important technological tools nowadays. It is considered a key "future-bearing technology" and its applications have contributed to the structuring of new economic and social systems [4]. In this sense, biotechnology appears to have the characteristics of a core technology, with the potential to underpin a new technoeconomic paradigm.

Biotechnology represents a set of technologies "that use biological systems, living organisms, or their derivatives to produce or modify products and processes for a specific use" [5]. Biotechnology features a multidimensional nature, involving different knowledges, scientific and technological. Indeed, biotechnology can be considered as the result of a scientific revolution that involves many disciplines. In other words, it is a "constellation of scientific revolutions" [6], which is based on different areas of knowledge such as molecular biology, biochemistry, computer science, biophysics, engineering, and others.

The large set of biotechnological methods along with its different forms of application results in different interpretations about the dimensions that compose the biotechnology [7]. It has helped to generate new high-impact services in various segments such as health area with impressive revolution in the treatment of diseases as well as the use of new drugs for humans. It is also possible to verify its impact on agriculture with the development of functional foods and the reproduction of plant species; in the chemical and petrochemical industries with treatments of wastes and wastewaters; and the sustainable use of biodiversity, among other areas [2].

When assessing the potential application of modern biotechnology for the next 20 years, some authors assert that areas such as genomics, proteomics (spatial identification of protein structures), biomaterials, bioengineering, pharmacogenomics, genetic engineering, genetically modified foods, and synthetic biology will have a great impact on the future [2, 8].

In view of these future prospects, the monitoring and research of scientific advances and trends in this area of knowledge have become essential for searching opportunities in research and development (R&D), as well as for potential innovations and business opportunities, in the developed countries and mainly in countries of emerging economies such as Brazil [2]. Countries with rich biodiversity, mainly South American countries such as Brazil and Colombia, have tried to develop national capacities in science and technology (through

improving their infrastructure, greater participation in network experts, training of human resources, the increase publications) that allows a better optimization of scientific resources and economics and the generation of technological products based on nationally developed biotechnology [2, 9].

In this situational scenario, in order to propose worldwide profile of knowledge management of trends in biotechnology in Brazil, this chapter presents science trends in this area, mapped out through a set of variables such as the identification of the actors involved as well as the institutional partnerships, the major journals, among others, using the analysis of all Brazilian scientific publications of international dissemination for the period of 1995–2014.

2. Biotechnology in the global market and Brazilian government programs

2.1. Biotechnology in the global market

For centuries, humans have used biotechnology in their daily lives. Biotechnology presents as the growth area and development products and technologies observed in different areas of economy, which use live microorganisms or parts of them [10], showing up as a promising area among the diverse emerging technological developments. Being a multidisciplinary area presents the possibility of new products and processes. The bioindustry has contributed to the economic and social growth by bringing new solutions to problems concerning to human and animal health, to agribusiness, the environment, and the creation of new materials. In these aspects, biotechnology is a technological innovation that also provides products or processes with social or commercial use [11].

As technological progress advances based on life sciences, the possibilities of obtaining products with higher added value or lower production costs increase. The most prospected areas of biotechnology are health (pharmaceutical) and agriculture (food) [2]. On the economic bias, biotechnology is a major focus of activities on research, development, and innovation in the developed countries and it is becoming increasingly central in developing countries due to the potential of exploiting great biodiversity [12].

For a long time, it has been emphasized about the biotechnology potential and the future full of innovations that it provides. However, today, we can observe that the largest potential innovators are in the United States and Europe (the United Kingdom, Germany, and France). Due to their investments in the chemical and pharmaceutical corporations they become owner of biotechnology companies, enlarge their capabilities of innovation, or maximized new products generation [13].

The biotechnology industry has been grown rapidly in recent years, more than doubled its worldwide revenue in the last decade, going from US\$ 8 billion in 1993 to US\$ 20 billion in 1999 [14]. According to the study presented in Ref. [15], the worldwide market for biotechnology resources moves expressive values of about \$ 410 billion per year. The worldwide market

for biotechnology between 2002 and 2006 grew at a compound annual growth rate (CAGR) of 13.4% [16]. Figures in this market are not very consistent. According to the study presented in Ref. [17], the United States alone moved the biotechnology industry resources of US\$ 3.7 billion in 2009, being the market that moves much capital in this biotechnology market.

Regarding innovative effort, we can affirm that the United States is the country that spent more on R&D in biotechnology. In 2009, it spent only in entrepreneurial sector, US\$ 22 billion on R&D, followed by France, Germany, and Canada, with the total expense on R&D about US\$ 2.5, 1.3, and 1 billion, respectively [18].

In Asia and Pacific, among the countries that are more prominent, Japan is the most advanced in the biotechnology area, due mainly to the cooperation work developed by Japanese government, universities and private sector, directed especially to build an adequate infrastructure for innovative business *start-ups* [18]. After that the focus turns on Singapore, Taiwan, China, India, and Australia. The Chinese government, for example, invested about US\$ 40 billion in the biotechnology industry in 2012 and, the Singapore government hopes to increase 20% biotechnology investments in the next 5 years, that is, it is intended to spend about US\$ 12.5 billion in R&D in this sector [18].

2.2. Biotechnology in Brazilian government programs

The biotechnology segment holds a special place among the priorities at government policies, as much for the developed country and as for developing countries. The wide set of opportunities created by biotechnology, especially in health and agriculture areas, show the essentiality of its development as competitive strategy and expansion input in international market, mainly in developing countries such as Brazil. Brazil is a country with great potential for the development of agricultural biotechnology because it has a wide biological diversity and is rich in plants, animals, and microorganisms [19]. Since the 1970s, the biotechnology applied in agriculture has been productive and economic relevance in the country.

This chapter focuses on the characteristics and development of biotechnology in Brazil and refers to the public investment, with the main aspects of the most noteworthy programs and financing structure. There is a governmental structure in the country directed for the development of the area. This structure is composed of the following ministries: the Ministry of Science, Technology and Innovation (MST&I); Environment and Agriculture; Production and Trade; and Health and Social Development. At the same time, the government also acts through its government agencies, the private sector, and academia [2].

The government policies to support the development and financing on biotechnology in Brazil have started since 1980 with the promotion of several programs dedicated to the area, the Integrated Genetic Program, which aimed at introducing some specific actions on genetic engineering. In 1981, the government officially unveiled the National Program for Biotechnology (Pronab) which was to consolidate public investments for maintenance of the research groups in areas related to the program.

In 1984, the Ministry of Science and Technology (MST) had created a wide program in order to support, finance, and develop strategic areas in the country. This program was the development

support to Scientific and Technological Program (PADCT), cofinanced by International Bank for Reconstruction and Development (BIRD). Since its beginning, biotechnology has been seen as a strategic area for the scientific development in the country and, thus, established a specific subprogram to support it, the Biotechnology Subprogram (SBIO). PADCT started in 1985 and its actions were continued until the 2000s. At first, PADCT prioritized the development of some activities such as molecular biology, genetic engineering, and biosafety, without forgetting areas such as biochemistry, microbiology, and agronomy.

From the wide vision, PADCT aimed to build physical structures for research centers and development projects in cooperation in order to attract private investment, promote dissemination, and technology transfer from academic centers to the production sector. During this period, the first biotech products emerged that were human insulin, biodegradable plastics, biofilm, genetically modified plants, among others.

Early in the 1990s, both the government and the private sector have reduced the resources applied to biotechnology due to low commercial return of biotech products developed so far.

In 1999, the government changed the pathway of financing in Brazil through the creation of sources of fiscal financing from various economic sectors, called sectoral funds. In 2001, the biotechnology sector fund was created and its main objective was to ensure the continuity of biotechnology research, especially those considered as strategic for the country. Somehow, the creation of Biotech fund strengthens the National Program of Biotechnology and Genetic, established in 2000 and under MST&I responsibility. The program emphasized actions to "conservation genetic resources and development of biotechnological products and processes with applications in industries, agriculture, and human health". Several of these opportunities have been adopted as strategy of the project management in networking way in order to increase the flow of innovation and the results to society.

Among the many action plans applied from 2002, it is worth mentioning the following: Brazilian Genome Project, Structural Biology Network, Brazilian Proteomics Network, the development of biopharmaceuticals and immunobiology, and the development of new technological routes.

In 2004, the government using the Brazil's Industrial, Technological and Foreign Trade Policy (PITCE) pointed out biotechnology as the "future-bearing technology." On that political occasion, the Biotechnology Competitiveness Forum had created in order to put together all actors involved in the production chain in favor of strengthening industrial competitiveness.

Biotechnology research is funded by federal, private, and international agencies. The Organic Law of Science, Technology and Innovation (Decree No. 10.973 of 2/12/2004) defines some coordination organizations on the national level, and the Ministry for Science, Technology and Innovation (MST&I) represents the leading national organization [2].

From Biotechnology Competitiveness Forum, in 2007, begun the Biotechnology Development Policy (PDB), which established the National Biotechnology Committee (CNB). On the whole policy, structuring actions were defined in order to promote transfer of technology, investments, training of human resources, strengthen networks, regulatory framework, and to improve infrastructure for research centers and R&D. Examples of priority areas supported

were: plant and animal breeding, food technology, bioinformatics and immunology, diagnosis and prospecting on biotechnology, and detection of genetically modified organisms.

3. Methodology and recovery documents

3.1. Database

The research was based on the selection of scientific publications collected in ISI/Web of KnowledgeSM database. It is international bibliographic data and is used as reference for the generation of indicators of S&T and Innovation [2].

3.2. Strategies for recovery documents

The strategy for the recovery of publications on a particular scientific topic must be carried out carefully to avoid damaging in the analysis of scientific literature. For the analysis of results in an accurate overview of the subject, ideally, the "search expression" must promote the recovery of all relevant publications on the subject present in the database, and at the same time, exclude nonrelevant publications. However, such situation is difficult to achieve since the multidisciplinary issues receive contributions from several areas of knowledge.

The solution for recovery of significant and coherent set of publications that allows an analysis of the scientific production in the biotechnology area was the elaboration of a complex search expression, consisting a large of set selected and tested keywords.

At first, we tried to develop search strategies in the mode "ISI-general search," using keywords from the available literature and descriptors present on the *Platform-Lattes*/National Council for Scientific and Technological Development (CNPq) - Brazil and *Portal Innovation*/ MSCT-Brazil related to research groups and ongoing research in the biotechnology area. These search expressions were just an exploratory character, that is, serve as a basis for elaborate more refined expressions and evaluate how the area is organized into database [20]. A selection from bibliographic references was also used.

In order to achieve the purpose of mining scientific production on biotechnology in Brazil over a period of 21 years (1995–2015), 60 distinct descriptors (**Table 1**) were used, selected by experts in biotechnology area, interviewed and research groups in the biotechnology area sought in the base *Plataforma Lattes CNPq/Brazil* and *Portal Innovation/*MSCT-Brazil.

From our previous experiences, we found that the best way to do this recovery would be a combination of different expressions. For this, we used the advanced search mode of ISI/Web of Science and the advanced search. Once defined the strategy, we used all these keywords, and various search expression and their combinations in the database ISI/Web of Science only looking for indexed publication using document types "article." The fields "Title, Summary, and Keyword" were used to analyze the themes of scientific publications with international dissemination [21].

Antisense	Biomaterial	Proteins engineering	GMO	
Recombinant antigen	Biopolymer	Genetic engineering	Protein	
Biodiversity	Bioprocess	Metabolic engineering	Recombinant protein	
Biocatalyst	Bioprospection	Molecular engineering	Proteome	
Biofuel	Bioreactor	Gene expression	Proteomic	
Bioeconomy	Bioremediation	Pharmacogenomics	PCR	
Bioengineering	Biosensor	Phytoremediation	RNA	
Bioethics	Biosorption	Gene	Microarray DNA	
Biofiltration	Biosurfactant	Genetic	Microarray RNA	
Bioindustry	Biosulfurization	Genome	Transcriptome	
Bioinformatics	Biotechnology	Genomic	Transgenic	
Biolixiviation	Stem cells	Microbiota	Cellular therapy	
Computational biology	Cloning	Biology modeling	Gene therapy	
Biome	T cells	Nanobiotechnology	Molecular therapy	
Biomass	DNA	Peptide	Vaccine	

Table 1. Biotechnology-related terms used to search articles in database ISI/Web of Science.

The software VantagePoint[®] was used to perform the processing of information from articles retrieved in order to expand and enrich the results. VantagePoint[®] allows management of big data and information in order to present correlations of distinct variables of interest. It also allows us to identify Who, What, When, and Where, help us clarify relationships and find critical patterns, among other possibilities [2].

For visualization of the data, we used the software VOSviewer[®]. It is a latest free software for the representation and analysis of information, which appears as an alternative to the traditional techniques of multidimensional representation and network display. VOSviewer combines the visualization and clustering techniques, favoring analysis while bypassing unnecessary technical complications. It make possible to view the maps of collaboration between institutions, countries, and keywords [22].

After this step, macroindicators were generated for providing a global overview of the scientific production on biotechnology in Brazil and raising the main issues as follows: (1) the total number of articles published per year (1995/2015) and the trend of publication of the most frequently used terms; (2) major journals that institutions often publish their articles in this specific area; (3) key areas of knowledge and the number of articles published on the used terms; and (4) the number of papers published by institutions and maps of network (institutions and countries) for the purpose of identifying partnership [2].

The trends for biotechnology were mapped out through a set of variables such as the identification of the actors involved as well as the institutional partnerships and networking.

4. Overview of the global scientific development in biotechnology in Brazil

4.1. Analysis of results and cowords networks

Government policies to incentive and promote R&D activities in biotechnology in Brazil have started since the 1980s. However, from the late 1990s and the early 2000s, especially in 2001, with the creation of the national fund for biotechnology, the area aimed at real conditions of productivity and competitiveness in R&D. Biotechnology has its amended financing path taking up considerable resources of 97% between 2002 and 2003, with the resources allocation from the National Fund for Scientific and Technological/FNDCT, managed by the MSCT, CNPq, and Studies and Projects Financing Institution (FINEP) [23].

From the data collected in the ISI database, bibliometric indicators have been produced that will help us to understand the scientific activity in the biotechnology area.

With regard to the scientific literature on biotechnology for the period between 1995 and 2015, it was retrieved 102,326 documents containing terms selected by experts in the fields of title, abstract, and keywords, and 69,977 documents where Brazil as the home country (1st author). **Figure 1** shows the evolution of publication number in biotechnology area since 1995.

By analysis, the number of articles published throughout this period was possible to observe that there is an increase in publication trends, which indicates an intensive scientific activity over the past 15 years, mainly in the last 8 years. An exponential growth with increased production greater than 6500% was observed, as shown in **Figure 1**. However, this scenario



Figure 1. Historical evolution of biotechnology scientific publishing indexed in Brazil since 1995.

of accelerated growth should be modified in the coming years because of the reduction of R&D investments by government agencies in recent two years (2014–2016); due to the severe economic crisis that Brazil has been facing since 2013, which will reflect in a slowdown of the scientific production in the country for the next years.

A probable cause of this growth seems to be the key role that the activities of biotechnology have gained notoriety on a worldwide scale in recent decades. In Brazil, this increase observed reflects the research government incentives, established the financial investments in the area and the implementation of government policies, as aforementioned.

With respect to the most frequently used terms to screen the biotechnology area in Brazil, it was possible to find a large dispersion in relation to them [2], which means that there are more than 7000 distinct descriptors (keywords) described by the authors in the scientific publications as examples: Vaccine, PCR, DNA, Genome, Leishmania, *Trypanosoma cruzi*, HIV, among others. Some of them were not used as initial descriptors when searching the database.

In the specific evaluation of these terms, we noted that terms such as DNA and PCR are often used since the beginning of the period analyzed, which means that since 1998, these are being associated with modern biotechnology techniques [2]. Other terms such as "genomic," (72 articles) "proteomics," (32 articles), and "stem cell" (37 articles) are further frequent, however, they are related to more advanced future-bearing technologies of modern biotechnology, being mentioned in Brazil by the National Biotechnology Committee as the frontier areas of biotechnology. This observed result is associated with the Brazil collaboration in the genome projects, as example the mapping of the *Xylella fastidiosa* genome in 1997, and later, working in international project such as Human Genome Project (HGP) in 1999. *X. fastidiosa* is a bacterium that attacks citrus agriculture producing "little yellow" and decreasing agricultural productivity, so this project was essential for the control of this pest in Brazilian farming. This project was much important because it was the first phytopathogen sequenced in the world.

According to Santana et al. [2] and Pisano [6], some areas of biotechnology will have a significant impact in the near future on the development of new technologies and its applications, which are the genomics, pharmacogenomics, transcriptomics, biomaterials, bioengineering, and synthetic biology. Some themes related to these terms were detected in this dataset. **Table 2** presents the terms mentioned by the authors (keywords' author) with more than 50 citations related to the field of biotechnology.

From the total of scientific publications found, it can be observed the terms like *Leishmania*, *T. cruzi*, *Schistosoma mansoni*, and Chagas disease, which refer to a potential application. It should be emphasized that the data collected point out that scientific research on biotechnology in Brazil is directly related to the study of some relevant social problems of a country mainly related to tropical diseases.

Neglected tropical disease (NTD) has become an extremely important issue in public health in Brazil, as they profoundly affect the quality of life and generate negative socioeconomic impacts for the population of the poorest countries. Although not unique to developing countries, they arouse little financial appeal from the large pharmaceutical industry, since they do not reach the large consumer market that is the developed countries. In Brazil, the

Articles	Top terms (keywords)	Articles	Top terms (keywords)	Articles	Top terms (keywords)
380	Vaccines	99	Cell	71	Molecular marker
324	PCR	83	Diagnosis	69	gene
232	DNA	82	Expressed gene	65	HIV
183	Genetic	85	Genotype	64	Microsatellite
134	Brazil	79	Trypanosoma cruzi	66	Chromosome
120	Genome	78	Oxidative stress	61	Apoptosis
115	Polymorphic	75	Cytokine	63	Bovine diseases
131	Leishmania	72	Genomics	57	Cancer
56	Schistosoma mansoni	56	Inflammation	54	Fish
54	Chagas disease	53	Epidemiology	50	Drugs

Table 2. Terms with more than 50 citations used by authors in the fields title, abstract and keywords in the articles published by Brazil.

Ministries of Health, Science and Technology and the Health Surveillance Secretariat defined seven neglected tropical diseases based on epidemiological criteria, disease impact, and demographic data. They are dengue, Chagas disease, leishmaniasis, malaria, schistosomiasis, leprosy, and tuberculosis [24].

Figure 2 presents a network map where the main lines of research stand out. This map representation was obtained using VOSviewer®, taking as the matrix of cooccurrence of the 7000 keywords' and authors standardized by the measure of strength of association [21].

Although the keywords, which represent the domain of cells and genome, show a highly interrelated distribution, the structure can be clearly seen. Analysis of the relations among the most frequent descriptors reveals four well-defined groups, but with a variable degree of dispersion. In the upper part (1), around the nodes infection and vaccine, we find the most compact group of the network. It includes descriptors that represent documents in the areas of immunology, infectious diseases, and tropical medicine, related fundamentally with the application and clinical research into tropical diseases.

This scenario is compounded by the lack of innovation in drug R&D programs in the area of NTD; thus, the situation requires a concentrated global for the creation and maintenance of R&D programs focused on the discovery of new alternative therapies for the control and treatment of these diseases [25, 26]. Modern biotechnological tools (e.g., genomics, functional genomics, proteomics, metabolomics, and cytometry) have provided valuable insights for the discovery and development of new drugs that are extremely useful in coping with these NTDs. Extremely important initiatives are being successfully implemented to include Brazil in an increasingly significant science and technology scenario. According to Guido et al. [25, 26], three examples are presented to illustrate the breadth and diversity of networks and partnerships that have provided great opportunities and challenges in the area of NTD.



Figure 2. Map of keywords based on a co-occurrence network on biotechnology in Brazil. *Source:* Map presentation software VOSviewer®.

To the right, cluster 2, the keywords genome, PCR, and DNA connect with a well-defined cluster of descriptors related to genetics, and molecular biology research related fundamentally with the application and clinical research into the genome of *X. fastidiosa* begun in the 1990s.

Genomics is also closely related to agriculture not only in genetic improvement of species such as transgenic crops (with resistance to pests and tolerance to pesticides), but also in the product quality changes (plants that produce hormone, eucalyptus with higher production cellulose). More recently, Brazil has been developing research on feasibility of plants, animals, and microorganisms as biofactories of molecules of agricultural, pharmaceutical and industrial interest; identification and applications of genes and biological functions that promote tolerance to abiotic and biotic stresses and elimination of contaminants in food; identification and characterization (structural and functional) of new molecules to increase the production capacity of biologically based products with low environmental impact; and identification and applications of genes and biological functions that promote tolerance to abiotic and biological functions that promote tolerance to abiotic and biological functions in food; identification and applications of genes and biological functions that promote tolerance to abiotic and biological functions that promote tolerance to abiotic and biological functions in food [27].

Meanwhile, the terms of biomass, diversity, and conversation compose the third cluster 3, far right, including areas of biotechnology and biodiversity. To the left, cluster 4, we have the keywords (cell, gene expression, and stem cell) associated with molecular biology and cell biology areas related to the processes of modern biotechnology associated with a medicine, oncology, and neuroscience therapies.

The information obtained through keywords analysis is much more comprehensive and precise. For this reason, even though the two largely coincide or overlap, subject categories (see **Figure 3**) are more general or superficial than the information based on keywords (**Figure 2**). The latter shows, with a greater level of desegregation, the distribution of the descriptors that specifically configure each thematic profile, plus the less productive or incipient research that would otherwise remain hidden [28].

4.2. Profile of the scientific sector: biotechnology

By its intrinsic multidisciplinary characteristics, biotechnology permeates many areas of knowledge. Therefore, when analyzing the frequency of publication of scientific areas, there is a predominance of biochemistry and molecular Biology (12.4%), followed by genetics and heredity (10.94%), immunology (8.66%), microbiology (7.57%), and veterinary science (6.22%) together representing 45.79% of total articles published. However, this study highlights the wide dispersion of scientific publications by all thematic areas. **Figure 3** demonstrates the percentage distribution for all the 20 areas with the number of articles greater than 200 indexed in ISI/Web of Science.



Figure 3. Percentage distribution of scientific publications on biotechnology by areas of knowledge indexed on database. Period 1995–2015.

It can be observed that most of the articles found are in the areas like life sciences and health, highlighting specific areas such as molecular biology, genetics, immunology, and microbiology. The predominance of these areas was already expected due to the characteristics of biotechnology, but it is interesting to note that areas such as molecular biology and genetics have a significant interest toward other traditional areas such as chemical and pharmaceutical industries.

Note, also, that a considerable number of articles are classified as multidisciplinary, confirming that research in biotechnology, as mentioned previously, brings together researchers from different areas of knowledge. This, probably, highlights the increasingly important role of biotechnological techniques for the development of new products.

Of the 73,125 articles published in internationally indexed journals it was possible to realize a trend of publication in two main areas: health and life science. It is worth mentioning that both thematic areas coincide with the prevailing keywords previously identified. Besides this fact, an interesting aspect observes the concerns of the regional coverage of the top 20 journals, which means that the majority of Brazilians scientific articles are submitted in Brazilian journals (60%), or particularly in indexed American journals (30%). This indicator illustrates the little dynamics of international cooperation of Brazilian research groups. Among the indexed journals, the Vaccine journal has the highest number of articles (12.65%), followed by other journals such as Memórias Instituto Oswaldo Cruz (10.98%), and genetics and molecular biology (10.58%).

Figure 4 presents the top 20 journals with a number of scientific publications greater than 40 articles. The list of journals is in accordance with classification presented in **Figure 3**.

An interesting fact observed is the leadership position of Vaccine Journal, which justly gives priority, an analysis of information and knowledge about human vaccines (infectious diseases and noninfectious diseases) and veterinary vaccines, molecular biology, immunology, production and manufacturing, regulatory, and legislation aspects. It is followed by others publications in areas such as microbiology, biology, genetics, life sciences, and others.

Considering the authors' affiliation, it is observed that there are 441 institutions, showing a high dispersion among the authors' institutions in Brazil. It is important to highlight the significant number of articles produced by institutions such as the University of São Paulo (USP) representing 32.71%, followed by the Oswaldo Cruz Foundation (FIOCRUZ) with 8.92%, and the Federal University of Rio Grande do Sul (UFRGS) with 7.58% articles, respectively. Another significant topic to be observed is the prevalence of public governmental institutions such as research centers or universities, evidencing a concentration of activities in biotechnology by public institutions or nonprofits corporations at research levels. It is also worth mentioning a few numbers of scientific publications indexed by the Brazilian biotechnology companies in the research period, less than 1%, which indicates a small number of companies working in that area or that they are still consolidating their capacities, according Santana et al. [2]. It is should be highlighted the public institution Brazilian Agricultural Research Corporation (EMBRAPA) Genetic Resources with the 8th institution in the ranking, which is the public company with



Figure 4. Main internationally indexed journals with biotechnology-related articles.

leadership in transgenic research in the agricultural area in the country. **Figure 5** shows the top 20 institutions that have published more than 100 articles in the observed period.

The research results present that the most of institutions are located in South/Southeast regions, which are more developed regions of the country. There is little representation of institutions in the Center-West/ Northwest regions, presented only for 4 universities: University of Brasília, Federal University of Pernambuco, University of Goiás and Federal University of Ceará.

In addition, there is the significant presence of the public universities of São Paulo (USP, UNESP, and UNICAMP), which together account for a significant share of national scientific output in the biotechnology area, that is, 46.31% of published articles in journals are from USP. This observed result was expected because since the beginning of projects development in the biotechnology area (*X. fastidiosa* genome and HGP/Brazil), there was an intensive participation of research groups of these universities. Furthermore, we can mention the continuous financial support from FAPESP foundation for the development of projects in the areas such biotechnology and bioprospection.

4.3. Maps of knowledge and network collaboration

According to Santana et al. [2], regarding the analysis of relationships between composing agents of the National Organizational System of S,T&I, many authors refer to the central idea



Figure 5. Leading Brazilian institutions with major publications in biotechnology for the period 1995–2015.

of networking and the valuable interactions between system components and its main actors in order to promote the dissemination and use of new scientific knowledge [29].

In accordance with these premises, we sought to analyze the links established between the organizations by observing how these actors relate to each other using the number of publications of scientific cooperation between institutions and countries; on a macro level, the international relationships developed by these institutions for R&D. The best representation of these collaborations is the visualization of maps of knowledge, where the existence or not of correlations and the degree of its intensity are clear, which provide to decision-makers some strategic subsidies in future planning of national activities of ST&I [2].

Considering the network map of articles' authorship in Brazil, for the biotechnology area, it is possible to verify an intensive cooperation between diverse institutions, especially among Brazilian institutions, showing that 66.9% of articles are written with internal collaboration, which subsidize the relevance to develop endogenous capacity of Brazilian groups and publish papers with little support from international cooperation [2].

It is possible to note groups with strong collaboration such as the University of São Paulo (USP), Federal University of Rio de Janeiro (UFRJ), the University of Campinas (UNICAMP), and Oswaldo Cruz Foundation (FIOCRUZ). These collaborations between Brazilian institutions demonstrate that relationship is based more heavily on the model of interaction between universities, research centers, and nongovernmental research centers and of little interaction with companies. It should be emphasized that this scenario will modify since the implementation of the Organic Law of Science, Technology and Innovation in January 2016, where the private sector was encouraged to invest more in partnerships with public institutions and/or in internal activities of ST&I, aiming to promote alliances between the private sector with scientific research center and universities [2, 29]. **Figure 6** shows the collaboration network among institutions with more than 60 articles published.

From a macro perspective, there is a large international collaboration, particularly with five countries: the United States (12.81%), the United Kingdom (2.27%), France (2.22%), Germany (1.43%), and Spain (1.30%). As seen previously, when focusing on Latin America, the block represents only 1.89% of all countries' collaboration, it is possible to identify that Brazil has networks of scientific collaboration and research with almost all countries. Inside this block, Argentina is a major coauthored number of articles (41.94%), followed by Colombia (19.47%), Chile (12.17%), Uruguay (9.92%), and Venezuela (8.80%). **Figure 7** shows the map collaboration network among 40 major countries and Brazil in the biotechnology area.



Figure 6. Map of institutional relations on biotechnology for Brazil. Source: Map presentation software VOSviewer®.


Figure 7. Map of network based on a co-authorship between Brazil and the 40 major countries on the biotechnology area. *Source*: Map presentation software VOSviewer[®].

5. Conclusions

Traditionally, biotechnology is a technology that is strongly dependent on the studies of the basic research area. Its great success in any country is strongly related to government policies regarding the incentive of science and its technological diffusion. This chapter presents the growing of scientific research on biotechnology in Brazil, over the number of scientific articles published in the area, showing a higher growth of 1.930%, in the last 20 years, more specifically in the last 5 years. With respect to searched terms, "Vaccine," "PCR," and "DNA" are among the terms most frequently applied, being cited, since 1995, by the authors as keywords in scientific journals as well as in relevant journals such as Vaccine and Mem Inst Oswaldo Cruz. Terms such as proteomics and stem cells, related to the frontier area of knowledge, have also appeared in this scientific research. This fact indicates that the research in this area in Brazil is consistent with worldwide trends. Another significant topic is that the biotechnology research in Brazil is very important for studying the country's social problems, especially related to tropical diseases, which presents itself as an important starting point for formulating policies on ST&I, since decision-makers should encourage links between scientific institutions and companies. It is important to point out that Brazilian agricultural biotechnology research has taken significant steps toward the development and use of innovations for sustainable production systems that provide safer food (biofortified foods with vitamins, minerals, and better quality proteins).

A wide dispersion in relation to the thematic areas of biotechnology is observed, however, concentrated mainly in areas such as biochemistry and molecular biology and genetics and heredity. Three institutions are highlighted here: São Paulo University (USP), Oswaldo Cruz Foundation (FIOCRUZ), and Federal University of Rio Grande do Sul (UFRGS). It is worth mentioning that institutions that carry out scientific research in this area are primarily governmental. Regarding the analysis of relationships between agents that compose the National System of Science, Technology and Innovation in Brazil, it was found that for biotechnology, there is a large network of cooperation among international and national institutions, as well as networking among many countries that present a positive factor in the biotechnological development of a country [2]. In conclusion, it was possible to identify the United States as a major coauthor of scientific publications relating the subject, but there are also partnerships with other Latin American countries such as Colombia and Argentina.

Acknowledgements

Parts of this chapter are reproduced from the authors' previous publications [30].

Author details

Maria de Fátima Ebole Santana

Address all correspondence to: mfebole@gmail.com

Federal University of Rio de Janeiro/Chemical School, Rio de Janeiro, Brazil

References

- [1] Antunes AMS, Canongia C, Bahrut E, Rodrigues HT, Pio M, Gianinne R. Prospección Tecnologica - Gestión del Conocimiento e Inteligencia Competitiva - Modelos de Gestión para la toma de decisiones y cosntrucción de futuro. In: Vásquez JM; Torres JMS, editors. Sinergía entre la prospectiva tecnológica y la vigilancia tecnológica e inteligencia competitiva. 1st ed. Colombia: Bogotá, Colciencias, 2010. p. 49-83
- [2] Santana MFE, Martínez RG, Pereira N Jr, Antunes AMS. Knowledge management and analysis of scientific biotechnology trends in Venezuela. Journal of Technology Management and Innovation. 2012;7(1):145-158. DOI: 10.4067/S0718-27242012000100010
- [3] Gertler MS, Yael ML. Local nodes in global networks: The geography of knowledge flows in biotechnology innovation. Industry and Innovation. 2005;**12**:487-507 Available from: http://digitalcommons.ilr.cornell.edu/articles/122
- [4] Ministério do Desenvolvimento, Indústria e Comércio Exterior. Diretrizes de política industrial, tecnológica e de comércio exterior. Brasília. 2003. Available from: http:// www.anped11.uerj.br/diretrizes.pdf [Accessed: April 20, 2004]

- [5] Massambani O, Campiglia SS. Convention on Biological Diversity. Conferência das Nações Unidas sobre o Meio Ambiente e Desenvolvimento -Rio-92. Revista Estudos Avançados. 1992;6(15):193-223. [Available from: http://www.scielo.br/pdf/ea/v6n15/ v6n15a15.pdf]. [Accessed: 2017-07-10]
- [6] Pisano G. editor. Science Business: The promise, the reality and the future of Biotech. 1st ed. Harvard Business School Press. United State: Massachusetts, 2006. 237 p. ISBN-13: 978-1591398400
- [7] Krattiger AF. Public-private partnerships for efficient proprietary biotech management and transfer, and increase private sector investments. In: A Briefing Paper with Six Proposals Commissioned by UNIDO. IP Strategy Today. Vol. 4. 2002. p. 42. Available from: www.biodevelopments.org/ip/index.htm [Accessed: December 15, 2012]
- [8] Torres OD, Velho L. Capacidades Científicas y Tecnológicas de Colombia para adelantar Prácticas de Bioprospección. Revista CTS. 2009;4(12):55-68
- [9] Silveira J, Borges I. Um panorama da biotecnologia moderna. In: Silveira J, Poz M, Assad A. editors. Biotecnologia e recursos genéticos: desafios e oportunidades para o Brasil. 1st ed. Unicamp, Instituto de Economia; 2004. p. 412. ISBN: 8587120050 9788587120052
- [10] Fortuin FTJM. Aligning innovation to business strategy: Combining cross-industry and longitudinal perspectives on strategic alignment in leading technology-based companies [thesis]. Waginingen University: Waginingen University and Research Center; 2006
- [11] Baker S, Aston, A. The Business of Nanotech. Bloomberg Business Week. 2005;2: p14-20
- [12] Esposito RS, Ostro MJ. Where have all the investors gone? The case for consolidation. Nature Biotechnology. Supplement: Bioentrepreneurship – Building a biotechnology company from the ground up. 1998;16:63. Available from: http://nature.com/cgitaf [Accessed: August 10, 2004]
- [13] Mascarenhas P, editor. Parque Nacional de Empresas de Biotecnologia. 1ª ed. Fundação Biominas. Brasil: Belo Horizonte, 2001. 250p. Available from: http://biominas.org.br/wpcontent/uploads/2015/06/Parque-Nacional-de-Empresas-de-Biotecnologia.pdf
- [14] Biotechnology Industry Organization. Bio Editor's and Reports' Guide to Biotechnology [Internet]. 2007. Available from: http://www.bio.org [Accessed: October 1, 2012]
- [15] ABDI and CGEE. Biotecnologia Iniciativa Nacional de Inovação. Panorama da Biotecnologia no Brasil e no Mundo [Internet]. 2008. Available from: http://www.abdi.com. br/Estudo/Panorama%20Setorial%20Biotecnologia.pdf [Accessed: November 7, 2016]
- [16] Ernest & Young LLP. Beyond Borders Global Biotechnology Report 2012 Internet. 2012. pp. 1-116. Available from: https://www.eycom.ch/en/Publications/20120619-Beyond-Borders-2012-Global-biotechnology-report/download [Accessed: July 30, 2016]
- [17] OCDE. OECD, Biotechnology Statistics Database [Internet]. 2011. Available from: http:// www.oecd.org/science/inno/49303992.pdf [Accessed: December 2, 2012]. OCDE. OECD, Main Science and Technology Indicators Database [Internet]. 2011b. Available from: http://www.oecd.org/science/inno/49303992.pdf [Accessed: December 2, 2012]

- [18] Ernst & Young, LLP. Beyond Borders Global Biotechnology Report 2013 [Internet].
 2013. pp. 1-85. Available from: http://www.ey.com/Publication/vwLUAssets/Beyond_ borders/\$FILE/.pdf [Accessed: August 3, 2016]
- [19] Silveira JMFJ, Borges IC, Buainain AM. Biotecnologia e Agricultura da Ciência e Tecnologia aos Impactos da Inovação. São Paulo em Perspectiva. 2005;19(2):101-114. abr./jun. 2005. DOI: 10.1590/S0102-88392005000200009
- [20] Lima RA, Velho LML, Faria LIL. Delimitação de uma área multidisciplinar para análise bibliométrica de produção científica: o caso da Bioprospecção. Revista Transformação. 2007;19(2):153-168. maio/ago, 2007. DOI: 10.1590/S0103-37862007000200006.
- [21] Van Eck NJ, Waltman L. Software survey: VOSviewer, a computer program for bibliometric mapping. Scientometrics. 2010;84(2):523-538. DOI: 10.1007/s11192-009-0146-3
- [22] Van Eck NJ, Waltman L. How to normalize cooccurrence data? An analysis of some wellknown similarity measures. Journal of the American Society for Information Science and Technology. 2009;60(8):1635-1651. DOI: 10.1002/asi.21075
- [23] Santana MFE, Antunes AMS. O Perfil da Biotecnologia no Brasil: Investimentos, Recursos Humanos e a Indústria de Biotecnologia. Antunes, AMS, Pereira-Jr N, Santana MFE, editors. In: Gestão em Biotecnologia. 1st ed. Ed. Brasil: Rio de Janeiro, 2006, p. 17-42. ISBN 85-76500-76-0
- [24] Jesus AR. O Impacto Social Das Doenças Negligenciadas No Brasil e No Mundo. Salvador, 2012. Universidade Federal da Bahia. pp. 1-39. Available from: https://repositorio.ufba.br/ri/bitstream/ri/7983/1/Anderson%20de%20Jesus%20Rocha%20(2012.1).pdf [Accessed: December 15, 2016]
- [25] Guido RV, Andricopulo AD, Oliva G. Drug design, biotechnology and medicinal chemistry: Applications to infectious diseases. Estudos avançados. 2010;24(70):81-98. DOI: 10.1590/S0103-40142010000300006
- [26] Empresa Brasileira de Pesquisa Agropecuária Ministério da Agricultura, Pecuária e Abastecimento. Visão 2014-2034: O Futuro do Desenvolvimento Tecnológico da Agricultura Brasileira – Síntese. 2014. pp. 1-53. Available from: https://www.embrapa.br/ documents/1024963/1658076/O+Futuro+de+Desenvolvimento+Tecnol%C3%B3gico+da+ Agricultura+Brasileira+-+s%C3%ADntese.pdf/ddb0a147-234d-47f1-8965-1959ef82311d [Accessed: December 15, 2016]
- [27] Cantos-Mateos GB, Vargas-Quesada Z, Chinchilla-Rodríguez MAZ. Stem cell research: Bibliometrics analysis of main research areas through KeyWords Plus. Aslib Proceedings. 2012;64(6):561-590. DOI: 10.1108/00012531211281698
- [28] Freeman C, editor. Technology Policy and Economic Performance: Lessons from Japan. 2nd ed. USA: NY, 1987. 155 p. ISBN-13: 978-0861879281
- [29] Presidência da República. Casa Civil. LEI Nº 13.243. De January, 11st 2016. Marco Legal da Ciência, Tecnologia e Inovação no Brasil
- [30] Santana MFE, Pereira N Jr, Antunes AMS, Oliveira MS. Evidences in scientific knowledge management of biotechnology in Brazil. International Journal of Managerial Studies and Research (IJMSR). 2016;4(9):52-60. DOI: 10.20431/2349-0349.0409007

Exploring the Impact of Online Clinical Guidelines on Individual Knowledge Management Behaviors and Individual Net Benefits

Shofang Chang, Tain-Junn Cheng and Chung-Hsien Chan

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/intechopen.70069

Abstract

Health care is an industry of high knowledge intensity; the knowledge and skills of the medical staff are the key advantages for hospital competitiveness. This research aims to explore how the use of online clinical guidelines influences individual KM behaviors and the net benefits. It investigates the users, including 120 physicians and 80 physician assistants (PAs) who used the online clinical guidelines in a medical center of southern Taiwan. A total of 195 of the 200 questionnaires surveyed including 111 physicians and 70 PA are effective, leading to 97.5% effective response rate. This research uses a structural equation model, partial least squares (PLSs) to verify the research model and hypotheses. The R^2 of the overall model is 0.68, which implies good reliability and validity of this model. The results reveal that the use of online clinical guidelines and user satisfactions has positive effects on individual KM behaviors and individual net benefits. Individual KM behaviors have partial effects on the individual net benefits. In addition, there are mediating effects of the individual KM behaviors and user satisfactions on the use of online clinical guidelines and the individual net benefits.

Keywords: online clinical guidelines, individual knowledge management (KM) behavior, partial least squares (PLSs)

1. Introduction

In the era of knowledge economy, knowledge has gradually become one of the most important assets for most enterprises. In the health-care industry, clinical knowledge, especially physicians' knowledge, has been recognized as an important asset and a core competence



© 2017 The Author(s). Licensee InTech. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. area for hospitals. Information technology's rapid transformation and the rise of the Internet have resulted in rapid changes in data and knowledge. The establishment of knowledge portals in a health-care management system could lower health-care costs and foster diagnostic accuracy, thus improving the quality of patient care. Hence, within the knowledge-intensive health-care industry, the management of clinical knowledge is evidently the hospital's most important challenge.

In 1990, the Institute of Medicine proposed the clinical practice guideline as a systematic operational guide to help clinical researchers or patients select suitable treatment strategies when experiencing specific clinical conditions [1]. These clinical guidelines serve to improve the quality of treatment, decrease health-care costs, and reduce medical heterogeneity [2–5].

In this information and cloud-computing age, online clinical guideline may improve the accessibility, reliability, validity, and velocity for searching best practice, thus enhancing the efficiency and effectiveness of medical practice. The implementation of an online clinical guideline system is an important way to promote medical knowledge management, develop knowledge management ability, and improve individual net benefit. Future users of the online clinical guideline system will be able to break through traditional space and time restrictions to enhance medical expertise and technology, change knowledge management behaviors, and improve health-care quality and capability to maintain their core competences.

2. Research objectives

The research objectives were as follows: (1) to explore the interactions among the satisfaction level of medical professionals who use the online clinical guideline system, individual knowledge management behavior, and net benefit; (2) to compare disparities between all aspects of various modes of interactions across different medical professionals; and (3) to provide relevant suggestions regarding the online clinical guideline system's infrastructure and implementation.

3. Literature review

3.1. Online clinical guideline

Isern and Moreno [6] pointed out that an online clinical guideline system benefits clinical physicians and patients but remains an area of ongoing research with insufficient implementation and has yet to be integrated into existing management systems of health-care protocols and used in routine medical infrastructures. Isabelle et al. [7] noted that the integration of clinical guidelines into the online network will solve problems faced by physicians in patient care and will increase the likelihood that these guidelines will be used by physicians.

Isern and Moreno [6] also proposed that the compilation of the clinical guidelines database should include a web-based version. Moreover, collaborative features are very helpful in

improving and creating various systems of clinical guidelines for different medical professionals. Studies have stated that, when computerized, a clinical decision support systems combine decisive strategies and clinical protocols and can provide concrete support for patients, improving clinical practice guideline compliance and treatment results for patients [8].

3.2. Individual knowledge management behavior

Nonaka and Takeuchi [9] argued that only individuals can create knowledge, and, through interactions with one another, individuals can develop organizational knowledge. Knowledge internalization and socialization should be the major knowledge management behavior of individuals. Rogers [10] categorized employees' knowledge behavioral conducts into three types regarding measurement of various aspects of cooperative knowledge behavior between employees, including knowledge acquisition, knowledge creating, knowledge sharing, knowledge hoarding, knowledge shirking, and knowledge appropriation. This study used knowledge socialization and internalization as the measuring variables of individual knowledge management behaviors.

3.3. Information system success model

Many studies use academics Delone and Mclean's [11] Information System (IS) success model for information system importing and effectiveness evaluation. This study focused on evaluating the effectiveness of an online clinical guideline system. The variables include system use, user satisfaction, user intention, and net benefits. Thus, we chose Delone and Mclean IS success model to evaluate the system, since they [11] have divided the actual system use and use intention. This study used two variables to measure the construct of system use.

4. Research methodology

4.1. Research hypothesis

This study's main focus was to investigate the satisfaction and usage levels associated with the online clinical diagnosis and treatment guideline system and the relationship between medical professional employees' knowledge management behavior and net benefit and to further explore relationships between various dimensions.

This study presented the following hypotheses:

H1: An online clinical guideline system use has a significant positive impact on its user satisfaction.

H2: An online clinical guideline system use has a significant positive impact on individual knowledge management behavior.

H3: An online clinical guideline system use has a significant positive impact on individual net benefit.

H4: An online clinical guideline system's user satisfaction has a significant positive impact on individual knowledge management behavior.

H5: An online clinical guideline system's user satisfaction has a significant positive impact on individual net benefit.

H6: Individual knowledge management behavior has a significant positive impact on individual net benefit.

4.2. Research design

This study investigated a medical center in Tainan, primarily focusing on some of the center's resident physicians, part-time physicians, practicing physicians, medical interns, and specialist assistants, and conducted questionnaire-based surveys and investigation. It investigates the users, including 120 physicians and 80 physician assistants (PAs) who used the online clinical guidelines in a medical center of southern Taiwan. A total of 195 of the 200 questionnaires surveyed including 111 physicians and 70 PA are effective, leading to 97.5% effective response rate.

5. Research results

5.1. Model reliability and validity test

In every dimension of the physicians and specialist assistant category, advertising value equivalency (AVE) values were all greater than the standard value of 0.5; CR values in all cases were greater than the standard value of 0.7. The overall model regression analysis R^2 -value was 0.68, and Cronbach's α falls above 0.84, exceeding Fornell and Larcker's [12] recommended high reliability index value (>0.7). This suggests that this study's structural model has good reliability and validity (**Table 1**).

	AVE	CR	R^2	Cronbach's α
User intention	0.81	0.92		0.88
Actual system use	0.68	0.89		0.84
User satisfaction	0.80	0.96		0.95
Knowledge socialization	0.64	0.91		0.89
Knowledge internalization	0.69	0.91		0.88
Individual net benefit	0.76	0.96	0.68	0.96

Table 1. Reliability and validity.

5.2. Path analysis

This study utilized partial least squares (PLSs) to evaluate the study. Using covariance structure as a basis, structural equation modeling (SEM) recommended that sample size must be at least 200 or above or 10 times the parameter [13]. Second, using covariance structure as a basis, SEM's predicted value must comply with the norm; otherwise, a disassociation with reality effect will ensue [14, 15]. In addition, a covariance-based SEM can only process reflective indicators and is unable to process formative indicators. As this study was conducted with a sample size of 195, which did not meet the standard 10 times of the parameter, PLS was selected as the method for analysis (**Figure 1, Table 2**).

This study found that online clinical guideline system usage had a significant positive impact on the system's user satisfaction ($\beta = 0.228$, p < 0.05). Additionally, regarding the relationship between individual knowledge management behavior and individual net benefit, only a small portion had a significant positive impact. Knowledge internalization has a significant impact on individual net benefit ($\beta = 0.183$, p < 0.05), while knowledge socialization reveals no impact on individual net benefit ($\beta = 0.05$, p > 0.1). The online clinical guideline system's user satisfaction had a significant positive impact on both individual knowledge management behavior (user satisfaction→knowledge socialization, $\beta = 0.191$, p < 0.1; user satisfaction→knowledge internalization, $\beta = 0.306$, p < 0.05) and individual net benefit ($\beta = 0.549$, p < 0.05), and individual knowledge management behavior had a partial significant positive impact on individual net benefit (knowledge socialization→individual net benefit, $\beta = 0.05$, p > 0.1; knowledge internalization→individual net benefit, $\beta = 0.183$, p < 0.05).



Figure 1. PLS model.

	Path coefficient	t-Value	
User intention \rightarrow user satisfaction	0.526	7.45**	
Actual system use \rightarrow user satisfaction	0.228	3.06**	
User intention→knowledge socialization	0.192	2.21*	
Actual system use →knowledge socialization	0.185	2.29*	
User intention→knowledge internalization	0.205	2.35*	
Actual system use→knowledge internalization	0.008	0.11	
User intention→individual net benefit	0.068	1.2	
Actual system use→individual net benefit	0.149	2.90**	
User satisfaction→knowledge socialization	0.191	1.80*	
User satisfaction→knowledge internalization	0.306	3.48**	
User satisfaction→individual net benefit	0.549	8.27**	
Knowledge socialization→individual net benefit	0.050	0.83	
Knowledge internalization→individual net benefit	0.183	2.83**	

Sample size: 500 **p* < 0.05; ***p* < 0.01.

Table 2. Structural method's path coefficient testing result.

Thus, hypotheses H1, H4, and H5 are supported. However, system use did not significantly influence knowledge internalization, user intention did not significantly influence individual net benefits, and individual knowledge socialization did not have a significant impact on individual net benefit. As a result, hypotheses H2, H3, and H6 are partially supported.

6. Conclusions and discussions

The results of this study showed that medical professionals had a certain tendency toward supportive systems that help their own fields and concurred that this type of system will always increase usage frequency but will also create dependence on the system to assist their medical strategies. It is evident, then, that medical professionals view usage of the online clinical guideline system as important to provide support and guidance for health-care work, which corresponds the viewpoints of Isern and Moreno [6] and Dongwen et al. [8]. On the other hand, with regard to medical decisions, they will still depend on their own professional knowledge.

This study also found that medical professionals have a certain degree of satisfaction with the online clinical guideline system. Online clinical guidance system's user satisfaction with the user's conscious satisfaction has successfully increased the functionality and practicality of using the system and indirectly assisted professional health-care employees to improve individual work performance. Simultaneously, users' satisfaction with the system is also an important factor determining continual usage of the system.

Regarding individual knowledge management behavior, this study developed two dimensions, "Knowledge Socialization" and "Knowledge Internalization," measuring personal knowledge management behaviors. Analysis of the results indicated that medical professionals expressed a higher degree of individual knowledge internalization related to socialization. When agreeing with individual knowledge internalization through "browsing of related professional websites," "obtaining expert knowledge," "educational training," "working," "observation," and related methods, individual knowledge can be increased. In individual knowledge socialization, through methods such as "internal and external department meetings," "sharing of ideas with colleagues," "team exchange of ideas," "discussion," and "seminars," a knowledge socialization effect can be achieved.

For individual net benefit, this study showed that medical professionals view the online clinical guideline system's usage to have a considerable impact on individuals' work and decision-making performance; notably, they expressed, "Using online clinical guideline system allowed for the quality of health-care strategies to improve," and "An online clinical guideline system can increase my work efficiency." From this, it is evident that the usage of the online clinical guideline system has a positive effect on medical professionals' work performance.

Based on this study's results, the usage of the online clinical guideline system has a positive impact on individual knowledge management behavior and individual net benefit; moreover, individual knowledge management behavior has a positive impact on individual net benefit. In addition, the online clinical guideline system's satisfaction level has an impact on individual net benefit that is greater than that of the online guidance clinical system or individual knowledge management behavior.

This study uses PLS and constructs a full model of online clinical guideline behaviors. Besides, the integration of the clinical guidance system into the Internet allowed us to know medical professionals' level of acceptance of the system by measuring their degree of satisfaction with using the system. This subsequently proved that medical professionals' acceptance level of the online clinical guideline system was recognized. Additionally, medical professionals, through the system's practical operation and usage, discovered that employees strongly value the practicality and reliability of data systems.

This study utilized a survey method focusing on only one medical center to investigate online clinical behavior and individual net benefit. Future research would be suggested to conduct the survey in multiple hospitals so as to improve the generalization of the research. In addition, due to the time restriction, this study fails to further explore the impact of online clinical guideline on patient satisfaction and health-care quality. Future research is suggested to extend the effect on patient level so as to build a more comprehensive model toward the effectiveness of online clinical guidelines.

Author details

Shofang Chang^{1*}, Tain-Junn Cheng² and Chung-Hsien Chan³

*Address all correspondence to: ctpaul@ms10.hinet.net

1 Department of Hospital and Health Care Administration, Chia-Nan University of Pharmacy & Science, Tainan, Taiwan, ROC

2 Department of Medical Information Management, Chi Mei Medical Center, Tainan, Taiwan, ROC

3 Continuing and Extension Education Division, National Pingtung University, Pingtung, Taiwan, ROC

References

- Field MJ, Lohr KN, editors. Clinical Practice Guidelines: Directions for a New Program. Washington, DC: Institute of Medicine, National Academy Press; 1990
- [2] Grilli R, Penna A, Zola P, Liberati A. Physicians' view of practice guidelines. A survey of Italian physicians. Social Sciences & Medicine. 1996;**43**:1283-1287
- [3] Warner BW, Kulick RM, Stoops MM, Mehta S, Stephan M, Kotagal UR. An evidencebased clinical pathway for acute appendicitis decreases hospital duration and cost. Journal of Pediatric Surgery. 1998;33:1371-1375
- [4] Carnett WG. Clinical practice guidelines: A tool to improve care. Quality Management for Health Care. 1999;8:13-21
- [5] Kohn LT, Corrigan JM, Donaldson MS. To Err is Human: Building a Safer Health System. Washington, DC: Institute of Medicine, National Academy Press; 1999
- [6] Isern D, Moreno A. Computer-based execution of clinical guidelines: A review. International Journal of Medical Informatics. 2008;77:787-808
- [7] Isabelle C, Angel R. Aguirre J, Sylvain Z, Marie CJ, Laurence L, Gilles C. Electronic implementation of guidelines in the EsPeR system: A knowledge specification method. International Journal of Medical Informatics. 2005;74:597-604

- [8] Dongwen W, Mor P, Samson WTu, Boxwala AA, Greenes RA, Patel VL, Shortliffe EH. Representation primitives, process models and patient data in computer-interpretable clinical practice guidelines: A literature review of guideline representation models. International Journal of Medical Informatics. 2002;68:59-70
- [9] Nonaka I, Takeuchi H. The Knowledge Creating Company: How Japanese Companies Create the Dynamics of Innovation. New York, NY: Oxford University Press; 1995
- [10] Rogers EW. Cooperative knowledge behavior in high tech organizations: Examining the relationship between employee perceptions of the employment game, cooperative knowledge behavior and firm performance [PhD dissertation]. Cornell University; 2000 (unpublished)
- [11] DeLone WH, McLean ER. The DeLone and McLean model of information systems success: A ten-year update. Journal of Management Information Systems. 2003;19(4):9-30
- [12] Fornell C, Larcker DF. Evaluating structural equation models with unobservable variables and measurement error. Journal of Marketing Research. 1981;**18**(3):39-50
- [13] Hoelter JW. The effects of role evaluation and commitment on identity salience. Social Psychology Quarterly. 1983;46:140-147
- [14] Norman GR, Streiner DL. PDQ Statistics. 3rd ed. BC Decker Inc.; 2003
- [15] Fornell C, Bookstein FL. Two structural equation models: Lisrel and PLS applied to customer exit-voice theory. Journal of Marketing Research. 1982;19:440-452

Knowledge-Based Assignment Model for Allocation of Employees in Engineering-to-Order Production

Matjaz Roblek, Maja Zajec and Benjamin Urh

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/intechopen.70073

Abstract

In today's rapidly changing business environment, it is necessary to react promptly in response to the product changes that happen constantly in an Engineering-to-Order production environment. Very often, there is not sufficient time to educate employees regarding new and necessary knowledge. If we insist on the standardization of a process execution, the process always requires appropriate knowledge from among available employees. In this chapter, an option for adjusting processes to available knowledge is studied. Following calculations, it was concluded that a partial corruption of a perfect process leads to a better knowledge alignment of employees. At first, with the corruption of a perfect process, its efficiency is decreased, but with better knowledge alignment, process efficiency is consequently increased to a level better than the original one. The optimization model presented in this chapter is based on a modified classic assignment problem and it includes a numerical example based on the data of ETO company. We proved our findings from the aspects of balance, employee capacity load and process efficiency.

Keywords: knowledge allocation, optimization model, activity-cutting principle, ETO

1. Introduction

Global competitiveness requires constant innovations of products and processes, which inherently require changes on the part of production companies. Management of these changes is especially important for those companies for which the production of new products is a regular business, that is, for which every customer requirement is so unique that it requires for the integration of research and development (R&D) department employees to a certain level. Linking of sales, R&D and production in such way is called an 'Engineering-to-Order



production strategy' (ETO). Products in ETO production have a complex structure and a customer-specified production that is treated as a project. These projects are generally unique and were never previously executed. Therefore, it is impossible that they be handled with existing standard project activities. Problems with the allocation of employees appear in the first activities of the ETO production project, in which activities require a high level of innovation, and the project requires a proper knowledge allocation prior to capacity allocation. Of course, the management needs both allocation views, but the knowledge aspect is more important when dealing with new product or technology changes. The typical question before executing each ETO project is: Do we have appropriate knowledge to do that?

Knowledge is an element of the employees and also an element of the activities of business processes [1]. In Make-to-Stock (MTS), production activities are highly specialized and require a small set of required knowledge. In ETO production, employees execute many activities with a large set of required knowledge. Due to salary requirements, the human-resource-required knowledge is linked to the work position definitions [2]. The management goal is to optimize the required knowledge of work positions and the current knowledge of employ-ees. With every product or process change, the knowledge structure of the work position is changed. If changes are permanent, there will be a continuous searching for new appropriate employees. However, what if the process of change was adjusted so that it took into consideration currently available knowledge? These employees are the only source that is available at the time a new product requires new knowledge in the process. What if the capacity load of each employee's knowledge and not just the employee's capacity in general were taken into consideration?

2. Literature review

In literature, this kind of optimization problem is classified as the worker assignment problem [3]. Applications of this problem are matching employees on work positions, where the required knowledge of work positions is compared to the actual knowledge of known employees [4]. The optimal solution (objective function) depends on the global minimum of the current knowledge deficit or the global maximum of the current knowledge surplus.

In a real environment, production processes are complicated and diverse. Almost every product and its production technology require modification of its objective function or modification of the entire optimization problem. Even if there is production of the same product in different locations, there will be modification needs, despite work standardization efforts. During process execution (over several years), the optimization problem also changes because of expected and unexpected events, such as production errors, economic opportunities and new arrangements. These events are sometimes very important for optimization. In the case of the presence of a more important and/or urgent business event, their importance for optimization disappears, and their priorities for optimization are changed. Therefore, there are many specific solutions for the worker assignment problem in the literature. Some solutions are case specific while other are made in an attempt to be universally applicable. Depending on the complexity of the worker assignment problem, researchers implement different optimization methods: mathematic programming models (linear, non-linear, integer), genetic algorithms and heuristics.

The following research has been used as a background for the worker assignment problem in this chapter. From the perspective of tasks, Azizi and Liang [5] developed an integrated approach to the worker assignment problem. Their dominant assignment problem includes workforce flexibility acquisition and task rotation. They used a constructive-search heuristic method and set the objective to minimizing the total cost including the incremental cost of new training cost, flexibility cost and productivity loss cost. The learning effect in the worker assignment model was also the subject of research in a project task scheduling problem [6]. They used a mixed non-linear integer program, solved by a proposed genetic algorithm. The objective function was to minimize outsourcing costs. From the task perspective, there is optimization model of task allocation and knowledge worker scheduling [7]. The purpose of this model is to assign knowledge workers to every task and arrange them (the tasks) in order to minimize the total time required to finish all projects. Their optimization is based on the Ant Colony algorithm as an optimization technique [8]. Nembhard [9] uses a heuristic approach for assigning workers to tasks that is based on individual learning rates.

There are also worker assignment models originating in production layout and shifts. McDonald et al. [10] developed a worker assignment model to evaluate a lean manufacturing cell, using a binary integer programming model that is solved using a branch-and-bound approach. The objective of this model is to minimize net present costs (initial training costs, incremental training costs, inventory costs and cost of poor quality). Previously, a model of worker assignment considering technical and human skills in cellular manufacturing was developed [11]. It is classified as mixed-integer programming problem. The objective of the model is to maximize profit, where profit has three components: productivity, quality costs and training costs. Ingolfsson et al. [12] combined integer programming heuristic to generate schedules; they used the randomization method to compute service levels. They described a method to find low cost shift schedules with a time-varying service level that is always above a specified minimum.

There are worker-assigning models that deal with the satisfaction of workers. Brusco and Johns [13] defined a model of staffing a multi-skilled workforce with varying levels of productivity. They applied integer linear programming model with the objective of minimizing workforce staffing costs subject to the satisfaction of minimum labour requirements across the planning horizon of a single work shift. Mohan [14] created a model of scheduling part-time personnel with availability restrictions and preferences to maximize employee satisfaction. He proposed an integer programming model to maximize employee satisfaction (while considering their seniority and availability) and to meet the demand requirements for each shift. A branch-and-bound algorithm was used for this.

From the perspective of competencies [15], there is a competence-driven staff assignment approach that is based on a stochastic working status model. This model seeks to minimize

employee wages and maximize strategic gains of the company from the increment of desirable competencies. The authors used a genetic algorithm as the optimization method. Competencies are also used in a model that seeks to maximize a weighted average of economic gains from projects and strategic gains from the increment of desirable competencies. As a sub-problem, the scheduling and staff assignment for a candidate set of selected projects is also optimized [16]. The authors used non-linear mixed-integer program formulation for the overall problem and then proposed heuristic solution techniques composed of a greedy heuristic for the scheduling and staff assignment, and alternative 'meta' heuristics for the project selection.

Recent studies are showing that the worker assignment problem is still important subject of research. Grosse et al. [17] designed a framework for integrating human factors into planning models. Crawford et al. [18] showed application of worker assignment problem in project scheduling and they innovated optimization approach using hyper-cube framework. A similar problem that discuses assignment of health care staff to tasks using fuzzy evaluation method was presented by Mutingi et al. [19]. Olivella et al. [20] gave emphasis on the cross-training goals, while Senjuti et al. [21] optimized the assignment of tasks to workers by proposing efficient adaptive algorithms. Current efforts are dealing with additional variables in creating the perfect optimization framework (knowledge, cross-training, etc.), or in finding the best optimization algorithms for solving worker assignment problem. They still assume that tasks are allocated to workers as 'they are'. Our effort was to study the effect of task redefinition in the meaning of splitting tasks on smaller parts with the goal of better knowledge alignment. From the organizational view, especially when the creative job must be done (like in ETO companies), the list of required tasks is created according to the available knowledge of workers, and the new definition of tasks is a subject of optimization output. This was our main theoretical issue that is described as real business example as follows:

- At first, there is an optimal worker assignment on the work position requirements of ETO company.
- Then, one or many workers leave the company at their own initiative. Because of the high level of customer demand, there is no time to re-educate the existing employees, and management will not approve recruiting new employees.
- The quality of process output (product) must remain at the same quality level. It is assumed that the quality can be reached only with proper knowledge.
- The quantity of process output may be reduced.

This is a typical example of a company that needs to increase the use of its internal sources. Many cases have been found in practice in ETO companies in which the management solved the problem of outgoing knowledge with reorganization of internal employees rather than with the simple extension of employees' existing capacities, for example, overtime work [22]. We also set two assumptions that were not subjects of this research: first, we accepted that in ETO production, business processes are constantly changing and, therefore, knowledge requirements are also changing. Second, because these are simulations, the relation between knowledge and the process efficiency was accepted: if employees have proper knowledge for

the execution of activities, then these activities are performed faster. This has an impact on better efficiency of the whole process if that activity is simultaneously a process bottleneck [23].

3. Method

The key solution of adjusting processes to the current knowledge lies in the theory of business process management [24], in which the main problem of achieving a short process throughput time lies in the waiting times among different work positions that are the consequence of unbalanced work. This problem is insignificant if the entire process is executed by only one employee who occupies one work position, because there are no work position breaks [25]. This works only in small companies. Large business systems are complicated: they have many business processes with diverse knowledge requirements (e.g. ETO production) and require many employees with different types and levels of knowledge. Work is divided into activities between different work positions. Each work position has its own knowledge requirements. In this case, management needs control over the specific knowledge and over the number of the work position changes, and must keep them at the 'desired' minimum level so that the optimal process efficiency and the work balance are reached. The problem is also in the required and actual capacity of the specific knowledge. The process output quantity reflects the frequency of activity executions [26]. From a previous description of the principle of minimization work position breaks, when the capacity of one employee is exceeded, an additional employee who can perform all activities in the process is required. Such a broadly educated employee is too expensive, and this solution is thus irrational. Therefore, the process is divided into activities (tasks) among many work positions with the least expensive employees. Management creates work positions with a simple and complex knowledge structure. However, dividing work in too many work positions slows down the process: the throughput time is extended because of the additional waiting time each time the work position is switched.

Regarding the theory of work position breaks, work position knowledge structure and employee knowledge capacity, we modified our previously published model [22]. **Figure 1** shows the steps of upgraded conceptual model. In the new model, we are measuring the effect of the partial corruption of a perfect process regarding better current knowledge alignment from the perspective of employee capacity load and from that of process efficiency; with corruption of the process, we are decreasing its efficiency due to new additional work position breaks, but with better knowledge alignment we are again increasing the process efficiency.

3.1. Measuring optimal knowledge alignment

We can observe in practice that if the current knowledge deficit is below the required knowledge, the result is less efficient work. Surprisingly, even an excess of actual knowledge over



Figure 1. Knowledge-based assignment conceptual model.

the required level of knowledge has the same result of over-educated and intelligent employees becoming bored when they are executing routine activities [22]. Therefore, we modified a classic assignment linear integer problem of Kolman and Beck [3]. In the original optimization model (Eq. (1)), the value c_{ij} represents the added value if employee *i* is allocated to work position *j* and the optimization function maximizes a profit.

$$max \ z \ = \ \sum_{i=1}^{n} \sum_{j=1}^{n} c_{ij} \cdot x_{ij} \tag{1}$$

We replaced the added value with the minimal knowledge deficit/surplus (absolute) gap of *n* key required knowledge K_k . That means if we allocate an employee with his/her actual knowledge that is nearest to required knowledge on the work position (neither below nor above) then we have attained optimal knowledge alignment. The idea is to minimize the overall absolute key knowledge gap in the processes of the specific company (Eq. (2)).

$$\min z = \sum_{i=1}^{n} \sum_{j=1}^{n} \left\{ \left(\sum_{k=1}^{n} \frac{|K_k|}{n} \right)_{ij} \cdot x_{ij} \right\}$$
(2)

where *i*... n = number of compared employees; *j*... n = number of different work positions; k... n = number of compared key knowledge; and $|K_k|$ = absolute difference between required and actual knowledge K.

In case of a new required ETO production change, this model can be used in the following situations:

- If there is an 'open' set of available employees, all potential candidates in the optimization function can be matched. If the candidate knowledge gap is excessive (the appropriate level was not a subject of this research) the candidate is inappropriate for the work position because the performed work will be less efficient. This action has certain inherent costs (hiring, firing).
- If there is time to provide additional education to employees, then the knowledge deficit can be decreased with additional knowledge. This action has additional education and training costs.
- Existing employees can also be re-assigned on existing work positions so that the company knowledge alignment is optimal.

Are these all the possible management actions?

3.2. Measuring the corruption of a perfect process

As an innovation, the effect of a partial corruption of a perfect process was tested, including its impact on a better knowledge alignment with the limitation that the set of employees must remain untouched. The hypothesis was that with a corruption of the process, a better knowledge alignment can be achieved and, consequently, the process efficiency can be increased, despite a simultaneous decrease of its efficiency due to new additional work position breaks. Moreover, there must be a point in the process corruption procedure after which the inefficiency of the process exceeds the benefits of better knowledge alignment.

The effect of work position breaks in the process is measured by structural index K_{wpb} (Eq. (3)) [27]. This is a common key performance indicator in the theory of analysing business processes.

$$K_{wpb} = \frac{C_{wp}}{P_a} \cdot 100 \tag{3}$$

 C_{up} counts all work position breaks in a specific process. P_a counts all activities in that process. In this theory, the process slightly stops each time the next process activity is performed by different employee (on a different work position). This is one of practical causes for additional waiting time in the structure of throughput time of the process. There can be up to n - 1 work position breaks in a process of n sequential activities. According to the total number of all process activities, a small number of work position breaks means that the process is more efficient. In practice, poor work quality can be found in the process due to inappropriate knowledge alignment. This generates additional feedback loops, activities are repeated and the result is additional work position breaks. Determining the causes of additional activity breaks is not a subject of this research.

3.3. Linking knowledge optimization and work position breaks

From the perspective of real business in ETO production, especially in this time of global economic crisis, accessibility to newly required knowledge is greatly limited due to extra educational costs. Downsizing also means that processes must be executed with fewer employees but at the same time the level of product quality must remain equal to previous process executions. Management typically reacts with reorganization of employees on activities. Furthermore, because we cannot split 'the human body', his or her structure of knowledge and the time capacity of that knowledge cannot be optimal for current (ideal) process. In the theory, the problem can be easily solved if we have all current employees with all required knowledge of the process.

In ETO production, there are many specialists (e.g. electrical engineers, mechanical engineers, software engineers) with one or two dominant fields of knowledge of very high quality or strength, and few employees with wide spectra of high quality knowledge (senior engineers, mechatronics), because the latter are too expensive. However, they are also key employees for the ETO production; they have the big picture over each new product, and they can control the efficiency and quality of the overall production process. They are never 'bottlenecks' in the process with regard to knowledge, but they can be problematic with regard to the available time capacity of his/her specific required knowledge, because they are involved in many processes (ETO projects).

This phenomenon is also a result of the accumulation of many small organizational changes in processes over time. When the company was established (or after process re-engineering project), processes and work positions were optimally designed for execution, employees were carefully selected and their knowledge was appropriate for knowledge requirements of work positions (**Figure 2**).

Over time, new activities were slowly added to work positions, thus generating newly required knowledge. These changes were so small at the beginning that the management did not recognize them as knowledge problems or capacity problems. They had no effect on the employees except that the work position received one or two new key pieces of knowledge that employees had to obtain. After a few years of small changes, the work position and their key knowledge structure had expanded in such a way that the management and the employee did not know which pieces of knowledge of the work position were key for business success (e.g. a designer in ETO production is working 30% of his capacity on designing, 40% of the time he is occupied with routine paper work and another 30% he is attending meetings; if we require 100% design work, then this person's design knowledge is a capacity bottleneck).



Figure 2. Explanation of cutting activities when employee leaves the process.

For such cases, we created a process and knowledge algorithm that is connected with a Key performance indicators (KPI) that measures process corruption as follows:

- **1.** We must have input data of current processes (As-Is), their activities and times, current work positions, required knowledge, current employees and their actual knowledge.
- **2.** Then, we test the impact of employee reduction on the knowledge structure of process. We can start with required knowledge that is recognized as a process bottleneck or with knowledge that is missing at the new activity executor.
- **3.** In first case, we reduce the process activity until only work with knowledge that was bottlenecked remains (i.e. knowledge that is available by only one employee). The removed parts of activity with removed knowledge are distributed among other employees in the process until the optimal knowledge alignment is reached (Eq. (2)). If some knowledge is insufficient with one employee, the part of activity requiring this knowledge is given to an employee who can cover it successfully. Then, we repeat this procedure until optimal process knowledge alignment is reached.
- **4.** At the same time, we measure the impact of the activity-cutting principle on the process (Eq. (3)). Because the better knowledge alignment improves the process efficiency, and the activity-cutting principle reduces the process efficiency, the algorithm serves as a 'trading' point when we are balancing and allocating employee knowledge on activities within his/ her available time capacity (**Figure 3**)
- 5. The final result (output) is a new process (To-Be) that is feasible.

Such a reorganized process is reengineered on the basis of knowledge.

To-Be solution without activity cuts



This is common solution. The management wants efficient process and balanced capacities. The knowledge alignment is not optimal: $k_3 \notin W_3$.

To-be solution with one activity cut



The solution is optimal from process efficiency and knowledge view. It is not optimal from capacity view.





The solution is not optimal from process efficiency view: one additional switch of work positions. It is optimal from capacity and knowledge view.

Figure 3. Possible outputs of algorithm for optimal knowledge alignment in ETO production.

4. Input data

4.1. Processes, process activities, work positions and required knowledge

In ETO production, at first sight, almost every product has its own and unique production process (routing). The fact is that activities (operations) among different processes are almost the same with regard to required knowledge. They differ mostly in the time required for execution. Because each product has its unique structure (bill of material), the process is named in practice as a project and its operations are named as activities. However, from the top-down approach, each project in ETO production has almost the same set and the same sequence of project phases (with many sub-activities), for example, (1) preparation, (2) design, (3) construction, and (4) testing. Therefore, it can be assumed that we have a standard form of the process (with activities) for almost all new products.

The same process activity could appear in a structure of many different processes and it is usually performed by the same work position (e.g. the same quality control activity with the same control parameters and tools for the whole product group). Moreover, one work position executes many activities. Until the system is well organized, a work position aggregates activities with approximately the same required set of knowledge. We defined that the required knowledge of a specific work position is represented as a set of knowledge from all executed activities. The sets of required knowledge of specific activity and their strength (Likert scale from 1 to 5; 5 meaning very important) are defined by the company's internal and external experts. If a specific piece of knowledge is required for the execution of many activities, the model uses its maximal value as a required strength.

Complex work positions have a wide range of required knowledge, many of unimportant strength. Reducing the amount of various required knowledge can simplify the calculations. Simplification was achieved with the definition of key knowledge K_k for each work position. If the strength of specific knowledge is above a specific level, it is treated as key knowledge of that work position.

In practice, the above-described idea of capturing process activities and their required knowledge can be used for documenting As-Is processes and, more importantly, for predicting future products, To-Be processes and their expected required knowledge. This is of great importance for planning required knowledge of future ETO production. We can analyse the following:

- Which activity among all activities of specific process is the most important from the key knowledge aspect, for example, to find the activity that is the 'knowledge bottleneck' in a process. Then we can combine this information with activity throughput rate and find an activity that is the real-time capacity bottleneck in the process.
- Which process (from among all of them) is the most important from the aspect of key knowledge, for example, for ranking all processes on the basis of the knowledge required (i.e. which process is currently the most important/crucial for the company from the knowledge view; this is important information for any ETO company in addition to the information regarding which process is crucial from capacity aspect).
- In ETO production, each work position typically executes many different activities in many different processes. Therefore, we are interested which work position has the highest required strength of all key knowledge, for example, we can use this information as a basis for creating salary grades.
- Which work positions in the company are exceptional from the knowledge aspect; a work position that has only one key type of knowledge but with a high required strength (e.g. CNC programmer) and which work positions are universal, that is, have many key types of required knowledge (e.g. ETO project manager).
- Which type of knowledge is dominant (repeats at every executed activity) for the specific process (short-term view) and for the whole company (long-term view).

If we have proper data on all the above mentioned entities (processes, activities, work positions, knowledge requirements with required strength) for the present time, and if we have good knowledge requirements (definitions) of new products (especially required technology and activities), we can then simulate all future knowledge requirements in advance. Therefore, we can determine differences, for example, which work position must be knowledge-reconstructed in the future; consequently, we can define projected mandatory changes in a structure of actual knowledge (employees).

4.2. Employees, actual knowledge and knowledge gap

Employees represent the basis for gathering current knowledge. There are many approaches to prove that an employee possesses specific knowledge and what the quality of it is (strength, level). In our approach, the 360° feedback method [28] was used. We used a list of all key required knowledge and assessed all employees (Likert scale from 0 to 5; 0 means knowledge not available). We gave employees the opportunity to extend this explicit knowledge with their tacit knowledge. In the context of our model, the term 'tacit' means the knowledge of an employee that is currently unknown to the company. Knowing about tacit knowledge is essential information when new processes have requirements for new types of knowledge. In practice, for optimization, it is also recommended that we have the knowledge data about potential candidates for employees.

The last step of input data preparation is a calculation of the key knowledge gap: each employee is compared to all work positions. We used the criterion $c_{ij'}$ explained in Eq. (2). Any deviation of actual knowledge over and below the required knowledge is considered to be inappropriate and will lower process efficiency (**Table 1**).

Table 1 shows a numerical example of matching the actual knowledge from k1 to k10 of employee E1 on activities from a1 to a7 of work position W1 (e.g. Product Manager of ETO project). The example is based on the real data of ETO company, Iskratel. Negative values (grey cells) represent deficits of employee knowledge strength compared to the required knowledge of a work position. The top rows represent activities of the work position with a

	Activities	a ₁	a ₂	a ₃	a ₄	a ₅	a ₆	a ₇
Knowledge		-2.48	-2.98	-6.64	-1.2	-0.11	-0.14	-0.27
k ₁	-0.40	0.60	-0.40	2.27	0.00	1.77	1.03	2.04
k ₂	0.0	1.19	1.18	1.57	0.60	2.40	2.26	2.73
k ₃	-0.62	-0.24	-0.38	0.23	0.60	1.62	0.26	1.51
k ₄	-4.54	-0.79	-1.00	-1.50	-1.00	-0.11	-0.14	1.44
k ₅	-1.20	1.77	-1.20	1.70	2.20	1.37	0.20	0.31
k ₆	-0.97	0.41	0.98	-0.97	0.40	0.59	0.06	0.64
k ₇	-1.54	0.76	0.40	-1.27	1.40	1.23	0.69	-0.27
k ₈	-2.45	-1.45	1.00	-1.00	0.60	1.00	1.57	1.33
k ₉	-1.07	0.20	0.47	-0.87	-0.20	1.08	2.80	2.58
k ₁₀	-1.03	1.15	0.13	-1.03	0.40	1.41	2.80	2.36

Table 1. Matching required and actual knowledge.

sum of negative values. We can identify activities that the employee is not suitable to execute (e.g. *a*1, *a*2, *a*3). The left column represents the required knowledge with the sum of negative values. We can identify the lack of employee knowledge (e.g. *k*4, *k*8).

In practice, we could integrate in our model the effect of learning and forgetting knowledge over time (decreasing knowledge strength if employee is not using that type of knowledge in processes for a long time). Because of model simplicity, this was not a subject of this research.

5. Results

We demonstrated the capabilities of our model on a small section of the real process that was described in **Figure 3**. This numerical example is based on the data of company Iskratel. We performed simulations of this example with the same tools as the calculations of real cases (**Tables 2** and **3**). Definitions of processes were recorded in the repository of Aris Toolset software [29]. Definitions of actual and required knowledge were recorded with MS Share Point and MS SQL. All data were then exported to the MS Excel analytical tool and solved with the WhatsBest [30] add-on. MS Excel was also used as reporting tool.

5.1. Input data of simulation scenarios

We prepared four simulation scenarios as follows:

- Scenario 0: As-Is situation. In the current state, there are three employees assigned to their own work positions, and the processing of four activities with four different types of knowledge.
- Scenario 1: employee on work position w_2 left the company. His/her activity a_2 is assigned to w_1 and a_3 to w_3 . This is typical management decision that does not generate an additional work position switch in the sequence of activities.
- Scenario 2: use of our algorithm: achieving better knowledge alignment. Employee on w_3 has no knowledge K_3 that is required for execution of activity a_3 ; therefore, we split activity on a'_3 and a'_3 .
- Scenario 3: is same as scenario 2, with one additional activity cut: we are searching for better balance of capacities between w_1 and w_3 . We split activity a_2 and we add knowledge K_2 to work position w_3 .

We can observe the things as follows:

- (i) In scenario 1, the result of management action on knowledge distribution among work positions: Knowledge K_1 and K_2 are moved from w_2 to w_1 . Knowledge K_3 and K_4 are moved from w_2 to w_3 . In case this is the same knowledge, we used the maximal strength as the required strength.
- (ii) In scenario 2, the result of optimization algorithm: according to As-Is situation, we moved from w_2 to w_1 knowledge K_1 and K_3 . This caused the rise of the strength of both

Simulation scenarios		_	
Scenario 0	Scenario 1	Scenario 2	Scenario 3
I. Input data: processes (ARIS EPC) and activities (a _n), work positions (w _n) an	d required knowledge (K _n)
$\begin{array}{c c} I \\ \hline \\$	$\begin{array}{c} I \\ a_1 \\ W_1 \\ W_1 \\ W_1 \\ W_1 \\ W_2 \\ W_3 \\ W_3 \\ W_3 \\ W_3 \\ W_3 \end{array}$	$ \begin{array}{c} $	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$
II. Input data: actual kno	wledge strength of employ	ees (Likert scale 0 to 5)	
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{cccccc} & E_1 & E_3 \\ K_1 & 3 & 0 \\ K_2 & 2 & 5 \\ K_3 & 1 & 0 \\ K_4 & 0 & 2 \end{array}$	$\begin{array}{cccccc} E_1 & E_3 \\ K_1 & 3 & 0 \\ K_2 & 2 & 5 \\ K_3 & 1 & 0 \\ K_4 & 0 & 2 \end{array}$	$\begin{array}{cccccc} E_1 & E_3 \\ K_1 & 3 & 0 \\ K_2 & 2 & 5 \\ K_3 & 1 & 0 \\ K_4 & 0 & 2 \end{array}$
III. Input data: required	knowledge strength of wor	k positions (Likert scale 0	to 5)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{ccccc} & w_1 & w_3 \\ K_1 & \textbf{5(ii)} & 0 \\ K_2 & 2 & 5 \\ K_3 & \textbf{3(ii)} & 0 \\ K_4 & 0 & \textbf{2(ii)} \end{array}$	$\begin{array}{ccccc} & w_1 & w_3 \\ K_1 & \textbf{5(iii)} & 0 \\ K_2 & 2 & \textbf{5(iii)} \\ K_3 & \textbf{3(iii)} & 0 \\ K_4 & 0 & \textbf{2(iii)} \end{array}$

Table 2. Input data of simulation scenarios.

types of knowledge for w_1 . We moved from w_2 to w_3 only knowledge K_4 , because the newly required strength is below the current required strength so it remains as it was for w_3 .

(iii) In scenario 3, the new activity cut did not cause any change in knowledge requirements (and strength) of w_1 and w_3 according to scenario 2.

5.2. Simulation results

We can see in Scenario 2 (implementing activity-cutting principle) that we decreased the knowledge gap in Scenario 1. Now, we must 'merge' the results of optimal knowledge alignment to determine the impact of using the activity-cutting principle on classic production optimization parameters (Scenario 3). Otherwise, we will break some lean manufacturing principles, for example, work balancing or eliminating waiting times. We added additional input data of As-Is process in **Table 4**.

Knowledge-Based Assignment Model for Allocation of Employees in Engineering-to-Order... 231 http://dx.doi.org/10.5772/intechopen.70073

Scena	ario O			Scenario 1			Scenario	2		Scenario 3			
I. Co	mpariso	on of ac	tual a	nd requir	ed knowled	ge (ab	solute kno	owledge	gap ; c _{i,j})				
$\begin{matrix} w_1 \\ K_2 \\ K_3 \\ K_4 \\ C_{i,j} \\ w_2 \\ K_3 \\ K_4 \\ C_{i,j} \\ w_3 \\ K_1 \\ K_2 \\ K_3 \\ K_4 \\ C_{i,j} \\ \end{matrix}$	E ₁ -1 0 0	E ₂ 1 1 1 / / 1.0 0 1 -1 0 0.5 / -2 / -1 1.5	E ₃ -4 3 -1 / / 2.7 -5 3 3 -5 3 3 1 3.0 / 0 0 0.0	${ \begin{array}{c} w_1 \\ K_1 \\ K_2 \\ K_3 \\ K_4 \\ C_{ij} \\ w_3 \\ K_1 \\ K_2 \\ K_3 \\ K_4 \\ C_{ij} \end{array} } }$	E ₁ E ₃ -2 0 / 0.7 3 / -3 -2 -2 2.3 1	-5 3 -1 / 3.0 / 0 -3 0 l.0	$\begin{matrix} w_1 \\ K_2 \\ K_3 \\ K_4 \\ C_{ij} \\ w_3 \\ K_1 \\ K_2 \\ K_3 \\ K_4 \\ C_{ij} \end{matrix}$	E ₁ -2 0 -2 / / 1.3 / -3 2.5	E ₃ -5 3 -3 / 3.7 / 0 0 0.0	$\begin{matrix} w_1 \\ K_2 \\ K_3 \\ K_4 \\ C_{i,j} \\ w_3 \\ K_1 \\ K_2 \\ K_3 \\ K_4 \\ C_{i,j} \end{matrix}$	E ₁ 1 -2 0 -2 / 1.3 / -3 / -2 2.5	E ₃ -5 3 -3 / 3.7 / 0 / 0 0.0	
II. Employees allocation optimization (problem specification)													
Variab $x_1 = E_1$ $x_2 = E_2$ $x_3 = E_3$ $x_4 = E_1$ $x_5 = E_2$ $x_6 = E_3$ $x_7 = E_1$ $x_8 = E_2$ $x_9 = E_3$ Optimiz $Z_{min} = 0$ Bound: E_1 can E_2 can E_3 can W_1 mus W_2 mus W_3 mus	les: occupies occupies occupies occupies occupies occupies occupies occupies occupies occupies axish funct $occupy: xoccupy: xoccupy$	W_1 W_1 W_2 W_2 W_2 W_3 W_3 W_3 W_3 W_3 ection: $+2,7x_3+1,2,5x_7+1,3x_7+1,$	$3x_{4} + 5x_{8} + 0x_{9}$ i i $x_{2} + x_{3} = I$ $x_{5} + x_{6} = I$ $x_{8} + x_{9} = I$	Variables: $x_1 = E_1 \operatorname{occr}$ $x_2 = E_3 \operatorname{occr}$ $x_3 = E_1 \operatorname{occr}$ $x_4 = E_3 \operatorname{occr}$ Optimizatio $Z_{min} = 0, 7x_1$ Boundaries $E_1 \operatorname{can occr}$ $E_3 \operatorname{can occr}$ $W_1 \operatorname{must}$ be $W_3 \operatorname{must}$ be	<i>upies</i> W_1 <i>upies</i> W_1 <i>upies</i> W_3 <i>upies</i> W_3 <i>upie</i>	x_{4} $x_{2} = I$ $x_{4} = I$	Variables: $x_1 = E_1 \circ ccu$ $x_2 = E_3 \circ ccu$ $x_3 = E_1 \circ ccu$ $x_4 = E_3 \circ ccu$ $Z_{min} = 1, 3x_1$ Boundaries $E_1 \ can \ occu$ $E_3 \ can \ occu$ $W_1 \ must \ be$ $W_3 \ must \ be$	pies W_1 pies W_1 pies W_3 pies W_3 pies W_3 on Function $py: x_1+x_3=$ $py: x_2+x_4=$ occupied:	$ \sum_{x_{3}+0x_{4}} \sum_{x_{4}+1} \sum_{x_{5}+x_{4}=1} \sum_{x_{5}+x$	Variables: $x_1 = E_1 \circ cctu$ $x_2 = E_3 \circ cctu$ $x_3 = E_1 \circ cctu$ $x_4 = E_3 \circ cctu$ $Z_{min} = 1, 3x_1$ Boundaries $E_1 \ can \ occtu$ $E_3 \ can \ occtu$ $W_1 \ must \ be$ $W_3 \ must \ be$	ppies W_1 ppies W_1 pies W_3 pies W_3 pon Function: $+3,7x_2+2,5x$ Functions: $py: x_1+x_3=1$ $py: x_2+x_4=1$ $occupied: x_3$ $occupied: x_3$	s+0x₄ r+x₂=1 s+x₄=1	
III. O	ptimiza	ation re	esults (minimal :	absolute kn	owled	ge gap an	d emplo	yees alloc	ation)			
$Z_{\min} = x_1 = 1.$	= 0.8 if x ₅ =1 an	d x9=1		$Z_{\min} = 1.2$ $x_1=1$ and	7 if x ₄ =1		$Z_{\min} = 1.3$ $x_1=1$ and	3 if x₄=1		$Z_{min} = 1.3$ if $x_1=1$ and $x_4=1$			

Table 3. Simulation results.

- The first assumption (i) in our evaluation is the amount of time that is added to process throughput time each time we change the work position (sending work from me to you etc.). In a real case, this could be measured exactly but in our demonstration we assumed a fixed value of 3 min.
- The second assumption (ii) in our evaluation is the amount of time that is added to process throughput time because of non-optimal knowledge alignment. In the As-Is process, we

Process Times		Activity sequence \rightarrow						
in minutes)	Knowledge sequence ↓	a_1	a ₂	a3	a4			
	K ₁	12						
	K2	15						
	K3	3						
	K ₁		5					
	K ₂		10					
	K3			7				
	K4			8				
	K ₂				10			
	<u>K4</u>		1 45		20			
	Total activity time	30	15	15	30			
	Work position	w ₁	W3					
	(i)Moving time	3 3						
	(i)Planned	$30+3+15+15+3+30 = 96 \text{ min}$ $(0.8/0.1)\times1\% = 8\% =>$ $96 \text{ min} \times 1.08 = 103.68 \text{ min}$ $\left(\begin{array}{c} w_{12} \\ 450 \\ w_{13} \\ \hline 95 \\ 100 \\ 100 \\ \hline 95 \\ 100 \\ 100 \\ \hline 95 \\ 100 $						
	throughput time							
	(ii)Expected							
	throughput time							
	Process			30				
	Throughput rate	min	≅ 13					
	per shift (450		* 3 +	15 + 15)			
	mm)		$w_{3};\frac{1}{3}$	$\frac{450}{+30} \cong$	13			
	Capacity Load per shift (%)	86.7 95.3 95.3						
	Expected	(9	0min/103	3.68min)	×100 =			
	Efficiency				86.8%			

Table 4. Production parameters of As-Is process.

know that we have 0.8 by the Likert non-optimal knowledge alignment. If the times in this table were measured without being aware of this knowledge gap then the real throughput time is longer. In a real case, we could measure this by comparing the knowledge gap and the difference between planned and real production times (we have to exclude other causes for time extension first). In our demonstration, we assumed that every 0.1 of knowledge gap adds 1% to planned process throughput time.

6. Discussion

The main specialty of our model is that we permit changes of the process because the actual knowledge is not appropriate for it. However, we do not allow changes in the sequence of activities; we allow only changes in the sequence of using employees. The results are new partial activities in the process; consequently, the process workflow is jumping forwards and backwards between employees.

In our model, we removed all unnecessary knowledge from the work positions that were process 'bottlenecks' and replaced it with the new process structure; this was done by taking into consideration the availability of the actual knowledge of employees. The entire individual employee time capacity is now focused only on the utilization of knowledge that is bottlenecked. Other required knowledge in the process that is also present in other employees is removed from that work position. Employee capacity is now free of all non-bottleneck knowledge, and this raises its capacity availability.

In our simulations, we used process time indicators to verify our assumption, even if we know, on the basis of real projects [31, 32], that the best improvements in the ETO production are achieved on the process quality indicators. Time indicators are improved indirectly as a result of better product quality: fewer aftermarket repairs means less additional invested time in the total production time of the specific product. The starting point of all scenarios is the departure of one employee from the original process (Scenario 0). In Scenario 1, we reacted by implementing the lean manufacturing principle of capacity balancing: the work of the lost employee is divided among remaining employees on the basis of capacity levelling without additional work position breaks. This is a common management decision, and it is expressed as a load capacity per shift (%) indicator in **Table 4**. This decision produced the knowledge gap of 1.7 (**Table 3**).

In Scenario 2, we used our model with the activity-cutting principle, and we reduced the knowledge gap by 0.4 or 23.5% (**Table 3**). Most time indicators were also improved (**Table 5**), except for the unbalanced load capacity per shift (%) indicator, and a lower process throughput rate (from 9 to 8 products per shift). Both indicators would have negative impact in mass or serial production, but according to the requirements of the ETO production it is more important that we achieved the desired quality of knowledge for production process because there are no repetitions (rather only unique, one-time process executions). Management can balance these indicators and make the decision that is adopted for a specific process 'case'.

In Scenario 3, we tested the total ignorance of the Lean Manufacturing principles, and we performed additional activity cuts for searching for even better knowledge alignment. We did not achieve a lower knowledge gap (**Table 3**); we also worsened all time indicators according to Scenario 2 (**Table 5**). This indicated that there is a point in the repetition of activity-cutting procedure after which the process becomes so inefficient that is better to hire a new employee if the knowledge gap is still too high for achieving the appropriate quality of ETO products. Where that point is, what the gap should be and whether its value is of universal use or case sensitive are all subjects of future research.

	5	Scenar	io 1		Scenario 2					Scenario 3						
	Activity sequence →					Activity sequence →					Activity sequence →					
Knowledge sequence 1	aı	a ₂	a3	a4	aı	a2	a3'	a3''	a4	aı	a2'	a2"	a3'	a3''	a4	
K1 K2 K3 K4 K4 K4 K4 K4 K4	12 15 3	5 10	7 8	10 20	12 15 3	5 10	7	8	10 20	12 15 3	5	10	7	8	10 20	
Total activity time	30	15	15	30	30	15	7	8	30	30	5	10	7	8	30	
Work position	W1	W1 W3)	w1 W3			W1 W3			W1 W3					
Moving time			3	·				3				3	3	3		
Planned throughput time		93 mi	n		93min				99 min							
Expected throughput time		108.8 n	nin		105.1 min				111.9 min							
Process Throughput rate per shift (450 min)	min	$\begin{cases} w_1; \frac{43}{4} \\ w_3; \frac{4}{4} \end{cases}$	$\frac{50}{5} = 10$ $\frac{50}{18} \cong 9$)	$\min\begin{cases} w_1: \frac{450}{52} \cong 8\\ w_3: \frac{450}{39} = 11 \end{cases}$				m		$\frac{450}{45} = \frac{450}{54} \cong$	10 8				
Capacity Load per shift (%)	90.0	%	96.	0%	92.4%		72.9%		9%	80.	0%			96.	0%	
Expected Efficiency		82.79	6			85.6%						\$ 0.	5%			
KPI		25%			20%			50%								

Table 5. The impact of activity-cutting principle on production parameters in scenarios from 1 to 3.

7. Conclusions

In Make-to-Stock, Assemble-to-Order and Make-to-Order production, assignment models for the allocation of employees assume that tasks of production processes (or routings) are of a fixed structure. Managers believe they found the most 'efficient' process of producing products and, therefore, all current optimization models are searching for appropriate employees for that process. Small deviations between the required and the actual knowledge are resolved with alternative routing; its structure is also known and fixed in advance. All of this is possible because extra time is invested for testing and preparing optimal processes for many repetitions. Extra time is also invested for finding employees with proper knowledge for that processes. This is the case of known theoretical and practical solutions of worker assignment problem.

However, in ETO production, and consequently in all knowledge-intensive processes or caselike processes, we determined that processes are structured around the available knowledge of employees. Otherwise, the cost of searching for missing knowledge in the form of a new employee could exceed all the added value to the business. Process 'cases' are never the same and each process 'repetition' requires a process structure that is adapted to the actual knowledge and its capacity in the company; the bottleneck is not the capacity of the employee but the capacity of his/her specific actual knowledge. With the activity-cutting principle in our assignment model, we proved that we can release the 'hidden' time capacity of employee who is the bottleneck so that we could remove all activities and consequently the knowledge that is also available with other employees from the work position. We recommend that this principle can be an option of all assignment models for the allocation of employees for ETO production and all other knowledge-intense companies. This is our main contribution to the theory of modelling worker assignment problem.

Of course, this research raises additional questions for our future work, especially in the field of practical application: is knowledge the right category in our assignment model or is it better to use all measureable work habits and personal skills [33]? There are also assumptions in **Table 4** that will need additional research and explanation. Nevertheless, our concept of redefining tasks with the goal of reaching optimal worker knowledge alignment could be used as a 'smart' reorganization principle for dynamic and real-time redefinition of processes in companies, where the standardization of tasks is not the main factor of reaching efficiency.

Author details

Matjaz Roblek*, Maja Zajec and Benjamin Urh

*Address all correspondence to: matjaz.roblek@um.si

Faculty of Organizational Sciences, University of Maribor, Kranj, Slovenia

References

- [1] Kern T, Roblek M, Mayer J, Urh B. Business processes and human resources competence profiles. In: Khosrowpour M, editor. Managing Modern Organization with Information Technology; 15-18 May 2005. San Diego, CA, USA: Idea Group; 2005. pp. 1202-1204
- [2] Meglic J, Kern T, Urh B, Balkovec J, Roblek M. Influence of polyvalence professionals on product development process efficiency. Strojarstvo. 2009;**51**(2):105-121
- [3] Kolman B, Beck R. Elementary Linear Programming with Aplications. San Diego: Academic Press; 1995
- [4] Gartner B, Mataušek J. Understanding and Using Linear Programing. Berlin: Springer; 2006
- [5] Azizi N, Liang M. An integrated approach to worker assignment, workforce flexibility acquisition, and task rotation. Journal of Advanced Manufacturing Technology. 2013;64(2):260-275
- [6] Wu MC, Sun SH. A project scheduling and staff assignments model considering learning effect. International Journal of Advanced Manufacturing Technology. 2006;28(11-12): 1190-1195

- [7] Wang Q, Liu M. Optimization of task allocation and knowledge workers scheduling based on ant colony algorithm. In: International Symposium on Intelligence Information Processing and Trusted Computing; 2010. pp. 386-389
- [8] Doringo M, Birattari M, Stutzle T. Ant colony optimization, artificial ants as a comutational intelligence technique. IEEE Computational Intelligence Magazine. 2006;1(4):28-39
- [9] Nembhard DA. Heuristic approach for assigning workers to tasks based on individual learning rates. International Journal of Production Research. 2001;**39**(9):1955-1968
- [10] McDonald T, Kimberly PE, Eileen MV, Koelling CP. Development and application of a worker assignment model to evaluate a lean manufacturing cell. International Journal of Production Research. 2009;47(9):2427-2447
- [11] Norman BA, Tharmmaphomphilas W, Needy KL, Bidanda B, Warner RC. Worker assignment in cellular manufacturing considering technical and human skills. International Journal of Production Research. 2002;40(6):1479-1492
- [12] Ingolfsson A, Campello F, Wu X, Cabral E. Combining integer programming and the randomization method to schedule employees. European Journal of Operational Research. 2010;202(1):153-163
- [13] Brusco MJ, Johns TR. Staffing a multiskilled workforce with varying levels of productivity: An analysis of cross-training policies. Decision Sciences. 1998;29(2):499-515
- [14] Mohan S. Scheduling part-time personnel with availability restrictions and preferences to maximize employee satisfaction. Mathematical and Computer Modelling. 2008;48(11-12):1806-18013
- [15] Du L, Zhang Y, Li H. A competence-driven staff assignment model based on stochastic working status. In: Zhu M, editor. Business Economics and Financial Sciences; Berlin: Springer 2012. pp. 463-470
- [16] Gutjahr WJ, Katzensteiner S, Reiter P, Stummer C, Denk M. Competence-driven project portfolio selection, scheduling and staff assignment. Central European Journal of Operations Research. 2008;16(3):281-306
- [17] Grosse EH, Glock CH, Jaber MY, Neumann WP. Incorporating human factors in order picking planning models: Framework and research opportunities. International Journal of Production Research. 2015;53(3):695-717
- [18] Crawford B, Soto R, Johnson F, Monfroy E, Paredes F. A max-min ant system algorithm to solve the software project scheduling problem. Expert Systems with Applications. 2014;41(15):6634-6645
- [19] Mutingi M, Mbohwa C. A fuzzy-based particle swarm optimisation approach for task assignment in home healthcare. South African Journal of Industrial Engineering. 2014;25(3):84-95

- [20] Olivella J, Corominas A, Pastor R. Task assignment considering cross-training goals and due dates. International Journal of Production Research. 2013;51(3):952-962
- [21] Senjuti BR, Ioanna L, Saravanan T, Sihem AY, Gautam D. Task assignment optimization in knowledge-intensive crowdsourcing. The VLDB Journal. 2015;24(4):467-491
- [22] Roblek M, Zajec M, Kern T. The lack of knowledge? Change the way you work. In: Sprague RH, editor. Proceedings of the 44th Annual Hawaii International Conference on System Sciences; 4-7 January; Koloa, Kauai, Hawaii. Los Alamitos: California: IEEE - The Institute of Electrical and Electronics Engineers; 2011. p. 6
- [23] Goldratt EM. The Goal: A Process of Ongoing Improvement. Third Edition. NewYork: Routledge; 2016
- [24] Sheer AW, Nuttgens M. Business process management: Models techniques and empirical Studies. In: ARIS Architecture and Reference Models to Business Process Management. Berlin: Springer-Verlag; 2000
- [25] Valiris G, Glykas M. Business analysis metrics for business process redesign. Business Process Management Journal. 2004;10(4):445-480
- [26] Anupindi R, Chopra S, Deshmukh SD, Van MIeghem JA, Zemel E. Managing Business Process Flows. New Jersey: Prentice Hall; 1999
- [27] Bauer K. KPIs: Not all metrics are created equal. DM Review. 2009;14(12):42
- [28] Maylett T. 360-Degree feedback Revisited: The transition from development to appraisal. Compensation & Benefits Review. 2009;41(5):52-59
- [29] Davis R. ARIS Design Platform: Advanced Process Modelling and Administration. London: Springer Verlag; 2008
- [30] Lindo Systems Inc. What's Best Software for Microsoft Excel [Internet]. 2007
- [31] Kern T, Roblek M, Urh B. Analysis of MORS processes, organization and personnel systemization. Project CRP M5-0048. 2006. Ljubljana
- [32] Kern T, Roblek M, Urh B, Kokalj S. Development of MORS model for a selection of candidates for special tasks execution. Project CRP M5-0171. 2008. Ljubljana
- [33] Green PC. Building Robust Competencies: Linking Human Resource Systems to Organizational Strategies. San Francisco: Jossey Bass, Inc. Publisher; 1999
An Examination of the Knowledge Management Process in the Emerging Chinese Hotel Industry

Tommy Wong, Linda French and Mark Wickham

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/67105

Abstract

Since the Open-Door Reforms of the Chinese market to foreign investment in the late 1970s, numerous Western firms have sought to enter and establish their presence as a market leader in their particular industry. Establishing a presence and generating growth in the emergent Chinese market has proven challenging for Western firms; operating effectively and efficiently in this 'new' market environment (characterised by significant psychographic, cultural, legal and political differences) has highlighted a number of significant research opportunities. One such opportunity has manifest in a call to develop a finer-grained understanding of the knowledge management processes required by Western hotel chains attempting to enter and establish their operations in China. This chapter presents an exploratory analysis of the knowledge management issues and processes experienced by the Marriott Group during its entry and growth in the Chinese hospitality industry. This analysis is based on a series of semi-structured interviews with senior managers present during Marriott's entry into China. The chapter concludes with a discussion of the managerial and theoretical implications emanating therefrom.

Keywords: hospitality industry, knowledge management, China

1. Introduction

The strategic management of organisational knowledge requires a comprehensive understanding of the process by which the various forms of tacit and explicit knowledge are generated, coded, stocked and leveraged for competitive gain (see Refs. [1, 2]). The process of extracting value from the careful management of knowledge-based assets has been investigated in multiple disciplines, industries and markets (see Ref. [3]); however, according to



© 2017 The Author(s). Licensee InTech. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. Refs. [2, 4], investigation into what constitutes appropriate knowledge management processes within emerging/developing markets remains limited. Many Western-based organisations seeking to enter emerging economies (such as those in South East Asia) have achieved mixed success, which in part has been attributed to the inherent limitations of transferring extant Western-based knowledge management processes to new international contexts (see Refs. [5-7]). Such attempts can be seen in the highly competitive Chinese hotel sector, which has become an important investment platform for leading multinational hotel groups including, but not limited to Marriott International, Accor, Inter-continental and Shangri-La (see Ref. [8]). Despite the heavy investment of these Western hotel groups, their ability to compete effectively in China remains an issue of ongoing concern (see Refs. [8–10]). As a result, Refs. [11–14] have called for a finer-grained understanding of what constitutes 'effective knowledge management' by Western firms in the highly competitive Chinese hotel sector. Focussing on one major Western hotel chain operating in China (i.e. Marriott International), this chapter presents an empirical exploration into the issues inherent to knowledge management processes in the Chinese hospitality industry. In addition, the potential implications identified during this exploration have been used to prepare and present an agenda for future research into knowledge management practices within the Chinese hospitality industry.

2. Knowledge management processes and the Chinese hotel industry

The knowledge management concept has been extolled across multiple disciplines as a control process whereby organisational knowledge is generated, retained and leveraged to enhance competitiveness and organisational growth (see Refs. [11, 15]). Organisational efforts to strategically manage their store of knowledge continue to intensify as the business environment becomes progressively competitive, and new, dynamic markets continue to emerge (see Refs. [14, 16]). The rapid expansion speed of new markets has allowed little time for Western organisations investing in markets such as China to reflect on the applicability of their existing managerial strategies to this emerging-market context (see Ref. [4]). Indeed, current theories related to knowledge management have predominantly been developed in Western markets where business dealings are traditionally transactional based; to date, little consideration has been given to the generalisability of these theories to Western organisations operating in emerging Asian economies where relational based business dealings are the norm (see Refs. [2, 4, 17, 18]). In order to improve our understanding of what constitutes effective knowledge management for Western organisations seeking to enter and compete in the Chinese hotel sector, we felt it appropriate to adopt a knowledge management process model to guide our data gathering and analysis.

For this study, the Knowledge Management Cycle (KMC) model has been chosen; the KMC consists of seven phases: identify and/or create; store; share; use; learn; improve; and create 'organisational knowledge' (see Ref. [1]). If a request for new knowledge is triggered within an organisation, the first stage is to identify if existing codified and encapsulated knowledge assets and subjectively held tacit knowledge exist, or if new knowledge assets need to be created. The KMC model, therefore, suggests that managing the knowledge process is

primarily reliant on (a) an organisation's executives' definition of what constitutes relevant 'knowledge' and 'knowledge gaps' in their organisation and (b) their ability to implement efficient and systematic methods to gather and/or communicate new knowledge to employees (see Refs. [18, 19]). Organisational specific rules, cultures and evaluative criteria are able to be used as a lens for interpreting, analysing and applying existing or newly created knowledge to decision making (see Ref. [20]). If deemed valuable, any new knowledge generated is codified and systematically stored in organisational memory from which it can be drawn, shared and protected (see Ref. [1]). Fulfilment of these tasks can embed a learning culture within the organisation (see Ref. [21]); learning cultures enable employees to generate their own internal store of knowledge and leverage their own understanding to make innovative decisions and contribute to their organisation's dynamic capabilities (see Refs. [19, 22]).

Research has determined that the existence of flexible and dynamic organisational networks, or communities of practice, provide the most effective means for encouraging and facilitating the dissemination of explicit and tacit organisational knowledge (see Refs. [14, 18, 20]). Once shared, the value from knowledge assets can be extracted and applied (see Ref. [20]). However, this stage can prove difficult as the ability of employees to extract value from the complex interplay of codified, encapsulated and tacit knowledge may also be influenced by additional factors, such as national culture and the institutional context (see Refs. [2, 14, 23]). The ability to simultaneously engage in continual organisational learning and develop a fluid store of country-specific knowledge (that supports distinctive culture-specific business practices) would seem, therefore, to be critical to Western organisational success in the Chinese market context. The success of knowledge management processes is therefore predicated on an organisation's Human Resource Management (HRM) practices and its ability to recruit, develop and motivate employees (see Refs. [2, 17]). This in itself can prove problematic as studies into the HRM practices of Western firms in China have identified the inability to find talented employees (see Ref. [23]), and the lack of employee initiative and acceptance of responsibility (see Ref. [17]) to be major inhibitors for organisational performance. An additional complication is employee leakage from the industry due to open competition for skilled/Western-trained labour between economic sectors in emerging economic settings (see Ref. [20]). Given these complicating factors, this chapter seeks to contribute to the knowledge management literature by exploring one Western organisation's (i.e. Marriott International's) successful entry into the Chinese hotel industry and provide a discussion of the managerial and theoretical implications emanating from the organisation's experience in assessing and developing its knowledge management practices in this context.

3. Sample and methodology

In order to explore the knowledge management process in the Chinese hotel industry context, the case study method was adopted. This method allowed the researchers to explore an existing theoretical concept (knowledge management) in a new context (a Western organisation operating in the Chinese hospitality industry) and to refine and elucidate previously unidentified issues. To achieve this, the authors conducted a series of semi-structured interviews with senior managers in the Marriott Group's Chinese operation. The choice of Marriott was driven by three factors; the first being that it presents a revelatory case, whereby senior management can deliberate those aspects of knowledge management critical to the strategic management of its business. Secondly, Marriott represents a renowned global organisation seeking to leverage its brand into the expanding Chinese market. Finally, the researchers were able to secure access to the organisation's senior management team through a key informant, who is recognised in the industry as one of the longest serving General Managers in China. In total, twelve Marriott senior managers were recruited to provide primary data relating to the organisation's knowledge management processes in the Chinese hospitality industry. Of the twelve senior managers: three were General Managers, three were Marketing Directors (that had been working with Marriott since its entry into China in the mid-1990s), three were Human Resource Directors and three were Sales Directors.

A longitudinal approach to primary data collection was adopted whereby each respondent was interviewed four times, over a period of 4 years (i.e. 2009–2012). Given the complexity of the phenomenon, this approach was adopted for two reasons: firstly, it gave respondents the time needed to reflect on knowledge management issues idiosyncratic to the Chinese hospitality industry. Secondly, it gave the authors' time to probe evidence gathered from other sources and focus interviews on factors identified as specific to knowledge management processes in the Chinese context. The semi-structured interview questions were formulated to cover the necessary knowledge management issues, yet framed in an open-ended manner. This design provided respondents with ample scope for introspection, and the freedom to pursue and openly report on matters they considered important. In total, 46 semi-structured interviews were conducted over the 4 years, each with a duration of 60–120 min. Secondary data were sourced from historical publications pertaining to China's economic and hospitality industry development, annual reports of the Marriott group, local and international newspaper archives, and Chinese government agency publications. This collection of data from various sources provided triangulation and subsequently assisted in the aggregation, analysis, and validation of information.

A rigorous content analysis process using the five-stage protocol forwarded by Refs. [24–26] was applied to each interview transcript and all secondary data. The content analysis and the verification of the conclusions drawn were facilitated by the use of the NVIVO software package. During Stage One of the content analysis, the aims and objectives of the research were identified. During Stage Two, the interview transcripts and secondary data were converted into MS Word[®] format and entered into a codified NVIVO database. Stage Three comprised an interrogation of the coded data to detect any significant themes emerging with regard to knowledge management issues in Marriott's strategic management of operations in China. These emergent trends and themes provided the basis for the second round of data coding categories. Stage Four of the content analysis involved a refinement of the second-round coding results, while Stage Five involved finalisation of the research findings. These findings are discussed in the following sections, along with implications for future research in the cross-cultural knowledge management field.

4. Marriott's knowledge management process in China's hotel industry

The second stage coding of the annual report data indicated that Marriott's knowledge management (as developed for the Chinese hotel industry) followed a five-stage process: (1) assessing the organisation's cross-cultural knowledge-base, (2) accessing domestic labour-market knowledge, (3) developing domestic labour-market/domestic employee knowledge, (4) developing domestic supply-chain knowledge, and (5) managing domestic employee satisfaction and commitment. Each of these five stages will be discussed in the following sections.

4.1. Stage 1: assessing the organisation's cross-cultural knowledge-base

During its entry stage into the Chinese hotel industry (i.e. the early 1990s), Marriott's senior management identified that a cross-cultural knowledge deficiency gap existed with regards to its Chinese operations. To address this deficiency and to simultaneously improve its image in the Chinese labour market, Marriott's senior management implemented strategies to (1) increase its capacity to understand the Chinese culture through its labour market and (2) establish its reputation as an 'employer of choice' in the Chinese hotel industry. Initially, Marriott implemented a two-tier brand reputation building and recruitment strategy; at the first-tier level, Marriott used their pool of existing domestic managers to work closely with (and gather data from) domestic hotel owners, recruitment agents and government officials regarding operational matters. At the second-tier level, Marriott used experienced expatriate managers to focus on working with the higher level national and local governments on its brand and reputation:

We always have a pool of well-trained and experienced managers ready to be posted anywhere in the world. The combination of this pool of foreign managers supported by our selected domestic managers is one of Marriott's strengths (Human Resources Director 3, Personal interview, 2010).

... its two-tier brand [reputation] building strategy, Marriott able to depend on management capability to effectively negotiate between domestic hotel owners, governments and Marriott's head office in the United States (General Manager 1, Personal Interview, 2011).

As part of this two-tier recruitment strategy, Marriott sourced skilled and qualified domestic workers to fill essential positions across its hotels and overcome its deficiency in terms of local cultural knowledge. The central part of this strategy was leveraging the personal networks of experienced domestic managers and using them to identify and recruit qualified and appropriately skilled workers. These personal networks comprised past work colleagues, family members and relatives, friends and government officials. Marriott's senior management also drew upon the negotiation expertise and comprehensive understanding of the domestic employment context possessed by their experienced domestic managers. These insights provided key inputs into public relations activities that were designed to promote Marriott's reputation and assist in the negotiation of employment contracts: We have a better understanding of domestic employment regulations and what governments are trying to achieve in the labour market...in a much better position to negotiate with relevant authorities regarding employment terms and conditions (Human Director 1, Personal Interview, 2010).

Marriott realised that if it were to continually improve its reputation as an employer of choice in the Chinese labour market (and, therefore, be able to attract skilled and qualified domestic workers), it would need to continue to improve its understanding of the Chinese market and establish a competitive advantage distinct from its international competitors. To achieve this aim, the Marriott website was used as a channel to promote the company to the domestic Chinese labour market and increase its brand reputation in the domestic market:

In the last three years [2002–2005], developed seventeen international sites...to learn about different cultures, and how to tailor sites for specific markets. We're interested in understanding Chinese consumer habits and upcoming market trends (Director of User Experience, in Ref. [27]).

4.2. Stage 2: accessing domestic labour-market knowledge

During the early 1990s, the Chinese hotel industry was considered an attractive industry to work in by the domestic workforce. However, with the influx of international hotels and consistent double-digit gross domestic product growth, China developed into a highly competitive labour market for hospitality workers. As the skilled and educated workers sought more secure and lucrative careers with government departments and multinational organisations in other industries, the hotel industry was faced with ongoing skill-shortages. Marriott recognised that in order to recruit enough skilled workers, they needed to improve their domestic management expertise and knowledge (to compete with other industries for domestic talent). In order to achieve this outcome, Marriott implemented a modified version of one of their successful recruitment strategies employed in the United States during the 1960s; Marriott actively recruited veteran Chinese hotel operators from surrounding countries, such as Hong Kong and Singapore, to improve their domestic management expertise and knowledge. According to Marriott's senior management, the modified recruitment strategy enabled Marriott to effectively source many of its skilled and qualified workers from across various provinces and cities:

[Domestic] managers are all Marriott people and the first job they do is to find domestic talent, as they know the domestic market (General Manager 1, Personal Interview, 2012).

...skilled employees with good language skills are one of the most important success factors in the Chinese hotel industry today...our extensive domestic networks assist us in finding our potential associates to fill positions across different functions and in different cities (General Manager 3, Personal Interview, 2012).

In order to build on its reputation as an employer of choice, Marriott's senior management also implemented a series of strategies to promote and retain its existing experienced domestic Chinese managers. One such strategy was to provide experienced domestic managers with a distinctive career succession plan and promoting them to the position of general manager in newly opened hotels in second and third-tier cities. This succession strategy was designed to leverage their experienced domestic managers' expertise, knowledge and networks to serve the emerging domestic segments in the second and third-tier cities and provide a retention incentive:

Our US managers not very effective in communicating with our Chinese key stakeholders [suppliers, hotel owners, and domestic government officials], needed domestic managers who speak Chinese and understand local culture to work closely with stakeholders to achieve desired outcomes for all parties (Human Resources Director 1, Personal Interview, 2009).

Marriott uses the second-tier cities as part of their domestic executive promotion strategy...it gives Marriott the domestic representation it needs in these cities, it also acts as an incentive to retain managers and to increase staff loyalty (Key Informant 1, Personal Interview, 2012).

4.3. Stage 3: developing domestic labour-market/domestic employee knowledge

During the early stages of its operation in China, Marriott's senior management found that the level of trust and productivity between the domestic employees and management was substantially lower in China than in the US. From their experience, Marriott recognised that if they were to successfully build trust and productive relationships with their domestic employees, they needed to gain their loyalty through organisational culture-based training and development initiatives. Marriott leveraged their experienced managers to work closely with employees (whom they term 'associates') to achieve required organisational standards and at the same time, to instil its organisational culture of the 'Marriott's Way':

...in the early 1990s, hotel employees had no clue and no ideas of how to handle their own grooming and personal hygiene, not to mention providing quality service to guests... started with very basic training to get them up to our minimal standards, then slowly introduced standard operating procedures (General Manager 1, Personal Interview, 2009).

...our domestic associates needed to be shown how to perform tasks and to be reminded how to work within the boundaries of procedures and processes (Human Resources Director 2, Personal Interview, 2010).

According to Marriott, the 'Marriott's Way' enabled the company to build strong relationships with its employees that encouraged them to share tacit knowledge across the company's structure:

Our competitive advantage is that we care about our people, our associates...the "Marriott's Way" is the core capability to our success for over eighty years, based on the philosophy of if you treat your associates well, in turn they will treat your customers well (General Manager 2, Personal Interview, 2011).

Overall, good experienced managers are still looked upon as the person who set the brand standards in a country where human capital is still at a development stage and the "Marriott's Way" is still very effective in associates' training and development. Hopefully, we can keep them for as long as possible before our competitors poach them away (General Manager 1, Personal Interview, 2012).

Marriott attempted to reinforce a strong bond with its domestic Chinese employees by conducting a series of ongoing targeted training and development programs throughout their careers. For example, an average 78 h of on-the-job training (plus another 34 h of professional development) is provided to each associate every year; for more senior employees, Marriott provided a specialised career and training paths to ensure they were prepared for future management positions: ...in the 1990s and even in the early 2000s investing in your employees and treating them well were not common management practices in China...Marriott is perceived as a very good employer in the Chinese market (Human Resources Director 1, Personal Interview, 2012).

Marriott is very good with in-house training programs....we spending around US\$800.00 per manager per year in training expense back in the early 2000s...we value our skilled associates and in China where hospitality skills are very hard to find, we use our in-house training programs to increase our available pool of skilled human resources (General Manager 1, Personal Interview, 2011).

4.4. Stage 4: developing domestic supply-chain knowledge

To continually and systematically promote Marriott as an employer of choice within the labour market, its senior management actively sought to engage with supply-chain members of the hospitality industry. For example, Marriott entered into arrangements with domestic hospitality training schools across China (as well as selectively inviting suitable candidates to participate in their internship program). Marriott's senior management state that their talent building campaigns focused on both the internal and external labour markets have provided them with a skilled labour force providing consistent delivery of its brand promises:

We [Marriott] work closely with domestic hospitality training schools in China...we get good students from some of these schools...our in-house training programs help students to get up to our standards pretty quickly once they have started with us (Human Resources Director 3, Personal Interview, 2011).

Marriott also promoted the 'Marriott's Way' organisational culture (on their "Destination Marriott" career website) to local supply companies to reinforce the quality standards required by the company. The 'Destination Marriott' career website was designed to assist potential employees to apply for the various careers available at Marriott worldwide and to highlight the various benefits and rewards enjoyed by Marriott's associates. For the more senior positions within Marriott, senior management continued to leverage their domestic managers' networks to seek out and appoint experienced managers from other industry sectors, as well as from direct competitors.

A fluent Mandarin speaking expatriate manager is still a rare resource in today's market, but we commonly transfer them across our hotels internally....one of our immediate issues is finding enough English speaking associates...English language skills are still considered to be low in China and that causes us concern with regard to our service quality level (Human Resources Director 1, Personal Interview, 2011).

The country [China] and its cities can build the world's most outstanding or tallest buildings...but the skills level is still very much behind most of the developed markets (General Manager 5, Personal Interview, 2011).

As part of Marriott's strategy in China, its senior management continued to actively seek high growth opportunities in second and third-tier cities outside the major urban areas. These cities are located outside key gateway cities such as Shanghai and Beijing in different provinces across China. The local governments, supply chain members and customers were the three major market segments identified in these second and third-tier cities. Marriott's senior management recognised that if they were to compete effectively in these emerging market segments,

they needed to communicate key messages to these segments which focussed on how the company could better serve and satisfy their needs. To achieve this, Marriott implemented a modified recruitment strategy that they had employed when first expanding into China (i.e. by appointing experienced senior domestic managers to work closely with these three major emerging market segments in second and third-tier cities):

Our growth has been fast in emerging markets like China, but we have to do things differently to succeed. Manpower is an issue. Finding experienced associates is difficult....Awareness is an issue, so public relations and local marketing efforts are important to establish the brand (Senior Vice President of International Marketing, in Ref. [28]).

...we get relocated to second-tier cities to manage our major clients [domestic governments and key stakeholders] and to train our management team to ensure our domestic customers get the services they want (General Manager 4, Personal Interview, 2009).

4.5. Stage 5: managing domestic employee satisfaction and commitment

As skill shortages continued to be one of the major challenges faced by the hotel industry in China, Marriott implemented a number of strategies designed to build staff loyalty with both its internal and external labour markets. Marriott's senior management implemented internal employment strategies which were designed to promote and retain experienced domestic managers. Further, as part of their internal staff loyalty building campaigns in China, a variety of in-house multi-disciplinary training and development programs were implemented which were designed to up skill all associates. For example, Marriott continued to develop and promote their experienced domestic female managers to senior management positions as part of its Global Diversity and Inclusion policy and Women's Leadership Development Initiative:

Human capital capabilities are becoming more important than having a western manager...western management strategy no longer requires a certain number of foreigners working as staff to operate in China....Foreign managers no longer have the monopoly of first pick of all the good positions (Human Resources Director 3, Personal Interview, 2012).

...with good support and training from Marriott's international office, we able to achieve good results from some of the newly implemented processes and systems in areas such as hotel operation standards, overall food and beverage offerings and management and general customer service...but up-skilling our associates took longer...a continuous process (General Manager 1, Personal Interview, 2010).

As previously discussed, according to Marriott's senior management, the use of action and policies designed to communicate and instil the ethos of "Marriott's Way" enabled the company to build productive relationships with its associates as well as to motivate them to improve their performance:

... if you see some strangers talking to your top managers, you can be certain that your top managers are being poached by your competitors (General Manager 4, Personal Interview, 2011).

It is becoming more difficult to employ and retain younger, quality staff because there are so many opportunities now open to young people in China (Director of User Experience, in Ref. [28]).

To maintain its market leadership position in China, Marriott continued to strategically build and market their brand as the employer of choice in the domestic market. Being the employer of choice enabled Marriott to select and recruit a skilled workforce to maintain and deliver its quality service level and to combat protected domestic competitors:

...the one child policy has contributed to the current skill-shortage situation in China. However, hotels are also competing with the domestic government departments for skilled employees, as the domestic governments are now providing benefits such as welfare, less hours and higher salaries to attract talents (Human Resources Director 3, Personal Interview, 2009).

Marriott was awarded the top employer in China by the Corporate Research Foundation Institute in 2012 (Human Resources Director 1, Personal Interview, 2012).

Marriott builds on its success by using initial strategy coupled with domestic experience and learning...Marriott's new processes and services in China are blended with domestic culture to create a unique quality standard (General Manager 1, Personal interview, 2012).

5. Concluding remarks

Given the results of this qualitative analysis, we propose the following five-stage knowledge management process model (see **Table 1**) for the emerging Chinese hotel industry. This process model illustrates Marriott's experience in assessing its knowledge management deficiencies in its Chinese operations, and the human resource/strategic management practices used to address them.

Stage One	Assessing the organisation's cross-cultural knowledge-base
Stage Two	Accessing domestic labour-market knowledge
Stage Three	Developing domestic labour-market/domestic employee knowledge
Stage Four	Developing domestic supply-chain knowledge
Stage Five	Managing domestic employee satisfaction and commitment

Table 1. A knowledge management process model for the Chinese hotel industry.

In terms of assessing the organisation's cross-cultural knowledge base, Marriott utilised its human resource inventory systems not only to audit the store of human resources it controlled in its Chinese hotel operations but also to gauge the extent to which the organisation had the capacity to access the tacit knowledge it possessed. Strategically, therefore, the initial stage of the knowledge management process required not only an understanding of what information/ knowledge deficiencies the organisation possessed, but also what data-gathering deficiencies it had in this regard. In terms of accessing domestic labour-market knowledge, Marriott recognised the importance of its recruitment and selection processes; in particular, it recognised that it had to adapt its recruitment and selection processes (that had been effective in the Western context) to account for the idiosyncrasies of the Chinese labour market. Strategically, Marriott recognised the need to establish itself as an "employer of choice" within the Chinese hotel industry to attract, motivate and retain the highest quality employees. In terms of developing its domestic labour-market/domestic employee knowledge, Marriott sought to develop strong employee-employer relationships with its domestic Chinese employees via its training and development programs. In addition to the organisation's standard programs (which focused on improving operational performance), the company also sought to improve the levels of trust in the organisation through training and development programs focused on indoctrination into the organisation's culture. In terms of developing domestic supply-chain knowledge, Marriott ensured that it maintained dynamic and mutually beneficial relationships with domestic suppliers, educational institutions and government departments. Lastly, Marriott found that one of the most effective ways of protecting the store of knowledge they had generated in the Chinese hotel industry was to take measures to ensure that their domestic employees were satisfied with their employment relationship and reported high levels of organisational commitment. Indeed, in the face of intense competition from within China's expanding hotel industry (and from employees in other industry sectors), the raising of the 'exit barriers' for their domestic employees represented a strategic priority.

Given these findings and the potential implications for Western organisations' knowledge management practices within the Chinese hospitality industry, we suggest the following avenues for future cross-cultural knowledge management research within the Chinese context (and emerging Asian economies generally): firstly, we believe that there needs to be further in-depth analysis of the link between knowledge management practices and 'employer-of-choice' and 'industry-of-choice' concepts within the Chinese hospitality/ emerging market context. A finer-grained understanding of knowledge management practices within China may enable Western organisations to become employers-of-choice and foster a strong organisational culture. For example, at an organisational level such investigation could inform the design of superior human resource and employer-marketing strategies that communicate the expected international standards of skill and behaviour to both existing and potential domestic employees. At an industry level, this research suggests the need for industry-wide marketing efforts formulated to promote hospitality as an industry of choice providing valuable long-term career options. The absence of such industry-based marketing efforts may result in the hospitality industry being seen as nothing more than a training ground for any alternate industry better able to market their career opportunities. It is anticipated that outputs from this research agenda would not only benefit both domestic operators looking to improve their competitive stance within this highly dynamic market and international hospitality operators engaged in (or considering) investment in the Chinese market, but importantly, it would contribute valuable insights and extend the current store of knowledge management theory.

Author details

Tommy Wong*, Linda French and Mark Wickham

*Address all correspondence to: tommy.wong@utas.edu.au

University of Tasmania, Hobart, Tasmania, Australia

References

- Evans MM, Dalkir K, Bidian C. A holistic view of the knowledge life cycle: the knowledge management cycle (KMC) model. The Electronic Journal of Knowledge Management. 2014;12:75-97.
- [2] Wang P, Tong TW, Koh CP. An integrated model of knowledge transfer from MNC parent to China subsidiary. World Business. 2004;**39**:168-182.
- [3] Hall DJ, Paradice D. Philosophical foundations for a learning-oriented knowledge management system. The Journal of Decision Support System. 2005;**39**:445-461.
- [4] Luo Y. Dimensions of knowledge: comparing Asian and Western MNEs in China. Asia Pacific Journal of Management. 1999;16:75-93.
- [5] Bose TK. Advantages and disadvantages of FDI in China and India. Journal of International Business Research. 2012;5:164-174.
- [6] Enderwick P. Large emerging markets LEMs and international strategy. International Marketing Review. 2009;26:7-16.
- [7] Wong T, Wickham M. An examination of Marriott's entry into the Chinese hospitality industry: a brand equity perspective. Tourism Management. 2015;**48**:439-454.
- [8] Guillet BD, Zhang HQ, Gao BW. Interpreting the mind of multinational hotel investors: future trends and implications in China. International Journal of Hospitality Management. 2011;30:222-232.
- [9] Hays J. Foreign companies and foreign investment in China [Internet]. 2008. Available from: http://factsanddetails.com/china.php?itemid=349 [Accessed on 2014-07-03].
- [10] Tsui-Auch LS, Möllering G. Wary managers: unfavourable environment, perceived vulnerability, and the development of trust of foreign enterprises in China. Journal of International Business Studies. 2010;41:1016-2506.
- [11] Cooper C. Knowledge management and tourism. Annals of Tourism Research. 2006;33: 47-64.
- [12] Hallin CA, Marnburg E. Knowledge management in the hospitality industry: a review of empirical research. Tourism Management. 2008;**29**:366-381.
- [13] Peng J, Moffett S, McAdam R. Knowledge management in China: a review. Journal of Technology Management. 2010;5:158-175.
- [14] Nieves J, Quintana A, Osorio J. Knowledge-based resources and innovation in the hotel industry. International Journal of Hospitality Management. 2014;**38**:65-73.
- [15] Diakoulakis IE, Georgopolous NB, Koulouriotis DE, Emiris DM. Towards a holistic knowledge management model. The Journal of Knowledge Management. 2004;8:32-46.

- [16] Tuomi I. The future of knowledge management. Lifelong Learning in Europe. 2002;VII: 69-79.
- [17] Bjorkman I, Xiucheng F. Human resource management and the performance of western firms in China. The International Journal of Human Resource Management. 2002;13: 853-864.
- [18] Hitt MA, Lee HU, Yucel E. The importance of social capital to the management of multinational enterprises: relational networks among Asian and Western firms. Asia Pacific Journal of Management. 2002;19:353-372.
- [19] Nieves J, Haller S. Building dynamic capabilities through knowledge resources. Tourism Management. 2014;40:224-232.
- [20] Cooper C. Managing tourism knowledge. Tourism Recreation Research. 2015;40: 107-119.
- [21] Werner K, Dickson G, Hyde KF. Learning and knowledge transfer processes in a megaevent context: the case of the 2011 Rugby World Cup. Tourism Management. 2015;48: 174-187.
- [22] Martinez-Martinez A, Cegarra-Navarro JG, Garcia-Perez A. Environmental knowledge management: a long-term enabler of tourism development. Tourism Management. 2015;50:281-291.
- [23] Hartmann E, Feisel E, Schober H. Talent management of western MNCs in China: balancing global integration and local responsiveness. Journal of World Business. 2010;45: 169-178.
- [24] Finn M, Elliott-White EM, Walton M. The analysis of qualitative data: content analysis and semiological analysis. In Tourism and leisure research methods: data collection, analysis and interpretation. Harlow: Pearson Education; 2000.
- [25] Hodson R. Analyzing documentary accounts. Thousand Oaks: Sage Publications; 1999.
- [26] Neumann WL. Social research methods (5th edition). Upper Saddle River: Prentice Hall; 2003.
- [27] PR Newswire. Marriott International launches water conservation program in China [Internet]. 2010. Available from: http://www.prnewswire.com/news-releases/marriottinternational-launches-water-conservation-program-in-china-94714449.html [Accessed on 2015-06-22].
- [28] Babitch S, Chen J. Design for the emerging markets: interview with Marriott International. Chicago: Institute of Design; 2005.

Widening the Understanding of Risk Approaches by Comparing Definitions from Different Disciplines

Gabriele Berg-Beckhoff, Peter Wiedemann,

Balázs Ádám, Joachim Schüz,

Kristian Breum Ølgaard,

Pernille Tanggaard Andersen,

Steven Ndugwa Kabwama and Jesper Bo Nielsen

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/intechopen.70074

Abstract

The aim of this chapter is to critically reflect definitions of hazard, risk, and risk perception and their assessments used in different scientific disciplines and give examples of the potential implications for scientific discussions, knowledge management, and risk communication. Scientists with backgrounds in public health, psychology, environmental health, occupational health, engineering, sociology, and medicine were asked for a definition of hazard, risk, risk assessment, and risk perception seen from their specific scientific disciplines. Hazard is generally seen as an adverse event or condition. For most risk definitions, probability and severity are important aspects. Often a quantification of risk is desired, whereas risk perception is seen as a subjective appraisal and a cognitive construct. As risk perceptions are based on a combination of knowledge and individual values and affects, it may not provide a reliable guidance for risk management decisions on a societal level. Discipline differences are mainly connected to terminology and interpretation of key concepts, but the differences are based on different tasks and perspectives. For dealing with controversies in science across disciplines, an acceptance and appreciation of terminology and perspectives from different scientific disciplines are needed to ensure a transparent risk assessment process.

Keywords: risk, hazard, risk perception, risk assessment, multidisciplinary



1. Introduction

The aim of this chapter is to critically reflect on definitions used in different disciplines during the procedure of risk management. Knowledge management is an important part of risk management and can be defined as a collaborative and integrated approach to the creation, capture, organization, access, and use of an intellectual asset [1]. This definition underlines the importance of knowledge management in risk management when different disciplines work together to identify the hazard, assess the risk, and finally predict mitigation mechanism [2]. To facilitate communication between different disciplines, we compared how risk management was approached in different disciplines. We recognized that harmonization of these approaches or definitions would be contraproductive as it would undermine the variety of knowledge and the tasks and perspectives of the different disciplines during the process of risk management. We, therefore, suggest to widen the understanding through appreciation of these differences between disciplines and not to harmonize definitions.

2. Method

The following disciplines were involved: public health, psychology, environmental health, occupational health, engineering, sociology, and medicine. The selection was made to get a wide variety of disciplines working in different areas but all with a link to risk and risk management. All selected participants have coauthored this chapter. They were recognized experts of risk-related disciplines. They expressed their opinions and synthetized conclusion in a reiterative process. Each scientist was asked to present his/her own views on the given topics. The following questions were developed to get comparable responses from the participants: What is hazard and risk? How is hazard/risk assessed and which practice of hazard identification and risk assessment is used? and How is risk perception defined? The results were summarized, commented, and discussed by all authors. This method is appropriate as it allows and discusses different opinions. The aim was not to agree on one set of definitions, but critically reflect on the discipline-related definitions of hazard, risk, and risk perception, and their assessments.

3. Introduction in risk research

Risk is human being's attempt to understand and deal with life's dangers [3]. Thus, the main reason for talking about hazard and risk is to have a sufficiently accurate perception of the situation to make decisions and manage situations in a manner minimizing the probability of adverse effects. In some instances, these decisions are strictly personal and have implications for the individual only, but in other cases, the decision may have implications for larger groups of people, even for the whole population, or for the environment. Disciplines predominantly dealing with numbers and quantifications define risks based on a calculable phenomenor; biological, natural, and technological scientists define risks as objective reality (mostly also in quantitative ways); sociologists view it as a social and cultural construct, whereas psychology looks at it as a cognitive and behavioral phenomenon. Each discipline

commonly utilizes their own terms about risks, while keeping different perspectives on what it is [4]. Also, risk communication is hampered by this semantic ambiguity not only in the communication between stakeholders and lay people, but also between different scientific disciplines working together to manage risk [4, 5].

Knowledge management plays an important role in risk issues, for example risk analysis and risk regulations, and knowledge transfer and knowledge sharing are important parts in risk communication. Explicit knowledge is easier to communicate which clarifies the particular problem with newly emerging risks, where explicit or generalized knowledge is unfortunately not available [6].

4. Knowledge about hazard and risk

In the historical perspective, no notion of risk is to be found in traditional cultures: preindustrial hazards or dangers like famines, plagues, or natural disasters were experienced as pregiven. They came from "others" — gods, nature, or demons [7]. However, when carefully reinterpreting the historical perspective, it appears that the notion of risk was already there, albeit implicitly, because one does not have to know the origin of a hazard in order to apply mitigation measures to avoid harm and suffering. One could even argue that sacrificing a virgin to please the god to avoid plagues or nature catastrophes involved an understanding and/or management of a hazard. It may well be that preventive actions taken today will in the future be regarded as similarly useless and surprisingly little evidence-based as sacrificing virgins to prevent plagues is regarded today.

In the historical perspective, first with the beginning of societal attempts to control risk, and particularly with the idea of steering toward a future of predictable security, the consequences of risk became political issues. Thus, it is a societal intervention—in the form of decision-making attempts to transform incalculable hazards into calculable risks.

Nowadays, it is necessary to separate the notions of risk and hazard. Knowledge about the hazard should be present before labeling any event as "risky." Hazard can be defined as an event [8] but it can also be considered as a condition or factor with a potential for causing an event, thus, as a synonym to danger. In this way, hazard is a qualitative term that tells whether exposure to a chemical or drug or certain behavior such as physical inactivity has the potential to cause an undesirable outcome on human health or other things we value, e.g., the environment. The evaluation of an event as a hazard is a mix of objective and subjective data, with the latter depending on individual or at least cultural preferences. The scientific component of the evaluation of the hazard aims to be objective and with a dichotomous (qualitative) outcome, namely being hazardous or not. Scientific uncertainty adds to complexity, if for instance studies on the carcinogenicity of a chemical are controversial, which may add some subjective elements (different researchers interpreting the same studies in different ways) and a categorical scale replacing the dichotomous one to express the level of scientific uncertainty. Nevertheless, researchers then continue to collect further evidence until ultimately the hazard can be established or rejected. Objectively, hazards can be fatal or life-threatening, leading to disability or only to temporary discomfort; hence, classified according to dimensions of impact on life and of reversibility. The assessment of severity is a quantitative part of hazard evaluation, and the subjective acceptance is the qualitative part. It is subjective because it is the individual's own judgment reflecting their values and preferences.

On the basis of these facts, a dichotomous hazard will be transformed into a quantitative risk term [9]. In all disciplines, there are two most known concepts of risk definition: the probability of occurrence and severity of the undesirable outcome. Under this approach, risk is the probability function that reflects in quantitative terms the likelihood that a hazard manifests itself while as we have alluded to earlier, the undesirable outcome is the hazard.

5. Dealing with knowledge during risk assessment

Normally, scientific risk assessment should be performed before labeling any event as "risky." Identification of a hazard is the first step in the risk assessment process. Before identifying the hazard, potential disease or outcome clusters need to be found. For example, before 1989 only one disease outbreak per year was identified in a population of over 60 million people in the Philippines. A surveillance system was set up in 1989 and in 1995 more than 80 disease outbreaks were identified [10]. This means that risks cannot be described as such unless there are procedures to identify or measure outcome clusters or hazards (perceived or real) and our vulnerability to them. Another point worth mentioning is that if you do not know the hazard, you cannot perceive the related risk. The second stage of the risk assessment process entails an estimate of the associated level and extent of potential harm which together with the expected probability of an unwanted event are important, because the evaluation of the acceptability of the options for mitigating the identified risk will usually depend on how much harm the hazard we identify can be estimated to make. For the assessment procedure, two forms of assessment are discussed in detail: the technical procedure and the observational approach.

In the technical procedure like *engineering*, material properties, technical performances, or technical measurements are based on tests or experiments. The results are countable from numerous versions of repeated experiments. The experimental approach allows estimating a small confidence interval considering probabilistic calculations. One example is the overall construction procedure in the design of the roof for a building. First, the total design load on the roof must be computed. The different loads could typically be: self-weight, wind load, and snow load. A design load can then be computed as the sum of these loads. However, as the loads are stochastic in nature, each load is multiplied by a coefficient that accounts for the lower probability that all loads are maximal at the same time. Strengthening the construction is intended to reduce the risk of failure, but will never eliminate it, and the design of a construction is, therefore, a balance between strength and acceptable risk for failure.

In the observational approaches of *epidemiological and public health* studies, the outcomes of interest are often death or diseases like cancer, myocardial infarction, or infectious diseases. There may be several causes for having that outcome and several hazards may need to be considered for the risk assessment. One cause hardly ever leads to only one single outcome. Stress at work, for example, is a risk factor for myocardial infarction. But other risk factors like hypertension, overweight, unhealthy blood lipid profile, or lack of physical activity are

also risk factors for myocardial infarction and all of them may be related to stress at work. On the other hand, overweight has more than one adverse consequence, including a higher risk of some cancers or diabetes. The complexity is topped by the inherent and much higher heterogeneity in susceptibility to an adverse outcome in a population of humans than among construction parts in engineering. Risk assessment is still possible; and for the quantitative estimation of absolute or relative risk, epidemiological measures are used by calculating disease rate and comparing rates of exposed and unexposed groups [11].

Risk assessment in *occupational health* is traditionally based on a four-stage process: hazard identification, exposure assessment, dose-response assessment, and risk characterization. However, risk assessment is typically put into the broader context of managing occupational risks and often includes ongoing monitoring and evaluation of the effectiveness of any initiative to decrease risk to an acceptable level. An acceptable risk level is defined not solely by health arguments but, especially in the case of carcinogenic agents that have stochastic health effects, such as hexavalent chromium exposure in electroplating causing lung cancer, also includes considerations on financial and technical feasibility. Since the main reason of risk assessment is to determine what, if any, measures should be taken by the employer to prevent adverse health effects in workers, it is common to integrate risk assessment and risk management in occupational health practice [12, 13]. Occupational health and safety is, therefore, one of the most regulated areas in terms of legal requirements for risk assessment [14].

Medical and environmental risk assessments to guide risk management become increasingly challenging, when certain exposures have both risks and benefits, or when the hazard is intended or inherent in a specific scenario. Medical application is a common example of balancing benefit and harm. The use of computed tomography (CT) with its high resolution can clearly lead to better diagnosis and planning of treatment and then becomes lifesaving, but given its exposure to ionizing radiation, a well-known carcinogen, endorses recommendations that unnecessary examinations need to be avoided and optimal dose adjustment, for example for children, is to be applied [15]. However, there may clearly be a huge risk related to avoiding the hazard of undergoing a CT examination, for instance, due to a delayed detection of cancer. Also, in environmental health not all hazards are entirely avoidable, but risk assessment can guide policy decisions insofar as where to set priorities for an acceptable level of risk and of how much risks can be reduced.

6. Dealing with knowledge in risk perception

Identifying hazard and severity of a damage has direct consequences for risk perception: "Perception is a process by which individuals select, organize and interpret stimuli to generate a coherent and meaningful picture" [16]. In contrast to risk assessment, which can have a scientific basis with a structure and procedures, risk perception is subjective, involves affects and there are many factors that can influence the way a risk is perceived.

Literally, risk cannot be perceived. Therefore, *psychological research* focusing on risk perception explores how judgments about the riskiness of an event, substance, agent, or technology are made in which cognitive, affective, and moral processes are involved. Although the first risk perception studies were conducted more than 50 years ago, the concept of "perceived risk"

only began to gain popularity in the mid-1970s [17]. Slovic's approach known as the psychometric paradigm aims at the "quantitative description of the cognitive and evaluative mental structure of risk and its determinants" [18]. His research focuses on "what people mean when they say that something is (or is not) 'risky', and to determine what factors underlie those perceptions" [19]. It focuses on the beliefs laypersons hold with regard to how chemicals cause health risks. For instance, laypersons are less sensitive to dose-response relationships compared to toxicologists suggesting that laypersons do not differentiate between the qualitative hazard and the quantitative risk, and therefore do not see risk as something that can be increased or reduced due to interventions or changes in exposure levels. This misconception is a continuing challenge in medical risk communication because most interventions, especially health promoting interventions, reduce but do not eliminate risks. Risk of an adverse event can generally only be reduced, but specific risks related to a certain hazard might be eliminated if the hazard is eliminated, though it may take some time. If you quit smoking, after 20 smokeless years, you will have vascular disease mortality comparable with never smokers [20], but like the rest of the population you still may die of vascular disease because of other hazards leading to the same adverse event.

From the rational point of view, to evaluate a risk, you need to know something about the hazard, assess probabilities as well as consequences of the negative outcome. In fact, however, intuitive risk judgment has a strong affective component, and it is associated with moral concerns and may neglect any probability consideration in risk appraisal. Doing something about your personal risk depends on how you understand the risk. This is a subjective exercise including the compilation of the more objective information on hazard and risk in combination with individual assessments on severity of risk and potential side effects, and last but not least on personal values and preferences.

From the sociological point of view, probabilistic risk assessment alone is unsociological. Sociological understanding of risk needs additional social, cultural, and historical contexts. Sociology explains how risks are handled in a society and might thereby clarify potential public misperceptions of risk. When technically defined risks are not contextually generated, public risks understanding will be defined as wrong, if it differs from the factual experts' opinion. From the sociological point of view, it is an error not considering contextual factors like traditions, culture, norms, etc. even though that does not mean that the factual knowledge is wrong. The role of sociology, when it comes to risk, is to explain why errors in dealing with risk in a society occur but not the truth behind them. Implication of this perspective deals with risk communication. The factual knowledge about specific risks needs to be understood and communicated. Sociology emphasizes that when dealing with communication it is important to consider different segments of the public understanding of risks to avoid misperception. Sociological amplification of risk explains how risks and events or accidents interact with psychological, social, institutional, and cultural processes, and often increase or decrease risk perception, public concern, and thereby risk behavior. The objective risk assessment provides information about the risk, but for a sociologist the task is to analyze how this information is understood and reflected in practical social actions in the public. Risks and hazards are sociocultural events and they are not only neutral facts that generate specific signals. It should be considered that also an assessment of a risk can be seen as a construction of a risk and hazard. This implies for sociology to see and explain risk as an overall product of social and cultural processes [21].

7. Knowledge management considering different disciplines

All risk concepts of the different disciplines have one element in common; the distinction between reality and uncertainty. **Figure 1** illustrates the tasks of the different disciplines in the process of risk management. Engineering and occupational health focuses on hazard and risk identification (explicit knowledge). Sociology looks at the risk on humans and society (social knowledge). Psychology works with risk perception and manages risk on a personal level (individual knowledge). The latter two disciplines rely on the first to identify and describe risk as objective as possible, and the former depend on the psychology and sociology to translate their observations and calculations into something usable for individuals and society. Therefore, it is essential that these disciplines keep their definition and taxonomy of risk knowledge. However, interaction and knowledge sharing is important for the overall risk management process.

The engineer tells the occupational hygienist technical details for the risk assessment; the occupational hygienist communicate risk assessment results to medicine to allow the best treatment for the exposed individual; the doctor needs to explain the information to the patient. Finally, the patient should perceive the risk in that manner to do the right action. Already for a straightforward treatment, a long chain of knowledge transfer needs to work. However, risk



Figure 1. Concepts of considered disciplines while defining risk.

assessment and management to reduce the risks is rarely straightforward, especially when scientific uncertainty and, at the same time, competing benefits and risks come into play. A recent example is management of potential risks related to the use of electronic cigarettes [22, 23], for which on the one hand, long-term adverse health effects are expected, but on the other hand, electronic cigarettes are also expected to be safer than conventional cigarettes to facilitate to quit smoking. Public health researchers mainly request strict regulation to minimize potential future health effects and avoiding electronic cigarettes to become a new socially acceptable way of smoking, especially among youngsters, and thereby torpedoing the efforts for achieving a smoke-free environment.

Therefore, it is very important to understand underlying terminology from different disciplines. But even more important is to understand their tasks and their views on the topic. For example, the role of sociology "to explain why errors occur but not the truth behind them" clarifies that communication is more important for the sociological perspective than the risk assessment procedure itself, even though both parts are necessary for the overall risk management. If these different views are not taken into account, conflicts and misunderstandings can happen.

In knowledge management, knowledge creation tools were often criticized. One critique is that most tools hide or eliminate important contextual information [24]. Furthermore, it could be shown that individual communication skills have more importance for perceived quality of risk knowledge sharing than the used technical systems [6]. Additionally, lack of time or awareness for the importance of sharing knowledge, missing communication skills, as well as differences in education and culture are seen as the most important barriers for interdisciplinary work [25]. In our opinion, collaborating researchers need to understand their used terms but also have to understand why there are differences. It is very important to know the terminology, tasks, and perspectives from different disciplines. Different terminology can be dealt with in a wiki development, a website that provides collaborative work on terminologies. However, further research is necessary on how to deal with different discipline-related aims or perspectives in the process of risk management.

Most differences between the disciplines are dealing with risk assessment. All disciplines have the theoretical wish to estimate the real quantitative risk and all disciplines agree that this is difficult to achieve or sometimes not even possible. Disciplines typically using probabilistic approaches, such as engineering or occupational health, define risks mathematically and more objectively. Consequently, these disciplines emphasize the importance of external validity and standardized procedures. Observational research is often connected with a black box phenomenon and is sometimes evaluated as giving less support to evidence than experiments. The presentation of results is more complex and difficult to understand and to trust. Consequently, in these disciplines, problems in objectivity and communication are more obvious. The cognitive approach seen in psychology and social sciences focuses more on the perception of risk. They evaluate the understanding of different risk presentations and look into emotions and cognitive processes while collecting risk assessment information. Due to the fact that risk assessment is seen as less objective, risks are seen as expectations or predictions. However, all disciplines agree that risk assessment should be as objective as possible. To allow for this, evidence-based information must be used to identify hazards and to assess

dose-response relationships, while exposure assessment should be based on documented, preferably quantitative, measurable, and representative data.

8. Conclusion

There is a broad common denominator between the disciplines; hazard is a negative event or condition; for most risk definitions, probability and severity of the risks are important; and finally risk perception is seen as subjective and a mental construct. Risk assessment should be evidence-based, preferably quantitative, measurable, and based on representative data. Assessment of concerns and perceptions is important as well but should be done separately.

Differences can be seen with regard to risk assessment between disciplines with cognitive and probabilistic approaches. However, differences are connected to the interpretation and terminology but not to procedures. Coping with scientific controversies is an important factor in risk management. Different scientific institutions often come to different conclusions and it is easy to select a specific opinion that supports only one view of risk. A transparent risk assessment framework considering different scientific perspectives is important to deal with controversies in risk science.

Knowledge management in risk management needs to account for different use of terminologies by different disciplines. It is important to consider the diversity of tasks and perspectives of various fields when defining terminologies and distributing work. A transparent risk assessment process can only be ensured with an acceptance and appreciation of terminologies and perspectives from different disciplines.

Author details

Gabriele Berg-Beckhoff^{1*}, Peter Wiedemann^{2,3}, Balázs Ádám⁴, Joachim Schüz⁵, Kristian Breum Ølgaard⁶, Pernille Tanggaard Andersen¹, Steven Ndugwa Kabwama¹ and Jesper Bo Nielsen⁷

*Address all correspondence to: gbergbeckhoff@health.sdu.dk

1 Unit for Health Promotion Research, University of Southern Denmark, Odense, Denmark

2 Science Forum EMF, Berlin, Germany

3 University of Wollongong, Wollongong, Australia

4 Department of Preventive Medicine, Faculty of Public Health, University of Debrecen, Hungary

5 Section of Environment and Radiation, International Agency for Research on Cancer (IARC), France

6 Ramboll Energy-Offshore Wind, Denmark

7 Research Unit for General Practice, University of Southern Denmark, Odense, Denmark

References

- Grey. What is Knowledge Management? The Knowledge Management Forum [Internet]. 1996. Available from: http://www.km-forum.org/t000008.htm [Accessed: July 14, 2016]
- [2] Zipperer L, Amori G. Knowledge management: An innovative risk management strategy. Journal of Healthcare Risk Management. 2011;30(4):8-14. DOI: 10.1002/jhrm.20064
- [3] Slovic P. Trust, emotion, sex, politics, and science: surveying the risk-assessment battlefield. Risk Analysis. 1999;19:689-701
- [4] Althaus CE. A disciplinary perspective on the epistemological status of risk. Risk Analysis. 2005;25:567-588. DOI:10.1111/j.1539-6924.2005.00625.x
- [5] German Institute for Risk Evaluation Bundesinstitut für Risikobewertung (BfR). Ulbig E, Hertel RF, Böl GF (Eds) Kommunikation von Risiko und Gefährdungspotenzial aus Sicht verschiedener Stakeholder. Berlin; BfR Hausdruckerei Dahlem; 2010
- [6] Rodriguez E, Edwards J. People, technology, processes and risk knowledge sharing. Electronic Journal of Knowledge Management. 2010;8(1):139-150
- [7] Beck U. Risikogesellschaft. Auf dem Weg in eine andere Moderne. Frankfurt a.M: Suhrkamp Verlag; 1986
- [8] Stewart M, Melchers RM. Probabilistic Risk Assessment of Engineering Systems. London, Champan & Hall. 1997
- [9] Luhmann N. Risk A Sociological Theory. Meuchen, DEU: Walter de Gruyter; 2011
- [10] White ME, Mac Donnel SM. Public health surveillance in low and middle-income countries. In: Teutsch SM, Churchill RE, editors. Principles and Practices of Public Health Surveillance. New York: Oxford University Press; 2000. pp. 287-315
- [11] Gordis L. Epidemiology. Philadelphia: Elsevier Saunders; 2014
- [12] Aw TC, Gardiner K, Harrington JM. Occupational Health. 5th ed. Oxford: Blackwell Publishing; 2007
- [13] International Labour Organization (ILO). Occupational Safety and Health Convention, 1981 (No. 155). Geneva: ILO; 1981
- [14] International Labour Organization (ILO). OHS Management System: A Tool for Continual Improvement. Geneva: ILO; 2011
- [15] Thierry-Chef I, Dabin J, Friberg EG, Hermen J, Istad TS, Jahnen A, Krille L, Lee C, Maccia C, Nordenskjöld A, Olerud HM, Rani K, Rehel JL, Simon SL, Struelens L, Kesminiene A. Assessing organ doses from paediatric CT scans a novel approach for an epidemiology study (the EPI-CT study). International Journal of Environmental Research and Public Health. 2013;10:717-728. DOI: 10.3390/ijerph10020717

- [16] Zimbardo PG, Gerrig R. Psychology and Life. United States: Allyn & Bacon, Incorporated; 2011
- [17] Slovic P, Fischhoff B, Lichtenstein S. Cognitive processes and societal risk taking. In: Jungermann H, Zeeuw Gd, editors. Decision Making and Change in Human Affairs. Dordrecht: Riedel; 1977. pp. 7-36
- [18] Jungermann H, Slovic P. Charakteristika individueller Risikowahrnehmung. In: Krohn G, Krücken G, editors. Riskante Technologien. Reflexion und Regulation. Einführung in die sozialwissenschaftliche Risikoforschung. Frankfurt am Main; Surhkamp: 1993
- [19] Slovic P. Perception of risk. Science. 1987;236:280-285
- [20] Kenfield SA, Stampfer MJ, Rosner BA, Colditz GA. Smoking and smoking cessation in relation to mortality. JAMA. 2008;299:2037-2047. DOI: 10.1001/jama.299.17.2037
- [21] Lidskog R, Sunderqvist G. Sociology of risk. In: Roeser S, Hillerbrand R, Sandin P, Peterson M, editors. Handbook of Risk Theory - Epistemology, Decision Theory, Ethics, and Social Implication of Risks. Dordrecht, Netherlands: Springer; 2012. pp. 1001-1028
- [22] WHO World Health Organization. Electronic Nicotine Delivery Systems [Internet]. 2004. Available from: http://apps.who.int/gb/fctc/PDF/cop6/FCTC_COP6_10-en.pdf [Accessed: July 7, 2014]
- [23] Hawkes N. WHO review exaggerated dangers of e-cigarettes, say specialists. British Medical Journal. 2014;349:g5501. DOI: 10.1136/bmj.g5501
- [24] Cooper LP. A research agenda to reduce risk in new product development through knowledge management: A practitioner perspective. Journal of Engineering and Technology Management. 2003;20:117-140
- [25] Ruzic-Dimitrijevic L. Risk assessment of knowledge management system. Online Journal of Applied Knowledge Management. 2014;3(2):114-126

Edited by Muhammad Mohiuddin, Norrin Halilem, SM Ahasanul Kobir and Cao Yuliang

Knowledge management (KM) has become an important business strategy in an era of accelerated globalization, digitalization, and servitization of products and services. Maximizing the use of organizational resources becomes fundamental for continuous growth and prosperity. Organizations of various kinds such as resource-based organizations, product-based organizations, as well as knowledge-intensive serviceoriented organizations require specific policies and support services to improve the knowledge management in their respective organizations. Knowledge Management Strategies and Applications focuses on the way organizations can create knowledge, share existing or new knowledge, and disseminate them among the stakeholders, most importantly among the employees, managers, customers, and suppliers. The selected topics are drawn from several fields of studies and give a multidisciplinary outlook. The book will be interesting not only for the researchers and students but also for the managers who want to improve knowledge sharing and innovation capabilities in their organizations.



Photo by Avosb / iStock

IntechOpen